Form 19b-4 Information

Item 1. <u>Text of Proposed Rule Change</u>

(a) Banque Centrale de Compensation, which conducts business under the name LCH SA ("LCH SA"), is proposing to expand its CDSClear service to provide clearing services for additional index and single name credit default swaps ("CDS"). Specifically, LCH SA is proposing to provide clearing services with regard to the iTraxx Asia ex Japan Index, the Markit CDX Emerging Markets ("CDX.EM") Index and the single names that comprise each index, as well as a list of additional sovereign single names which are not constituent of an index (all together the "New Products"). To expand its clearing services in this way, LCH SA is proposing to amend its CDS Clearing Supplement (the "Supplement") and Section 2 of the CDS Clearing Procedures (the "Procedures") to accommodate these additional indices and single names. LCH SA is further proposing to amend its CDS Margin Framework and Default Fund Methodology to reflect the addition of the New Products in the scope of instruments eligible for clearing by members of LCH SA CDSClear service (all together the "Proposed Rule Change").

The text of the Proposed Rule Change is in Exhibit 5.¹

The launch of the various initiatives reflected in the Proposed Rule Change will be contingent upon LCH SA's receipt of all necessary regulatory approvals, including the approval by the Commission of the Proposed Rule Change described herein.

- (b) Not applicable.
- (c) Not applicable.

Capitalized terms used but not defined herein shall have the meaning specified in the CDS Clearing Rule Book or the Clearing Supplement, as applicable.

Item 2. Procedure of the Self-Regulatory Organization

LCH SA has completed all of the required action to be taken to approve the Proposed Rule Change. The Proposed Rule Change was approved by the LCH SA Executive Risk Committee ("ERCo") on December 9, 2021, by the LCH SA Risk Committee ("RiskCo") on February 9, 2022, by the LCH SA Board of Directors ("Board") on February 22, 2022 and by LCH SA Rule Change Committee ("RCC") on December 23, 2021. No further approvals are necessary.

Questions should be addressed to François Faure, Chief Compliance Officer, at françois.faure@lseg.com or +33 1 70 37 65 96; or Mohamed Meziane, Senior Regulatory Advisor, Compliance Department, at mohamed.meziane@lseg.com or +33 1 70 37 65 62.

Item 3. Self-Regulatory Organization's Statement of Purpose, and Statutory Basis for, the Proposed Rule Change

(a) Purpose

The Proposed Rule change is being adopted to expand LCH SA's CDSClear service to provide clearing services for additional index and single name CDS. Specifically, LCH SA is proposing to provide clearing services with regard to the iTraxx Asia ex Japan Index, the CDX.EM Index and the single names that comprise each index, as well as a list of additional sovereign single names which are not constituent of an index.

LCH SA has determined that the existing CDSClear risk model currently appropriately takes into account the risk associated with the New Products but is proposing to amend both its CDS Margin Framework and CDS Default Fund Methodology (Guide Stress testing) in order to reflect the addition of the New Products to the list of instruments eligible for clearing. To accommodate the New Products, LCH SA is further proposing to amend the Supplement and Section 2 of the Procedures.

a) Proposed amendments to the Clearing Supplement

To accommodate the New Products, LCH SA is proposing to amend the following definitions set out in Section 1.2 of Part B of the Supplement: (i) "Compression Cut-off Date"; (ii) "Novation Cut-off Date"; (iii) "Index Cleared Transaction Confirmation"; and (iv) "Transaction Business Day".

Specifically, the definitions of "Compression Cut-off Date" and "Novation Cut-off Date" are each being amended to add two additional credit events that are taken into consideration in determining the "Compression Cut-off Date" and "Novation Cut-off Date": (i) the "Obligation Acceleration Credit Event"; (ii) and "Repudiation/Moratorium Credit Event". These credit events, which are both standard under the 2014 ISDA Credit Derivatives Definitions, are not credit events that apply to any of the transaction types referenced by CDS that are currently eligible for clearing at LCH SA and, therefore, did not previously need to be addressed in the Supplement. These credit events apply to certain transaction types for sovereigns, and are proposed to be added as a result of index comprising of and single name CDS referencing sovereign reference entities becoming eligible for clearing.

In addition, the definition of "Index Cleared Transaction Confirmation" is proposed to be revised to provide that: (i) with regard to any index cleared transaction that references a Markit iTraxx ex Japan Index Series [27] or above, the confirmation will be the form of confirmation that incorporates the iTraxx Asia/Pacific Untranched Standard Terms Supplement; and (ii) with regard to any index cleared transaction that references a Markit CDX.EM Index Series [27] or above, the form of confirmation that incorporates the CDX

Emerging Markets Untranched Transactions Standard Terms Supplement, in each case being the latest version in force as published by Markit North America, Inc.

The definition of a "Transaction Business Day" is currently defined to mean a "Business Day", as defined in the Index Cleared Transaction Confirmation or the Single Name Cleared Transaction Confirmation, as applicable. This term is proposed to be amended to take into account the situation where such confirmations could include different definitions of the term "Business Day" depending on the circumstances by providing that, "if the relevant Index Cleared Transaction Confirmation or Single Name Cleared Transaction Confirmation defines such term differently depending upon its use, such distinction shall also apply to the use of the term Transaction Business Day herein."

In Section 2 of Part B of the Supplement, LCH SA is proposing to amend Section 2.2 (*Index Cleared Transaction Confirmation*) which specifies the manner in which an Index Cleared Transaction Confirmation is amended, supplemented and completed depending on the index CDS that is cleared to include, in addition to the indices currently set out in the section, the iTraxx Asia ex Japan Index and the CDX.EM Index and provide for the necessary amendments to be made to the relevant confirmations depending on the index. Section 2.2. is also proposed to be amended to provide that "*The applicable Physical Settlement Matrix is the version of the Physical Settlement Matrix which is in force on the Clearing Day on which the Index Cleared Transaction is registered by LCH SA*" in a new indent (i) of paragraph (f). The purpose of this amendment is to ensure that the Additional Provisions for Certain Russian Entities published by ISDA on March 25, 2022 will apply to the relevant cleared trades, including the trades submitted through the backloading cycle that could have been entered into before the implementation date of these Additional

Provisions and updated Physical Settlement Matrix and for which one of the parties, or both, did not adhere to the ISDA 2022 Russia Additional Provisions Protocol published by ISDA on March 29, 2022.

In Section 4 of Part B of the Supplement, LCH SA is proposing to amend Section 4.1(b) to add a "Repudiation/Moratorium Extension Notice" to the types of notices that neither LCH SA nor a clearing member is entitled to deliver with regard to an M(M)R Restructuring in accordance with the terms of any Restructuring Cleared Transaction. As above, a "Repudiation/Moratorium Extension Notice" is standard under the 2014 ISDA Credit Derivatives Definitions and is being proposed to be added as a result of index comprising of and single name CDS referencing sovereigns becoming eligible for clearing.²

In Section 6 of Part B of the Supplement, Section 6.5(c) is proposed to be amended to add "Package Observable Bond" to the types of asset packages that can be identified in a Notice of Physical Settlement ("NOPS") or a NOPS Amendment Notice. The Package Observable Bond provisions in the 2014 ISDA Credit Derivatives Definitions only apply to transactions referencing sovereigns. As a result, they did not previously need to be referenced in the Supplement.

LCH SA is also proposing to add a new section 6.8(c) entitled "Buy-in of Bonds – Cap on Settlement" for the purposes of clarifying how the "60 Business Day Cap on Settlement", which is relevant for transactions derived from the CDX EM Index amongst others, will apply to CCM Client Transactions in respect of the Matched Contracts of a

For the same reason, "Repudiation/Moratorium Extension Notice" is proposed to be added to Section 5(b) of Appendix XIII of Part B of the Supplement (*CCM Client Transaction Requirements*).

Settlement Matched Pair. This proposed amendments consist in making an adjustment as to the manner in which Section 9.10 of the 2014 ISDA Credit Derivatives Definitions works between Matched Buyer and Matched Seller to ensure that the extension of the Termination Date provided for by Section 9.10 will apply when there has been a notice delivered to Matched Seller by its client under a CCM Client Transaction. This is to ensure that the Termination Date of the Cleared Transactions and related CCM Client Transaction is the same.³

(b) Proposed amendments to Section 2 of the Procedures

LCH SA is also proposing to make one minor technical amendment to Section 2 of the Procedures (*Margin, NPV Payment and Price Alignment*). Specifically, the initial sentence of Section 2.7(c) currently provides, *inter alia*, that, where a Clearing Member is acting as a CDS Seller, Short Charge Margin will be required to cover the risk that the Clearing Member is subject to an event of default at the same time that a credit event occurs "with respect to a Reference Entity". Recognizing that a credit event may occur with respect to more than one Reference Entity, this sentence is proposed to be revised to refer to "one or more Reference Entities".

For the same reason, the provisions of section 6.8(c) are effectively repeated in Section 7.8 and Section 7.18 of Appendix XIII of Part B of the Supplement (*CCM Client Transaction Requirements*). Separately, Section 7.15 of Appendix XIII, Alternative Procedures relating to Loans in respect of Matched Contracts, and Section 7.17 of Appendix XIII, Alternative Procedures relating to Assets Not Delivered, are proposed to be amended to remove as unnecessary the phrase "for the purposes of the Matched Contracts of the related Settlement Matched Pair" and also to use the correct defined term "Settlement Matched Pair".

(c) Proposed amendments to the Reference Guide: CDS Margin Framework

LCH SA is proposing to amend the Margin Framework to reflect the addition of the new single names. For example, Section 3.4.5, Portfolio Margining, which, *inter alia*, lists the various combinations of instruments that can constitute an index basis package, is proposed to be revised to add to the list (i) the CDX.EM Index vs All Single Names Constituents of the index and (ii) the iTraxx Asia ex Japan vs All Single Names Constituents of the index. In addition, LCH SA is proposing to amend Section 3.1.1, Recovery Rate for Short Charge to note that the recovery rate for state-owned enterprises ("SOE") is 70 percent. LCH is also proposing to move the provisions of current Section 3.5.2, Short Charge Calculation, to a new Section 3.5.3. A new Section 3.5.2, Sovereign Exposures, is proposed to be added, which notes the high level of correlation between SOEs and their sovereign entities. As a result, an SOE that is more than 50 percent owned by a sovereign entity would be defaulted jointly with its sovereign entity when the positions are not risk reducing. Further, exposures for SOEs will be calculated using a fixed 70 percent recovery rate.

LCH SA is also proposing to amend Section 3.8.1, Offsets inter-region, to expand the regional pairs that LCH SA will consider in calculating wrong way risk to include: (i) Europe/US; (ii) Europe/Australia; (iii) Europe/Asia; (iv) US/Australia; (v) US/Asia; and (vi) Asia/Australia.

LCH SA is proposing to amend Section 4.1.1, Liquidity Charge for Linear Portfolio, to note that the liquidation cost of a sub-portfolio composed of a single 5 year position in the principal on the run index is simply the sum of the macro hedging cost.

Further, single names without a parent index are considered a sub-portfolio for which LCH SA charges the cost of unwinding a non-hedged sub-portfolio. Finally, Section 4.1.2, Macro Hedging Phase, which, *inter alia*, sets out a list of sub-portfolios corresponding to indices and their components is proposed to be revised to add: (i) the CDX.EM sub-portfolio; (ii) the iTraxx Asia ex Japan IG sub-portfolio, and (iii) the No parent index sub-portfolio.

LCH SA is proposing to amend Section 4.1.7 to update the existing thresholds and include more cleared indexes in the table for volume thresholds based on calibrations done in December 2021. A dedicated liquidity grid has also been added for sovereign single names in order to reflect their tighter bid-ask spreads and higher liquidity profiles.

LCH SA is also proposing to amend the CDS Default Fund Methodology (Guide Stress testing) in a number of sections, to reflect the extension of the product offer as well as to introduce a Sovereign Stressed Short Charge component aimed to capture a potential joint default of a member and its country:

- the last paragraph of section 2.2 adds to the list of index families covered to reflect the addition of CDX.EM and iTraxx Asia. It also adds iTraxx Australia, as this should have been updated when introducing that index.
- section 2.4.1 details how State-Owned Entities' exposures should be added to the exposure on the sovereign name only if risk increasing
- section 2.4.2 introduces a Sovereign Stressed Short Charge, considering jointly the top exposure across the portfolio and if relevant the exposure on the sovereign name corresponding to the member's jurisdiction

- section 2.4.3. and 2.7.2 describe the same Sovereign Stressed Short Charge with formulas instead of plain text
- section 2.6.1. and 2.6.3 extend the logic of exercise decisions to consider the Sovereign Stressed Short Charge when relevant.

(b) Statutory Basis

LCH SA believes that the Proposed Rule Change is consistent with the requirements of Section 17A of the Act⁴ and regulations thereunder applicable to it, including Commission Rule 17Ad-22(e).⁵ In particular, Section 17A(b)(3)(F) of the Act requires, *inter alia*, that the rules of a clearing agency be designed to "promote the prompt and accurate clearance and settlement of . . . derivatives agreements, contracts, and transactions".⁶ By proposing to amend its CDS Clearing Supplement to authorize the expansion of LCH SA's CDSClear Service to provide clearing services with regard to the New Products, on the terms and conditions set out in the Proposed Rule Change, LCH SA considers that this would encourage Clearing Members to clear additional indices and single name CDS through its CDSClear service, which, in turn, should promote the prompt and accurate clearance and settlement of those instruments within the meaning of Section 17A(b)(3)(F) of the Act.⁷. The Proposed Rule Change, in particular, the amendments to the CDS Clearing Supplement, therefore, are consistent with the requirements of Section 17A(b)(3)(F) of the Act.

⁴ 15 U.S.C. 78q-1.

⁵ 17 C.F.R. §240.17Ad-22.

^{6 15} U.S.C. 78q-1(b)(3)(F).

⁷ 15 U.S.C. 78q-1(b)(3)(F).

Further, from the perspective of financial risk management and margin requirements, the clearing of the New Products would not require changes to LCH SA's existing margin methodology, default management policies and procedures and operational process, as LCH SA determined that the current margin framework for its CDSClear service already appropriately captures the risk associated to the New Products. The New Products would be cleared pursuant to LCH SA's existing clearing arrangements and related financial safeguards, protections and risk management procedures which are consistent with Exchange Act Rule 17Ad-22(e)(17),8 requiring a covered clearing agency to establish, implement, maintain and enforce written policies and procedures reasonably designed to manage the covered clearing agency's operational risks by, among other things, identifying the plausible sources of operational risk, both internal and external, and mitigating their impact through the use of appropriate systems, policies, procedures, and controls.

Adopting rules to facilitate the clearing of the New Products would also be consistent with other relevant requirements of Rule 17Ad-22(e),⁹ as set forth in the following discussion.

<u>Margin Requirements</u>. Rule 17Ad-22(e)(4)¹⁰ requires LCH SA to establish, implement, maintain, and enforce written policies and procedures reasonably designed to effectively identify, measure, monitor, and manage its credit exposures to participants and those arising from its payment, clearing, and settlement processes, among other

^{8 17} CFR 240.17Ad-22(e)(17).

⁹ 17 CFR 240.17Ad-22(e).

¹⁰ 17 CFR 240.17Ad-22(e)(4).

requirements. In terms of financial resources, LCH SA would apply its existing margin methodology to the New Products. LCH SA believes that the proposed rules that would apply this risk model to the New Products will provide sufficient margin requirements to cover its credit exposure to its clearing members from clearing such contracts, consistent with the requirements of Rule 17Ad-22I(4).¹¹

Financial Resources. Rule 17Ad-22I(4)(i)¹² requires LCH SA to establish, implement, maintain, and enforce written policies and procedures reasonably designed to effectively identify, measure, monitor, and manage its credit exposures to participants and those arising from its payment, clearing, and settlement processes by maintaining sufficient financial resources to cover its credit exposure to each participant fully with a high degree of confidence. To the extent not already maintained pursuant to paragraph (e)(4)(i), Rule 17Ad-22(e)(4)(ii)¹³ requires LCH SA's policies and procedures be reasonably designed to maintain additional financial resources at the minimum to enable it to cover a wide range of foreseeable stress scenarios that include, but are not limited to, the default of the two participant families that would potentially cause the largest aggregate credit exposure for the covered clearing agency in extreme but plausible market conditions. As explained above, LCH SA is proposing to make some changes to its CDS Default Fund Methodologydocumentation (Guide Stress testing) in order to reflect the extension of the product offer as well as to introduce a Sovereign Stressed Short Charge component aimed to capture a potential joint default of a member and its

¹¹ 17 CFR 240.17Ad-22(e)4.

¹² 17 CFR 240.17Ad-22(e)(4)(i).

¹³ 17 CFR 240.17Ad-22(e)(4)(ii).

country. LCH SA believes that with the proposed changes in its stress testing framework, its Default Fund will, together with the required margin, provide sufficient financial resources to support the clearing of the New Products, consistent with the requirements of Rules 17Ad22(e)(4)(i) and (ii).

Operational Resources. Rule 17Ad-22(e)(3)¹⁴ requires LCH SA to establish, implement, maintain, and enforce written policies and procedures reasonably designed to maintain a sound risk management framework for comprehensively managing legal, credit, liquidity, operational, general business, investment, custody, and other risks that arise in or are borne by the covered clearing agency. LCH SA believes that its existing operational and risk management resources will be sufficient for clearing of the New Products, consistent with the requirements of Rule 17Ad-22(e)(3)¹⁵, as these new contracts are substantially the same from an operational and risk management perspective as the existing CDS contracts cleared by LCH SA CDSClear.

LCH SA will also apply its existing default management policies and procedures for the New Products. As with current CDSClear products with similar risk profile, LCH SA believes that these procedures allow for it to take timely action to contain losses and liquidity pressures and to continue meeting its obligations in the event of clearing member insolvencies or defaults in respect of the additional single names, in accordance with Rule 17Ad-22(e)(3). ¹⁶

¹⁴ 17 CFR 240.17Ad-22(e)(3).

¹⁵ 17 CFR 240.17Ad-22(e)(3).

¹⁶ 17 CFR 240.17Ad-22(e)(3).

Exchange Act Rule 17Ad-22(e)(1)¹⁷ requires that a covered clearing agency establish, implement, maintain, and enforce written policies and procedures reasonably designed to provide for a well-founded, clear, transparent, and enforceable legal basis for each aspect of its activities in all relevant jurisdictions. As described above, the Proposed Change is also modifying the Supplement to take into account the New Products and provide for a clear and transparent legal basis for LCH SA's CDS Clearing rules consistent with the requirements of Exchange Act Rule 17Ad-22(e)(1)¹⁸.

Credit default swap (CDS) is an over-the-counter (OTC) market on which participants can be active at any time in the context of market stress. The LCH SA CDSClear risk model is considering 5-d moves of unhedged portfolios and the back testing results confirmed that the margins for the New Products were sufficient to cover the exposure in the interval between the last margin collection and the close out of the portfolio a defaulting cleating member which is consistent with the requirements of SEC Rule 17Ad-22(e)(6)(iii)¹⁹.

Item 4. Self-Regulatory Organization's Statement on Burden on Competition

Section 17A(b)(3)(I) of the Act requires that the rules of a clearing agency not impose any burden on competition not necessary or appropriate in furtherance of the purposes of the Act.²⁰

LCH SA does not believe that its proposed clearing of the New Products will adversely affect competition in the trading market for those contracts or CDS generally.

¹⁷ 17 CFR 240.17Ad-22(e)(1).

¹⁸ 17 CFR 240.17Ad-22(e)(1).

¹⁹ 17 CFR 240. 17Ad-22(e)(6)(iii).

²⁰ 15 U.S.C. 78q-1(b)(3)(I).

By allowing LCH SA to clear the New Products, market participants will have additional choices on where to clear and which products to use for risk management purposes, which, in turn, will promote competition and further the development of CDS for risk management.

In addition, LCH SA will continue to apply its existing fair and open access criteria to the clearing of these additional products and will apply the same criteria to every clearing member or client who proposes to enter into this clearing activity.

Accordingly, LCH SA does not believe that the Proposed Rule Change would impose any burden on competition that is not necessary or appropriate in furtherance of the purposes of the Act.

Item 5. <u>Self-Regulatory Organization's Statement on Comments on the Proposed Rule Change Received from Members, Participants or Others</u>

Written comments relating to the Proposed Rule Change have not been solicited or received. LCH SA will notify the Commission of any written comments received by LCH SA.

Item 6. Extension of Time Period for Commission Action

LCH SA does not consent to the extension of the time period listed in Section 19(b)(2) of the Exchange Act for Commission action.²¹

Item 7. Basis for Summary Effectiveness Pursuant to Section 19(b)(3) or for Accelerated Effectiveness Pursuant to Section 19(b)(2) or Section 19(b)(7)(D) of the Exchange Act

- (a) Not applicable.
- (b) Not applicable.

²¹ 15 U.S.C. 78s(b)(2).

- (c) Not applicable.
- (d) Not applicable.

Item 8. Proposed Rule Change Based on Rules of Another Self-Regulatory Organization or the Commission

The Proposed Rule Change is not based on the rules of another self-regulatory organization or the Commission.

Item 9. Security-Based Swap Submissions Filed Pursuant to Section 3C of the Exchange Act

Not applicable.

Item 10. Advance Notices Filed Pursuant to Section 806(e) of the Payment, Clearing and Settlement Supervision Act

Not applicable

Item 11. Exhibits

Exhibit 1 – Not Applicable

Exhibit 1A – Notice of proposed rule change for publication in the Federal

Register.

Exhibit 2 – Not Applicable

Exhibit 3 – CDSClear Risk methodology document including back testing results.

Omitted and filed separately with the Commission. Confidential treatment pursuant to 17 CFR 240.24b-2 being requested

Exhibit 4 – Not Applicable

Exhibit 5 – Text of the proposed rule change:

Exhibit 5a - LCH SA CDS Clearing Supplement

Exhibit 5b - LCH SA CDS Clearing Procedures – Section 2

Exhibit 5c - LCH SA Methodology Services Reference Guide: CDS

Margin Framework (V3.14): Omitted and filed separately with the

Commission. Confidential treatment pursuant to 17 CFR 240.24b-2

Exhibit 5d - LCH SA CDS Default Fund Methodology – V1.9: Omitted and filed separately with the Commission. Confidential treatment pursuant to 17 CFR 240.24b-2 being requested

SIGNATURES

being requested

Pursuant to the requirements of the Exchange Act, Banque Centrale de Compensation has caused this filing to be signed on its behalf by the undersigned hereunto duly authorized.

BANQUE CENTRALE DE COMPENSATION

By:______Francois Faure

Chief Compliance Officer

EXHIBIT 1A

SECURITIES AND EXCHANGE COMMISSION
(Release No; File No. SR-LCH SA-2022-007)
, 2022
Self-Regulatory Organizations; LCH SA; Notice of Filing of Proposed Rule Change Relating to Providing Clearing Services for Additional Index and Single Name CDS
Pursuant to Section 19(b)(1) of the Securities Exchange Act of 1934 ("Act"), and
Rule 19b-4 thereunder, ² notice is hereby given that on, 2022, Banque
Centrale de Compensation, which conducts business under the name LCH SA ("LCH
SA"), filed with the Securities and Exchange Commission ("Commission") the proposed
rule change ("Proposed Rule Change") described in Items I, II and III below, which Items
have been primarily prepared by LCH SA. The Commission is publishing this notice to
solicit comments on the Proposed Rule Change from interested persons.

I. <u>Clearing Agency's Statement of the Terms of Substance of the Proposed Rule Change</u>

LCH SA is proposing to expand its CDSClear service to provide clearing services for additional index and single name credit default swaps ("CDS"). Specifically, LCH SA is proposing to provide clearing services with regard to the iTraxx Asia ex Japan Index, the Markit CDX Emerging Markets ("CDX.EM") Index and the single names that comprise each index, as well as a list of additional sovereign single names which are not constituent of an index (all together the "New Products"). To expand its clearing services in this way, LCH SA is proposing to amend its CDS Clearing Supplement (the "Supplement") and

¹ 15 U.S.C. 78s(b)(1).

² 17 C.F.R. § 240.19b-4.

Section 2 of the CDS Clearing Procedures (the "**Procedures**") to accommodate these additional indices and single names. LCH SA is further proposing to amend its CDS Margin Framework and CDS Default Fund Methodology (Guide Stress Testing) to reflect the addition of the New Products in the scope of instruments eligible for clearing by members of LCH SA CDSClear service.

The text of the Proposed Rule Change is in Exhibit 5.³

The launch of the various initiatives reflected in the Proposed Rule Change will be contingent upon LCH SA's receipt of all necessary regulatory approvals, including the approval by the Commission of the Proposed Rule Change described herein.

II. <u>Clearing Agency's Statement of the Purpose of, and Statutory Basis for, the Proposed Rule Change</u>

In its filing with the Commission, LCH SA included statements concerning the purpose of and basis for the Proposed Rule Change and discussed any comments it received on the Proposed Rule Change. The text of these statements may be examined at the places specified in Item IV below. LCH SA has prepared summaries, set forth in sections A, B, and C below, of the most significant aspects of such statements.

A. <u>Clearing Agency's Statement of the Purpose of, and Statutory Basis for, the Proposed Rule Change</u>

1. <u>Purpose</u>

The Proposed Rule change is being adopted to expand LCH SA's CDSClear service to provide clearing services for additional index and single name CDS. Specifically, LCH SA is proposing to provide clearing services with regard to the iTraxx Asia ex Japan Index,

Capitalized terms used but not defined herein shall have the meaning specified in the CDS Clearing Rule Book or the Clearing Supplement, as applicable.

the CDX.EM Index and the single names that comprise each index, as well as a list of additional sovereign single names which are not constituent of an index.

LCH SA has determined that the existing CDSClear risk model currently appropriately takes into account the risk associated with the New Products but is proposing to amend both its CDS Margin Framework and CDS Default Fund Methodology (Guide Stress testing) in order to reflect the addition of the New Products to the list of instruments eligible for clearing. To accommodate the New Products, LCH SA is further proposing to amend the Supplement and Section 2 of the Procedures.

(a) The CDS Clearing Supplement

To accommodate the New Products, LCH SA is proposing to amend the following definitions set out in Section 1.2 of Part B of the Supplement: (i) "Compression Cut-off Date"; (ii) "Novation Cut-off Date"; (iii) "Index Cleared Transaction Confirmation"; and (iv) "Transaction Business Day".

Specifically, the definitions of "Compression Cut-off Date" and "Novation Cut-off Date" are each being amended to add two additional credit events that are taken into consideration in determining the "Compression Cut-off Date" and "Novation Cut-off Date": (i) the "Obligation Acceleration Credit Event"; and (ii) the "Repudiation/Moratorium Credit Event". These credit events, which are both standard under the 2014 ISDA Credit Derivatives Definitions, are not credit events that apply to any of the transaction types referenced by CDS that are currently eligible for clearing at LCH SA and, therefore, did not previously need to be addressed in the Supplement. These credit events apply to certain transaction types for sovereigns, and are proposed to be added as a result of index comprising of and single name CDS referencing sovereign reference entities becoming eligible for clearing.

In addition, the definition of "Index Cleared Transaction Confirmation" is proposed to be revised to provide that: (i) with regard to any index cleared transaction that references a Markit iTraxx ex Japan Index Series [27] or above, the confirmation will be the form of confirmation that incorporates the iTraxx Asia/Pacific Untranched Standard Terms Supplement; and (ii) with regard to any index cleared transaction that references a Markit CDX.EM Index Series [27] or above, the form of confirmation that incorporates the CDX Emerging Markets Untranched Transactions Standard Terms Supplement, in each case being the latest version in force as published by Markit North America, Inc.

The definition of a "Transaction Business Day" is currently defined to mean a "Business Day", as defined in the Index Cleared Transaction Confirmation or the Single Name Cleared Transaction Confirmation, as applicable. This term is proposed to be amended to take into account the situation where such confirmations could include different definitions of the term "Business Day" depending on the circumstances by providing that, "if the relevant Index Cleared Transaction Confirmation or Single Name Cleared Transaction Confirmation defines such term differently depending upon its use, such distinction shall also apply to the use of the term Transaction Business Day herein."

In Section 2 of Part B of the Supplement, LCH SA is proposing to amend Section 2.2 (*Index Cleared Transaction Confirmation*) which specifies the manner in which an Index Cleared Transaction Confirmation is amended, supplemented and completed depending on the index CDS that is cleared to include, in addition to the indices currently set out in the section, the iTraxx Asia ex Japan Index and the CDX.EM Index and provide

for the necessary amendments to be made to the relevant confirmations depending on the index. Section 2.2. is also proposed to be amended to provide that "The applicable Physical Settlement Matrix is the version of the Physical Settlement Matrix which is in force on the Clearing Day on which the Index Cleared Transaction is registered by LCH SA" in a new indent (i) of paragraph (f). The purpose of this amendment is to ensure that the Additional Provisions for Certain Russian Entities published by ISDA on March 25, 2022 will apply to the relevant cleared trades, including the trades submitted through the backloading cycle that could have been entered into before the implementation date of these Additional Provisions and updated Physical Settlement Matrix and for which one of the parties, or both, did not adhere to the ISDA 2022 Russia Additional Provisions Protocol published by ISDA on March 29, 2022.

In Section 4 of Part B of the Supplement, LCH SA is proposing to amend Section 4.1(b) to add a "Repudiation/Moratorium Extension Notice" to the types of notices that neither LCH SA nor a clearing member is entitled to deliver with regard to an M(M)R Restructuring in accordance with the terms of any Restructuring Cleared Transaction. As above, a "Repudiation/Moratorium Extension Notice" is standard under the 2014 ISDA Credit Derivatives Definitions and is being proposed to be added as a result of index comprising of and single name CDS referencing sovereigns becoming eligible for clearing.⁴

Transaction Requirements).

For the same reason, "Repudiation/Moratorium Extension Notice" is proposed to be added to Section 5(b) of Appendix XIII of Part B of the Supplement (CCM Client

In Section 6 of Part B of the Supplement, Section 6.5(c) is proposed to be amended to add "Package Observable Bond" to the types of asset packages that can be identified in a Notice of Physical Settlement ("NOPS") or a NOPS Amendment Notice. The Package Observable Bond provisions in the 2014 ISDA Credit Derivatives Definitions only apply to transactions referencing sovereigns. As a result, they did not previously need to be referenced in the Supplement.

LCH SA is also proposing to add a new section 6.8(c) entitled "Buy-in of Bonds – Cap on Settlement" for the purposes of clarifying how the "60 Business Day Cap on Settlement", which is relevant for transactions derived from the CDX EM Index amongst others, will apply to CCM Client Transactions in respect of the Matched Contracts of a Settlement Matched Pair. This proposed amendments consist in making an adjustment as to the manner in which Section 9.10 of the 2014 ISDA Credit Derivatives Definitions works between Matched Buyer and Matched Seller to ensure that the extension of the Termination Date provided for by Section 9.10 will apply when there has been a notice delivered to Matched Seller by its client under a CCM Client Transaction. This is to ensure that the Termination Date of the Cleared Transactions and related CCM Client Transaction is the same.⁵

⁻

For the same reason, the provisions of section 6.8(c) are effectively repeated in Section 7.8 and Section 7.18 of Appendix XIII of Part B of the Supplement (*CCM Client Transaction Requirements*).

Separately, Section 7.15 of Appendix XIII, Alternative Procedures relating to Loans in respect of Matched Contracts, and Section 7.17 of Appendix XIII, Alternative Procedures relating to Assets Not Delivered, are proposed to be amended to remove as unnecessary the phrase "for the purposes of the Matched Contracts of the related Settlement Matched Pair" and also to use the correct defined term "Settlement Matched Pair".

(b) Section 2 of the Procedures

LCH SA is also proposing to make one minor technical amendment to Section 2 of the Procedures (*Margin, NPV Payment and Price Alignment*). Specifically, the initial sentence of Section 2.7(c) currently provides, *inter alia*, that, where a Clearing Member is acting as a CDS Seller, Short Charge Margin will be required to cover the risk that the Clearing Member is subject to an event of default at the same time that a credit event occurs "with respect to a Reference Entity". Recognizing that a credit event may occur with respect to more than one Reference Entity, this sentence is proposed to be revised to refer to "one or more Reference Entities".

(c) The Reference Guide: CDS Margin Framework

LCH SA is proposing to amend the Margin Framework to reflect the addition of the new single names. For example, Section 3.4.5, Portfolio Margining, which, *inter alia*, lists the various combinations of instruments that can constitute an index basis package, is proposed to be revised to add to the list (i) the CDX.EM Index vs All Single Names Constituents of the index and (ii) the iTraxx Asia ex Japan vs All Single Names Constituents of the index. In addition, LCH SA is proposing to amend Section 3.1.1, Recovery Rate for Short Charge to note that the recovery rate for state-owned enterprises ("SOE") is 70 percent. LCH is also proposing to move the provisions of current Section 3.5.2, Short Charge Calculation, to a new Section 3.5.3. A new Section 3.5.2, Sovereign Exposures, is proposed to be added, which notes the high level of correlation between SOEs and their sovereign entities. As a result, an SOE that is more than 50 percent owned by a sovereign entity would be defaulted jointly with its sovereign entity when the positions

are not risk reducing. Further, exposures for SOEs will be calculated using a fixed 70 percent recovery rate.

LCH SA is also proposing to amend Section 3.8.1, Offsets inter-region, to expand the regional pairs that LCH SA will consider in calculating wrong way risk to include: (i) Europe/US; (ii) Europe/Australia; (iii) Europe/Asia; (iv) US/Australia; (v) US/Asia; and (vi) Asia/Australia.

LCH SA is proposing to amend Section 4.1.1, Liquidity Charge for Linear Portfolio, to note that the liquidation cost of a sub-portfolio composed of a single 5 year position in the principal on the run index is simply the sum of the macro hedging cost. Further, single names without a parent index are considered a sub-portfolio for which LCH SA charges the cost of unwinding a non-hedged sub-portfolio. Finally, Section 4.1.2, Macro Hedging Phase, which, *inter alia*, sets out a list of sub-portfolios corresponding to indices and their components is proposed to be revised to add: (i) the CDX.EM sub-portfolio; (ii) the iTraxx Asia ex Japan IG sub-portfolio, and (iii) the No parent index sub-portfolio.

LCH SA is proposing to amend Section 4.1.7 to update the existing thresholds and include more cleared indexes in the table for volume thresholds based on calibrations done in December 2021. A dedicated liquidity grid has also been added for sovereign single names in order to reflect their tighter bid-ask spreads and higher liquidity profiles.

LCH SA is also proposing to amend the CDS Default Fund Methodology (Guide Stress Testing) in a number of sections, to reflect the extension of the product offer as well as to introduce a Sovereign Stressed Short Charge component aimed to capture a potential joint default of a member and its country:

- the last paragraph of section 2.2 adds to the list of index families covered to reflect the addition of CDX.EM and iTraxx Asia. It also adds iTraxx Australia, as this should have been updated when introducing that index.
- section 2.4.1 details how State-Owned Entities' exposures should be added to the exposure on the sovereign name only if risk increasing
- section 2.4.2 introduces a Sovereign Stressed Short Charge, considering jointly the top exposure across the portfolio and if relevant the exposure on the sovereign name corresponding to the member's jurisdiction
- section 2.4.3. and 2.7.2 describe the same Sovereign Stressed Short Charge with formulas instead of plain text
- section 2.6.1. and 2.6.3 extend the logic of exercise decisions to consider the Sovereign Stressed Short Charge when relevant

2. <u>Statutory Basis</u>

LCH SA believes that the Proposed Rule Change is consistent with the requirements of Section 17A of the Act⁶ and regulations thereunder applicable to it, including Commission Rule 17Ad-22(e).⁷ In particular, Section 17A(b)(3)(F) of the Act requires, *inter alia*, that the rules of a clearing agency be designed to "promote the prompt and accurate clearance and settlement of . . . derivatives agreements, contracts, and transactions".⁸ By proposing to amend its CDS Clearing Supplement to authorize the expansion of LCH SA's CDSClear Service to provide clearing services with regard to the

⁶ 15 U.S.C. 78q-1.

⁷ 17 C.F.R. §240.17Ad-22.

⁸ 15 U.S.C. 78q-1(b)(3)(F).

New Products, on the terms and conditions set out in the Proposed Rule Change, LCH SA considers that this would encourage Clearing Members to clear additional indices and single name CDS through its CDSClear service, which, in turn, should promote the prompt and accurate clearance and settlement of those instruments within the meaning of Section 17A(b)(3)(F) of the Act.⁹. The Proposed Rule Change, in particular, the amendments to the CDS Clearing Supplement, therefore, are consistent with the requirements of Section 17A(b)(3)(F) of the Act.

Further, from the perspective of financial risk management and margin requirements, the clearing of the New Products would not require changes to LCH SA's existing margin methodology, default management policies and procedures and operational process, as LCH SA determined that the current margin framework for its CDSClear service already appropriately captures the risk associated to the New Products. The New Products would be cleared pursuant to LCH SA's existing clearing arrangements and related financial safeguards, protections and risk management procedures which are consistent with Exchange Act Rule 17Ad-22(e)(17),10 requiring a covered clearing agency to establish, implement, maintain and enforce written policies and procedures reasonably designed to manage the covered clearing agency's operational risks by, among other things, identifying the plausible sources of operational risk, both internal and external, and mitigating their impact through the use of appropriate systems, policies, procedures, and controls.

⁹ 15 U.S.C. 78q-1(b)(3)(F).

¹⁰ 17 CFR 240.17Ad-22(e)(17).

Adopting rules to facilitate the clearing of the New Products would also be consistent with other relevant requirements of Rule 17Ad-22(e),¹¹ as set forth in the following discussion.

Margin Requirements. Rule 17Ad-22(e)(4)¹² requires LCH SA to establish, implement, maintain, and enforce written policies and procedures reasonably designed to effectively identify, measure, monitor, and manage its credit exposures to participants and those arising from its payment, clearing, and settlement processes, among other requirements. In terms of financial resources, LCH SA would apply its existing margin methodology to the New Products. LCH SA believes that the proposed rules that would apply this risk model to the New Products will provide sufficient margin requirements to cover its credit exposure to its clearing members from clearing such contracts, consistent with the requirements of Rule 17Ad-22I(4).¹³

Financial Resources. Rule 17Ad-22I(4)(i)¹⁴ requires LCH SA to establish, implement, maintain, and enforce written policies and procedures reasonably designed to effectively identify, measure, monitor, and manage its credit exposures to participants and those arising from its payment, clearing, and settlement processes by maintaining sufficient financial resources to cover its credit exposure to each participant fully with a high degree of confidence. To the extent not already maintained pursuant to paragraph (e)(4)(i), Rule

¹⁷ CFR 240.17Ad-22(e).

¹² 17 CFR 240.17Ad-22(e)(4).

¹³ 17 CFR 240.17Ad-22(e)4.

¹⁷ CFR 240.17Ad-22(e)(4)(i).

17Ad-22(e)(4)(ii)¹⁵ requires LCH SA's policies and procedures be reasonably designed to maintain additional financial resources at the minimum to enable it to cover a wide range of foreseeable stress scenarios that include, but are not limited to, the default of the two participant families that would potentially cause the largest aggregate credit exposure for the covered clearing agency in extreme but plausible market conditions. As explained above, LCH SA is proposing to make some changes to its CDS Default Fund Methodology documentation (Guide Stress Testing) in order to reflect the extension of the product list as well as to introduce a Sovereign Stressed Short Charge component aimed to capture a potential joint default of a member and its country. LCH SA believes that with the proposed changes in its stress testing framework, its Default Fund will, together with the required margin, provide sufficient financial resources to support the clearing of the New Products, consistent with the requirements of Rules 17Ad22(e)(4)(i) and (ii).

Operational Resources. Rule 17Ad-22(e)(3)¹⁶ requires LCH SA to establish, implement, maintain, and enforce written policies and procedures reasonably designed to maintain a sound risk management framework for comprehensively managing legal, credit, liquidity, operational, general business, investment, custody, and other risks that arise in or are borne by the covered clearing agency. LCH SA believes that its existing operational and risk management resources will be sufficient for clearing of the New Products, consistent with the requirements of Rule 17Ad-22(e)(3)¹⁷, as these new contracts are

¹⁵ 17 CFR 240.17Ad-22(e)(4)(ii).

¹⁶ 17 CFR 240.17Ad-22(e)(3).

¹⁷ 17 CFR 240.17Ad-22(e)(3).

substantially the same from an operational and risk management perspective as the existing CDS contracts cleared by LCH SA CDSClear.

LCH SA will also apply its existing default management policies and procedures for the New Products. As with current CDSClear products with similar risk profile, LCH SA believes that these procedures allow for it to take timely action to contain losses and liquidity pressures and to continue meeting its obligations in the event of clearing member insolvencies or defaults in respect of the additional single names, in accordance with Rule 17Ad-22(e)(3). 18

Exchange Act Rule 17Ad-22(e)(1)¹⁹ requires that a covered clearing agency establish, implement, maintain, and enforce written policies and procedures reasonably designed to provide for a well-founded, clear, transparent, and enforceable legal basis for each aspect of its activities in all relevant jurisdictions. As described above, the Proposed Change is also modifying the Supplement to take into account the New Products and provide for a clear and transparent legal basis for LCH SA's CDS Clearing rules consistent with the requirements of Exchange Act Rule 17Ad-22(e)(1)²⁰.

Credit default swap (CDS) is an over-the-counter (OTC) market on which participants can be active at any time in the context of market stress. The LCH SA CDSClear risk model is considering 5-d moves of unhedged portfolios and the back testing results confirmed that the margins for the New Products were sufficient to cover the exposure in the interval between the last margin collection and the close out of the

¹⁸ 17 CFR 240.17Ad-22(e)(3).

¹⁹ 17 CFR 240.17Ad-22(e)(1).

²⁰ 17 CFR 240.17Ad-22(e)(1).

portfolio a defaulting cleating member which is consistent with the requirements of SEC Rule 17Ad-22(e)(6)(iii)²¹.

B. Clearing Agency's Statement on Burden on Competition

Section 17A(b)(3)(I) of the Act requires that the rules of a clearing agency not impose any burden on competition not necessary or appropriate in furtherance of the purposes of the Act.²²

LCH SA does not believe that its proposed clearing of the New Products will adversely affect competition in the trading market for those contracts or CDS generally. By allowing LCH SA to clear the New Products, market participants will have additional choices on where to clear and which products to use for risk management purposes, which, in turn, will promote competition and further the development of CDS for risk management.

In addition, LCH SA will continue to apply its existing fair and open access criteria to the clearing of these additional products and will apply the same criteria to every clearing member or client who proposes to enter into this clearing activity.

Accordingly, LCH SA does not believe that the Proposed Rule Change would impose any burden on competition that is not necessary or appropriate in furtherance of the purposes of the Act.

C. Clearing Agency's Statement on Comments on the Proposed Rule Change Received from Members, Participants or Others

²¹ 17 CFR 240. 17Ad-22(e)(6)(iii).

²² 15 U.S.C. 78q-1(b)(3)(I).

Written comments relating to the proposed rule change have not been solicited or received. LCH SA will notify the Commission of any written comments received by LCH SA.

III. Date of Effectiveness of the Proposed Rule Change and Timing for Commission Action

Within 45 days of the date of publication of this notice in the Federal Register or within such longer period up to 90 days (i) as the Commission may designate if it finds such longer period to be appropriate and publishes its reasons for so finding or (ii) as to which the self-regulatory organization consents, the Commission will:

- (A) by order approve or disapprove such proposed rule change, or
- (B) institute proceedings to determine whether the proposed rule change should be disapproved.

IV. Solicitation of Comments

Interested persons are invited to submit written data, views, and arguments concerning the foregoing, including whether the proposed rule change is consistent with the Act. Comments may be submitted by any of the following methods:

Electronic Comments:

- Use the Commission's Internet comment form (<u>http://www.sec.gov/rules/sro.shtml</u>); or
- Send an e-mail to <u>rule-comments@sec.gov</u>. <u>Please include File Number SR-LCH</u>
 <u>SA-2022-007 on the subject line</u>.

Paper Comments:

Send paper comments in triplicate to Secretary, Securities and Exchange
 Commission, 100 F Street, NE, Washington, DC 20549-1090.

All submissions should refer to File Number SR-LCH SA-2022-007. This file number should be included on the subject line if e-mail is used. To help the Commission process and review your comments more efficiently, please use only one method. The Commission will post all comments on the Commission's Internet website (http://www.sec.gov/rules/sro.shtml). Copies of the submission, all subsequent amendments, all written statements with respect to the proposed rule change that are filed with the Commission, and all written communications relating to the proposed rule change between the Commission and any person, other than those that may be withheld from the public in accordance with the provisions of 5 U.S.C. 552, will be available for website viewing and printing in the Commission's Public Reference Room, 100 F Street, NE, Washington, DC 20549 on official business days between the hours of 10:00 am and 3:00 pm. Copies of the filing also will be available for inspection and copying at the principal office of LCH SA and on LCH SA's website at: https://www.lch.com/resources/rulebooks/proposed-rule-changes. All comments received will be posted without change. Persons submitting comments are cautioned that we do not redact or edit personal identifying information from comment submissions. You should submit only information that you wish to make available publicly. All submissions should refer to File Number SR-LCH SA-2022-007 and should be submitted on or before

[Commission to insert date 21 days from publication in the <u>Federal Register</u>].

For the Commission, by the Division of Trading and Markets, pursuant to delegated authority. 23

Secretary

²³ 17

File No. SR-LCH SA-2022-007 Page 36 of 439

Exhibit 5a

Formatted: Font: Times New Roman
Formatted: Font: Times New Roman
Formatted: Space Before: 0 pt
Formatted: Font: Times New Roman
Formatted: Font: Times New Roman
Formatted: Font: Times New Roman, 10 pt

Style Definition: Schedule 6: Indent: Left: 1.9 cm, Hanging: 0.63 cm, Tab stops: Not at 2.54 cm



CDSClear

LCH SA CDS Clearing Supplement

[•]

Classification: Confidential

File No. SR-LCH SA-2022-007 Page 37 of 439

PUBLIC

This document is for use with the clearing of index linked credit derivative transactions, single name credit derivative transactions and swaption transactions in respect of index linked credit derivative transactions. The CDS Clearing Supplement is split into three parts, Part A, Part B and Part C.

Part A of the CDS Clearing Supplement shall only be used in connection with any single name credit derivative transactions or components of index linked credit derivative transactions that incorporate the 2003 ISDA Credit Derivatives Definitions, as published by the International Swaps and Derivatives Association, Inc.

Part B of the CDS Clearing Supplement shall only be used in connection with any single name credit derivative transactions or components of index linked credit derivative transactions that incorporate the 2014 ISDA Credit Derivatives Definitions, as published by the International Swaps and Derivatives Association, Inc.

Part C of the CDS Clearing Supplement shall only be used in connection with any swaption transactions, which give to a buyer the right to enter into an index linked credit derivative transaction with a seller, that incorporate the 2006 ISDA Definitions and the 2014 ISDA Credit Derivatives Definitions, as published by the International Swaps and Derivatives Association, Inc.

Classification: Confidential

File No. SR-LCH SA-2022-007 Page 38 of 439

PUBLIC

PART A

CDS CLEARING SUPPLEMENT

FOR INDEX CLEARED TRANSACTIONS AND SINGLE NAME TRANSACTIONS INCORPORATING THE 2003 ISDA CREDIT DERIVATIVES DEFINITIONS

CONTENTS

SECTION		PAGE	
1	GENERAL PROVISIONS	38	Formatted: English (United Kingdom), Not All caps
			Formatted: Not All caps
2.	TERMS OF CLEARED TRANSACTIONS	49	Formatted: Not All caps
3.	PAYMENTS AND DELIVERIES	57	Formatted: Not All caps
4.	CREDIT EVENTS, SUCCESSION EVENTS AND RENAME EVENTS	58	Formatted: Not All caps
5.	RESTRUCTURING	62	Formatted: Not All caps
6.	PHYSICAL SETTLEMENT	64	Formatted: Not All caps
7.	DELIVERY OF NOTICES AND FALLBACKS	76	Formatted: Not All caps
8.	MATCHED PAIR DESIGNATIONS AND NOTICES	85	Formatted: Not All caps
9.	SELF-REFERENCING TRANSACTIONS	89	Formatted: Not All caps
10.	MANDATORY PROVISIONS FOR CCM CLIENT TRANSACTIONS	93	Formatted: Not All caps
11.	AMENDMENTS	93	Formatted: Not All caps
12.	FORMS OF NOTICES	93	Formatted: Not All caps
13.	EXCLUSION OF LIABILITY	94	Formatted: Not All caps
14.	DISPUTE RESOLUTION	94	Formatted: Not All caps
15	GOVERNING I AW	95	Formatted: Not All cans

Classification: Confidential

1. GENERAL PROVISIONS

1.1 Incorporation of Defined Terms

Capitalised terms used in this CDS Clearing Supplement and not otherwise defined herein shall have the meaning given pursuant to the Index Cleared Transaction Confirmation, Single Name Cleared Transaction Confirmation, the 2003 ISDA Credit Derivatives Definitions or the CDS Clearing Rule Book, as applicable. In the case of any such terms defined in the CDS Clearing Rule Book, such terms shall be interpreted in accordance with the governing law specified therefore in the CDS Clearing Rule Book.

1.2 Terms defined in the CDS Clearing Supplement

For the purposes of the CDS Clearing Documentation, the following capitalised terms shall, unless otherwise specified, have the respective meanings set out below:

2003 ISDA Credit Derivatives Definitions: The 2003 ISDA Credit Derivatives Definitions published by ISDA as supplemented by the 2009 ISDA Credit Derivatives Determinations Committees, Auction Settlement and Restructuring Supplement to the 2003 ISDA Credit Derivatives Definitions, including the DC Rules and Credit Derivatives Auction Settlement Terms (each as defined therein).

2014 ISDA Credit Derivatives Definitions: The 2014 ISDA Credit Derivatives Definitions published by ISDA.

Affected Cleared Transaction: Following a DC Credit Event Announcement or a publication by the DC Secretary of:

- (a) a Succession Event Resolution; or
- (b) a DC Resolution that a Substitute Reference Obligation has been determined,

a Cleared Transaction that references the affected Reference Entity and/or Reference Obligation, as applicable.

Buy-in Effective Date: As set out at Section 6.8 (*Buy-in of Bonds – Matched Seller has entered into CCM Client Transaction*).

CCM Client Cleared Transaction: A Cleared Transaction between a CCM and LCH SA registered in a CCM Client Trade Account of a CCM.

CCM Client Transaction: A Transaction between a CCM and a CCM Client which is on the same economic terms as its corresponding CCM Client Cleared Transaction.

CCM Client Transaction Documents: The documentation entered into by a CCM and its CCM Client to document a CCM Client Transaction.

CDSClear Preferred Reference Obligation This term shall have the meaning set out in Section 4 of the Procedures.

Classification: Confidential

File No. SR-LCH SA-2022-007 Page 39 of 439

PUBLIC

CDSClear Product Committee: A committee composed of representatives of LCH SA and representatives of Clearing Members for the purposes of carrying-out the tasks apportioned to it by the CDS Clearing Documentation as further described in the terms of reference for such committee, agreed in consultation with the Clearing Members and as may be amended from time to time in consultation with the Clearing Members.

CDS Clearing Rule Book: The document entitled "CDS Clearing Rule Book" published by LCH SA, as amended from time to time.

CDS Type: A class of Cleared Transactions that are identical as to their terms, except that they may differ as to:

- (a) the Trade Date;
- (b) in respect of Index Cleared Transactions, the Original Notional Amount;
- (c) in respect of Single Name Cleared Transactions, the Effective Date (provided that the current and future Fixed Rate Payer Calculation Periods for such Single Name Cleared Transactions are the same), the Floating Rate Payer Calculation Amount and the Reference Obligation (provided that LCH SA determines that the Reference Obligations of the Cleared Transactions are equivalent for the purposes of this CDS Clearing Supplement);
- (d) the Initial Payment Payer;
- (e) the Initial Payment Amount;
- (f) the Initial Payment Date; and
- (g) the identity of the relevant Buyer and Seller.

CEN Triggering Period: In relation to any Restructuring Cleared Transaction, the period during which the parties thereto may deliver a Restructuring Credit Event Notice in relation to all or part of such Restructuring Cleared Transaction, such period starting at 9:00 a.m. on and including the earlier to occur of:

- (a) the Transaction Business Day following publication of the related Final List; and
- (b) the tenth calendar day following the No Auction Announcement Date,

and ending on and including the Exercise Cut-off Date.

Clearing Member Acknowledgement: As set out at Section 7.7 (*Clearing Member Acknowledgements*).

Clearing Member Communications Failure Event: As set out at Section 7.4 (*Notification of DTCC Failure and Resolution*).

File No. SR-LCH SA-2022-007 Page 40 of 439

PUBLIC

Clearing Member Self Referencing Transaction: A Single Name Cleared Transaction which is registered in the House Account Structure of the Clearing Member and in respect of which, the Reference Entity is either the relevant Clearing Member or an Affiliate of such Clearing Member.

Client Self Referencing Transaction: A Single Name Cleared Transaction (a) that is registered in the Client Account Structure of a Clearing Member; and (b) in respect of which, the Reference Entity is either the relevant Client or an Affiliate of such Client.

Compression Cut-off Date: The last date on which a Clearing Member may submit a request for any Cleared Transaction to be compressed pursuant to the ad hoc compression methodology and on which a daily automatic compression cycle will be run by LCH SA, in each case in accordance with Chapter 3 (*Compression*) of Title III (*Clearing Operations*) of the CDS Clearing Rule Book and Section 5 of the Procedures, being:

- in respect of any Index Cleared Transaction, the date falling one Transaction Business
 Day prior to the Novation Cut-off Date in respect of the relevant Eligible Index Version;
- (b) in respect of any Single Name Cleared Transaction and:
 - (i) a Restructuring Credit Event, the earlier of (A) the date of publication of the relevant Initial List (as defined in the DC Rules), (B) the date falling two Transaction Business Days prior to the relevant date on which the related RMP Notification Deadline falls and (C) such other date falling between the dates in (A) and (B), as determined by LCH SA in consultation with the CDSClear Product Committee and notified to the relevant Clearing Members prior to such date;
 - (ii) a Failure to Pay Credit Event or a Bankruptcy Credit Event, the date falling one Transaction Business Day prior to the calendar day following the related Auction Final Price Determination Date, Auction Cancellation Date or No Auction Announcement Date;
 - (iii) a Succession Event, a date determined by LCH SA in consultation with the CDSClear Product Committee, which shall be not later that the Transaction Business Day before the date on which DTCC will amend its records in respect of such Single Name Cleared Transaction to take into account the occurrence of such Succession Event, and notified to the relevant Clearing Members prior to such date; and
 - (iv) a Rename Event, the Transaction Business Day before the date on which DTCC will amend its records in respect of Single Name Cleared Transactions to take into account the occurrence of such Rename Event.

For the avoidance of doubt, where a Clearing Member has specified automatic compression on a weekly basis then the last such weekly automatic compression cycle performed by LCH

40

File No. SR-LCH SA-2022-007 Page 41 of 439

PUBLIC

SA will be performed on the Clearing Day falling on the Thursday on or before the relevant Compression Cut-off Date.

DC Restructuring Announcement Date: The date on which the DC Credit Event Announcement of a Restructuring Credit Event is made, *provided that* where such DC Credit Event Announcement is made after 6.30 p.m. on a Business Day or on a day which is not a Business Day, the DC Restructuring Announcement Date will be the first following Business Day.

DC Rules: This term shall have the meaning given to the term "Rules" in the 2003 ISDA Credit Derivatives Definitions.

DC Secretary: This term shall have the meaning set out in the DC Rules.

Deemed Buy-in Period: As set out at Section 6.8(b)(i).

Dispute: This term shall have the meaning set out in the CDS Dispute Resolution Protocol.

DTCC Failure Event: As set out at Section 7.4 (Notification of DTCC Failure and Resolution).

DTCC Failure Event Time: As set out at Section 7.4 (*Notification of DTCC Failure and Resolution*).

DTCC Notice Facility: A facility made available pursuant to the DTCC Rules for the delivery of Credit Event Notices relating to Restructuring Credit Events or Notices to Exercise Movement Ontion

DTCC Resolution Time: As set out at Section 7.4 (Notification of DTCC Failure and Resolution).

DTCC Rules: The "Operating Procedures", as published by DTCC and as amended from time to time

Failed Amount: As set out at Section 6.10 (Failure to pay Physical Settlement Amount).

First Novation Date: In respect of:

- (a) an Index Cleared Transaction, the first date on which LCH SA will accept Original Transactions referencing the relevant Eligible Index Version for clearing pursuant to the CDS Clearing Documentation, being, in the case of any such version published pursuant to the occurrence of a Succession Event or Credit Event, the date on which credit default swap transactions referencing such version of such index are accepted for registration in the TIW in accordance with the DTCC Rules;
- (b) a Single Name Cleared Transaction in respect of which a Novation Cut-off Date has previously occurred pursuant to the occurrence of a Restructuring Credit Event in respect of the relevant Reference Entity, the calendar day following the Transaction Business Day following the latest possible Exercise Cut-off Date for the relevant Restructuring Credit Event or such other date on which LCH SA determines in

- consultation with the CDSClear Product Committee that LCH SA will begin to again accept Original Transactions referencing the relevant Reference Entity; and
- (c) a Single Name Cleared Transaction (other than in the circumstances set-out in (b) above), the first date on which LCH SA determines in consultation with the CDSClear Product Committee that LCH SA will accept or will begin to again accept (as applicable) Original Transactions referencing the relevant Reference Entity.

Index Cleared Transaction: A Cleared Transaction which references a portfolio of Reference Entities specified in a credit default swap index and consists of a Component Transaction (as defined in the Index Cleared Transaction Confirmation) in respect of each such Reference Entity, the terms of which are as evidenced by an Index Cleared Transaction Confirmation.

Index Cleared Transaction Confirmation: For

- (a) any Index Cleared Transaction which references a Markit iTraxx® Europe Index Series 21 or below, the form of confirmation which incorporates the iTraxx® Europe Legacy Untranched Standard Terms Supplement, each as published on 20 September 2014 by Markit Group Limited; and
- (b) any Index Cleared Transaction which references a Markit CDX™ Index Series 22 or below, the form of confirmation which incorporates the CDX Legacy Untranched Transactions Standard Terms Supplement, each as published on 22 September 2014 by Markit Group Limited,

in each case as amended by this CDS Clearing Supplement and as completed by reference to the relevant Transaction Data (or such other form of confirmation as may be adopted in respect of any CDS Type in accordance with Section 1.2.2 (*Modification*) of the CDS Clearing Rule Book).

Index CCM Client Transaction: A CCM Client Transaction which references a portfolio of Reference Entities specified in a credit default swap index and consists of a Component Transaction (as defined in the Index Cleared Transaction Confirmation) in respect of each such Reference Entity and which is on the same economic terms as an Index Cleared Transaction registered in the CCM Client Trade Account in the name of such CCM for such CCM Client.

Initial Re-couponing Notice: As set-out in Section 4.4 (*Re-couponing of Restructuring Cleared Transactions*).

Initial Single Name Cleared Transaction: A Cleared Transaction entered into following the novation of an Original Transaction and which references a single Reference Entity, the terms of which are evidenced by a Single Name Cleared Transaction Confirmation.

Initial Single Name CCM Client Transaction: A CCM Client Transaction which references a single Reference Entity and which is on the same economic terms as an Initial Single Name

File No. SR-LCH SA-2022-007 Page 43 of 439

PUBLIC

Cleared Transaction registered in the CCM Client Trade Account in the name of such CCM for such CCM Client.

ISDA: The International Swaps and Derivatives Association, Inc. and any successor thereto.

Mandatory Provisions: As set-out in Appendix XIII (CCM Client Transaction Requirements).

Matched Buyer: A CDS Buyer comprised in a Matched Pair.

Matched Buyer Contract: A Cleared Transaction (or part thereof) between a Matched Buyer and LCH SA which is the subject of a Matched Pair having the Restructuring Matched Pair Amount or the Settlement Matched Pair Delivery Amount, as the case may be, relating to that Matched Pair.

Matched Contract: A Matched Seller Contract or a Matched Buyer Contract, as applicable.

Matched Pair: A Restructuring Matched Pair or Settlement Matched Pair, as applicable.

Matched Seller: A CDS Seller comprised in a Matched Pair.

Matched Seller Contract: A Cleared Transaction (or part thereof) between a Matched Seller and LCH SA which is the subject of a Matched Pair having the Restructuring Matched Pair Amount or the Settlement Matched Pair Delivery Amount, as the case may be, relating to that Matched Pair.

Matching Information Notification Deadline: In respect of a Restructuring Cleared Transaction, 9.00 a.m. on the first day of the CEN Triggering Period, *provided that* the Matching Information Notification Deadline shall fall no earlier than the fifth Transaction Business Day following the relevant DC Credit Event Announcement.

Matrix Re-versioning Date: As set-out in Section 2.5 (Physical Settlement Matrix Updates).

NEMO Triggering Period: In relation to any Restructuring Cleared Transaction for which either "Restructuring Maturity Limitation and Fully Transferable Obligation Applicable" or "Modified Restructuring Maturity Limitation and Conditionally Transferable Obligation Applicable" is applicable, the period starting at 9:00 a.m. on the day falling two Transaction Business Days prior to the relevant Movement Option Cut-off Date for the relevant Credit Event and ending at the last time for delivery of a valid Notice to Exercise Movement Option under the terms of the relevant Cleared Transaction.

No Physical Settlement Confirmation: As set out in Section 6.19 (*Miscellaneous Provisions relating to Physical Settlement*).

No Physical Settlement Confirmation Deadline: As set out in Section 6.19 (*Miscellaneous Provisions relating to Physical Settlement*).

Non-Deliverable Obligation: As set out at Section 6.12 (*Fallback to Cash Settlement in respect of Non-Deliverable Obligations*).

File No. SR-LCH SA-2022-007 Page 44 of 439

PUBLIC

Non-DVP Obligation: In respect of any Physically Settled Cleared Transaction, a Deliverable Obligation validly specified in a Notice of Physical Settlement which does not, in accordance with then current market standards, settle on a delivery-versus-payment basis through the books and records of a central securities depository or securities settlement system.

Notice Acknowledgement Deadline: As set out in Section 7.7 (*Clearing Member Acknowledgements*).

Notice Reconciliation Deadline: As set out in Section 7.7 (*Clearing Member Acknowledgements*).

Novation Cut-off Date: The date with effect from which LCH SA will no longer accept Original Transactions referencing an Eligible Index Version or Eligible Reference Entity for novation, being:

- (a) following the occurrence of a Restructuring Credit Event, the earlier of:
 - a date determined by LCH SA in consultation with the CDSClear Product Committee, which shall not be earlier than the DC Restructuring Announcement Date; and
 - (ii) the calendar day immediately following the No Auction Announcement Date,

provided that LCH SA in consultation with the CDSClear Product Committee may, but is not required to, determine that the Novation Cut-off Date for Index Cleared Transactions and the Novation Cut-off Date for Initial Single Name Cleared Transactions are to occur on different days;

- (b) following the occurrence of a Failure to Pay Credit Event or a Bankruptcy Credit Event, the calendar day following the related Auction Final Price Determination Date, Auction Cancellation Date or No Auction Announcement Date, as applicable;
- (c) following the occurrence of a Succession Event:
 - (i) in respect of an Index Cleared Transaction, the date notified by DTCC to LCH SA as being the date on which DTCC will amend its records in respect of such Cleared Transaction to take into account a new version of such index published by the relevant index publisher taking into account the occurrence of such Succession Event; or
 - (ii) in respect of a Single Name Cleared Transaction, the day after the date on which ISDA publicly announces that the relevant Determinations Committee has resolved that a Succession Event has occurred;
- (d) following the occurrence of a Rename Event in respect of Single Name Cleared Transactions, the date determined by LCH SA in consultation with the CDSClear Product Committee, which such date shall not be later than the date on which DTCC

File No. SR-LCH SA-2022-007 Page 45 of 439

PUBLIC

amends its records in respect of Single Name Cleared Transactions to take into account the occurrence of such Rename Event; or

(e) otherwise, as determined by LCH SA in consultation with the CDSClear Product Committee.

Partial Cash Settlement Terms: As set out in Section 9.8 (*Partial Cash Settlement Terms*) of the 2003 ISDA Credit Derivatives Definitions, as amended by this CDS Clearing Supplement.

Payer: As set out at Section 6.17 (Physical Settlement Costs).

Physical Settlement Confirmation: As set out in Section 6.19 (*Miscellaneous Provisions relating to Physical Settlement*).

Physical Settlement Confirmation Deadline: As set out in Section 6.19 (*Miscellaneous Provisions relating to Physical Settlement*).

Physically Settled Cleared Transaction: As set out at Section 6.3 (*Physically Settled Cleared Transactions*).

Physical Settlement Matrix: The Credit Derivatives Physical Settlement Matrix (as defined in the 2005 Matrix Supplement to the 2003 ISDA Credit Derivatives Definitions published on March 7 2005).

Recipient: As set out at Section 6.17 (*Physical Settlement Costs*).

Re-couponing Date: As set-out in Section 4.4 (*Re-couponing of Restructuring Cleared Transactions*).

Re-couponing Notice: As set-out in Section 4.4 (*Re-couponing of Restructuring Cleared Transactions*).

Rename Event: As set out in Section 4.7 (Rename Events).

Restructuring Cleared Transaction: A Cleared Transaction which references a single Reference Entity and is created following the creation of Restructuring Matched Pairs pursuant to Section 5.2 (*Creation of Restructuring Cleared Transactions*), the terms of which are evidenced by a Single Name Cleared Transaction Confirmation.

Restructuring CCM Client Transaction: A CCM Client Transaction between a CCM and a CCM Client which references a single Reference Entity and which is on the same economic terms as a Restructuring Cleared Transaction registered in the CCM Client Trade Account in the name of such CCM for such CCM Client.

Restructuring Matched Pair: A pairing comprised of a Matched Buyer and a Matched Seller in respect of Restructuring Cleared Transactions created by LCH SA for the purposes of delivery of Credit Event Notices in respect of the relevant Restructuring Credit Event under Section 5.1 (*Creation and Notification of Restructuring Matched Pairs*).

File No. SR-LCH SA-2022-007 Page 46 of 439

PUBLIC

Restructuring Matched Pair Amount: In respect of a Restructuring Matched Pair, the amount of the Floating Rate Payer Calculation Amount in respect of the relevant Restructuring Cleared Transactions allocated by LCH SA to such Restructuring Matched Pair under Section 8.1 (*Creation of Matched Pairs*).

Resulting Single Name Cleared Transaction: A Cleared Transaction which references a single Reference Entity and is created following the termination of a Restructuring Cleared Transaction pursuant to Section 5.15 (*Recouponing*) of the Procedures, the terms of which are evidenced by a Single Name Cleared Transaction Confirmation.

RMP Notification Deadline: In respect of a notification by LCH SA to the relevant Clearing Members in relation to Restructuring Matched Pairs, 10.00 a.m. on the Transaction Business Day immediately prior to the first day of the CEN Triggering Period *provided that* the RMP Notification Deadline shall in no event fall prior to the second Transaction Business Day following the occurrence of the related DC Credit Event Announcement.

Self Referencing Transaction: A Clearing Member Self Referencing Transaction or a Client Self Referencing Transaction, as applicable.

Settlement Matched Pair: A pairing comprised of a Matched Buyer and a Matched Seller in respect of Physically Settled Cleared Transactions deemed to have been created by LCH SA under Section 6.2 (*Creation and Notification of Settlement Matched Pairs*).

Settlement Matched Pair Delivery Amount: In respect of a Settlement Matched Pair, the amount of the Floating Rate Payer Calculation Amount in respect of the relevant Physically Settled Cleared Transactions allocated by LCH SA to such Settlement Matched Pair under Section 8.1 (*Creation of Matched Pairs*).

Single Name Cleared Transaction: An Initial Single Name Cleared Transaction, a Spin-off Single Name Cleared Transaction, a Restructuring Cleared Transaction and a Resulting Single Name Cleared Transaction.

Single Name Cleared Transaction Confirmation: The form of confirmation for use with the Physical Settlement Matrix that incorporates the 2003 ISDA Credit Derivatives Definitions, as amended by this CDS Clearing Supplement and as completed by reference to:

- (a) in the case of an Initial Single Name Cleared Transaction, the relevant Transaction Data; or
- (b) in the case of any other type of Single Name Cleared Transaction, the relevant Cleared Transaction or Cleared Transactions that existed immediately prior to the relevant event that resulted in the creation of such Single Name Cleared Transaction (with such amendments as are required pursuant to the terms of the CDS Clearing Documentation),

or such other form of confirmation as may be adopted in respect of any CDS Type in accordance with Section 1.2.2 (*Modification*) of the CDS Clearing Rule Book.

File No. SR-LCH SA-2022-007 Page 47 of 439

PUBLIC

Single Name CCM Client Transaction: A CCM Client Transaction between a CCM and a CCM Client which is on the same economic terms as a Single Name Cleared Transaction registered in the CCM Client Trade Account in the name of such CCM for such CCM Client.

SMP Notification Deadline: In respect of a notification by LCH SA to the relevant Clearing Members in relation to Settlement Matched Pairs, noon on the Transaction Business Day following the day on which the Fallback Settlement Method first becomes applicable.

Spin-off Single Name Cleared Transaction: The separate Single Name Cleared Transaction formed in respect of a Reference Entity from a Component Transaction (as defined in the Index Cleared Transaction Confirmation) of an Index Cleared Transaction following the occurrence of a DC Credit Event Announcement in respect of a Restructuring Credit Event with respect to the Reference Entity of such Component Transaction in accordance with the terms of such Index Cleared Transaction, the terms of which are evidenced by a Single Name Cleared Transaction Confirmation.

Transaction Business Day: A "Business Day", as defined in the Index Cleared Transaction Confirmation or the Single Name Cleared Transaction Confirmation, as applicable.

Transaction Data: In respect of an Original Transaction to be novated pursuant to Title III (*Clearing Operations*) of the CDS Clearing Rule Book and cleared by LCH SA as an Index Cleared Transaction or Initial Single Name Cleared Transaction, the data provided by an Approved Trade Source System to LCH SA for such purposes, which includes, without limitation:

- in respect of an Index Cleared Transaction, the relevant index, including details of the index name, series and version, the annex date, the Original Notional Amount and the currency of the Original Notional Amount;
- (b) in respect of an Initial Single Name Cleared Transaction, the Reference Entity, Reference Obligation, applicable Transaction Type, Floating Rate Payer Calculation Amount, the currency of the Floating Rate Payer Calculation Amount and the Fixed Rate;
- (c) the Trade Date;
- (d) the Scheduled Termination Date;
- (e) the Floating Rate Payer;
- (f) the Fixed Rate Payer;
- (g) the Fixed Rate Payer Payment Dates;
- (h) the Initial Payment Payer;
- (i) the Initial Payment Amount; and

(j) whether the 2003 Credit Derivatives Definitions or the 2014 Credit Derivatives Definitions are applicable to such Original Transaction (or, in the case of an Index Cleared Transaction, each component of such Original Transaction).

1.3 Inconsistency

To the extent of any conflict between:

- any definition or provision contained in Appendix 1 (CDS Default Management Process) of the CDS Clearing Rule Book;
- (b) the remaining sections of the CDS Clearing Rule Book;
- (c) the CDS Admission Agreement;
- (d) this CDS Clearing Supplement;
- (e) an Index Cleared Transaction Confirmation or a Single Name Cleared Transaction Confirmation;
- (f) the Procedures; or
- (g) any Clearing Notices,

the first referenced document shall prevail except in relation to determining the existence and amount of any payment and delivery obligations under any Cleared Transactions, in respect of which this CDS Clearing Supplement, the Index Cleared Transaction Confirmation or the Single Name Cleared Transaction Confirmation, as applicable, shall prevail to the extent permitted by law.

1.4 Timing

Pursuant to Article 1.2.8 (*Time reference*) of the CDS Clearing Rule Book, any reference to a time of day herein shall be deemed to be a reference to Central European Time unless otherwise provided herein.

1.5 Third Party Rights

Unless otherwise provided in this CDS Clearing Supplement or in the CDS Clearing Rulebook, a person who is not a party to a Cleared Transaction does not have any right under the Contracts (Rights of Third Parties) Act 1999 to enforce any provision of such Cleared Transaction or this CDS Clearing Supplement as it relates to such Cleared Transaction.

1.6 Recording of Conversations

Each of LCH SA and each Clearing Member consents to the recording of telephone conversations between the trading, marketing and other relevant personnel of the parties in connection with any Cleared Transaction and agrees to obtain any necessary consent of, and give any necessary notice of such recording to, its relevant personnel and agrees, to the

extent permitted by applicable law, that such recordings may be submitted as evidence in any related court or arbitral proceedings.

1.7 Application to FCM/BD Clearing Members

- (a) Upon the taking of any action pursuant to this CDS Clearing Supplement by an FCM/BD Clearing Member in respect of an FCM/BD Cleared Transaction entered into as agent for the account of an FCM/BD Client (as described in FCM/BD CDS Clearing Regulation 1(c)), such FCM/BD Clearing Member shall be deemed to represent to LCH SA that it has the power and authority to, and has been duly authorised to, take such action for the account of such FCM/BD Client.
- (b) For purposes of this CDS Clearing Supplement, with regard to any Cleared Transaction entered into by an FCM/BD Clearing Member acting as agent for the account of an FCM/BD Client (as described in FCM/BD CDS Clearing Regulation 1(c)):
 - (i) references herein to "CDS Buyer" or "CDS Seller", as the case may be, shall be understood to be references to such FCM/BD Client (which shall not, for the avoidance of doubt, prejudice LCH SA's right to deal solely with the FCM/BD Clearing Member pursuant to Article 6.1.1.3(vi) of the CDS Clearing Rule Book);
 - (ii) references herein to: (1) a Clearing Member entering into a Cleared Transaction with LCH SA; and (2) a Clearing Member forming part of a Matched Pair, shall each be understood as such FCM/BD Clearing Member acting as agent for the account of such FCM/BD Client (as described in FCM/BD CDS Clearing Regulation 1(c)).

2. TERMS OF CLEARED TRANSACTIONS

2.1 General Terms of Cleared Transactions

(a) Terms of Index Cleared Transactions and Initial Single Name Cleared Transactions

Upon novation of an Original Transaction at the Novation Time in accordance with Title III (*Clearing Operations*) of the CDS Clearing Rule Book, each resulting Index Cleared Transaction or Initial Single Name Cleared Transaction is entered into by LCH SA and the relevant Clearing Member on the terms of the related Index Cleared Transaction Confirmation or Single Name Cleared Transaction Confirmation (as applicable).

(b) Terms of Spin-off Single Name Cleared Transactions, Restructuring Cleared Transactions and Resulting Single Name Cleared Transactions

If any Spin-off Single Name Cleared Transaction is deemed to have been entered into by the parties in accordance with the terms of any Index Cleared Transaction or if any Restructuring Cleared Transaction is created pursuant to Section 5.2 (*Creation of Restructuring Cleared Transactions*) or if any Resulting Single Name Cleared

Transaction is created pursuant to Section 5.15 (*Recouponing*) of the Procedures, such Spin-off Single Name Cleared Transaction, Restructuring Cleared Transaction or Resulting Single Name Cleared Transaction is entered into by LCH SA and the relevant Clearing Member on the terms of the related Single Name Cleared Transaction Confirmation.

(c) Trade Date of Cleared Transactions following Compression

Notwithstanding paragraphs (a) and (b) above, where Cleared Transactions are subject to compression in accordance with Chapter 3 (*Compression*) of Title III (*Clearing Operations*) of the CDS Clearing Rule Book, the Trade Date of any resulting Cleared Transaction(s) shall be, in respect of any Cleared Transaction subject to:

- (i) ad hoc compression (as described in Chapter 3 (Compression) of Title III (Clearing Operations) of the CDS Clearing Rule Book), the date on which the request for compression was effectively received and processed in accordance with Section 5 of the Procedures, which shall be:
 - (A) the Clearing Day on which such request is submitted and uploaded by the relevant Clearing Member provided that such request for compression was received by LCH SA before 7.00 p.m. on such Clearing Day (if such request is submitted and uploaded by the relevant Clearing member via any means of access specified in a Clearing Notice or 5.00 p.m. (if such request is not submitted via any means of access specified in a Clearing Notice in the case of a disruption of the relevant means of access):
 - (B) the Clearing Day on which such request is submitted if such request is not submitted via any means of access specified in a Clearing Notice in the case of disruption of the relevant means of access and is submitted after 5.00 p.m. but LCH SA, in its sole discretion, processes such request on the Clearing Day on which such request is submitted; and
 - (C) unless the relevant Clearing Member instructs the Operations Department to withdraw such request, the Clearing Day following the Clearing Day on which such request is submitted if such request is not submitted via any means of access specified in a Clearing Notice in the case of disruption of the relevant means of access and is submitted after 5.00 p.m. and LCH SA, in its sole discretion, does not process such request on the Clearing Day on which such request is submitted; or
- (ii) automatic compression (as described in Chapter 3 (Compression) of Title III (Clearing Operations) of the CDS Clearing Rule Book), the Clearing Day on which such Cleared Transaction is automatically compressed by LCH SA in accordance with Section 5 of the Procedures.

2.2 Index Cleared Transaction Confirmation

The Index Cleared Transaction Confirmation is amended, supplemented and completed as follows:

- (i) if the Index Cleared Transaction references a Markit iTraxx® Europe Index, by deleting the words "between [●] (Party A) and [●] (Party B)" in the third line of the first paragraph; and
 - (ii) if the Index Cleared Transaction references a Markit CDX™ Index, by deleting the words "between [Party A] ("Party A") and [counterparty's name] ("Party B")" in the third and fourth lines of the first paragraph,

and in each case replacing them with:

- "between LCH SA ("Party A") and Clearing Member, as identified in the relevant CDS Admission Agreement between such Clearing Member and Party A ("Party B")";
- (i) if the Index Cleared Transaction references a Markit iTraxx® Europe Index, by deleting the fifth paragraph thereof; and
 - (ii) if the Index Cleared Transaction references a Markit $CDX^{\text{\tiny{IM}}}$ Index, by deleting the third paragraph thereof,

in each case replacing it with the following:

- "This Confirmation supplements, forms a part of, and is subject to, the CDS Clearing Documentation, as defined in the CDS Clearing Rule Book.";
- (f) if the Index Cleared Transaction references a Markit iTraxx® Europe Index, by deleting the sixth paragraph thereof and replacing it with the following:
 - "The terms of the iTraxx® Master Transaction, which is an Index Cleared Transaction, to which this Confirmation relates are as follows:";
- (g) if the Index Cleared Transaction references a Markit CDX™ Index, by deleting the fourth paragraph thereof and replacing it with the following:
 - "The terms of the Master Transaction, which is an Index Cleared Transaction, to which this Confirmation relates are as follows:";
- (h) by specifying that the "Calculation Agent" is Party A;
- (i) (i) if the Index Cleared Transaction references a Markit iTraxx® Europe Index, by inserting the following "Additional terms":
 - "The Legacy Standard Terms Supplement is amended for the purposes of this Transaction by deleting paragraph 5.4 (*De Minimis Cash Settlement*) of Section A in its entirety": and

(ii) if the Index Cleared Transaction references a Markit CDX™ Index, by inserting the following "Additional terms":

"The CDX Legacy Standard Terms Supplement is amended for the purposes of this Transaction by deleting paragraph 5.3 (*De Minimis Cash Settlement*) of Section A in its entirety":

- (j) by deleting the contact details for notices and the account details; and
- (k) by deleting the signature blocks.

2.3 Single Name Cleared Transaction Confirmation

The Single Name Cleared Transaction Confirmation is amended, supplemented and completed as follows:

- (a) by deleting the words "between us" from the first paragraph thereof and replacing them with:
 - "between LCH SA ("Party A") and Clearing Member, as identified in the relevant CDS Admission Agreement between Clearing Member and Party A ("Party B")";
- (b) by deleting the words "as supplemented by each of the May 2003 Supplement and the 2005 Matrix Supplement to the 2003 ISDA Credit Derivatives Definitions" in the second paragraph thereof and replacing them with the words "as supplemented by each of the 2005 Matrix Supplement to the 2003 ISDA Credit Derivatives Definitions and the 2009 ISDA Credit Derivatives Determinations Committees, Auction Settlement and Restructuring Supplement to the 2003 ISDA Credit Derivatives Definitions (published on July 14, 2009)";
- (c) by deleting the third paragraph thereof and replacing it with the following:
 - "This Confirmation supplements, forms a part of and is subject to the CDS Clearing Documentation, as defined in the CDS Clearing Rule Book.";
- (d) by specifying the Matrix Publication Date as the date of publication of the Relevant Physical Settlement Matrix;
- (e) by specifying that the "Calculation Agent" is Party A;
- notwithstanding the terms of the relevant Original Transaction, by specifying that the Reference Obligation is the CDSClear Preferred Reference Obligation;
- (g) by deleting references to the Initial Payment Payer and the Initial Payment Amount in part 2 in relation to Single Name Cleared Transactions in respect of which there is no Initial Payment Payer and Initial Payment Amount;
- (h) by deleting parts 4, 5, 6 and 7 in their entirety;

52

- (i) by inserting the following provision in the "Additional Terms" section of the Confirmation:
 - "Section 2.31 (Merger of Reference Entity and Seller) of the Credit Derivatives Definitions shall not apply."; and
- (j) by deleting the signature blocks.

2.4 Amendments to 2003 ISDA Credit Derivatives Definitions

- (a) For the purposes of this CDS Clearing Supplement, Section 9.2(c)(iv) of the 2003 ISDA Credit Derivatives Definitions as incorporated in any Cleared Transaction shall be amended such that, where LCH SA is the designator in relation to any Restructuring Cleared Transaction or Physically Settled Cleared Transaction, as applicable, it is permitted to designate any relevant CDS Buyer or CDS Seller, as applicable, in accordance with Section 8 (Matched Pair Designations and Notices) as its designee, notwithstanding that such relevant CDS Buyer or CDS Seller, as applicable, is not its Affiliate and for the additional purposes of:
 - delivering or receiving any Credit Event Notice relating to a Restructuring Credit Event or Notice to Exercise Movement Option (in each case, to the extent not delivered via a DTCC Notice Facility);
 - (ii) delivering or receiving any Notice of Physical Settlement or NOPS Amendment Notice:
 - (iii) making or receiving payment of any Physical Settlement Amount and any costs and expenses of Physical Settlement; and
 - (iv) delivering or receiving any notice and otherwise exercising any rights or performing any obligations of LCH SA for the purposes of Sections 9.9 (Buy-in of Bonds Not Delivered) or 9.10 (Alternative Procedures Relating to Loans Not Delivered) of the 2003 ISDA Credit Derivatives Definitions.
- (b) Section 9.1(c)(iii) of the 2003 ISDA Credit Derivatives Definitions is amended by deleting both occurrences of the words "master agreement between Buyer and Seller that governs the relevant Credit Derivative Transaction" and replacing each with the words "CDS Clearing Documentation".
- (c) Section 9.2(a) of the 2003 ISDA Credit Derivatives Definitions is amended by deleting the words "any master agreement governing the Credit Derivative Transaction" at the end thereof and replacing them with the words "the CDS Clearing Documentation".
- (d) Section 9.2(c)(ii) of the 2003 ISDA Credit Derivatives Definitions is amended by deleting the words "any master agreement governing the Credit Derivative Transaction" at the end of the third sentence thereof and replacing them with the words "the CDS Clearing Documentation".

- (e) Section 9.2(c)(iii) of the 2003 ISDA Credit Derivatives Definitions is amended by deleting both occurrences of the words "a master agreement governing the Credit Derivative Transaction" and replacing each with the words "the CDS Clearing Documentation".
- (f) The Protocol Effectiveness Condition (as defined in the 2014 ISDA Credit Derivatives Definitions Protocol published by ISDA on 21 August 2014 and supplemented by the Supplement thereto published by ISDA on 15 September 2014 and as further supplemented from time to time (the "2014 Protocol")) being satisfied, then from and including the Implementation Date (as defined in the 2014 Protocol) the amendments set forth in Part 1 (Global Amendments) of Schedule 1 (Amendments to Protocol Covered Transactions) of the 2014 Protocol shall apply to each Cleared Transaction which incorporates the 2003 ISDA Credit Derivatives Definitions and each such Cleared Transaction shall be a Protocol Covered Transaction for the purposes of the 2014 Protocol provided that:
 - the First Novation Date in respect of such Cleared Transaction is prior to the Implementation Date;
 - (ii) such Cleared Transaction is not an Excluded Transaction (as defined in the 2014 Protocol) provided that if such Cleared Transaction relates to more than one Reference Entity and would otherwise be a Protocol Covered Transaction but for the inclusion of any Affected Portion (as defined in the 2014 Protocol) therein, such Affected Portion shall be excluded for the purposes of this Section 2.4(f) and the remaining portion of such Cleared Transaction shall be deemed to be a Protocol Covered Transaction;
 - the Scheduled Termination Date in respect of such Cleared Transaction is on or after the Implementation Date;
 - (iv) no Event Determination Date has occurred in respect of such Cleared Transaction prior to the Implementation Date;
 - (v) from and including the Implementation Date the provisions of Part B of the CDS Clearing Supplement shall apply to such Cleared Transaction or the portion of such Cleared Transaction other than the Affected Portion, as applicable and such Cleared Transaction shall be treated as if it had been entered into and is subject to the 2014 ISDA Credit Derivatives Definitions. In the event of any inconsistency between the provisions of Part B of the CDS Clearing Supplement and the amendments set forth in this paragraph 2.4, the provisions of Part B of the CDS Clearing Supplement shall prevail.

2.5 Physical Settlement Matrix Updates

(a) Publication of Revised Matrix

54

File No. SR-LCH SA-2022-007 Page 55 of 439

PUBLIC

Where ISDA publishes a new version (the "Revised Matrix") of the Physical Settlement Matrix that is the Relevant Physical Settlement Matrix in respect of an Eligible Reference Entity (the "Existing Matrix"), LCH SA will, in consultation with the CDSClear Product Committee, determine whether the Existing Matrix and the Revised Matrix are fungible for the purposes of Single Name Cleared Transactions referencing such Eligible Reference Entity.

If there are any changes between the terms of the Revised Matrix and the Existing Matrix that would apply to Single Name Cleared Transactions referencing the Eligible Reference Entity and the Existing Matrix were they to reference the Revised Matrix instead, then the Existing Matrix and the Revised Matrix shall be deemed not to be fungible and paragraph (c) below shall apply.

(b) Revised Matrix Fungible

If LCH SA determines in consultation with the CDSClear Product Committee that the Existing Matrix and the Revised Matrix are fungible for the purposes of Single Name Cleared Transactions referencing an Eligible Reference Entity (such date of determination, a **Matrix Re-versioning Date**), then:

- (i) LCH SA will promptly notify all Clearing Members of such determination; and
- (ii) each Single Name Cleared Transaction referencing such Eligible Reference Entity and the Existing Matrix will, as of the close of business on the Matrix Reversioning Date, be deemed to have been amended so as to reference the Revised Matrix, which shall become the Relevant Physical Settlement Matrix in respect of such Eligible Reference Entity.

For the avoidance of doubt, following the publication of a Revised Matrix, there may be different Matrix Re-versioning Dates in respect of Single Name Cleared Transactions referencing different Eligible Reference Entities.

(c) Revised Matrix not Fungible

If it is determined that the Revised Matrix and the Existing Matrix are not fungible for the purposes of Single Name Cleared Transactions referencing an Eligible Reference Entity, then existing Single Name Cleared Transactions referencing such Eligible Reference Entity and the Existing Matrix shall continue to reference the Existing Matrix.

(d) Original Transactions submitted prior to Matrix Re-versioning Date

Pursuant to Section 2.3(d) above, any Original Transaction referencing an Eligible Reference Entity submitted for clearing after a Matrix Re-versioning Date in respect of such Eligible Reference Entity and which references the Existing Matrix prior to such Matrix Re-versioning Date will, following novation, result in an Initial Single Name Cleared Transaction referencing the Revised Matrix.

2.6 Terms of STM Cleared Transaction

The following terms of a registered STM Cleared Transaction are designated as the "STM Cleared Terms".

If a Cleared Transaction is designated as an STM Cleared Transaction in accordance with Article 3.1.10.7 of the CDS Clearing Rule Book the terms of such Cleared Transaction will automatically, and without any further action by either party, include the STM Cleared Terms.

(a) Net present value

- (i) LCH SA shall, at least once per Cash Payment Day, determine the net present value of this STM Cleared Transaction in such manner and at such times as may be provided in Section 2 of the Procedures.
- (ii) Immediately upon LCH SA making such a determination of the net present value of this STM Cleared Transaction:
 - (A) An amount (determined in accordance with Section 2.14 (NPV Amount) of the Procedures) of cash may become due and payable by the Clearing Member or LCH SA (as applicable, and as determined in accordance with Section 2.14 (NPV Amount) of the Procedures) under these STM Cleared Terms; and
 - (B) the net present value of this STM Cleared Transaction shall for all purposes be reset to zero.
- (iii) The amount (if any) referred to in (ii)(A) above shall immediately become due and payable by the relevant party under the STM Cleared Terms applicable to this STM Cleared Transaction and shall be made in such manner and at such times as may be provided in Section 2 of the Procedures.
- (iv) In any circumstance preventing the NPV Payments or Variation Margin Collateral Transfers from being performed in US Dollar, LCH SA shall be entitled to convert any amount denominated in US Dollar into Euro in accordance with Article 4.2.3.2 of the CDS Clearing Rule Book.

(b) Price Alignment Amount

(i) LCH SA shall, at least once per Cash Payment Day, determine the Price Alignment Amount in respect of this STM Cleared Transaction in such manner and at such times as may be provided in Section 2 of the Procedures.

File No. SR-LCH SA-2022-007 Page 57 of 439

PUBLIC

- (ii) Immediately upon LCH SA making such a determination of the Price Alignment Amount in respect of this STM Cleared Transaction, an amount (determined in accordance with Section 2.17 (*Price Alignment Amount*) of the Procedures) of cash may become due and payable by the Clearing Member or LCH SA (as applicable, and as determined in accordance with Section 2.17 (*Price Alignment Amount*) of the Procedures) under these STM Cleared Terms.
- (iii) The amount (if any) referred to in (ii) above shall immediately become due and payable by the relevant party under the STM Cleared Terms applicable to this STM Cleared Transaction and shall be made in such manner and at such times as may be provided in Section 2 of the Procedures.

3. PAYMENTS AND DELIVERIES

3.1 Obligation to pay and deliver

Each of LCH SA and each Clearing Member will make each payment or delivery specified under the terms of each Cleared Transaction to be made by it, subject to the other provisions of the CDS Clearing Documentation.

Payments under any Cleared Transaction will be made on the due date for value on that date in the place of the account specified for the relevant party in the CDS Admission Agreement (or such other account as may be designated by it from time to time for such purpose in accordance with the CDS Admissions Agreement and/or the Procedures, as applicable) and otherwise in accordance with the CDS Clearing Documentation, in freely transferable funds and in the manner customary for payments in the required currency.

Deliveries under any Cleared Transaction will be made for receipt on the due date in the manner customary for the relevant obligation unless otherwise specified in or pursuant to the CDS Clearing Documentation.

3.2 Payments under Original Transactions

- (a) If any Initial Payment Amount or Fixed Amount is due and payable under the terms of an Original Transaction on or before the Clearing Day on which the related Cleared Transactions are created by novation pursuant to Title III (Clearing Operations) of the CDS Clearing Rule Book, such amount shall be payable under and in accordance with the terms of such Original Transaction. In such event, no corresponding Initial Payment Amount or Fixed Amount shall be payable in respect of such Cleared Transactions.
- (b) If the Initial Payment Date of an Original Transaction is a date falling after the Clearing Day on which the Cleared Transactions related to such Original Transaction are created by novation pursuant to Title III (Clearing Operations) of the CDS Clearing Rule Book then the corresponding Initial Payment Date for the related Cleared Transactions

shall occur on the Transaction Business Day which is also a Clearing Day immediately following the Clearing Day on which such related Cleared Transactions are created.

4. CREDIT EVENTS, SUCCESSION EVENTS AND RENAME EVENTS

4.1 Determination of Credit Events and Succession Events

Notwithstanding any provision of any Cleared Transaction to the contrary:

- (a) LCH SA (in its capacity as Calculation Agent with respect to such Cleared Transaction) shall not make any determination in respect of any matter which is or may be subject to resolution under Sections 3.5 (Succession Resolutions) or 3.6 (Substitute Reference Obligation Resolutions) of the DC Rules; and
- (b) neither LCH SA nor any Clearing Member shall be entitled to deliver a Succession Event Notice or a Credit Event Notice (other than Credit Event Notices in relation to a Restructuring Credit Event in accordance with the terms of any Restructuring Cleared Transaction and, where applicable, the DTCC Rules and Section 7.8 (Failure to notify Matched Pairs)).

4.2 Credit Event Timeline

(a) Publication of Credit Event Timeline

Upon a DC Credit Event Announcement, LCH SA will publish and make available to Clearing Members a timeline in respect of the relevant Credit Event and related Affected Cleared Transactions, to notify, among other things, the relevant Novation Cut-off Date(s), Compression Cut-off Date(s) and First Novation Date(s).

Any such timeline shall (i) be published and made available on the date of such DC Credit Event Announcement or, if LCH SA determines that such publication on such date is not practicably possible, as soon as practicable thereafter (but in no event later than the Transaction Business Day before the earlier of the relevant Novation Cut-off Date and the relevant Compression Cut-off Date) and (ii) in all cases be without prejudice to and consistent with the terms of the CDS Clearing Documentation and any relevant DC Resolutions.

(b) Amendment of Credit Event Timeline

Any such timeline may be subject to subsequent amendment by LCH SA, by means of a Clearing Notice to Clearing Members, only to reflect subsequent DC Resolutions, timing provisions of any relevant Transaction Auction Settlement Terms and/or actions of DTCC, or in each case any subsequent amendments thereto. Any such amendment shall be made by LCH SA as soon as reasonably practicable following the relevant event.

4.3 Novation and Compression following Credit Events

Any Restructuring Cleared Transaction (or portion thereof) in respect of which a valid Credit Event Notice is not delivered during the relevant CEN Triggering Period shall become eligible for compression in accordance with Chapter 3 (*Compression*) of Title III (*Clearing Operations*) of the CDS Clearing Rule Book on the Transaction Business Day following the related Exercise Cut-off Date applicable to the relevant Buyer.

By way of clarification to Chapter 3 (*Compression*) of Title III (*Clearing Operations*) of the CDS Clearing Rule Book, LCH SA shall effect compression of relevant Cleared Transactions if and to the extent requested by a Clearing Member.

4.4 Re-couponing of Restructuring Cleared Transactions

Following a Restructuring Credit Event, LCH SA may perform re-couponing in accordance with Section 5 of the Procedures on any Restructuring Cleared Transaction (or portion thereof) in respect of which a valid Credit Event Notice has not been delivered during the relevant CEN Triggering Period, provided that:

- (a) LCH SA notifies (such notice, the "Initial Re-couponing Notice") all of the relevant Clearing Members of its intention to perform such re-couponing and the intended date of such re-couponing (such date, the "Re-Couponing Date");
- (b) the Re-Couponing Date is no earlier than the later of (i) the end of the relevant CEN Trigger Period; and (ii) eight Transaction Business Days after the effective date of the Initial Re-couponing Notice;
- (c) all of the relevant Clearing Members have subsequently consented to the proposed re-couponing, subject to confirmation of the terms of such re-couponing, within five Transaction Business Days of the effective date of the Initial Re-couponing Notice;
- (d) after receiving consent from all relevant Clearing Members in accordance with paragraph (c) above, LCH SA notifies (such notice, the "Re-couponing Notice") the relevant Clearing Members, no later than three Transactions Business Days prior to the Re-Couponing Date, of the proposed terms of the Resulting Single Name Cleared Transactions determined by LCH SA in accordance with Section 5 of the Procedures; and
- (e) all of the relevant Clearing Members have subsequently consented to the proposed terms of such re-couponing process as set out in the Re-couponing Notice within two Transaction Business Days of the effective date of the Re-couponing Notice.

For the avoidance of doubt, if, in respect of a proposed re-couponing process, LCH SA does not obtain consent from all of the relevant Clearing Members in accordance with paragraphs (c) or (e) above, then LCH SA may elect to propose a re-couponing process again in accordance with this Section 4.4 for so long as any relevant Restructuring Cleared Transactions remain outstanding.

4.5 Succession Events and Cleared Transactions

If LCH SA determines that any Original Transaction submitted for novation or any Cleared Transaction subject to compression would have been subject to a Succession Event but will no longer be subject to such Succession Event upon novation or compression (as applicable) because of the Trade Date that would be specified with respect to the relevant Cleared Transactions, LCH SA may take such action as it deems necessary to ensure that such Succession Event is given effect with respect to such Cleared Transactions, including, without limitation, specifying an alternate Trade Date for purposes of Section 2.1 (*Reference Entity*) of the 2003 ISDA Credit Derivatives Definitions with respect to each relevant Cleared Transaction or, where LCH SA determines that an alternative course of action is not practicable, declining to accept such Original Transaction for novation or Cleared Transaction for compression (as applicable).

4.6 Succession Event Timeline

(a) Publication of Succession Event Timeline

Following the publication of a resolution of a relevant Credit Derivatives Determinations Committee to the effect that one or more Successors have been identified with respect to any Reference Entity for the purposes of any Cleared Transaction, LCH SA will publish and make available to Clearing Members a timeline in respect of the relevant Succession Event and related Affected Cleared Transactions, to notify, amongst other things, the relevant Novation Cut-off Date(s), Compression Cut-off Date(s) and First Novation Date(s).

Any such timeline shall be published and made available as soon as practicable (but in no event later than the Transaction Business Day before the earlier of the relevant Novation Cut-off Date and the relevant Compression Cut-off Date) and shall in all cases be without prejudice to and consistent with the terms of the CDS Clearing Documentation and any relevant DC Resolutions.

(b) Amendment of Succession Event Timeline

Any such timeline may be subject to subsequent amendment by LCH SA, by means of a Clearing Notice to Clearing Members, to reflect subsequent DC Resolutions and any relevant actions of DTCC, or in each case any subsequent amendments thereto. Any such amendment shall be made by LCH SA as soon as reasonably practicable following the relevant event.

4.7 Rename Events

In respect of Single Name Cleared Transactions, if a Reference Entity changes its name (a "Rename Event"), LCH SA will publish and make available to Clearing Members as soon as practicable upon becoming aware of such Rename Event a timeline in respect of the relevant Rename Event and related Affected Cleared Transactions, to notify, amongst other things,

the relevant Novation Cut-off Date(s), Compression Cut-off Date(s) and First Novation Date(s).

Any such timeline may be subject to subsequent amendment by LCH SA, by means of a Clearing Notice to Clearing Members, to reflect any relevant actions of DTCC. Any such amendment shall be made by LCH SA as soon as reasonably practicable following the relevant event.

4.8 Updating Eligible Index Versions

(a) Publication of Revised Index

Where the Index Publisher of an Eligible Index Version publishes a revised version of such index following:

- (i) a DC Credit Event Announcement;
- a DC Resolution becoming effective which specifies that a Succession Event has occurred; or
- (iii) the determination of a Substitute Reference Obligation,

LCH SA will in consultation with the CDSClear Product Committee determine whether such revised index version is fungible with the Eligible Index Version after taking account of the relevant Credit Event, Succession Event or, as applicable, Substitute Reference Obligation.

(b) Index Version not Fungible

If LCH SA determines in consultation with the CDSClear Product Committee that such revised index version is not fungible with the Eligible Index Version after taking account of the relevant Credit Event, Succession Event or, as applicable, Substitute Reference Obligation, it shall notify DTCC accordingly so that DTCC does not automatically update the trade records for Index Cleared Transactions and Index CCM Client Transactions referencing the relevant Eligible Index Version in the TIW.

(c) Index Fungible

Unless LCH SA has notified DTCC in accordance with paragraph (b) above, following confirmation from DTCC that the trade records for Cleared Transactions referencing the relevant index have been updated in the TIW (such date a **DTCC Re-versioning Date**) so that such Cleared Transactions reference the revised index version (which such revised index version shall become the Eligible Index Version), LCH SA will update its corresponding records in the CDS Clearing System overnight following such DTCC Re-versioning Date. LCH SA will send the relevant reports referred to in Section 5 of the Procedures to the relevant Clearing Members on the Transaction Business Day following such DTCC Re-versioning Date which will be updated so as to refer to Cleared Transactions referencing the revised index version.

4.9 Reversal of DC Credit Event Announcements and Margining

If a DC Credit Event Announcement is reversed then, subject to Section 9.1(c)(iii)(B) of the 2003 ISDA Credit Derivatives Definitions, LCH SA shall be obliged in accordance with Section 3 of the Procedures to calculate and shall be entitled to call for Margin and/or be obliged to return Margin with respect to each Clearing Member on the basis that the DC Credit Event Announcement that is reversed had not been made.

5. **RESTRUCTURING**

5.1 Creation and Notification of Restructuring Matched Pairs

Following the occurrence of a Restructuring Credit Event, LCH SA will create (on one or, if the Novation Cut-off Date in respect of Index Cleared Transactions is prior to the Novation Cut-off Date in respect of Initial Single Name Cleared Transactions, two or more occasions) Restructuring Matched Pairs in accordance with Section 8.1 (*Creation of Matched Pairs*). Each such Restructuring Matched Pair shall be composed of two Restructuring Cleared Transactions created at the same time as and as a result of the termination of the relevant Single Name Cleared Transactions pursuant to Section 5.2 (*Creation of Restructuring Cleared Transactions*) below.

LCH SA shall notify the relevant Matched Buyer and Matched Seller comprised within each Restructuring Matched Pair of:

- (a) the identity of the other Clearing Member (together with the address, fax number, telephone number, email address and other applicable notice details of such other Clearing Member) of such Matched Pair; and
- (b) the associated Restructuring Matched Pair Amount,

as soon as reasonably practicable following the related Compression Cut-off Date, but in any event, at or prior to the RMP Notification Deadline.

LCH SA will additionally provide to DTCC all relevant "Matching Information" (as defined in the DTCC Rules) on or prior to the Matching Information Notification Deadline.

5.2 Creation of Restructuring Cleared Transactions

Upon the notification to the relevant Clearing Members of Restructuring Matched Pairs, the relevant Initial Single Name Cleared Transactions, Restructuring Cleared Transactions (created following the occurrence of a previous Restructuring Credit Event), Spin-off Single Name Cleared Transactions and Resulting Single Name Cleared Transactions entered into by each Clearing Member with LCH SA will be deemed terminated and new Restructuring Cleared Transactions of the same CDS Type will be deemed to be entered into between each such Clearing Member and LCH SA, with each such Restructuring Cleared Transaction having a Floating Rate Payer Calculation Amount corresponding to the Restructuring Matched Pair Amount of a Restructuring Matched Pair in which the relevant Clearing Member is comprised as a Matched Buyer or a Matched Seller, as applicable.

5.3 Triggering of Restructuring Cleared Transactions

Subject as set out in Section 7 (*Delivery of Notices and Fallbacks*), and notwithstanding anything to the contrary in the terms of any Cleared Transaction, Clearing Members may only deliver Credit Event Notices in relation to a Restructuring Credit Event during the CEN Triggering Period following notification of Restructuring Matched Pairs by LCH SA and subject always to the terms of the relevant Restructuring Cleared Transaction and, where applicable, the DTCC Rules.

Any Credit Event Notice delivered in respect of a Restructuring Matched Pair for an amount which is greater than the related Floating Rate Payer Calculation Amount shall be ineffective as to such excess.

5.4 Notice to Exercise Movement Option

Subject as set out in Section 7 (*Delivery of Notices and Fallbacks*), Clearing Members may only deliver a Notice to Exercise Movement Option during the NEMO Triggering Period, subject always to the other terms of the relevant Restructuring Cleared Transaction and, where applicable, the DTCC Rules.

5.5 Reversal of DC Credit Event Announcements

If a DC Credit Event Announcement made in relation to a Restructuring Credit Event is reversed then, subject to Section 9.1(c)(iii)(B) of the 2003 ISDA Credit Derivatives Definitions:

- (a) LCH SA shall have no obligation to create Restructuring Matched Pairs in accordance with Section 5.1 (Creation and Notification of Restructuring Matched Pairs), any such Restructuring Matched Pairs so created shall be deemed not to have been created and any Credit Event Notices delivered in connection with such Restructuring Matched Pairs shall be deemed to be ineffective;
- (b) LCH SA shall, where applicable, make relevant registrations in the TIW in order to reflect such reversal which shall also automatically result in such registrations being made in respect of any related CCM Client Transactions; and
- (c) Section 4.9 (Reversal of DC Credit Event Announcements and Margining) shall apply.

5.6 Reports

Without prejudice to the notification requirements set out elsewhere in the CDS Clearing Documentation, LCH SA will communicate to the relevant Clearing Members, on the basis of information received from DTCC or, as applicable, from Clearing Members, amongst other things:

(a) the aggregate Floating Rate Payer Calculation Amounts of Restructuring Cleared Transactions to which they are a party and which are the subject of Credit Event Notices; and

File No. SR-LCH SA-2022-007 Page 64 of 439

PUBLIC

(b) the results of the exercise of any Movement Option in relation to Cleared Transactions to which they are a party,

at or around 7.00 p.m. on each day during each CEN Triggering Period and NEMO Triggering Period, as applicable, through the reports referred to in Section 5 of the Procedures.

For the avoidance of doubt, such communication shall not affect the validity or effectiveness of any Credit Event Notice or Notice to Exercise Movement Option which shall be subject to the terms of the relevant Restructuring Cleared Transaction.

6. PHYSICAL SETTLEMENT

6.1 General Terms relating to Physical Settlement

Where Physical Settlement is applicable as the Fallback Settlement Method pursuant to Section 12.1 (*Auction Settlement*) or 12.17 (*Movement Option*) of the 2003 ISDA Credit Derivatives Definitions, each Cleared Transaction will be subject to settlement in accordance with its terms and this Section 6 (*Physical Settlement*).

LCH SA has implemented a process, as set-out in this Section 6 (*Physical Settlement*), pursuant to which Clearing Members will manage the physical delivery process bilaterally in respect of any Cleared Transactions for which Physical Settlement is applicable.

Notwithstanding such process, LCH SA shall remain the legal counterparty for any such Cleared Transactions for which Physical Settlement is applicable and shall continue to be liable with respect to its obligations relating to such Physical Settlement, subject to its terms and this Section 6 (*Physical Settlement*).

6.2 Creation and Notification of Settlement Matched Pairs

LCH SA will create Settlement Matched Pairs in accordance with Section 8.1 (*Creation of Matched Pairs*) and notify Matched Buyer and Matched Seller comprised within each Settlement Matched Pair of:

- the identity of the other Clearing Member (together with the address, fax number, telephone number, email address and other applicable notice details of each such other Clearing Member); and
- (b) the associated Settlement Matched Pair Delivery Amount,

at or prior to the SMP Notification Deadline.

Notwithstanding the above, the Settlement Matched Pair for a Restructuring Cleared Transaction shall be the Restructuring Matched Pair previously created by LCH SA in accordance with Section 5.1 (*Creation and Notification of Restructuring Matched Pairs*) in respect of such Restructuring Cleared Transaction.

6.3 Physically Settled Cleared Transactions

Following the actual or deemed creation of such Settlement Matched Pairs, the relevant Cleared Transactions in respect of which the Fallback Settlement Method applies will be construed as if they had been terminated and new Cleared Transactions of the same CDS Type will be deemed to have been entered into between each such Clearing Member and LCH SA (each such deemed Cleared Transaction being a "Physically Settled Cleared Transaction").

Each such Physically Settled Cleared Transaction shall have a Floating Rate Payer Calculation Amount corresponding to the Settlement Matched Pair Delivery Amount of a Settlement Matched Pair in which the relevant Clearing Member is comprised as a Matched Buyer or a Matched Seller, as applicable.

6.4 Matched Seller Calculation Agent

Notwithstanding any provision to the contrary in a Physically Settled Cleared Transaction (but subject as set out in Section 6.10 (Failure to pay Physical Settlement Amount), the "Calculation Agent" for the purposes of Article VIII (Terms relating to Physical Settlement) and Section 9.8 (Partial Cash Settlement Terms) of the 2003 ISDA Credit Derivatives Definitions shall be the relevant Matched Seller.

6.5 Notices of Physical Settlement

(a) No Notices of Physical Settlement until Notification of Settlement Matched Pairs

Subject as set out at Section 7.8 (Failure to notify Matched Pairs) and notwithstanding anything to the contrary in the terms of any Cleared Transaction, Clearing Members may not deliver any Notices of Physical Settlement in respect of any Affected Cleared Transactions until they have been notified of their Settlement Matched Pairs.

(b) Notice of Physical Settlement to contain Matched Buyer Account Information

Each Notice of Physical Settlement delivered by Matched Buyer shall contain, in addition to the information required under the terms of the relevant Physically Settled Cleared Transaction, Matched Buyer's account information.

(c) LCH SA not Liable

LCH SA shall have no liability with respect to any loss, cost, damage or expense suffered or incurred by a Matched Seller as a result of any error or inaccuracy in any Notice of Physical Settlement or any NOPS Amendment Notice sent by a Matched Buyer and shall have no duty to verify any such notice or the contents thereof.

6.6 **Delivery of Non-DVP Obligations**

The following shall apply in relation to any Settlement Matched Pair where the relevant Matched Buyer specifies a Non-DVP Obligation in a Notice of Physical Settlement or NOPS Amendment Notice:

- (a) The relevant Matched Buyer shall notify LCH SA that it is ready to Deliver the relevant Non-DVP Obligation at or prior to 5.30 p.m. on the Transaction Business Day prior to the date on which Delivery is scheduled to occur. In such notice, the relevant Matched Buyer shall also specify the amount of any expenses payable to it under Section 9.2(c)(vi) of the 2003 ISDA Credit Derivatives Definitions.
- (b) LCH SA shall notify the relevant Matched Seller at or prior to 6.30 p.m. on the Transaction Business Day prior to the date on which Delivery is scheduled to occur that it is obligated to pay LCH SA the amount of the relevant Physical Settlement Amount corresponding to the Outstanding Amount of such Non-DVP Obligation and any amounts in respect of expenses notified by the relevant Matched Buyer before 9.00 a.m. on the following Transaction Business Day.
- (c) The relevant Matched Seller shall pay to LCH SA the amount(s) so requested on or prior to 9.00 a.m. on the Transaction Business Day immediately following the date on which the relevant Matched Buyer notified LCH SA of its readiness to Deliver provided that the request for payment has been made in a timely manner as set out above.
- (d) LCH SA shall notify the relevant Matched Buyer that it has received payment at or prior to 5.30 p.m. on the Transaction Business Day in which LCH SA receives payment, provided that payment is received by LCH SA at or prior to 9.00 a.m. on such Transaction Business Day as set out above.
- (e) The relevant Matched Seller shall notify LCH SA that Delivery has occurred by 5.30 p.m. on the Transaction Business Day on which Matched Seller receives Delivery of the relevant Non-DVP Obligations. Notices received after 3.30 p.m. will be deemed received on the next following Transaction Business Day, unless LCH SA agrees otherwise.
- (f) If and to the extent that LCH SA has received payment from the relevant Matched Seller of the Physical Settlement Amount in full on a timely basis as set out above, LCH SA shall pay the Physical Settlement Amount and any expenses due to the relevant Matched Buyer under paragraph 9.2(c)(vi) of the 2003 ISDA Credit Derivatives Definitions on or prior to 9.00 a.m. on the Transaction Business Day following the Transaction Business Day on which LCH SA receives the notice referred to in sub-paragraph (e) above from the relevant Matched Seller.
- (g) If and to the extent that Delivery of the relevant Non-DVP Obligations has not been effected by the relevant Matched Buyer in accordance with terms of the relevant Physically Settled Cleared Transaction as of the expiry of the third Transaction Business Day following delivery of the relevant Matched Buyer's notice referred to

above, the relevant Matched Seller may request LCH SA to repay the Physical Settlement Amount not earlier than the first Transaction Business Day following the date on which such request is effectively delivered to LCH SA.

6.7 Alternative Delivery Procedure

A Matched Buyer and Matched Seller comprising a Settlement Matched Pair may elect to exercise their rights against and perform obligations to LCH SA in relation to the Settlement Matched Pair Delivery Amount (or any portion thereof) directly as between themselves. To exercise such option, the relevant Matched Buyer and Matched Seller must each notify LCH SA accordingly (in the form set out in Appendix V hereto) specifying the relevant Matched Contracts in respect of the related relevant Settlement Matched Pair Delivery Amounts and obtain the consent of LCH SA, which consent will not be unreasonably withheld or delayed by more than one Transaction Business Day following receipt of such notification by Matched Buyer and Matched Seller.

With effect from the time that LCH SA confirms its consent, the Settlement Matched Pair Delivery Amount will be reduced by the amount specified in the joint notice of the relevant Matched Buyer and Matched Seller, and the relevant Matched Buyer and Matched Seller shall each perform their obligations to each other and shall each acquire enforcement rights in respect of the other party's obligations to it pursuant to the Contracts (Rights of Third Parties) Act 1999 in respect of any such reduction as agreed between them.

6.8 Buy-in of Bonds – Matched Seller has entered into CCM Client Transaction

The following provisions shall solely be applicable to a Matched Seller that is a CCM in respect of the Matched Contracts of the Settlement Matched Pair if such Matched Seller notifies Matched Buyer and LCH SA that it has a CCM Client Transaction which corresponds to the Matched Seller Contract:

- (a) Buy-in of Bonds Settlement Matched Pair
 - Section 9.9 (*Buy-in of Bonds Not Delivered*) of the 2003 ISDA Credit Derivatives Definitions shall not apply in respect of the Matched Contracts of the Settlement Matched Pair.
- (b) Deemed Buy-in of Bonds resulting from CCM Client Transaction of Matched Seller that is a CCM

Provided that Physical Settlement has not already occurred in respect of the Matched Contracts of a Settlement Matched Pair, if:

(i) the Matched Seller that is a CCM receives a Buy-in Notice from its CCM Client in respect of the CCM Client Transaction between such Matched Seller and its CCM Client which is validly delivered pursuant to the terms of such CCM Client Transaction (including the Mandatory Provisions), then such Matched Seller shall notify Matched Buyer and LCH SA of its receipt of such Buy-in Notice and

File No. SR-LCH SA-2022-007 Page 68 of 439

PUBLIC

of the content thereof and Matched Buyer's right to Deliver the Relevant Bonds specified in the relevant Buy-in Notice shall be suspended until the fourth Business Day (inclusive) following the Buy-In Date specified in the relevant Buy-In Notice (the "Deemed Buy-in Period"); and

- (ii) such Matched Seller notifies Matched Buyer and LCH SA that it has been notified by its CCM Client pursuant to Section 9.9 (Buy-in of Bonds Not Delivered) of the 2003 ISDA Credit Derivatives Definitions that a Buy-in Price has been determined in respect of Relevant Bonds for the purposes of the CCM Client Transaction between such Matched Seller and its CCM Client, then on the third Business Day following receipt by Matched Seller of such notice from its CCM Client (which such date Matched Seller shall specify) (the "Buy-in Effective Date"):
 - (A) Matched Buyer will be deemed to have Delivered to such Matched Seller an Outstanding Principal Balance of the Deliverable Obligations specified in the Notice of Physical Settlement or any NOPS Amendment Notice, as applicable, for which a Buy-in Price was determined by the CCM Client; and
 - (B) the Physical Settlement Amount to be paid by such Matched Seller to Matched Buyer in respect of the Matched Contracts of the Settlement Matched Pair shall be reduced (but not below zero) by an amount equal to the amount by which the Physical Settlement Amount to be paid to such Matched Seller by its CCM Client in respect of the CCM Client Transaction has been reduced pursuant to Section 9.9 (Buy-in of Bonds Not Delivered) of the 2003 ISDA Credit Derivatives Definitions.

Such Matched Seller shall notify Matched Buyer and LCH SA of the Outstanding Principal Balance of the Deliverable Obligations and the Physical Settlement Amount reduction determined in respect of its CCM Client Transaction for the purposes of (A) and (B) above.

6.9 Alternative Procedures relating to Loans – Buyer Right to Deliver (Section 9.10(a) 2003 ISDA Credit Derivatives Definitions)

The following provision shall solely be applicable in respect of the Matched Contracts of the Settlement Matched Pair if a Matched Buyer that is a CCM notifies Matched Seller and LCH SA that it has a CCM Client Transaction which corresponds to the Matched Buyer Contract:

If a Matched Buyer that is a CCM notifies Matched Seller and LCH SA that it has received a Bond or Loan from its CCM Client in respect of its CCM Client Transaction pursuant to Section 9.10(a) (Alternative Procedures Relating to Loans Not Delivered) of the 2003 ISDA Credit Derivatives Definitions, then such Bonds or Loans shall be deemed specified in the Notice of Physical Settlement or any NOPS Amendment Notice for the purposes of the Matched Contracts of the Settlement Matched Pair.

6.10 Failure to pay Physical Settlement Amount

If, in relation to any Settlement Matched Pair, a Matched Seller fails to pay all or part of the Physical Settlement Amount (the **Failed Amount**) to the related Matched Buyer, as designee for LCH SA, when due:

- (a) the relevant Matched Buyer may and the relevant Matched Seller shall, as soon as practicable, give notice in writing to LCH SA, giving all material details of the Settlement Matched Pair involved, the relevant failure to pay and the Failed Amount;
- (b) such failure to pay shall not constitute or be deemed to constitute a breach of contract, event of default or failure to pay by LCH SA under the CDS Clearing Documentation or otherwise (but such failure to pay may, for the avoidance of doubt, constitute a breach of the CDS Clearing Documentation and/or a Payment Failure for the purposes of Article 4.3.1.1 of the CDS Clearing Rule Book by or with respect to the relevant Clearing Member);
- (c) if the relevant Matched Buyer elects to notify LCH SA of such failure to pay as contemplated above, such Matched Buyer may give any such notice as soon as reasonably practicable after the occurrence of such failure to pay by the relevant Matched Seller;
- (d) upon notice being given to LCH SA by the relevant Matched Buyer, "Cash Settlement" between the relevant Matched Buyer and LCH SA and the relevant Matched Seller and LCH SA pursuant to the Partial Cash Settlement Terms (set out in Section 9.8 (Partial Cash Settlement Terms) of the 2003 ISDA Credit Derivatives Definitions, as amended by this CDS Clearing Supplement) shall be deemed to apply to the relevant Physically Settled Cleared Transactions of the Settlement Matched Pair with respect to the Deliverable Obligations corresponding to the Failed Amount as though:
 - (i) the Deliverable Obligations not Delivered were Undeliverable Obligations;
 - the Latest Permissible Physical Settlement Date were the date on which the relevant Matched Buyer gave the relevant notice to LCH SA;
 - (iii) Indicative Quotations were not applicable;
 - the relevant Matched Buyer were the Calculation Agent in respect of the Physically Settled Cleared Transaction of the Settlement Matched Pair to which it is a direct party; and
 - (v) the Cash Settlement Amount determined in respect of the Physically Settled Cleared Transaction between Matched Buyer and LCH SA is also the Cash Settlement Amount in respect of the corresponding Physically Settled Cleared Transaction of the Settlement Matched Pair; and

(e) LCH SA and the relevant Matched Buyer will settle the relevant Physically Settled Cleared Transaction accordingly.

6.11 Partial Cash Settlement due to Impossibility or Illegality

Section 9.3 (Partial Cash Settlement due to Impossibility or Illegality) of the 2003 ISDA Credit Derivatives Definitions shall apply to a Physically Settled Transaction in the circumstances contemplated therein, provided that Matched Buyer or Matched Seller notifies the other Clearing Member comprised in the relevant Settlement Matched Pair and LCH SA accordingly.

In such case, the related Physically Settled Cleared Transaction entered into between LCH SA and the other Clearing Member comprised in the relevant Settlement Matched Pair shall likewise be subject to "Cash Settlement" pursuant to the Partial Cash Settlement Terms and Section 6.14 (*Consequences of Cash Settlement*).

6.12 Fallback to Cash Settlement in respect of Non-Deliverable Obligations

- (a) If Matched Buyer is not permitted to Deliver one or more Deliverable Obligations (such Deliverable Obligations, the Non-Deliverable Obligations) specified in the relevant Notice of Physical Settlement or NOPS Amendment Notice to Matched Seller as designee for LCH SA in the relevant Matched Pair because:
 - the amount of such Deliverable Obligation to be Delivered is less than the relevant minimum denomination of such Deliverable Obligation; or
 - (ii) such Matched Seller is not a permitted transferee under such Deliverable Obligation (and, in the case of this sub-section (ii), such circumstance would not constitute an illegality or impossibility outside the control of a relevant party for the purposes of Section 9.3 (*Partial Cash Settlement due to Impossibility or Illegality*) of the 2003 ISDA Credit Derivatives Definitions),

then it shall notify the relevant Matched Seller and LCH SA (in the form set out in Appendix VI hereto) accordingly describing in reasonable detail the relevant circumstances.

With effect from such notification, such occurrence shall be treated, in relation to each relevant Physically Settled Cleared Transaction, as an illegality or impossibility outside the control of a relevant party for the purpose of Section 9.3 (*Partial Cash Settlement Due to Impossibility or Illegality*) of the 2003 ISDA Credit Derivatives Definitions.

(b) Upon notice being given to LCH SA by Matched Buyer under sub-paragraph (a) of this Section 6.12 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations), "Cash Settlement" pursuant to the Partial Cash Settlement Terms shall be deemed to apply to the Matched Contracts in respect of the relevant Settlement Matched Pair with respect to the Non-Deliverable Obligations as though the Non-Deliverable Obligations were Undeliverable Obligations and the provisions set out in Section 6.14 (Consequences of Cash Settlement) of this CDS Clearing Supplement shall apply.

6.13 Cash Settlement Resulting from CCM Client Transaction of a Matched Buyer that is a CCM

The following provision shall solely be applicable in respect of the Matched Contracts of the Settlement Matched Pair if a Matched Buyer that is a CCM notifies Matched Seller and LCH SA that it has a CCM Client Transaction which corresponds to the Matched Buyer Contract:

If a Matched Buyer that is a CCM notifies Matched Seller and LCH SA that the corresponding CCM Client Transaction between such Matched Buyer and its CCM Client is to be settled (in whole or in part) by Cash Settlement pursuant to Section 9.3 (*Partial Cash Settlement Due to Impossibility or Illegality*) of the 2003 ISDA Credit Derivatives Definitions or Mandatory Provision 7.2 (*Fallback to Cash Settlement in respect of Non-Deliverable Obligations*), then:

- (a) "Cash Settlement" shall be deemed to apply (in whole or in part, as applicable) to the Matched Contracts of the Settlement Matched Pair pursuant to the Partial Cash Settlement Terms and Section 6.14 (Consequences of Cash Settlement);
- (b) the 'Undeliverable Obligations' for the purposes of the Partial Cash Settlement Terms shall be the Undeliverable Obligations or Non-Deliverable Obligations in respect of the corresponding CCM Client Transaction; and
- (c) the Cash Settlement Amount and the Cash Settlement Date in respect of the Matched Contracts of the Settlement Matched Pair shall be the same as the Cash Settlement Amount and the Cash Settlement Date determined in respect of the corresponding CCM Client Transaction.

6.14 Consequences of Cash Settlement

If the circumstances set out in either Section 6.11 (Partial Cash Settlement Due to Impossibility or Illegality) or paragraph (a) of Section 6.12 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) or Section 6.13 (Cash Settlement Resulting from CCM Client Transaction of a Matched Buyer that is a CCM) apply, then:

- (a) the Latest Permissible Physical Settlement Date in respect of the relevant Physically Settled Cleared Transaction will be deemed to be the first date on which the relevant Matched Buyer or Matched Seller effectively gave the relevant notice to both LCH SA and the other Clearing Member as referred to in Section 6.11 (Partial Cash Settlement Due to Impossibility or Illegality) or paragraph (a) of Section 6.12 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) or Section 6.13 (Cash Settlement Resulting from CCM Client Transaction of a Matched Buyer that is a CCM) above, as applicable, (and for these purposes, Section 9.7 (Latest Permissible Physical Settlement Date) of the 2003 ISDA Credit Derivatives Definitions shall not apply);
- (b) the relevant Matched Buyer will be deemed to be the Calculation Agent;
- (c) LCH SA and the relevant Matched Buyer will settle the applicable Matched Buyer Contract, and LCH SA and the relevant Matched Seller will settle the applicable Matched Seller Contract, accordingly; and

(d) where sub-paragraph (a)(ii) of Section 6.12 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) applies, Indicative Quotations shall not be applicable.

6.15 Amendments to Section 9.8(k) of 2003 ISDA Credit Derivatives Definitions

Solely for the purposes of Section 6.11 (Partial Cash Settlement due to Impossibility or Illegality), Section 6.12 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations), Section 6.13 (Cash Settlement Resulting from CCM Client Transaction of a Matched Buyer that is a CCM) and Section 6.14 (Consequences of Cash Settlement) of this CDS Clearing Supplement, Section 9.8(k) of the 2003 ISDA Credit Derivatives Definitions is amended by inserting the following at the beginning thereof:

- "(A) For the purposes hereof, in addition to the requirements of Section 7.10, each firm Quotation shall:
 - (1) be for a transaction with Matched Buyer (or its designee) (the Relevant Buyer) pursuant to which the Relevant Buyer agrees to Deliver the Deliverable Obligations to the Dealer submitting the firm quotation (the Quoting Dealer), which transaction shall be governed by documentation that is consistent with market practice applicable to the sale and purchase of Deliverable Obligations on the Valuation Date (which may be determined by the relevant Credit Derivatives Determinations Committee), including, without limitation, a representation that the Quoting Dealer has completed all "know your customer" or similar requirements under all applicable laws, regulations and internal compliance procedures relating to a transaction with the Relevant Buyer and in respect of the Reference Entity;
 - (2) be capable of acceptance by the Relevant Buyer (for such purposes, each firm Quotation must, inter alia, be obtained from a Dealer with whom the Relevant Buyer, in its sole and absolute discretion, determines that it is able, in accordance with all its internal compliance and policy requirements, to transact and to Deliver the Deliverable Obligations) and be open for acceptance by the relevant party for at least 30 minutes; and
 - (3) be obtained on the basis that if the Relevant Buyer agrees to Deliver the Deliverable Obligations to such Quoting Dealer on the terms set forth herein, such Quoting Dealer agrees to pay the settlement amount (calculated and payable for this purpose in accordance with the relevant market standard documentation and based on the price so quoted) that would be payable to the Relevant Buyer for such Deliverable Obligations.
- (B) Otherwise,".

6.16 Subsequent Determination by the Credit Derivatives Determinations Committee with respect to an Auction

If the Fallback Settlement Method applies in respect of a Cleared Transaction and a subsequent resolution of the Credit Derivatives Determinations Committee determines that Transaction Auction Settlement Terms will be published, LCH SA shall have no obligation to create Settlement Matched Pairs in accordance with Section 6.2 (*Creation and Notification of Settlement Matched Pairs*).

Unless settlement has occurred with respect to any Settlement Matched Pairs prior to such determination, any such Settlement Matched Pairs so created shall be deemed not to have been created and any notices delivered in connection with such Settlement Matched Pairs shall be deemed to be ineffective.

LCH SA shall effect any relevant registrations in the TIW in order to reflect the application of Auction Settlement as the Settlement Method in respect of the relevant Cleared Transactions.

6.17 Physical Settlement Costs

If the Fallback Settlement Method applies in respect of a Cleared Transaction and any amounts are payable in relation to any costs or expenses of Physical Settlement, then:

- (a) where such amount would otherwise be payable by LCH SA to Matched Buyer or Matched Seller (the **Recipient**), such amount shall be payable to the Recipient by the other party to the Matched Contract who is not the Recipient, as designee to pay such amount on behalf of LCH SA; and
- (b) where such amount would otherwise be payable to LCH SA by Matched Buyer or Matched Seller (the Payer), such amount shall be payable by the Payer to the other party to the Matched Contract who is not the Payer, as designee to receive such amount on behalf of LCH SA.

Any Matched Seller or Matched Buyer who is required to make any payment as designee on behalf of LCH SA pursuant to this Section 6.17 shall not be entitled to any reimbursement from LCH SA in respect of such amount.

Any Matched Seller or Matched Buyer who receives any payment pursuant to this Section 6.17 shall not be obliged to remit to LCH SA any such amount so received (without prejudice to any rights of LCH SA where there is a failure to Deliver).

LCH SA shall not be liable to a Matched Buyer or a Matched Seller for any of the costs and expenses of Physical Settlement of any Cleared Transaction.

Formatted: Indent: Left: 1.5 cm, Keep with next

6.18 Representations and Agreements relating to Physical Settlement

(a) Claims by Matched Seller against LCH SA in respect of Obligations Delivered by Physical Settlement

If a Matched Seller pursues a claim against LCH SA under Sections 9.2(a), (b), (c)(i) or (c)(iv) of the 2003 ISDA Credit Derivatives Definitions in respect of any obligations Delivered by way of Physical Settlement of any Matched Seller Contract, then:

- (i) notwithstanding any provision of the 2003 ISDA Credit Derivatives Definitions to the contrary, LCH SA shall only be liable to make payments to that Matched Seller in respect of that claim to the extent of amounts recovered, including, without limitation, any amounts recovered by way of set-off or netting, by LCH SA from or on behalf of the related Matched Buyer in respect of any corresponding claims under or in connection with the Matched Buyer Contract (including, without limitation, following the declaration of an Event of Default in respect of such Matched Buyer) and after deducting any costs and expenses reasonably incurred by LCH SA in pursuing such corresponding claims for recovery under or in connection with the Matched Buyer Contract;
- (ii) LCH SA will use reasonable efforts to pursue any claim it may have against the related Matched Buyer but, notwithstanding Section 6.18(a)(i) above, LCH SA will, in the pursuit of such claims, act as though its liability to Matched Seller under the Matched Seller Contract was not limited or restricted in any way; and
- (iii) references to indemnity provisions set out in Section 9.2(a), 9.2(b) and 9.2(c)(iv) of the 2003 ISDA Credit Derivatives Definitions shall, in connection with a Matched Buyer Contract, be interpreted without regard to the limitations to Matched Seller's recourse to LCH SA under the corresponding Matched Seller Contract imposed by the provisions of this Section 6.18(a) and such limitations shall not restrict a Matched Seller from making a claim or raising a Dispute.

(b) Right of Matched Seller to enforce against Matched Buyer

Without prejudice to paragraph (a) above, a Matched Seller shall be entitled to enforce Sections 9.2(a), (b), (c)(i) and (c)(iv) of the 2003 ISDA Credit Derivatives Definitions against the related Matched Buyer in respect of any obligations Delivered by way of Physical Settlement of any Physically Settled Cleared Transaction under the Contracts (Rights of Third Parties) Act 1999 as though Matched Seller were party to the relevant Matched Buyer Contract in place of LCH SA.

(c) Satisfaction of Claim by Matched Buyer discharges Liabilities owed to and by LCH SA

For the avoidance of doubt, if a Matched Buyer satisfies a claim made by a Matched Seller directly against the relevant Matched Buyer under Sections 9.2(a), (b), (c)(i) or (c)(iv) of the 2003 ISDA Credit Derivatives Definitions pursuant to the rights granted to

such Matched Seller under paragraph (b) above, such satisfaction shall also constitute complete satisfaction and discharge of the corresponding liability of such Matched Buyer to LCH SA and the corresponding liability of LCH SA to such Matched Seller in respect of such claim.

(d) Effect of Illegality or Tax or other Expense resulting from Designation through Creation of Matched Pairs

The last sentence of Section 9.2(c)(iv) of the 2003 ISDA Credit Derivatives Definitions shall not operate so as to prevent LCH SA from creating any Matched Pair in accordance with this CDS Clearing Supplement, and LCH SA shall have no obligation to grant any indemnity with respect to any Tax, loss or cost to any Matched Buyer or Matched Seller thereunder.

If the circumstances contemplated at Section 9.2(c)(iv)(A) or (B) of the 2003 ISDA Credit Derivatives Definitions apply in respect of any Physically Settled Cleared Transaction and any related Settlement Matched Pair (in the case of (B), as notified by the Clearing Member which is the non-designating party for such purpose to the other Clearing Member comprised in the relevant Settlement Matched Pair prior to the first Delivery Date), then an impossibility shall be deemed to have occurred with respect to Physical Settlement of the relevant Physically Settled Cleared Transactions, and Section 9.3 (Partial Cash Settlement due to Impossibility or Illegality) of the 2003 ISDA Credit Derivatives Definitions (as amended by this CDS Clearing Supplement) shall apply.

6.19 Miscellaneous Provisions relating to Physical Settlement

(a) Margin

For the avoidance of doubt, LCH SA will continue to call all Margin and such Margin will remain due in relation to any Cleared Transaction to which Physical Settlement applies until:

- LCH SA has received a Notice of Physical Settlement from each Clearing Member:
- (ii) LCH SA has received a No Physical Settlement Confirmation from each Clearing Member; or
- (iii) in each case, any related dispute is referred to and finally resolved by arbitration or by litigation, as applicable, in accordance with the CDS Dispute Resolution Protocol, subject to the provisions of Sections 8 and 9 of the Procedures.

(b) Notification of Completion of Physical Settlement

Upon completion of Physical Settlement by the relevant Matched Pair of a Physically Settled Cleared Transaction, Matched Buyer and Matched Seller shall inform LCH SA as soon as reasonably practicable and in any event before the date falling two Transaction Business Days following such completion (the **Physical Settlement Confirmation**) in the form set out at Appendix VII hereto.

(c) Notification that Physical Settlement will not occur

If no Notice of Physical Settlement has been delivered within the relevant time period permitted for such delivery in accordance with the terms of the relevant Physically Settled Cleared Transaction and, accordingly, Physical Settlement will not, under the terms of the relevant Physically Settled Cleared Transaction, occur, Matched Buyer and Matched Seller shall inform LCH SA as soon as reasonably practicable thereafter and in any event before the date falling two Transaction Business Days following the relevant date after which delivery of a Notice of Physical Settlement is no longer permitted in accordance with the terms of the relevant Physically Settled Cleared Transaction (the No Physical Settlement Confirmation Deadline) by notice (a No Physical Settlement Confirmation) in the form set out at Appendix VIII hereto.

(d) Dispute regarding Physical Settlement

If LCH SA receives a Physical Settlement Confirmation or No Physical Settlement Confirmation from one relevant Matched Buyer or Matched Seller only at or prior to the relevant Physical Settlement Confirmation Deadline or No Physical Settlement Confirmation Deadline, as the case may be, there shall be deemed to be a Dispute with respect to the Physically Settled Cleared Transactions between LCH SA and each relevant Clearing Member.

7. DELIVERY OF NOTICES AND FALLBACKS

7.1 General Rules relating to Notices

(a) Methods of Delivery and deemed Time of Delivery

Subject to Section 7.2 (*Oral Notices*) and without prejudice to Section 1.10 (*Requirements Regarding Notices*) and the remaining provisions of the 2003 ISDA Credit Derivatives Definitions (including, for the avoidance of doubt, in relation to notices permitted to be given orally), any notice or other communication in respect of any Cleared Transaction may be given in any manner described below and will be deemed effective as indicated:

- (i) if delivered in person or by courier, on the date and at the time it is delivered;
- (ii) if sent by facsimile transmission, on the date it is received by a responsible employee of the recipient in legible form (it being agreed that the burden of

proving receipt will be on the sender and will not be met by a transmission report generated by the sender's facsimile machine);

- (iii) if sent by certified or registered mail (airmail, if overseas) or the equivalent (return receipt requested), on the date it is delivered or its delivery is attempted; or
- (iv) if sent by electronic messaging system (including e-mail or any other electronic access solution established by LCH SA for such purpose), on the date it is received.

unless the date of that delivery (or attempted delivery) or that receipt, as applicable, is not a Transaction Business Day or that communication is delivered (or attempted) or received, as applicable, pursuant to the above after 6.00 p.m. in the location of the recipient on a Transaction Business Day, in which case that communication will be deemed given and effective on the first following day that is a Transaction Business Day.

(b) Notices from or to LCH SA

Any such notice or communication given by LCH SA to a Clearing Member or vice versa shall (except where delivered via a relevant DTCC Notice Facility) be given to the address or number previously specified in or previously notified for the relevant purpose in accordance with the CDS Admissions Agreement or the Procedures.

(c) Manual Notices between Clearing Members

Notices given by a Clearing Member to another Clearing Member comprised in a relevant Matched Pair and which are not delivered via a relevant DTCC Notice Facility shall be given to the address or number notified by LCH SA to the deliveror pursuant to Sections 5.1 (*Creation and Notification of Restructuring Matched Pairs*) or 6.2 (*Creation and Notification of Settlement Matched Pairs*).

Such notices shall only be deemed to be delivered effectively by LCH SA through the relevant Clearing Member as its designee as against the recipient where the address or number so notified by LCH SA corresponds in all material respects to the address or number, as applicable, specified by such recipient in or pursuant to the CDS Admissions Agreement.

(d) No Obligation on LCH SA to verify Signatories

LCH SA shall have no obligation to verify the authority of any signatory of any notice delivered by any Clearing Member directly pursuant to this Section 7 (*Delivery of Notices and Fallbacks*).

7.2 Oral Notices

Notwithstanding the provisions of the 2003 ISDA Credit Derivatives Definitions, where, by way of exception as contemplated in this Section 7, Credit Event Notices and Notices to Exercise Movement Option are to be delivered directly to LCH SA (and not via a relevant DTCC Notice Facility), such notices may not be delivered by telephone.

7.3 Credit Event Notices and NEMOs

(a) Credit Event Notices and NEMOs to be given via DTCC

Credit Event Notices and Notices to Exercise Movement Option shall be delivered by way of the relevant DTCC Notice Facility, save if and as expressly stated to the contrary in this Section 7 or expressly agreed by LCH SA. The deemed time of delivery of any such notices shall be as set out in the DTCC Rules from time to time.

(b) Credit Event Notices and NEMOs delivered in respect of CCM Client Transaction

If a Matched Buyer or a Matched Seller that is a CCM receives a valid Credit Event Notice or Notice to Exercise Movement Option from its CCM Client in respect of a CCM Client Transaction by way of the relevant DTCC Notice Facility, such notice shall be deemed also to be a Credit Event Notice or Notice to Exercise Movement Option (as applicable) for the purposes of the corresponding Matched Contracts of the Restructuring Matched Pair.

7.4 Notification of DTCC Failure and Resolution

(a) LCH SA to notify Clearing Members of DTCC Failure Event

If DTCC notifies LCH SA or LCH SA otherwise becomes aware that the relevant DTCC Notice Facility is or will be unavailable to process all or substantially all Credit Event Notices or Notices to Exercise Movement Option, as applicable, with respect to a Restructuring Credit Event in a timely manner (a DTCC Failure Event), then LCH SA will, as soon as reasonably practicable and in any event within one hour of such notification or of LCH SA becoming aware of such non-availability, as applicable, notify all relevant Clearing Members of such occurrence, including the time at which such failure occurred (or, if LCH SA is not notified of such time by or on behalf of DTCC, the time at which LCH SA received the relevant notification from or on behalf of DTCC or becomes aware of the relevant non-availability) (the DTCC Failure Event Time). LCH SA shall also publish such information on its Website as soon as reasonably practicable after becoming aware of a DTCC Failure Event.

(b) LCH SA to notify Clearing Members of Resolution of DTCC Failure Event

If, subsequent to a DTCC Failure Event, DTCC (or a third party designated under the DTCC Rules from time to time) notifies LCH SA that a DTCC Failure Event previously notified to Clearing Members is no longer in effect, LCH SA will as soon as reasonably

practicable notify all relevant Clearing Members accordingly, including the time (the **DTCC Resolution Time**) (which must be no earlier than 30 minutes following the time of such notification) at which such DTCC Failure Event is deemed to have been resolved and following which the relevant DTCC Notice Facility is operative for the purposes of delivery of relevant notices relating to Restructuring Cleared Transactions (including transactions in respect of which notices may have been delivered directly pursuant to Section 7.5 (*Consequences of DTCC failure*). LCH SA shall also publish such information on its Website as soon as reasonably practicable after notifying Clearing Members that the DTCC Failure Event is no longer in effect and of the DTCC Resolution Time.

(c) Notices given prior to DTCC Resolution Time to be confirmed to DTCC

LCH SA and, to the extent so requested by LCH SA, each Clearing Member shall, as soon as reasonably practicable and to the extent permitted by DTCC, provide or confirm to DTCC details of any relevant notices (in the case of a Clearing Member, being any relevant notices delivered directly by such Clearing Member) given in respect of Cleared Transactions prior to the DTCC Resolution Time, so as to permit delivery of subsequent notices in respect of such Cleared Transactions via the relevant DTCC Notice Facility.

7.5 Consequences of DTCC Failure

From (and including) the DTCC Failure Event Time to (but excluding) the DTCC Resolution Time:

- (a) Section 7.3(a) (*Credit Event Notices and NEMOs to be given via DTCC*) shall not apply and accordingly Credit Event Notices and Notices to Exercise Movement Option shall be delivered directly (and not via the relevant DTCC Notice Facility);
- (b) any notice delivered via the relevant DTCC Notice Facility prior to the DTCC Failure Event Time will be valid and will not be affected by such DTCC Failure Event; and
- (c) any notice delivered or purported to be delivered via the relevant DTCC Notice Facility at or following the DTCC Failure Event Time but prior to the DTCC Resolution Time will not be valid and effective.

Section 7.3(a) (*Credit Event Notices and NEMOs to be given via DTCC*) shall apply with effect from the DTCC Resolution Time and, accordingly, any notice thereafter delivered or purported to be delivered directly (and not via the relevant DTCC Notice Facility) will not be valid and effective.

7.6 Clearing Member Communications Failure Event

(a) Right to deliver Notices manually following Clearing Member Communications

Failure Event

If a Clearing Member is affected by a significant communications or information technology failure resulting in it being impossible or impractical for such Clearing Member to deliver any Credit Event Notice in relation to a Restructuring Credit Event or any Notice to Exercise Movement Option via a relevant DTCC Notice Facility (a Clearing Member Communications Failure Event) it may, notwithstanding Section 7.3(a) (Credit Event Notices and NEMOs to be given via DTCC) deliver Credit Event Notices and Notices to Exercise Movement Option directly (and not via the relevant DTCC Notice Facility).

(b) Clearing Member to notify LCH SA of Occurrence of Clearing Member Communications Failure Event

Following the occurrence of a Clearing Member Communications Failure Event, the affected Clearing Member shall, within one hour of delivering any Credit Event Notice or Notice to Exercise Movement Option directly, deliver to LCH SA a notice (in the form set out at Appendix IX hereto) signed by a senior officer (such as a managing director or equivalent) of such Clearing Member certifying that it is affected by a Clearing Member Communications Failure Event (or, if such Clearing Member is unable to deliver such notice in writing, orally by telephone).

LCH SA shall notify all Clearing Members accordingly as soon as reasonably practicable and in any event within one hour of receipt of any such notification.

(c) Notices to Clearing Member affected by Clearing Member Communications Failure Event

For the avoidance of doubt, Section 7.3(a) (*Credit Event Notices and NEMOs to be given via DTCC*) shall continue to apply in respect of notices given to the affected Clearing Member by Clearing Members comprised in any Matched Pair in respect of which the affected Clearing Member is a Matched Buyer or Matched Seller.

(d) Notification of Resolution of Clearing Member Communications Failure Event

As soon as reasonably practicable upon a Clearing Member ceasing to be subject to a Clearing Member Communications Failure Event, it shall notify LCH SA accordingly (in the form set out at Appendix X hereto) and thereupon Section 7.3(a) (*Credit Event Notices and NEMOs to be given via DTCC*) shall apply and, accordingly, any Credit Event Notice or Notice to Exercise Movement Option thereafter delivered or purported to be delivered directly (and not via the relevant DTCC Notice Facility) will not be valid and effective.

(e) Clearing Member Duty to Mitigate

A Clearing Member which is subject to a Clearing Member Communications Failure Event shall use reasonable endeavours to mitigate the operational impact on other Clearing Members and LCH SA of any Clearing Member Communications Failure Event, to cure such Clearing Member Communications Failure Event as soon as possible and

to ensure that the circumstances giving rise to the relevant Clearing Member Communications Failure Event do not recur.

(f) Breach by Clearing Member does not Invalidate Valid Notices

Without prejudice to any other rights or remedy of LCH SA, any breach by a Clearing Member of the provisions of this Section 7.6 shall not cause any Credit Event Notice or Notice to Exercise Movement Option delivered otherwise than in accordance with the relevant Restructuring Cleared Transaction, which would otherwise be valid and effective, to be invalid or ineffective.

7.7 Clearing Member Acknowledgements

(a) Duty to deliver Clearing Member Acknowledgements

Each Clearing Member shall notify LCH SA or deliver a copy to LCH SA of any notice delivered or received by such Clearing Member to or from another Clearing Member comprised in a Matched Pair, including, without limitation, any Credit Event Notice or Notice to Exercise Movement Option which was delivered or received directly (and not via a DTCC Notice Facility) pursuant to Sections 7.5 (Consequences of DTCC Failure) or 7.6 (Clearing Member Communications Failure Event), and which such Clearing Member asserts or acknowledges was effective for the purposes of this CDS Clearing Supplement and the relevant Cleared Transactions (such notification, or delivery of such copy, in respect of any relevant notice, a Clearing Member Acknowledgement) by no later than 6.00 p.m. on:

- in the case of a Notice of Physical Settlement or NOPS Amendment Notice, the date falling two Transaction Business Days following the date on which such notice was sent; or
- (ii) in the case of any other notice, on the last date on which such notice could validly be sent,

(in each case, the Notice Acknowledgement Deadline).

(b) Clearing Member Acknowledgement received from both Clearing Members

Where LCH SA receives a Clearing Member Acknowledgement in respect of any notice from both relevant Clearing Members comprised in a Matched Pair at or prior to the Notice Acknowledgement Deadline, LCH SA shall perform its obligations in respect of the relevant Cleared Transactions in accordance with and subject to the remaining provisions of the CDS Clearing Documentation.

(c) Clearing Member Acknowledgement received from one Clearing Member

Where LCH SA receives a Clearing Member Acknowledgement in respect of any notice from one relevant Clearing Member only at or prior to the Notice Acknowledgement Deadline, the provisions of Section 7.11 (*Disputes as to Notices*) shall apply and LCH

File No. SR-LCH SA-2022-007 Page 82 of 439

PUBLIC

SA and each relevant Clearing Member shall perform their obligations in respect of the relevant Cleared Transactions in accordance with and subject to the remaining provisions of the CDS Clearing Documentation and the terms of any final resolution of the relevant dispute, as agreed between the relevant parties or as determined by arbitration or by litigation, as applicable, in accordance with the CDS Dispute Resolution Protocol, subject to the provisions of Sections 8 and 9 of the Procedures.

In such case, LCH SA shall notify the Clearing Member from which it has not received a Clearing Member Acknowledgement of the asserted delivery or, as applicable, receipt of the relevant notice (in the case of a Credit Event Notice or Notice to Exercise Movement Option, through the reports referred to at Section 5.6 (*Reports*)).

(d) No Clearing Member Acknowledgement received

Where LCH SA does not receive Clearing Member Acknowledgement or confirmation of valid delivery in respect of any notice from either Clearing Member in the relevant Matched Pair on or prior to the relevant Notice Acknowledgement Deadline, the rights and obligations of LCH SA as against each relevant Clearing Member, and vice versa, shall be construed as if no such notice had been given.

(e) Consequences of no Clearing Member Acknowledgement being received

Where sub-section (d) above is applicable, the following provisions shall apply:

(i) Notices other than Notices of Physical Settlement and NOPS Amendment Notices

Save in the case of a Notice of Physical Settlement or NOPS Amendment Notices, an amount shall be payable between the Clearing Members equal to the difference between the value of the Matched Buyer Contract had Clearing Member Acknowledgement been given to LCH SA on a timely basis and the value of such contract in the absence of such acknowledgement; such difference in value shall be determined as of the earlier of the day on which notice is given by any relevant Clearing Member that such amount is due and payable and, in the case of a Credit Event Notice or Notice to Exercise Movement Option, the eighth Transaction Business Day following the Auction Settlement Date, no Auction Announcement Date or Auction Cancellation Date, as applicable or otherwise the eighth Transaction Business Day following the last day on which such notice would validly have been delivered, by reference to the relevant Auction Settlement Amount or end of day contributed prices, in each case if available.

(ii) Notices of Physical Settlement and NOPS Amendment Notices

Where the relevant notice is a Notice of Physical Settlement or a NOPS Amendment Notice, the relevant Clearing Members shall acquire rights as against the other as though party to a bilateral credit default swap transaction

File No. SR-LCH SA-2022-007 Page 83 of 439

PUBLIC

on the terms of the related Matched Buyer Contract (including, without limitation, as to the satisfaction of the applicable Conditions to Settlement) and the Notice of Physical Settlement shall be deemed to have been given in respect of such transaction. Any resulting payment shall be due and payable two Transaction Business Days following the giving of a notice that such amount is due and payable.

In each case, the relevant Clearing Members shall have enforcement rights as against each other pursuant to the Contracts (Rights of Third Parties) Act 1994 in respect of any resulting payments and deliveries; LCH SA shall have no liability in respect thereof.

7.8 Failure to notify Matched Pairs

Notwithstanding Section 5.3 (*Triggering of Restructuring Cleared Transactions*) and 6.5 (*Notices of Physical Settlement*), if LCH SA does not notify the relevant Clearing Members of Restructuring Matched Pairs created pursuant to Section 5.1 (*Creation and Notification of Restructuring Matched Pairs*) on or prior to the RMP Notification Deadline or Settlement Matched Pairs and related information specified in Section 6.2 (*Creation and Notification of Settlement Matched Pairs*) by the SMP Notification Deadline, as applicable:

- the relevant Clearing Members may deliver Credit Event Notices, Notices to Exercise Movement Option, Notices of Physical Settlement or NOPS Amendment Notices to LCH SA, and vice versa;
- (b) the relevant Cleared Transactions shall be subject to Physical Settlement in accordance with their terms; and
- (c) the provisions of Sections 2.4 (Amendments to 2003 ISDA Credit Derivatives Definitions), 6 (Physical Settlement) and 8 (Matched Pair Designations and Notices) shall not apply and the terms of this CDS Clearing Supplement shall be construed accordingly.

For such purpose, Section 7.3 (*Credit Event Notices and NEMOs*) shall not apply in respect of notices given by the affected Clearing Members and accordingly Credit Event Notices and Notices to Exercise Movement Option shall be delivered directly (and not via the relevant DTCC Notice Facility).

7.9 Failure to notify Matching Information

If LCH SA notifies relevant Clearing Members of Restructuring Matched Pairs created pursuant to Section 5.1 (*Creation and Notification of Restructuring Matched Pairs*) on or prior to the RMP Notification Deadline, but does not notify DTCC of relevant Matching Information on or prior to the RMP Notification Deadline, then Section 7.3(a) (*Credit Event Notices and NEMOs to be given via DTCC*) shall not apply in respect of notices to be delivered by affected Clearing Members and accordingly Credit Event Notices and Notices to Exercise Movement Option shall be delivered directly as between Clearing Members (as designees of LCH SA) (and not via the relevant DTCC Notice Facility).

7.10 Uncertain Delivery

(a) Manual Notice permitted if Delivery of Notice in DTCC uncertain

Notwithstanding Section 7.3(a) (*Credit Event Notices and NEMOs to be given via DTCC*), where such notices are permitted to be delivered by means other than the relevant DTCC Notice Facility pursuant to this Section 7, and a Clearing Member is uncertain as to whether or not a Credit Event Notice or Notice to Exercise Movement Option (as applicable) it (or, in the case of a CCM, its CCM Client) attempted to deliver via a DTCC Notice Facility has:

- (i) actually been delivered; or
- (ii) was delivered prior to the DTCC Failure Time,

that Clearing Member shall be entitled to deliver such a notice directly to any Clearing Member comprised in a relevant Matched Pair (as designee of LCH SA) specifying that such notice is only to be effective to the extent that the other purported notice is not effective.

(b) Details to be provided of Uncertain Notice

If a Clearing Member delivers a manual notice pursuant to sub-section (a) (Manual Notice permitted if Delivery of Notice in DTCC uncertain) above, such Clearing Member shall be required to provide (together with such notice) sufficient details of the notice attempted to be given by way of the relevant DTCC Notice Facility so as to allow the other Clearing Member and LCH SA to identify the communication concerned.

(c) DTCC Notice delivered successfully

If the first Credit Event Notice or Notice to Exercise Movement Option (as applicable) to which the manual notice delivered pursuant to sub-section (a) (Manual Notice permitted if Delivery of Notice in DTCC uncertain) above relates was actually delivered successfully, any subsequent Credit Event Notice or Notice to Exercise Movement Option delivered shall be deemed not to have been delivered.

7.11 Disputes as to Notices

If any Clearing Member comprised in a Matched Pair where one such party is acting as designee of LCH SA disputes the effective delivery in accordance with the terms of the relevant Cleared Transactions of any notice delivered directly (and not via a relevant DTCC Notice Facility) in accordance with this Section 7 (*Delivery of Notices and Fallbacks*) (and for such purposes, a dispute between the relevant Clearing Member and LCH SA shall be deemed to have arisen if LCH SA receives a Clearing Member Acknowledgement from one relevant Clearing Member only in respect of any such notice as contemplated at Section 7.7(c) (*Clearing Member Acknowledgement received from one Clearing Member*)):

File No. SR-LCH SA-2022-007 Page 85 of 439

PUBLIC

- (a) LCH SA shall be entitled in accordance with the Procedures to calculate and call for Margin with respect to each such Clearing Member on the basis of the maximum requirement for such Clearing Member that could result from any foreseeable outcome of such dispute;
- (b) following final resolution of such dispute by arbitration or by litigation, as applicable, in accordance with the CDS Dispute Resolution Protocol, subject to the provisions of Sections 8 and 9 of the Procedures, the Clearing Members comprised in the relevant Matched Pair shall take such actions with respect to the Cleared Transactions the subject of such dispute as LCH SA determines appropriate to give effect to any relevant binding resolution; and
- (c) without prejudice to its obligations upon final resolution of the dispute, pending final resolution of any such dispute, LCH SA shall not be obliged to take any step pursuant to the terms of the relevant Cleared Transactions which would be required to have been taken by it had the relevant notice been validly delivered.

Each relevant Clearing Member must promptly notify LCH SA of any such disputes (in the form set out at Appendix XI hereto).

8. MATCHED PAIR DESIGNATIONS AND NOTICES

8.1 Creation of Matched Pairs

LCH SA will create Matched Pairs where required to do so pursuant to Section 5.1 (*Creation and Notification of Restructuring Matched Pairs*) or 6.2 (*Creation and Notification of Settlement Matched Pairs*) using a matching procedure that matches CDS Sellers with CDS Buyers pursuant to an algorithm incorporating the following principles:

- (a) the procedure seeks to create Matched Pairs between the same Clearing Member to the extent it is possible to do so before creating Matched Pairs between different Clearing Members and, for this purpose, in the context of CCMs, the procedure will create Matched Pairs separately for CCMs and their CCM Clients (individually where the CCM Client is allocated to a single CCM Client Margin Account or together where a set of CCM Clients is allocated to a single CCM Client Margin Account) and Clearing Member will be construed accordingly;
- (b) the procedure seeks to minimise the number of Matched Pairs (and accordingly, largest positions will be matched first);
- (c) each Matched Pair will, to the extent possible, have an aggregate Restructuring Matched Pair Amount or, as applicable, Settlement Matched Pair Delivery Amount, which meets the criteria as set out in a Clearing Notice; and
- (d) LCH SA will allocate a Restructuring Matched Pair Amount or, as applicable, Settlement Matched Pair Delivery Amount to each Matched Pair such that:

File No. SR-LCH SA-2022-007 Page 86 of 439

PUBLIC

- (i) the sum of all Restructuring Matched Pair Amounts or, as applicable, Settlement Matched Pair Delivery Amounts, of each CDS Buyer is equal to the aggregate Floating Rate Payer Calculation Amounts in respect of all (A) Initial Single Name Cleared Transactions, Restructuring Cleared Transactions (created following the occurrence of a previous Restructuring Credit Event), Spin-off Single Name Cleared Transactions and Resulting Single Name Cleared Transactions in respect of which the Fallback Settlement Method applies to the CDS Type for such Cleared Transaction, as applicable, to which such CDS Buyer is a party; and
- (ii) the sum of all Restructuring Matched Pair Amounts or, as applicable, Settlement Matched Pair Delivery Amounts, of each CDS Seller is equal to the aggregate Floating Rate Payer Calculation Amounts in respect of (A) Initial Single Name Cleared Transactions, Restructuring Cleared Transactions (created following the occurrence of a previous Restructuring Credit Event), Spin-off Single Name Cleared Transactions and Resulting Single Name Cleared Transactions in respect of which the Fallback Settlement Method applies to the CDS Type for such Cleared Transaction, as applicable, to which such CDS Seller is a party.

Notwithstanding the above, if the Fallback Settlement Method applies in relation to a Cleared Transaction and a Restructuring Credit Event, the Restructuring Matched Pairs previously created pursuant to Section 5.1 (*Creation of Restructuring Matched Pairs*) and this Section 8.1 shall be deemed to be Settlement Matched Pairs created in accordance with Section 6.2 (*Creation and Notification of Settlement Matched Pairs*) and LCH SA shall have no obligation to create Settlement Matched Pairs in respect of such Cleared Transaction.

8.2 Registration of new Cleared Transactions and Removal of original Cleared Transactions

To the extent that any Cleared Transaction created pursuant to Section 5.2 (*Creation of Restructuring Cleared Transactions*) or Section 6.3 (*Physically Settled Cleared Transactions*) is not automatically registered in accordance with the DTCC Rules, LCH SA shall register such new Cleared Transaction in the TIW in accordance with the DTCC Rules prior to 6.00 p.m. on the date on which the RMP Notification Deadline or SMP Notification Deadline (as applicable) falls. In respect of CCMs and CCM Clients only, such registration by LCH shall also result in the automatic registration of any amendments made to the corresponding CCM Client Transactions.

In addition, LCH SA will, on behalf of the relevant Clearing Member, send an "Exit" message to the TIW in accordance with the DTCC Rules to terminate and remove the corresponding original Cleared Transaction(s) from the TIW prior to 6.00 p.m. on the date on which the RMP Notification Deadline or SMP Notification Deadline (as applicable) falls.

8.3 Matched Buyer Contracts

In respect of each Matched Buyer Contract which is the subject of a Matched Pair, LCH SA, pursuant to Section 9.2(c)(iv) of the 2003 ISDA Credit Derivatives Definitions (amended as set out at Section 2.4 (*Amendments to 2003 ISDA Credit Derivatives Definitions*)), as designator, shall be deemed to have designated Matched Seller in such Matched Pair as its designee:

- (a) to receive on its behalf from, and to deliver on its behalf to, Matched Buyer of the Matched Pair any applicable notices or certifications in accordance with the terms of the applicable Cleared Transaction (other than notices required to be delivered via a DTCC Notice Facility):
- (b) other than in respect of the Physical Settlement Amount relating to the settlement of Non-DVP Obligations as referred to in Section 6.6 (*Delivery of Non-DVP Obligations*), to pay on behalf of LCH SA any applicable Physical Settlement Amount in accordance with the terms of the applicable Physically Settled Cleared Transaction, and to pay to, and to receive from, Matched Buyer of the Matched Pair, in each case, on behalf of LCH SA, any other amounts due and payable (including costs and expenses of settlement due under the applicable Matched Buyer Contract); and
- (c) to take Delivery, on behalf of LCH SA, of Deliverable Obligations from Matched Buyer of the Settlement Matched Pair.

The relevant Matched Seller shall be deemed to have accepted such designation upon notification of the relevant Matched Pair created and notified in accordance with the provisions of this CDS Clearing Supplement.

8.4 Matched Seller Contracts

In respect of each Matched Seller Contract which is the subject of a Matched Pair, LCH SA, pursuant to Section 9.2(c)(iv) of the 2003 ISDA Credit Derivatives Definitions (as amended pursuant to Section 2.4 (*Amendments to 2003 ISDA Credit Derivatives Definitions*) above), as designator, shall be deemed to have designated Matched Buyer in such Matched Pair as its designee:

- to receive on its behalf from, and to deliver on its behalf to, Matched Seller of the Matched Pair any applicable notices or certifications in accordance with the terms of the applicable Cleared Transaction (other than notices required to be delivered via a DTCC Notice Facility);
- (b) other than in respect of any Physical Settlement Amount relating to the settlement of Non-DVP Obligations as referred to in Section 6.6 (*Delivery of Non-DVP Obligations*), to receive on behalf of LCH SA the applicable Physical Settlement Amount in accordance with the terms of any applicable Physically Settled Cleared Transaction, and to pay to, and to receive from, Matched Seller of the Matched Pair, in each case, on behalf of LCH SA, any other amounts due and payable (including costs and expenses of settlement due under the Matched Seller Contract); and

(c) to Deliver, on behalf of LCH SA, the relevant Deliverable Obligations to Matched Seller of the Settlement Matched Pair.

The relevant Matched Buyer shall be deemed to have accepted such designation upon notification of the relevant Matched Pair created and notified in accordance with the provisions of this CDS Clearing Supplement.

8.5 Exercise of Rights

In relation to each Matched Pair:

- (a) the exercise of any rights by Matched Buyer against LCH SA under a Matched Buyer Contract (other than the right to give any notice via DTCC Notice Facility) shall be deemed to constitute the exercise of equal and simultaneous rights by LCH SA against Matched Seller under the Matched Seller Contract of the relevant Matched Pair; and
- (b) the exercise of any rights by Matched Seller against LCH SA under a Matched Seller Contract (other than a right to give notice under a DTCC Notice Facility) shall be deemed to constitute the exercise of equal and simultaneous rights by LCH SA against Matched Buyer under the Matched Buyer Contract of the relevant Matched Pair.

8.6 Clearing Member matched with Itself

(a) Notices

In the event that:

- Matched Buyer and Matched Seller of a Matched Pair pursuant to this Section 8 (Matched Pair Designations and Notices) is the same Clearing Member; and
- (ii) notwithstanding Section 7.3(a) (Credit Event Notices and NEMOs to be given via DTCC), a notice or certification is permitted to be delivered in respect of one of the Cleared Transactions forming part of such Matched Pair by means other than the relevant DTCC Notice Facility pursuant to Section 7 (Delivery of Notices and Fallbacks),

such notice shall be deemed to be given upon such Clearing Member sending a Clearing Member Acknowledgement to LCH SA pursuant to Section 7.7(a) (*Duty to deliver Clearing Member Acknowledgements*) above in respect of such notice and Section 7.7(b) (*Clearing Member Acknowledgement received from both Clearing Members*) shall apply.

b) Payments and Deliveries

In the event that:

(iii) Matched Buyer and Matched Seller of a Matched Pair pursuant to this Section 8 (Matched Pair Designations and Notices) is the same Clearing Member; and

(iv) such Clearing Member is required to make a payment or delivery pursuant to the terms of one of the Cleared Transactions forming part of such Matched Pair as designate of LCH SA,

such payment or delivery shall be deemed to have been made upon such Clearing Member giving notice to LCH SA in accordance with Section 7.1 (*General Rules relating to Notices*) that such payment or delivery should be deemed to have been made for the purposes of such Cleared Transaction.

8.7 Notices

In relation to each Matched Pair:

- (a) where Matched Buyer validly delivers or serves any notice to Matched Seller as designee of LCH SA in accordance with the terms of a relevant Matched Buyer Contract, such notice shall additionally be effective as a notice given by such Matched Buyer as designee of LCH SA to Matched Seller for the purposes of the relevant Matched Seller Contract; and
- (b) where Matched Seller validly delivers or serves any notice to Matched Buyer as designee of LCH SA in accordance with the terms of a relevant Matched Seller Contract, such notice shall additionally be effective as a notice given by such Matched Seller as designee of LCH SA to Matched Buyer for the purposes of the relevant Matched Buyer Contract.

9. SELF-REFERENCING TRANSACTIONS

9.1 Occurrence of Clearing Member Self Referencing Transaction

(a) Duty to notify

In respect of a Single Name Cleared Transaction that is registered in the House Account Structure of a Clearing Member, the relevant Clearing Member shall, unless prohibited from so doing by applicable law or its internal policies, notify LCH SA as soon as reasonably practicable if:

- such Clearing Member is or consolidates or amalgamates with, or merges into, or transfers all or substantially all of its assets to, the Reference Entity in relation to such Single Name Cleared Transaction, or enters into any agreement in respect of any of the foregoing;
- (ii) such Clearing Member and the Reference Entity in relation to such Single Name Cleared Transaction are or become Affiliates; or
- (iii) in respect of a Restructuring Cleared Transaction, such Clearing Member is or becomes the Reference Entity in relation to such Restructuring Cleared Transaction as a result of the occurrence of the relevant Restructuring Credit Event.

(b) Auction of Affected Transactions

Following receipt of any such notification or occurrence of a Clearing Member Self Referencing Transaction due to a registration of Transfer Positions under Clause 6 of the CDS Default Management Process or a transfer of Client Cleared Transactions to the House Trade Account under TITLE V, Chapter 4 or TITLE VI, Chapter 4 of the CDS Clearing Rule Book, LCH SA will, unless otherwise agreed in consultation with the CDS Default Management Group and the affected Clearing Member, and with the assistance of the CDS Default Management Group, conduct an auction process to liquidate the relevant Single Name Cleared Transaction and enter into an equivalent Single Name Cleared Transaction with a Clearing Member other than the affected Clearing Member.

(c) Alternative Action

If LCH SA, after consultation with the CDS Default Management Group and the affected Clearing Member, believes that the circumstances are such that an auction may be inappropriate, LCH SA may take such other action in consultation with the Risk Committee as it considers reasonably necessary to achieve its primary aim in these circumstances of addressing the risks resulting from a Clearing Member being party to a Clearing Member Self Referencing Transaction, while endeavouring, as far as is reasonably practicable in the circumstances without prejudicing the achievement of the primary aim, to avoid materially and adversely affecting the relevant Clearing Member.

(d) Compression of Affected Transactions prior to Auction

Prior to determining the Single Name Cleared Transactions to be subject to any auction pursuant to paragraph (b) above, where the affected Clearing Member acts as CDS Buyer and CDS Seller in respect of fungible Single Name Cleared Transactions, LCH SA shall, in consultation with the CDS Default Management Committee as to the transaction sizes of resulting Single Name Cleared Transactions to be auctioned, compress such Single Name Cleared Transactions up to the extent that, following such compression, Single Name Cleared Transactions representing in aggregate the Open Position of the affected Clearing Member in respect of such fungible Single Name Cleared Transactions are recognised.

For these purposes, LCH SA will provide the affected Clearing Member with a report detailing the Single Name Cleared Transactions to be subject to such compression.

The affected Clearing Member will be deemed to have submitted a request to LCH SA prior to 5.00 p.m. on the day on which LCH SA carries out the compression for ad hoc compression of such Single Name Cleared Transactions in accordance with Section 5 of the Procedures and such compression shall be carried out in accordance with Section 5 of the Procedures on the basis of such deemed request for ad hoc compression.

(e) Auction Terms

LCH SA shall determine the timing and other particular characteristics of each such auction in consultation with the CDS Default Management Committee, including determining the size of the bid/offer spread and/or of the Single Name Cleared Transactions to be auctioned, whether one or more such auctions are to be held and the timing and structure of such auctions (including the frequency at which firm bid and firm offer quotations will be requested and the transaction size (that is, the Floating Rate Payer Calculation Amount)).

Clearing Members (excluding the affected Clearing Member) may be requested, and will not be required, to submit actionable quotations in such an auction.

(f) Creation of New Transactions and Termination of Existing Transactions

LCH SA will enter into Single Name Cleared Transactions with Clearing Members, other than the affected Clearing Member, in the amount and at the prices determined pursuant to such auction.

At the time of entering into such Single Name Cleared Transactions, the corresponding Single Name Cleared Transactions of the affected Clearing Member shall be terminated by reference to the prices at which LCH SA enters into such new Single Name Cleared Transactions.

The affected Clearing Member, LCH SA and the other Clearing Members, as applicable, shall submit such information as is required in accordance with the DTCC Rules so as to reflect the terms of any reduction to, termination of or entry into of any Single Name Cleared Transaction as a result of any such auction(s).

(g) Costs of LCH SA

The affected Clearing Member will bear the cost of the associated bid/offer spread and any reasonable out-of-pocket costs and expenses of LCH SA in connection with such auction(s) and its entering into such new Single Name Cleared Transactions.

(h) LCH SA to determine Amounts Owed and Payable

Amounts owed by the affected Clearing Member to, or receivable by it from, LCH SA in connection with any such auction shall be determined by LCH SA.

In addition, any amounts payable (and the dates of settlement with respect thereto) relating to any Single Name Cleared Transactions created, reduced or terminated pursuant to any such auction shall be determined by LCH SA.

9.2 Occurrence of Client Self Referencing Transactions

(a) Notification

In respect of a Single Name Cleared Transaction registered in the Client Account Structure of a Clearing Member, the relevant Clearing Member shall, unless prohibited from so doing by applicable law or its internal policies, notify LCH SA as soon as reasonably practicable if it receives a notice from the relevant Client that:

- the Client is or has consolidated or amalgamated with, or merged into, or has transferred all or substantially all of its assets to, the Reference Entity in relation to such Single Name Cleared Transaction or entered into any agreement in respect of any of the foregoing;
- (ii) the Client and the Reference Entity in relation to such Single Name Cleared Transaction are or have become Affiliates; or
- (iii) in respect of a Restructuring Cleared Transaction, the Client is or has become the Reference Entity in relation to such Restructuring Cleared Transaction as a result of the occurrence of the relevant Restructuring Credit Event.

(b) Auction Process

Following the giving of any such notification:

- (i) LCH SA will, unless otherwise agreed in consultation with the CDS Default Management Group and the relevant Clearing Member, conduct an auction process in consultation with the CDS Default Management Group and the relevant Clearing Member, to liquidate the relevant Single Name Cleared Transaction and enter into an equivalent Single Name Cleared Transaction with another Clearing Member; and
- (ii) the provisions of Section 9.1 (Occurrence of Clearing Member Self Referencing Transaction) will apply mutatis mutandis, provided that:
 - (A) Section 9.1(a) (Duty to Notify) and 9.1(b) (Auction of Affected Transactions) shall not apply;
 - (B) Section 9.1(d) (Compression of Affected Transactions prior to Auction) is amended by deleting the words "to paragraph (b) above" and replacing them with the words "as a result of Section 9.2 (Occurrence of Client Self Referencing Transactions)";
 - (C) Section 9.1(d) (Compression of Affected Transactions prior to Auction) is amended in the case of CCM Clients only by inserting the words "that have a corresponding CCM Client Transaction with the affected CCM Client" immediately after the words "in respect of fungible Single Name Cleared Transactions";

- (D) Section 9.1(e) (Auction Terms) is amended by deleting the words "(excluding the affected Clearing Member)" in the final paragraph thereof;
- (E) Section 9.1(f) (Creation of New Transactions and Termination of Existing Transactions) is amended by deleting the words "other than the affected Clearing Member" in the first paragraph thereof; and
- (F) Each reference to "affected Clearing Member" in Section 9.1 (Occurrence of Clearing Member Self Referencing Transaction) is deleted and replaced with a reference to "relevant Clearing Member".

(c) Costs of LCH SA

The cost of the associated bid/offer spread and any reasonable out-of-pocket costs and expenses of LCH SA in connection with such auction(s) or any alternative action shall be allocated to the CCM Client Collateral Account or the FCM/BD Client Collateral Account as applicable.

10. MANDATORY PROVISIONS FOR CCM CLIENT TRANSACTIONS

In Appendix XIII, certain provisions are set-out (the "Mandatory Provisions") for incorporation into a CCM Client Transaction between a CCM and its CCM Client that corresponds to a CCM Client Cleared Transaction registered in the CCM Client Trade Account in the name of such CCM for such CCM Client. The CDS Clearing Supplement and these Mandatory Provisions have been drafted so as to complement each other.

LCH SA shall not be responsible for any loss suffered or expense incurred by a CCM or any CCM Client as a result of the inclusion or non-inclusion of the Mandatory Provisions in the CCM Client Transaction Documents.

11. AMENDMENTS

LCH SA may amend the provisions of this CDS Clearing Supplement (including, without limitation, the Mandatory Provisions) from time to time so as to comply with any legal or regulatory developments or any recommendations adopted by the industry in respect of CDS or Cleared Transactions or CCM Client Transactions, as applicable, or so as to reflect any technological advancements, in each case in accordance with the provisions of Section 1.2.2 (*Modification*) of Chapter 2 (*General Provisions*) of the CDS Clearing Rule Book.

12. FORMS OF NOTICES

A form of Credit Event Notice, Notice to Exercise Movement Option, Notice of Physical Settlement, NOPS Amendment Notice, notice to exercise alternative delivery procedure pursuant to Section 6.7 (*Alternative Delivery Procedure*), notice to fallback to Cash Settlement in respect of Non-Deliverable Obligations pursuant to Section 6.12 (*Fallback to Cash Settlement in respect of Non-Deliverable Obligations*), Physical Settlement

File No. SR-LCH SA-2022-007 Page 94 of 439

PUBLIC

Confirmation, No Physical Settlement Confirmation, notice of Clearing Member Communications Failure Event and where no longer subject to a Clearing Member Communications Failure Event, in each case pursuant to Section 7.6 (*Clearing Member Communications Failure Event*), notice of dispute relating to any Matched Pair as contemplated by Section 7 (*Dispute as to Notices*) and notice relating to Self-Referencing Transactions as contemplated by Section 9 (*Self-Referencing Transactions*) is set out in Appendix I, II, III, IV, V, VI, VII, VIII, IX, X, XI and XII respectively hereto.

Any of the above referenced notices shall be delivered in substantially the form appended hereto, provided, for the avoidance of doubt, that such notices may refer to multiple transactions and may have certain firm-specific variations.

For the avoidance of doubt, the above referenced notices shall be governed by and construed in accordance with English law.

13. EXCLUSION OF LIABILITY

Without prejudice to the provisions of Article 1.2.10.3 of the CDS Clearing Rule Book:

(a) No liability for Failure of Designee to perform in respect of Matched Pair

Without prejudice to its obligations under or in respect of a Cleared Transaction, LCH SA shall not be liable for any loss or cost arising out of any failure of any Clearing Member comprised in a Matched Pair to perform its obligations as designee of LCH SA against a related Matched Buyer or Matched Seller, as applicable.

(b) No liability for Fault of Third Party or Force Majeure

LCH SA shall have no liability to any person where Restructuring Cleared Transactions are not or are improperly created, Restructuring Cleared Transactions are not or are improperly terminated or the Movement Option process is not or is improperly implemented, in each case for the purposes of the DTCC Rules, because of a third party's fault or a force majeure event. In particular, LCH SA shall not incur any liability arising as a result of any action or omission of DTCC.

(c) No Obligation to verify Notices received

LCH SA shall have no responsibility to verify the contents of any notice received by it from any Clearing Member under the terms of any Cleared Transaction.

14. **DISPUTE RESOLUTION**

For the avoidance of doubt, all Disputes shall be referred to and finally resolved by arbitration or by litigation, as applicable, in accordance with the CDS Dispute Resolution Protocol, subject to the provisions of Sections 8 and 9 of the Procedures.

File No. SR-LCH SA-2022-007 Page 95 of 439

PUBLIC

15. GOVERNING LAW

For the avoidance of doubt, the governing law applicable to this CDS Clearing Supplement (excluding the Mandatory Provisions to the extent that such terms are incorporated by reference in the CCM Client Transaction Documents entered into between a CCM and its CCM Client in respect of a CCM Client Transaction), the 2003 ISDA Credit Derivatives Definitions and any Cleared Transactions (and any related definitions or Clearing Notices issued in respect of the CDS Clearing Supplement, the 2003 ISDA Credit Derivatives Definitions or any Cleared Transactions) and any non-contractual obligations arising out of, relating to or having any connection with them shall be as set out in Section 1.2.14 (*Governing Law*) of the CDS Clearing Rule Book.

APPENDIX I: FORM OF CREDIT EVENT NOTICE

To: [Restructuring Matched Pair Counterparty Address and Contact Information]

[To/Copy to:]

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

CREDIT EVENT NOTICE:

Credit Derivative Transaction Details: As set out in the Schedule hereto¹.

Reference is made to the Credit Derivative Transaction[s] described in the Schedule hereto (the **Transaction[s]**). Capitalised terms used and not otherwise defined in this letter shall have the meanings given them in the confirmation of the relevant Transaction.

This letter is our Credit Event Notice to you that a [insert type] Credit Event occurred with respect to [insert name of Reference Entity] on or about [insert date], when [describe Credit Event].

Nothing in this letter shall be construed of a waiver of any rights we may have with respect to the Transaction.

Sincerely

[Clearing Member]	
Name:	
Title:	

<u>70-41024903</u>

Formatted: Line spacing: single

10219615309-v3

 $^{^{1}}$ A single Credit Event Notice may be submitted for multiple trades in respect of the same Counterparty

File No. SR-LCH SA-2022-007 Page 97 of 439

PUBLIC

Formatted: Line spacing: single

SCHEDULE

Credit Derivative Transaction Details

[Clearing Member] acting as Seller/Buyer	Restructuring Matched Pair ID	Trade ID	Reference Entity	Trade Date	Effective Date	Exercise Amount ²
[Seller] [Buyer]	[●]	[●]	[●]	[●]	[●]	[•]

70-41024903

10219615309-v3

 $^{^{\}rm 2}$ Where different to the outstanding Floating Rate Payer Calculation Amount

APPENDIX II: FORM OF NOTICE TO EXERCISE MOVEMENT OPTION

To: [Restructuring Matched Pair Counterparty Address and Contact Information]

[To/Copy to:]

LCH SA
18, rue du Quatre Septembre
75002 Paris
France
[Contact details]

Dear Sir/Madam

Notice to Exercise Movement Option

Credit Derivative Transaction Details: As set out in the Schedule hereto³.

Reference is made to: (a) the Credit Derivative Transaction[s] described in the Schedule hereto (the **Transaction[s]**) between [], as Seller, and [], as Buyer; (b) the Credit Event Notice dated insert date], previously delivered to you on [insert date]; and (c) the occurrence of the No Auction Announcement Date on [insert date] pursuant to Section 12.12(b) of the 2003 ISDA Credit Derivatives Definitions, as supplemented by the 2009 ISDA Credit Derivatives Determinations Committees, Auction Settlement and Restructuring Supplement to the 2003 ISDA Credit Derivatives Definitions (published on July 14, 2009) (the **Definitions**).

This letter constitutes a Notice to Exercise Movement Option. Any capitalised term not otherwise defined in this letter will have the meaning, if any, assigned to such term in the confirmation of the relevant Transaction or, if no meaning is specified therein, in the Definitions.

We hereby exercise the Movement Option, confirm that each Transaction will be settled in accordance with the relevant Credit Derivatives Auction Settlement Terms specified in the column entitled "Auction Settlement Terms" corresponding to such Transaction in the Schedule hereto and require performance by you in accordance therewith.

Yours faithfully,
[Matched Buyer/Matched Seller]
Name:
Title:

Formatted: Line spacing: single

<u>10219615309-v3</u> <u>70-41024903</u>

³ A single Notice to Exercise Movement Option may be submitted for multiple trades in respect of the same Counterparty

File No. SR-LCH SA-2022-007 Page 99 of 439

PUBLIC

SCHEDULE

Credit Derivative Transaction Details

Restructuring Matched Pair ID	Trade ID	Reference Entity	Trade Date	Effective Date	Auction Settlement Terms
[●]	[●]	[●]	[●]	[●]	[●]

<u>10219615309-v3</u> <u>70-41024903</u>

File No. SR-LCH SA-2022-007 Page 100 of 439

PUBLIC

APPENDIX III: FORM OF NOTICE OF PHYSICAL SETTLEMENT

To: Settlement Matched Pair Matched Seller Address and Contact Information

Copy to:

LCH SA
18, rue du Quatre Septembre
75002 Paris
France
[Contact details]

Notice of Physical Settlement

Credit Derivative Transaction Details: As set out in the Schedule hereto4.

Reference is made to: (a) the Credit Derivative Transaction[s] described in the Schedule hereto (the **Transaction[s]**) between [], as Seller, and [], as Buyer. Reference is also made to the Credit Event Notice [and Notice of Publicly Available Information] dated [insert date], previously delivered to you on [insert date].

This letter constitutes a Notice of Physical Settlement. Any capitalised term not otherwise defined in this letter will have the meaning, if any, assigned to such term in the confirmation of the relevant Transaction (the **Relevant Confirmation**) or, if no meaning is specified therein, in the 2003 ISDA Credit Derivatives Definitions, as supplemented by the 2009 ISDA Credit Derivatives Determinations Committees, Auction Settlement and Restructuring Supplement to the 2003 ISDA Credit Derivatives Definitions (published on July 14, 2009) (the **Definitions**).

We hereby confirm that we will settle the Transaction[s] and require performance by you in accordance with the provisions of the Relevant Confirmation and the Definitions relating to Physical Settlement. Subject to the terms of the relevant Transaction, we will Deliver to you on or before the Physical Settlement Date, an amount of the Deliverable Obligation(s) described in the column entitled "Deliverable Obligation(s)" in the Schedule hereto, corresponding to such Transaction:

[Further, in respect of each Transaction[s] we identify those Enabling Obligation(s) described in the column entitled "Enabling Obligation(s)" in the Schedule hereto, corresponding to such Transaction:]

Yours faithfully,						
[Matched Buyer]						
Name:						
Title:						

Formatted: Line spacing: single

⁴ A single Notice of Physical Settlement may be submitted for multiple trades in respect of the same Counterparty

File No. SR-LCH SA-2022-007 Page 101 of 439

PUBLIC

SCHEDULE

Credit Derivative Transaction Details

Settlement Matched Pair ID	Trade ID	Reference Entity	Trade Date	Effective Date	Deliverable Obligation(s) ⁵	Enabling Obligation(s) ⁶]
[●]	[●]	[●]	[●]	[●]	[Currency][Due and Payable Amount]][•]	[●]

10219615309-v3 70-41024903

Formatted: Line spacing: single

⁵ describe the Deliverable Obligation(s) to be Delivered, including the currency and Outstanding Principal Balanceor Due and Payable Amount for each such Deliverable Obligation and, if available and applicable, the CUSIP or ISIN number (or, if such identifying number is not available, the rate and tenor of the Deliverable Obligation)

⁶ describe each such Enabling Obligation, including the CUSIP or ISIN number, if available and applicable (or, if such identifying number is not available, the rate and tenor), of such Enabling Obligation, or any other information necessary to establish that such obligation is an Enabling Obligation.

File No. SR-LCH SA-2022-007 Page 102 of 439

PUBLIC

APPENDIX IV: FORM OF NOPS AMENDMENT NOTICE

To: Settlement Matched Pair Matched Seller Address and Contact Information

Copy to:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

Settlement Matched Pair ID: [●]

Trade ID: [●]

Yours faithfully

10219615309-v3

NOPS Amendment Notice

Credit Derivative Transaction Details: [Trade Date], [Effective Date], [Reference Entity]

Reference is made to the Credit Derivative Transaction described above (the **Transaction**) between [], as Seller, and [], as Buyer. Reference is also made to the Notice of Physical Settlement] NOPS Amendment Notice] dated [insert date], previously delivered to you on [insert date] dated [insert date].

This letter constitutes a NOPS Amendment Notice. Any capitalised term not otherwise defined in this letter will have the meaning, if any, assigned to such term in the Confirmation of the Transaction or, if no meaning is specified therein, in the CDS Clearing Supplement.

We hereby notify you that we are replacing the following Deliverable Obligation(s) specified in the Notice of Physical Settlement] NOPS Amendment Notice] specified above with the following Replacement Deliverable Obligation(s):

[describe the Deliverable Obligation(s) to be replaced, including the Replaced Deliverable Obligation Outstanding Amount for each such Deliverable Obligation and, if available and applicable, the CUSIP or ISIN number (or, if such identifying number is not available, the rate and tenor of the Deliverable Obligation) and the Replacement Deliverable Obligation(s) for each Replaced Deliverable Obligation Outstanding Amount so specified and, if available and applicable, the CUSIP or ISIN number (or, if such identifying number is not available, the rate and tenor of the Replacement Deliverable Obligation)].

, , , , , , , , , , , , , , , , , , , ,		
[Matched Buyer]		
Name:		
Title:		

70-41024903

Formatted: Space After: 11 pt

APPENDIX V: FORM OF NOTICE TO EXERCISE ALTERNATIVE DELIVERY PROCEDURE PURSUANT TO SECTION 6.7 (ALTERNATIVE DELIVERY PROCEDURE)

To:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

[date]

Dear Sir/Madam

Notice to Exercise Alternative Delivery Procedure⁷

Reference is made to: the Settlement Matched Pairs described in the Schedule hereto, being the Settlement Matched Pairs to which this notice relates and Section 6.7 (*Alternative Delivery Procedure*) of the CDS Clearing Supplement. Defined terms shall have the meanings assigned to them in the CDS Clearing Supplement, unless otherwise defined herein.

The Settlement Matched Pair Delivery Amount relating to each Settlement Matched Pair is that amount specified in the column entitled "Settlement Matched Pair Delivery Amount" in the Schedule hereto, corresponding to such Settlement Matched Pair.

In respect of each Settlement Matched Pair and in accordance with Section 6.7 (*Alternative Delivery Procedure*) we hereby elect to exercise our rights against and perform obligations to you in accordance with the alternative delivery procedure in relation to such percentage and amount of the Settlement Matched Pair Delivery Amount as set out in the column entitled "Percentage and Amount of Settlement Matched Pair Delivery Amount" in the Schedule hereto corresponding to such Settlement Matched Pair.

 $This \ notice \ may \ be \ executed \ in \ any \ number \ of \ counterparts \ which \ together \ shall \ constitute \ one \ notice.$

By countersigning this notice, you are deemed to have given your consent to the above in satisfaction of the requirement to obtain your consent contained in Section 6.7 (*Alternative Delivery Procedure*) of the CDS Clearing Supplement.

This notice (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this agreement or its formation) shall be governed by and construed in accordance with English law.

Yours faithfully,

[Matched Buyer/ Matched Seller]

⁷ A single Notice to Exercise Alternative Delivery Procedure may be submitted for multiple trades in respect of the same Counterparty Formatted: Line spacing: single

<u>10219615309-v3</u> <u>70-41024903</u>

File No. SR-LCH SA-2022-007 Page 104 of 439

PUBLIC

Name:
Title:
Countersigned by LCH SA
Name:
Title:

<u>10219615309-v3</u> <u>70-41024903</u>

File No. SR-LCH SA-2022-007 Page 105 of 439

PUBLIC

SCHEDULE

Credit Derivative Transaction Details

Settlement Matched Pair ID	Trade ID	Reference Entity	Trade Date	Effective Date	Settlement Matched Pair Delivery Amount	Percentage and amount of Settlement Matched Pair Delivery Amount
[●]	[●]	[●]	[●]	[●]	[●]	[●]

10219615309-v3 70-41024903

APPENDIX VI: FORM OF NOTICE OF FALLBACK TO CASH SETTLEMENT OF NON-DELIVERABLE OBLIGATIONS PURSUANT TO SECTION 6.12 (FALLBACK TO CASH SETTLEMENT IN RESPECT OF NONDELIVERABLE OBLIGATIONS)

To: Settlement Matched Pair Matched Seller Address and Contact Information

To:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

Settlement Matched Pair ID: [●]

Trade ID: [●]

Dear Sir/Madam

Notice of fallback to Cash Settlement in respect of Non-Deliverable Obligations pursuant to Section 6.12 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) of the CDS Clearing Supplement

Reference is made to Section 6.12 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) of the CDS Clearing Supplement and [insert details of the relevant Settlement Matched Pair(s)], being the Settlement Matched Pair[s] to which this notice relates and to the [Notice of Physical Settlement][NOPS Amendment Notice] previously delivered to you on [insert date] dated [insert date]. Defined terms shall have the meanings assigned to them in the CDS Clearing Supplement, unless otherwise defined herein.

In accordance with Section 6.12 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) of the CDS Clearing Supplement we hereby notify you we are not permitted to Deliver the Deliverable Obligations specified below (such Deliverable Obligations, the Non-Deliverable Obligations) as specified in the [Notice of Physical Settlement][NOPS Amendment Notice] specified above for the following reasons:

[Insert details of the relevant Non-Deliverable Obligations and reasonable detail of the relevant circumstances, as described paragraphs (a)(i) and (a)(ii) of Section 6.12 of the CDS Clearing Supplement].

This notice (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this agreement or its formation) shall be governed by and construed in accordance with English law.

Yours faithfully,

<u>10219615309-v3</u> <u>70-41024903</u>

File No. SR-LCH SA-2022-007 Page 107 of 439

IRI	

[Matched Buyer]		
Name:		
Title:		

<u>10219615309-v3</u> <u>70-41024903</u>

File No. SR-LCH SA-2022-007 Page 108 of 439

PUBLIC

APPENDIX VII: FORM OF PHYSICAL SETTLEMENT CONFIRMATION AS CONTEMPLATED BY SECTION 6.19 (MISCELLANEOUS PROVISIONS RELATING TO PHYSICAL SETTLEMENT)

To:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

[date]

Dear Sir/Madam

Physical Settlement Confirmation

Reference is made to Section 6.19(b) (Notification of Completion of Physical Settlement) of the CDS Clearing Supplement and the Settlement Matched Pair[s] described in the Schedule hereto, being the Settlement Matched Pair[s] to which this notice relates. Defined terms shall have the meanings assigned to them in the CDS Clearing Supplement, unless otherwise defined herein.

In accordance with Section 6.19(b) (*Notification of Completion of Physical Settlement*) of the CDS Clearing Supplement, we hereby notify you that we have completed Physical Settlement with respect to such Settlement Matched Pair[s].

This notice is delivered in satisfaction of the requirement in Section 6.19(b) (*Notification of Completion of Physical Settlement*) of the CDS Clearing Supplement to inform you of such completion.

This notice (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this agreement or its formation) shall be governed by and construed in accordance with English law.

Yours faithfully,	
[Matched Buyer]/[Matched Seller]	
Name:	
Title:	

File No. SR-LCH SA-2022-007 Page 109 of 439

PUBLIC

SCHEDULE

Settlement Matched Pair Details

Settlement Matched Pair ID	Trade ID	Reference Entity	Trade Date	Effective Date
[•]	[•]	[•]	[•]	[•]

File No. SR-LCH SA-2022-007 Page 110 of 439

PUBLIC

APPENDIX VIII: FORM OF NO PHYSICAL SETTLEMENT CONFIRMATION AS CONTEMPLATED BY SECTION 6.19 (MISCELLANEOUS PROVISIONS RELATING TO PHYSICAL SETTLEMENT)

To:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

[date]

Settlement Matched Pair ID: [●]

Trade ID: [●]

Dear Sir/Madam

No Physical Settlement Confirmation

Reference is made to Section 6.19(c) (*Notification that Physical Settlement will not occur*) of the CDS Clearing Supplement and the Settlement Matched Pair[s] described in the Schedule hereto, being the Settlement Matched Pair[s] to which this notice relates. Defined terms shall have the meanings assigned to them in the CDS Clearing Supplement, unless otherwise defined herein.

In accordance with Section 6.19(c) (Notification that Physical Settlement will not occur) of the CDS Clearing Supplement, we hereby notify you that no Notice of Physical Settlement has been delivered within the relevant time period permitted for such delivery in accordance with the terms of the relevant Physically Settled Cleared Transactions and, accordingly, that Physical Settlement will not, under the terms of such Physically Settled Cleared Transactions, occur.

This notice is delivered in satisfaction of the requirement in Section 6.19(c) (*Notification that Physical Settlement will not occur*) of the CDS Clearing Supplement to inform you of such completion.

This notice (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this agreement or its formation) shall be governed by and construed in accordance with English law.

Yours faithfully,	
[Matched Buyer]/[Matched Seller]	
Name:	
Title:	

File No. SR-LCH SA-2022-007 Page 111 of 439

PUBLIC

SCHEDULE

Settlement Matched Pair Details

Settlement Matched Pair ID	Trade ID	Reference Entity	Trade Date	Effective Date
[•]	[•]	[•]	[•]	[•]

File No. SR-LCH SA-2022-007 Page 112 of 439

PUBLIC

APPENDIX IX: FORM OF NOTICE OF CLEARING MEMBER COMMUNICATIONS FAILURE EVENT PURSUANT TO SECTION 7.6 (CLEARING MEMBER COMMUNICATIONS FAILURE EVENT)

To:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

[date]

Dear Sir/Madam

Notice certifying occurrence of a Clearing Member Communications Failure Event

Reference is made to Section 7.6(b) (Clearing Member to notify LCH SA of Occurrence of Clearing Member Communications Failure Event) of the CDS Clearing Supplement. Defined terms shall have the meanings assigned to them in the CDS Clearing Supplement, unless otherwise defined herein.

In accordance with Section 7.6(b) (Clearing Member to notify LCH SA of Occurrence of Clearing Member Communications Failure Event) of the CDS Clearing Supplement, notice is hereby given that we are affected by a Clearing Member Communications Failure Event [insert details of such failure].

This notice (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this agreement or its formation) shall be governed by and construed in accordance with English law.

itii Eligiisii iaw.
ours faithfully,
Clearing Member]

ame:
tle:
iigned by a senior officer (such as a managing director or equivalent) on behalf of the Clearing lember]

File No. SR-LCH SA-2022-007 Page 113 of 439

PUBLIC

APPENDIX X: FORM OF NOTICE FOR CEASING TO BE SUBJECT TO A CLEARING MEMBER COMMUNICATIONS FAILURE EVENT PURSUANT TO SECTION 7.6 (CLEARING MEMBER COMMUNICATIONS FAILURE EVENT)

To:

LCH SA
18, rue du Quatre Septembre
75002 Paris
France
[Contact details]

[date]

Dear Sir/Madam

Notice that a Clearing Member is no longer subject to Clearing Member Communications Failure

Reference is made to Section 7.6(d) (Notification of Resolution of Clearing Member Communications Failure Event) of the CDS Clearing Supplement and the notice certifying the occurrence of a Clearing Member Communications Failure Event delivered by us to LCH SA on [●] (the Notice of Clearing Member Communications Failure Event). Defined terms shall have the meanings assigned to them in the CDS Clearing Supplement, unless otherwise defined herein.

In accordance with Section 7.6(d) (*Notification of Resolution of Clearing Member Communications Failure Event*) of the CDS Clearing Supplement, notice is hereby given that we are no longer subject to the relevant Clearing Member Communications Failure Event described in the Notice of Clearing Member Communications Failure Event.

This notice (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this agreement or its formation) shall be governed by and construed in accordance with English law.

_		
ours faithfully,		
Clearing Member]		
Name:		
Γitle:		

File No. SR-LCH SA-2022-007 Page 114 of 439

PUBLIC

APPENDIX XI: FORM OF NOTICE OF DISPUTE RELATING TO ANY MATCHED PAIR

To:
LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]
[date]
[Restructuring][Settlement] Matched Pair ID: [●]
Trade ID: [●]
Dear Sir/Madam
Notice of dispute relating to [insert details of the relevant Matched Pairs subject to a dispute]
Reference is made to Section 7.11 (<i>Disputes as to Notices</i>) of the CDS Clearing Supplement. Defined terms shall have the meanings assigned to them in the CDS Clearing Supplement, unless otherwise defined herein.
In accordance with Section 7.11 (<i>Disputes as to Notices</i>) of the CDS Clearing Supplement, notice is hereby given of the following dispute(s):
[insert details of Matched Pair(s) affected and the relevant dispute].
This notice (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this agreement or its formation) shall be governed by and construed in accordance with English law.
Yours faithfully,
[Matched Buyer]/[Matched Seller]
·
Name:
Title:

File No. SR-LCH SA-2022-007 Page 115 of 439

PUBLIC

APPENDIX XII: FORM OF NOTICE RELATING TO SELF-REFERENCING TRANSACTIONS

To:

LCH SA
18, rue du Quatre Septembre
75002 Paris
France
[Contact details]

[date]

Dear Sir/Madam

Notice relating to Self-Referencing Transactions

Credit Derivative Transaction Details: As set out in the Schedule hereto⁸.

Reference is made to the Credit Derivative Transaction[s] described in the Schedule hereto (the **Transaction[s]**) between [●], as Seller and [●], as Buyer and to Section 9.1(a) (*Duty to notify*) of the CDS Clearing Supplement. Defined terms shall have the meanings assigned to them in the CDS Clearing Supplement, unless otherwise defined herein.

In accordance with Section 9.1(a) (*Duty to notify*) of the CDS Clearing Supplement, notice is hereby given of the following: [insert details of one or more of the relevant events, as set out in paragraphs (i), (ii) and (iii) of Section 9.1(a) (Duty to notify) of the CDS Clearing Supplement].

This notice (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this agreement or its formation) shall be governed by and construed in accordance with English law.

Yours faithfully,		
[Clearing Member]		
Name:		
Title:		

<u>10219615309-v3</u> <u>70-41024903</u>

Formatted: Line spacing: single

 $^{^{8}}$ A single Credit Event Notice may be submitted for multiple trades in respect of the same Counterparty

File No. SR-LCH SA-2022-007 Page 116 of 439

PUBLIC

SCHEDULE

Credit Derivative Transaction Details

[Restructuring] Matched Pair ID	Trade ID	Reference Entity	Trade Date	Effective Date
[●]	[●]	[●]	[●]	[●]

APPENDIX XIII: CCM CLIENT TRANSACTION REQUIREMENTS

The following provisions (the "Mandatory Provisions") are to be incorporated into a CCM Client Transaction between a CCM and its CCM Client that corresponds to a CCM Client Cleared Transaction registered in the CCM Client Trade Account in the name of such CCM for such CCM Client. The terms of the corresponding CCM Client Cleared Transaction will be governed by the CDS Clearing Supplement. The CDS Clearing Supplement and these Mandatory Provisions have been drafted so as to complement each other.

LCH SA shall not be responsible for any loss suffered or expense incurred by a CCM or any CCM Client as a result of the inclusion in the CCM Client Transaction Documents of the requirements set-out in this Appendix XIII.

The Mandatory Provisions, when they are incorporated into any CCM Client Transaction Documents, shall be governed by and construed in accordance with the governing law applicable to such CCM Client Transaction Documents of which they form part, or if different and applicable, in accordance with such CCM Client Transaction Documents, the governing law applicable to transactions entered into under such CCM Client Transaction Documents. The Mandatory Provisions shall be subject to such dispute resolution mechanisms and procedures and such courts or other forum for hearing disputes as are applicable in respect of such CCM Client Transaction Documents of which they form part. Each CCM and its CCM Client to which the Mandatory Provisions apply will waive any right to object to any such choice of law or proceedings on the basis of *forum non conveniens*, that the governing law or forum is not specified on the face of this document or otherwise.

In this Appendix XIII:

"CCM Client Buyer" means a CCM Client that is party to a CCM Client Transaction as protection buyer;

"CCM Client Seller" means a CCM Client that is party to a CCM Client Transaction as protection seller;

"CCM Buyer/Matched Seller" means a CCM that is party to a CCM Client Transaction as protection buyer and to the corresponding CCM Client Cleared Transaction as protection seller; and

"CCM Seller/Matched Buyer" means a CCM that is party to a CCM Client Transaction as protection seller and to the corresponding CCM Client Cleared Transaction as protection buyer.

1. Defined Terms

Terms used in the Mandatory Provisions and not otherwise defined herein or in the 2003 ISDA Credit Derivatives Definitions shall have the meanings given to them in the CDS Clearing Supplement.

2. Terms of CCM Client Transactions

2.1 2003 ISDA Credit Derivatives Definitions

The definitions and provisions contained in the 2003 ISDA Credit Derivatives Definitions published by the International Swaps and Derivatives Association as supplemented by the 2009 ISDA Credit Derivatives Determinations Committees, Auction Settlement and

Restructuring Supplement to the 2003 ISDA Credit Derivatives Definitions (the "2003 ISDA Credit Derivatives Definitions"), are incorporated into each CCM Client Transaction.

2.2 Single Name CCM Client Transactions - Reference Obligation

With effect from the date on which the CCM Client Cleared Transaction corresponding to a Single Name CCM Client Transaction is registered in the TIW, such Single Name CCM Client Transaction shall be deemed to reference the CDSClear Preferred Reference Obligation in respect of the relevant Reference Entity.

2.3 Single Name CCM Client Transactions - Updating Physical Settlement Matrix

With effect from the date on which the CCM Client Cleared Transaction corresponding to a Single Name CCM Client Transaction is registered in the TIW, such Single Name CCM Client Transaction shall be deemed to reference the Relevant Physical Settlement Matrix.

With effect from the close of business on any Matrix Re-versioning Date, any Single Name CCM Client Transaction referencing the Existing Matrix will be deemed to have been amended so as to reference the Revised Matrix.

2.4 Index CCM Client Transactions - Updating Eligible Index Versions

Upon the occurrence of a DTCC Re-versioning Date, any Index CCM Client Transaction shall be automatically amended so as to reference the portfolio of Reference Entities specified in the revised version of the index published by the Index Publisher and referenced in the TIW with effect from such DTCC Re-versioning Date.

2.5 Initial Payment Date

Notwithstanding anything to the contrary in the 2003 ISDA Credit Derivatives Definitions, if the Initial Payment Date specified in the CCM Client Transaction Documents in respect of any CCM Client Transaction is a date falling after the Clearing Day on which the Cleared Transactions related to such CCM Client Transaction are created by novation pursuant to Title III (Clearing Operations) of the CDS Clearing Rule Book, the Initial Payment Date in respect of such CCM Client Transaction shall be deemed to be the Transaction Business Day immediately following the Clearing Day on which the Cleared Transactions relating to such CCM Client Transaction are created.

2.6 2014 ISDA Credit Derivatives Definitions Protocol

The Protocol Effectiveness Condition (as defined in the 2014 ISDA Credit Derivatives Definitions Protocol published by ISDA on 21 August 2014 and supplemented by the Supplement thereto published by ISDA on 15 September 2014 and as further supplemented from time to time (the "2014 Protocol")) being satisfied, then from and including the Implementation Date (as defined in the 2014 Protocol) the amendments set forth in Part 1 (Global Amendments) of Schedule 1 (Amendments to Protocol Covered Transactions) of the 2014 Protocol shall apply to each Index CCM Client Transaction and each Single Name CCM Client Transaction which incorporates the 2003 ISDA Credit Derivatives Definitions and each

such CCM Client Transaction shall be a Protocol Covered Transaction for the purposes of the 2014 Protocol provided that:

- (i) the date on which the CCM Client Cleared Transaction corresponding to such CCM Client Transaction is registered in the TIW is prior to the Implementation Date;
- (ii) such CCM Client Transaction is not an Excluded Transaction (as defined in the 2014 Protocol) provided that if such CCM Client Transaction relates to more than one Reference Entity and would otherwise be a Protocol Covered Transaction but for the inclusion of any Affected Portion (as defined in the 2014 Protocol) therein, such Affected Portion shall be exclude for the purposes of this Mandatory Provision 2.6(a) and the remaining portion of such CCM Client Transaction shall be deemed to be a Protocol Covered Transaction:
- the Scheduled Termination Date in respect of such CCM Client Transaction is on or after the Implementation Date;
- (iv) no Event Determination Date has occurred in respect of such CCM Client Transaction prior to the Implementation Date;
- (v) from and including the Implementation Date the provisions of Appendix XIII of Part B of the CDS Clearing Supplement shall apply to such CCM Client Transaction or the portion of such CCM Client Transaction other than the Affected Portion, as applicable and the CCM Client Cleared Transaction corresponding thereto and such CCM Client Transaction shall be treated as if it had been entered into and is subject to the 2014 ISDA Credit Derivatives Definitions. In the event of any inconsistency between the provisions of Part B of the CDS Clearing Supplement and the amendments set forth in this paragraph 2.6, the provisions of Appendix XIII of Part B of the CDS Clearing Supplement shall prevail.

3. Additional CCM Client Transactions, Compression and Succession Events

3.1 Creation of Additional CCM Client Transactions

Immediately following:

- (a) the creation of Matched Pairs by LCH SA pursuant to Section 8.1 (*Creation of Matched Pairs*) of the CDS Clearing Supplement; or
- (b) the creation of Resulting Single Name Cleared Transactions pursuant to Section 4.4 (Re-couponing of Restructuring Cleared Transactions) of the CDS Clearing Supplement,

if a CCM Client Transaction has been specified to have been split into or replaced by two or more separate CCM Client Transactions in the TIW as a result of the creation of such Matched Pairs or Resulting Single Name Cleared Transactions, such CCM Client Transaction shall be split into or terminated and replaced by two or more (as applicable) corresponding CCM Client Transactions. The Floating Rate Payer Calculation Amount and Fixed Rate of each such CCM Client Transaction shall correspond to the Floating Rate Payer Calculation Amount and Fixed Rate specified in TIW for such CCM Client Transaction. In respect of CCM Client

Transactions created as a result of the creation of Resulting Single Name Cleared Transactions, the Trade Date of such new CCM Client Transactions shall be the same as the Trade Date of the equivalent Resulting Single Name Cleared Transactions. Otherwise, each new CCM Client Transaction shall have the same terms as the original CCM Client Transaction.

3.2 Reversal of Creation of Additional CCM Client Transactions

If a CCM Client Transaction has been split into two or more CCM Client Transactions pursuant to Mandatory Provision 3.1 (*Creation of Additional CCM Client Transactions*) above and the relevant DC Credit Event Announcement that led to the creation of the Matched Pairs is reversed such that Section 5.5 (*Reversal of DC Credit Event Announcements*) of the CDS Clearing Supplement applies, then, subject to Section 9.1(c)(iii)(B) of the 2003 ISDA Credit Derivatives Definitions, any additional CCM Client Transactions created pursuant to Mandatory Provision 3.1 (*Creation of Additional CCM Client Transactions*) above shall be deemed not to have been created and any Credit Event Notices delivered in connection with such CCM Client Transactions shall be deemed to be ineffective.

3.3 Compression of CCM Client Transactions

If two or more CCM Client Transactions are specified in TIW to have been compressed into a single CCM Client Transaction pursuant to Chapter 3 (*Compression*) of Title III (*Clearing Operations*) of the CDS Clearing Rule Book, such CCM Client Transactions shall be compressed into a single CCM Client Transaction with a Floating Rate Payer Calculation Amount equal to the aggregate Floating Rate Payer Calculation Amounts of the original CCM Client Transactions.

3.4 Succession Events and Cleared Transactions

If LCH SA takes any action with respect to a CCM Client Cleared Transaction pursuant to Section 4.5 (*Succession Events and Cleared Transactions*) of the CDS Clearing Supplement so as to give effect to a Succession Event, such action shall also be deemed to have been taken with respect to the corresponding CCM Client Transaction.

4. Notices

4.1 Validity of Notices

Save if and as expressly stated to the contrary in the Mandatory Provisions, any notice delivered by a CCM Client to its CCM in respect of a CCM Client Transaction (including, without limitation, a Credit Event Notice, Notice of Physical Settlement, Notice to Exercise Movement Option or NOPS Amendment Notice) at a time or in a manner in which the CCM would not be permitted to deliver such a notice to LCH SA (or to a relevant Matched Buyer or Matched Seller as designee of LCH SA(as applicable)) in respect of the corresponding CCM Client Cleared Transaction pursuant to the terms of the CDS Clearing Supplement shall be deemed not to have been delivered.

4.2 Credit Event Notices and NEMOs given via DTCC

(a) Credit Event Notices and NEMOs to be given via DTCC

Credit Event Notices and Notices to Exercise Movement Option shall be delivered by way of the relevant DTCC Notice Facility, save if and as expressly stated to the contrary in the Mandatory Provisions or otherwise agreed between the parties to the CCM Client Transaction. The deemed time of delivery of any such notices shall be as set out in the DTCC Rules from time to time.

(b) Credit Event Notices and NEMOs delivered in respect of corresponding CCM Client Cleared Transaction

In respect of a CCM Client Transaction, if:

- CCM Seller/Matched Buyer or CCM Buyer/Matched Seller delivers a valid Credit Event Notice or Notice to Exercise Movement Option in respect of the corresponding CCM Client Cleared Transaction by way of the relevant DTCC Notice Facility; or
- (ii) a Credit Event Notice or Notice to Exercise Movement Option is deemed to have been delivered in respect of the corresponding CCM Client Cleared Transaction pursuant to Section 7.3(b) (Credit Event Notices and NEMOs delivered in respect of CCM Client Transaction) of the CDS Clearing Supplement as a result of the receipt of a valid Credit Event Notice or Notice to Exercise Movement Option (as applicable) by way of the relevant DTCC Notice Facility in respect of the CCM Client Transaction between the other CCM of a Matched Pair and its CCM Client,

and a Credit Event Notice or Notice to Exercise Movement Option (as applicable) has not already been given in respect of such CCM Client Transaction in accordance with Mandatory Provision 4.4 (*Communications Failure Event*) such notice (or deemed notice) shall be deemed also to be a Credit Event Notice or Notice to Exercise Movement Option (as applicable) for the purposes of such CCM Client Transaction.

4.3 Consequences of DTCC Failure

If a DTCC Failure Event occurs, from (and including) the DTCC Failure Event Time to (but excluding) the DTCC Resolution Time:

- (a) Mandatory Provision 4.2(a) (Credit Event Notices and NEMOs to be given via DTCC) shall not apply and accordingly Credit Event Notices and Notices to Exercise Movement Option shall be delivered directly (and not via the relevant DTCC Notice Facility);
- (b) any notice delivered via the relevant DTCC Notice Facility prior to the DTCC Failure Event Time will be valid and will not be affected by such DTCC Failure Event; and

(c) any notice delivered or purported to be delivered via the relevant DTCC Notice Facility at or following the DTCC Failure Event Time but prior to the DTCC Resolution Time will not be valid and effective.

Mandatory Provision 4.2(a) (Credit Event Notices and NEMOs to be given via DTCC) shall apply with effect from the DTCC Resolution Time and, accordingly, any notice thereafter delivered or purported to be delivered directly (and not via the relevant DTCC Notice Facility) will not be valid and effective.

4.4 Communications Failure Event

(a) Right to deliver Notices manually following Communications Failure Event

If a party is affected by a significant communications or information technology failure resulting in it being impossible or impractical for such party to deliver any Credit Event Notice in relation to a Restructuring Credit Event or any Notice to Exercise Movement Option via a relevant DTCC Notice Facility (a "Communications Failure Event") it may, notwithstanding Mandatory Provision 4.2(a) (Credit Event Notices and NEMOs to be given via DTCC), deliver Credit Event Notices and Notices to Exercise Movement Option directly (and not via the relevant DTCC Notice Facility).

Such party shall deliver, together with any Credit Event Notice or Notice to Exercise Movement Option delivered by it directly, a notice signed by a senior officer (such as a managing director or equivalent) of such party certifying that it is affected by a Communications Failure Event (or, if such party is unable to deliver such notice in writing, orally by telephone).

(b) Notices to party affected by Communications Failure Event

For the avoidance of doubt, Mandatory Provision 4.2(a) (Credit Event Notices and NEMOs to be given via DTCC) shall continue to apply in respect of notices given by the party not affected by the Communications Failure Event to the party affected by the Communications Failure Event.

(c) Notification of Resolution of Communications Failure Event

As soon as reasonably practicable upon a party ceasing to be subject to a Communications Failure Event, it shall notify the other party accordingly and thereupon Mandatory Provision 4.2(a) (Credit Event Notices and NEMOs to be given via DTCC) shall apply and, accordingly, any notice thereafter delivered or purported to be delivered directly (and not via the relevant DTCC Notice Facility) will not be valid and effective.

(d) Duty to Mitigate

A party which is subject to a Communications Failure Event shall use reasonable endeavours to mitigate the operational impact on the other party of any Communications Failure Event, to cure such Communications Failure Event as soon as possible and to ensure that the circumstances giving rise to the relevant Communications Failure Event do not recur.

(e) Breach does not Invalidate Valid Notices

Without prejudice to any other rights or remedies of the parties, any breach by a party of the provisions of this Mandatory Provision 4.4 shall not cause any Credit Event Notice or Notice to Exercise Movement Option delivered otherwise than in accordance with the terms of the relevant CCM Client Transaction, which would otherwise be valid and effective, to be invalid or ineffective.

4.5 Uncertain Delivery

(a) Manual Notice permitted if Delivery of Notice in DTCC uncertain

Notwithstanding Mandatory Provision 4.2(a) (Credit Event Notices and NEMOs to be given via DTCC), where such notices are permitted to be delivered by means other than the relevant DTCC Notice Facility pursuant to this Mandatory Provision 4 (Notices), and a party is uncertain as to whether or not a Credit Event Notice or Notice to Exercise Movement Option (as applicable) it attempted to deliver via a DTCC Notice Facility has:

- (i) actually been delivered; or
- (ii) was delivered prior to the DTCC Failure Time,

that party shall be entitled to deliver such a notice directly to the other party specifying that such notice is only to be effective to the extent that the other purported notice is not effective.

(b) Details to be provided of Uncertain Notice

If a party delivers a manual notice pursuant to Mandatory Provision (a) (*Manual Notice permitted if Delivery of Notice in DTCC uncertain*) above, such party shall be required to provide (together with such notice) sufficient details of the notice attempted to be given by way of the relevant DTCC Notice Facility so as to allow the other party to identify the communication concerned.

(c) DTCC Notice delivered successfully

If the first Credit Event Notice or Notice to Exercise Movement Option (as applicable) to which the manual notice delivered pursuant to Mandatory Provision (a) (*Manual Notice permitted if Delivery of Notice in DTCC uncertain*) above related was actually delivered successfully, any subsequent Credit Event Notice or Notice to Exercise Movement Option delivered shall be deemed not to have been delivered.

5. Determination of Credit Events and Succession Events

Notwithstanding any provision to the contrary:

(a) the Calculation Agent shall not make any determination in respect of any matter which is or may be subject to resolution under Sections 3.5 (Succession Event Resolutions) or 3.6 (Substitute Reference Obligation Resolutions) of the DC Rules; and

(b) neither party shall be entitled to deliver a Succession Event Notice or a Credit Event Notice (other than Credit Event Notices in relation to a Restructuring Credit Event in accordance with the terms of any Restructuring CCM Client Transaction (including the Mandatory Provisions) and, where applicable, the DTCC Rules).

6. Timings for the Delivery of Manual Notices

The following provisions shall solely be applicable in respect of a CCM Client Transaction between a CCM Client Buyer and its CCM Seller/Matched Buyer:

6.1 Delivery of Manual Notices by CCM Client Buyer

For the purposes of the delivery by CCM Client Buyer of any notice in respect of a CCM Client Transaction which is permitted pursuant to the terms of such CCM Client Transaction (including the Mandatory Provisions) to be delivered manually (rather than via the relevant DTCC Notice Facility), Section 1.10 (*Requirements Regarding Notices*) of the 2003 ISDA Credit Derivatives Definitions shall be amended so as to provide that, solely in respect of the final day on which such manual notice could validly be delivered pursuant to the terms of such CCM Client Transaction (including the Mandatory Provisions), any such notice shall be required to be delivered on or prior to 2:00 p.m. (Calculation Agent City time) in order to be effective.

A manual notice (including, without limitation, a Credit Event Notice or a Notice to Exercise Movement Option) delivered after 2:00 p.m. (Calculation Agent City time) on the final day on which such notice could validly be delivered pursuant to the terms of the relevant CCM Client Transaction (including the Mandatory Provisions) shall be deemed not to have been delivered.

6.2 Onward Delivery of Certain Notices by CCM Seller/Matched Buyer to Matched Seller

Any Credit Event Notice, Notice to Exercise Movement Option, Notice of Physical Settlement, NOPS Amendment Notice, any notice given pursuant to Section 9.9 (*Buy-in of Bonds Not Delivered*) of the 2003 ISDA Credit Derivatives Definitions or any notice given pursuant to Section 9.10 (*Alternative Procedures Relating to Loans Not Delivered*) of the 2003 ISDA Credit Derivatives Definitions which is permitted to be and is delivered manually by CCM Client Buyer to CCM Seller/Matched Buyer in respect of and pursuant to the terms of a CCM Client Transaction (including the Mandatory Provisions) shall not be effective unless and until CCM Seller/Matched Buyer effectively delivers the relevant equivalent notice to the relevant Matched Seller in respect of and pursuant to the terms of the corresponding Restructuring Cleared Transaction or Physically Settled Cleared Transaction, as applicable.

CCM Seller/Matched Buyer undertakes to deliver such a notice to the relevant Matched Seller within two hours of its receipt of the equivalent notice from CCM Client Buyer if such notice is received between 9:00 a.m. (Calculation Agent City time) and 4:00 p.m. (Calculation Agent City time) on a Calculation Agent City Business Day. Any such notice received by CCM Seller/Matched Buyer after 4:00 p.m. (Calculation Agent City time) on a Calculation Agent City Business Day shall be deemed to have been received at 9:00 a.m. (Calculation Agent City time) on the following Calculation Agent City Business Day. Any such notice received by CCM Seller/Matched Buyer before 9:00 a.m. (Calculation Agent City time) on a Calculation Agent

City Business Day shall be deemed to have been received at 9:00 a.m. (Calculation Agent City time) on such Calculation Agent City Business Day. Any such notice delivered on a day that is not a Calculation Agent City Business Day shall be deemed to have been delivered at 9.00 a.m. (Calculation Agent City time) on the following Calculation Agent City Business Day.

The following provision shall solely be applicable in respect of a CCM Client Transaction between a CCM Client Seller and its CCM Buyer/Matched Seller:

6.3 Receipt of Certain Notices by CCM Buyer/Matched Seller deemed to be Receipt by CCM Client Seller

Any Credit Event Notice, Notice to Exercise Movement Option, Notice of Physical Settlement, NOPS Amendment Notice, any notice given pursuant to Section 9.9 (*Buy-in of Bonds Not Delivered*) of the 2003 ISDA Credit Derivatives Definitions or any notice given pursuant to Section 9.10 (*Alternative Procedures Relating to Loans Not Delivered*) of the 2003 ISDA Credit Derivatives Definitions which is permitted to be and is delivered manually by the relevant Matched Buyer to CCM Buyer/Matched Seller in respect of and pursuant to the terms of a Restructuring Cleared Transaction or Physically Settled Cleared Transaction (as applicable) relating to a CCM Client Transaction between such CCM Buyer/Matched Seller and CCM Client shall be deemed to constitute simultaneous delivery by CCM Buyer/Matched Seller to CCM Client Seller of such notice in respect of such CCM Client Transaction.

CCM Buyer/Matched Seller undertakes to deliver such a notice to CCM Client Seller within two hours of its receipt of the equivalent notice from the relevant Matched Buyer if such notice is received between 9:00 a.m. (Calculation Agent City time) and 4:00 p.m. (Calculation Agent City time) on a Calculation Agent City Business Day. Any such notice received by CCM Buyer/Matched Seller after 4:00 p.m. (Calculation Agent City time) on a Calculation Agent City Business Day shall be deemed to have been received at 9:00 a.m. (Calculation Agent City time) on the following Calculation Agent City Business Day. Any such notice received by CCM Buyer/Matched Seller before 9:00 a.m. (Calculation Agent City time) on a Calculation Agent City Business Day shall be deemed to have been received at 9:00 a.m. (Calculation Agent City time) on such Calculation Agent City Business Day. Any such notice delivered on a day that is not a Calculation Agent City Business Day shall be deemed to have been delivered at 9:00 a.m. (Calculation Agent City time) on the following Calculation Agent City Business Day.

7. Physical Settlement

7.1 Fallback to Cash Settlement resulting from corresponding CCM Client Cleared Transaction

If a CCM notifies its CCM Client that the Physically Settled Cleared Transaction corresponding to their CCM Client Transaction is to be settled (in whole or in part) by Cash Settlement pursuant to Section 6 (*Physical Settlement*) of the CDS Clearing Supplement and such CCM Client Transaction has not already been settled by Physical Settlement, such CCM Client Transaction shall also be settled (in whole or in part, as applicable) by Cash Settlement and the Cash Settlement Amount and the Cash Settlement Date shall be the same as the Cash Settlement Amount and the Cash Settlement Date determined in respect of the corresponding Physically Settled Cleared Transaction.

In respect of the CCM Client Transaction between CCM Client Buyer and CCM Seller/Matched Buyer, If CCM Client Buyer has already Delivered the Deliverable Obligations to CCM Seller/Matched Buyer, CCM Seller/Matched Buyer shall redeliver equivalent Deliverable Obligations to CCM Client Buyer in whole (if Cash Settlement applies) or in part (if Partial Cash Settlement applies in which case CCM Seller/Matched Buyer shall retain a proportion of the Deliverable Obligations equal to the proportion of the Transaction to be settled by Physical Settlement).

7.2 Fallback to Cash Settlement in respect of Non-Deliverable Obligations

If, in respect of a CCM Client Transaction, Buyer is not permitted to Deliver one or more Deliverable Obligations (such Deliverable Obligations, the Non-Deliverable Obligations) specified in the relevant Notice of Physical Settlement or NOPS Amendment Notice to Seller because:

the amount of such Deliverable Obligation is less than the relevant minimum denomination of such Deliverable Obligation; or

Seller is not a permitted transferee under such Deliverable Obligation (and, in the case of this subsection (ii), such circumstance would not constitute an illegality or impossibility outside the control of a relevant party for the purposes of Section 9.3 (*Partial Cash Settlement Due to Impossibility or Illegality*) of the 2003 ISDA Credit Derivatives Definitions),

then it shall notify Seller accordingly describing in reasonable detail the relevant circumstances.

With effect from such notification, such occurrence shall be treated, in relation to such CCM Client Transaction, as an illegality or impossibility outside the control of a relevant party for the purpose of Section 9.3 (*Partial Cash Settlement Due to Impossibility or Illegality*) of the 2003 ISDA Credit Derivatives Definitions and "Cash Settlement" pursuant to the Partial Cash Settlement Terms shall be deemed to apply to such CCM Client Transaction with respect to the Non-Deliverable Obligations as though the Non-Deliverable Obligations were Undeliverable Obligations and the provisions set out in Mandatory Provision 7.3 (*Consequences of Cash Settlement*) below shall apply.

7.3 Consequences of Cash Settlement

If the circumstances set out in either Section 9.3 (Partial Cash Settlement Due to Impossibility or Illegality) of the 2003 ISDA Credit Derivatives Definitions or Mandatory Provision 7.2 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) apply to a CCM Client Transaction, then:

(a) the Latest Permissible Physical Settlement Date in respect of such CCM Client Transaction will be deemed to be the first date on which the relevant Buyer or Seller effectively gave the relevant notice to the other pursuant to either Section 9.3 (Partial Cash Settlement Due to Impossibility or Illegality) of the 2003 ISDA Credit Derivatives Definitions or Mandatory Provision 7.2 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) above, as applicable, (and for these purposes, Section 9.7

- (Latest Permissible Physical Settlement Date) of the 2003 ISDA Credit Derivatives Definitions shall not apply); and
- (b) where sub-paragraph (ii) of Mandatory Provision 7.2 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) applies, Indicative Quotations shall not be applicable.

The following provisions shall solely be applicable in respect of a CCM Client Transaction between a CCM Client Buyer and its CCM Seller/Matched Buyer:

7.4 Delivery of Deliverable Obligations by CCM Client Buyer to CCM Seller/Matched Buyer

This Mandatory Provision 7.4 shall be applicable unless the CCM Client Buyer and CCM Seller/Matched Buyer agree that it shall not apply in respect of a specific CCM Client Transaction.

In respect of a CCM Client Transaction, the Delivery of any Deliverable Obligations to be Delivered by CCM Client Buyer to CCM Seller/Matched Buyer shall be deemed not to have occurred for the purposes of such CCM Client Transaction unless and until CCM Seller/Matched Buyer Delivers equivalent Deliverable Obligations to Matched Seller pursuant to the Physical Settlement of the corresponding Physically Settled Cleared Transaction.

Unless in respect of the corresponding Physically Settled Cleared Transactions:

- (a) Cash Settlement is applicable (in whole or in part);
- a Buy-in Period is applicable or Buy-in Price has been determined pursuant to Section
 9.9 (Buy-in of Bonds Not Delivered) of the 2003 ISDA Credit Derivatives Definitions;
- a Deemed Buy-in Period is applicable pursuant to Section 6.8(b)(i) of the CDS Clearing Supplement or Section 6.8(b)(ii) of the CDS Clearing Supplement is applicable; or
- (d) Section 9.10(a) of the 2003 ISDA Credit Derivatives Definitions is applicable or Matched Seller has required Matched Buyer to Deliver a Bond or Loan pursuant to Section 9.10(b) of the 2003 ISDA Credit Derivatives Definitions,

(each such event, a "Non-delivery Event"), CCM Seller/Matched Buyer undertakes to Deliver the Deliverable Obligations (or such portion of the Deliverable Obligations not affected by a Non-delivery Event) to Matched Seller not later than the first Business Day after the day on which a trade in such Deliverable Obligation would, if effected on the day on which CCM Seller/Matched Buyer received the Deliverable Obligations from CCM Client Buyer or on which the relevant Non-delivery Event ceased to apply, as applicable, (or if such day is not a Business Day, the following Business Day), be settled in accordance with then current market practice of such Deliverable Obligation, as determined by the Calculation Agent after consultation with the parties.

For the purposes of Article VIII (*Terms relating to Physical Settlement*) of the 2003 ISDA Credit Derivatives Definitions, the Physical Settlement Amount shall not be payable by CCM Seller/Matched Buyer to CCM Client Buyer in respect of the CCM Client Transaction until the

Physical Settlement of the corresponding Physically Settled Cleared Transaction has occurred.

7.5 Buy-in of Bonds not Applicable

Section 9.9 (Buy-in of Bonds Not Delivered) of the 2003 ISDA Credit Derivatives Definitions shall not apply.

7.6 Buyer's Right to Deliver suspended during Buy-in Period

If CCM Seller/Matched Buyer notifies CCM Client Buyer in respect of a CCM Client Transaction that (i) it has received a Buy-in Notice from Matched Seller in respect of the Matched Contracts of the related Settlement Matched Pair or (ii) it has been notified by Matched Seller in respect of the Matched Contracts of the related Settlement Matched Pair pursuant to Section 6.8(b) (Deemed Buy-in of Bonds resulting from CCM Client Transaction of Matched Seller) of the CDS Clearing Supplement that such Matched Seller has received a Buy-in Notice from its CCM Client in respect of the CCM Client Transaction between such Matched Seller and its CCM Client, such notice from CCM Seller/Matched Buyer specifying:

- i) the Buy-in Date;
- ii) the Relevant Bonds; and
- iii) the Outstanding Principal Balance thereof sought to be bought-in,

then CCM Client Buyer's right to Deliver the specified Relevant Bonds shall be suspended until the fourth Business Day (inclusive) following such Buy-in Date.

7.7 Buy-in of Bonds in respect of Matched Contracts of the Settlement Matched Pair

Provided that Physical Settlement has not already occurred in respect of a CCM Client Transaction, if CCM Seller/Matched Buyer notifies CCM Client Buyer that a Buy-in Price has been (i) determined or (ii) deemed to be determined pursuant to Section 6.8(b) (*Deemed Buy-in of Bonds resulting from CCM Client Transaction of Matched Seller*) of the CDS Clearing Supplement, in respect of Relevant Bonds for the purposes of the Matched Contracts of the related Settlement Matched Pair, then on the Buy-in Effective Date:

- i) CCM Client Buyer will be deemed to have Delivered to CCM Seller/Matched Buyer an Outstanding Principal Balance of the Deliverable Obligations equal to the Outstanding Principal Balance of the Deliverable Obligations Delivered or deemed to be Delivered by CCM Seller/Matched Buyer to Matched Seller in respect of the Matched Contracts; and
- ii) the Physical Settlement Amount to be paid by CCM Seller/Matched Buyer to CCM Client Buyer in respect of this CCM Client Transaction shall be reduced (but not below zero) by an amount equal to the amount by which the Physical Settlement Amount to be paid to CCM Seller/Matched Buyer by Matched Seller in respect of the Matched Contracts is to be reduced.

CCM Seller/Matched Buyer shall notify CCM Client Buyer of such Outstanding Principal Balance of the Deliverable Obligations and such Physical Settlement Amount reduction for the purposes of i) and ii) above and of the Buy-in Effective Date.

If CCM Client Buyer has already Delivered Deliverable Obligations to CCM Seller/Matched Buyer, CCM Seller/Matched Buyer shall redeliver equivalent Deliverable Obligations to CCM Client Buyer and such equivalent Deliverable Obligations shall have an Outstanding Principal Balance equal to the Outstanding Principal Balance of the Deliverable Obligation(s) specified by CCM Seller/Matched Buyer in the above notice and deemed to have been Delivered by CCM Client Buyer to CCM Seller/Matched Buyer pursuant to this Mandatory Provision 7.7.

7.8 Alternative Procedures relating to Loans – Seller Right to Select

Section 9.10(b) (Alternative Procedures Relating to Loans Not Delivered) of the 2003 ISDA Credit Derivatives Definitions shall not apply.

7.9 Alternative Procedures relating to Loans – Seller designates alternative Loan or Bond

In respect of a CCM Client Transaction, CCM Seller/Matched Buyer shall notify CCM Client Buyer if it has purchased Bond(s) and/or Loan(s) pursuant to Section 9.10(b) (*Alternative Procedures Relating to Loans Not Delivered*) of the 2003 ISDA Credit Derivatives Definitions on the instructions of the Matched Seller in respect of the corresponding CCM Client Cleared Transaction.

Following such notification, such Bond(s) and/or Loan(s) shall be deemed to have been Delivered by CCM Client Buyer to CCM Seller/Matched Buyer in Physical Settlement of the CCM Client Transaction and the Physical Settlement Amount payable by CCM Seller/Matched Buyer to CCM Client Buyer shall be reduced (but not below zero) by an amount equal to the price at which such Bond(s) and or Loan(s) were purchased.

CCM Seller/Matched Buyer shall provide CCM Client Buyer with details of such Bond(s) and/or Loan(s) and the related purchase price(s) in such notice.

If CCM Client Buyer has already Delivered Deliverable Obligations to CCM Seller/Matched Buyer, CCM Seller/Matched Buyer shall redeliver equivalent Deliverable Obligations to CCM Client Buyer and such equivalent Deliverable Obligations shall have an Outstanding Principal Balance equal to the Outstanding Principal Balance of the Bond(s) and/or Loan(s) specified by CCM Seller/Matched Buyer in the above notice and deemed to have been Delivered by CCM Client Buyer to CCM Seller/Matched Buyer pursuant to this Mandatory Provision 7.9.

The following provisions shall solely be applicable in respect of a CCM Client Transaction between CCM Client Seller and CCM Buyer/Matched Seller:

7.10 Delivery of Deliverable Obligations to CCM Client Seller

This Mandatory Provision 7.10 shall be applicable unless the CCM Client Seller and CCM Buyer/Matched Seller agree that it shall not apply in respect of a specific CCM Client Transaction.

Subject to the proviso below, in respect of a CCM Client Transaction, the Delivery of any Deliverable Obligations to be Delivered by CCM Buyer/Matched Seller to CCM Client Seller shall be deemed to have occurred for the purposes of such CCM Client Transaction upon receipt by CCM Buyer/Matched Seller of the Deliverable Obligations in respect of the Physical Settlement of the related Physically Settled Cleared Transaction; provided, however, that if the CCM Client Transaction is to be settled by Cash Settlement (in whole or in part) pursuant to Section 9.3 (Partial Cash Settlement Due to Impossibility or Illegality) of the 2003 ISDA Credit Derivatives Definitions or Mandatory Provision 7.2 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) then such portion of the CCM Client Transaction which is to be settled by Cash Settlement shall not be deemed to be settled until such Cash Settlement occurs.

Provided that Cash Settlement is not applicable, CCM Buyer/Matched Seller undertakes to Deliver the Deliverable Obligations to CCM Client Seller not later than the first Business Day after the day on which a trade in such Deliverable Obligation would, if effected on the day on which CCM Buyer/Matched Seller received the Deliverable Obligations from Matched Buyer (or if such day is not a Business Day, the following Business Day), be settled in accordance with then current market practice of such Deliverable Obligation, as determined by the Calculation Agent after consultation with the parties.

7.11 Alternative Procedures relating to Loans – Seller Right to Select

Section 9.10(b) (Alternative Procedures Relating to Loans Not Delivered) of the 2003 ISDA Credit Derivatives Definitions shall not apply.

7.12 Alternative Procedures relating to Loans in respect of Matched Contracts

In respect of a CCM Client Transaction, if CCM Buyer/Matched Seller notifies CCM Client Seller that a deemed amendment has been made to the Notice of Physical Settlement or any NOPS Amendment Notice in respect of the Matched Contracts of the Settlement Matched Pair that corresponds to such CCM Client Transaction pursuant to (i) Section 9.10(a) or (b) of the 2003 ISDA Credit Derivatives Definitions or (ii) Section 6.9 (Alternative Procedures relating to Loans — Buyer Right to Deliver (Section 9.10(a) 2003 ISDA Credit Derivatives Definitions) of the CDS Clearing Supplement, then the Notice of Physical Settlement or any NOPS Amendment Notice in respect of such CCM Client Transaction shall be deemed to have been amended by the addition of the Bonds or Loans which have been deemed to be added to the Notice of Physical Settlement or any NOPS Amendment Notice for the purposes of the Matched Contracts of the related Settlement Matched Pair.

8. Self Referencing Transactions

8.1 Section 2.31 (Merger of Reference Entity and Seller) of the 2003 ISDA Credit Derivatives

Definitions

Section 2.31 (Merger of Reference Entity and Seller) of the 2003 ISDA Credit Derivatives Definitions shall not apply.

8.2 Notification of Self Referencing Transactions

In respect of any Single Name CCM Client Transaction, the CCM Client shall, unless prohibited from so doing by applicable law, notify the CCM as soon as reasonably practicable if:

- the CCM Client is or consolidates or amalgamates with, or merges into, or transfers all or substantially all of its assets to, the Reference Entity in relation to such Single Name CCM Client Transaction or enters into any agreement in respect of any of the foregoing;
- ii) the CCM Client and the Reference Entity in relation to such Single Name CCM Client Transaction are or become Affiliates; or
- iii) in respect of a Restructuring CCM Client Transaction, the CCM Client is or becomes the Reference Entity in relation to such Restructuring CCM Client Transaction as a result of the occurrence of the relevant Restructuring Credit Event.

8.3 Termination of Self Referencing Transactions

A CCM Client Transaction shall be terminated (unless it has already been terminated) at the same time as the termination of the corresponding Single Name Cleared Transaction pursuant to Section 9.2 (*Occurrence of Client Self Referencing Transactions*) of the CDS Clearing Supplement and by reference to the price at which such Single Name Cleared Transaction is terminated and an amount will be payable:

- (a) if the CCM receives an amount from LCH SA in relation to such Single Name Cleared Transaction pursuant to Section 9.2 (Occurrence of Client Self Referencing Transactions), by the CCM to the CCM Client equal to such amount and on the Business Day following receipt by the CCM of such amount from LCH SA; and
- (b) if the CCM is obliged to pay an amount to LCH SA in relation to such Single Name Cleared Transaction pursuant to Section 9.2 (Occurrence of Client Self Referencing Transactions), by the CCM Client to the CCM equal to such amount and on the later of (I) the Business Day prior to the day on which the CCM is obliged to pay such amount to LCH SA and (II) the Business Day following the Business Day on which the CCM gives notices to the CCM Client of the relevant amount.

8.4 Costs of Terminating Self Referencing Transactions

Without prejudice to any other indemnity agreed between the CCM and the CCM Client in relation to CCM Client Transactions, the CCM Client agrees to indemnify and hold harmless the CCM from and against all costs and expenses that the CCM is obliged to bear pursuant to

File No. SR-LCH SA-2022-007 Page 132 of 439

PUBLIC

Section 9.2 (Occurrence of Client Self Referencing Transactions) of the CDS Clearing Supplement.

8.5 Compression of Self Referencing Transactions

Where the CCM acts as Matched Buyer and Matched Seller in respect of fungible Single Name Cleared Transactions that have a corresponding CCM Client Transaction in respect of which CCM Client has given notice to CCM pursuant to Mandatory Provision 8.2 (Notification of Self Referencing Transactions) and the relevant Single Name Cleared Transactions are compressed pursuant to Section 9.2 (Occurrence of Client Self Referencing Transactions) of the CDS Clearing Supplement, the CCM Client will be deemed to have submitted to CCM a request to compress the corresponding CCM Client Transactions.

9. Calculation Agent

9.1 Appointment of Calculation Agent

The Calculation Agent in respect of any CCM Client Transaction shall be the CCM.

9.2 Calculations and Determinations of Calculation Agent

In the event that the Calculation Agent is entitled or required to make any calculation or determination in respect of a CCM Client Transaction in respect of a matter that has already been or will be determined in respect of and pursuant to the terms of the corresponding CCM Client Cleared Transaction, the Calculation Agent in respect of the CCM Client Transaction shall be obliged to make the same calculation or determination in respect of such CCM Client Transaction as the determination in respect of the corresponding CCM Client Cleared Transaction (including, without limitation, any determination of any Cash Settlement Amount payable in respect of the CCM Client Transaction).

10. Amendments

The Mandatory Provisions may be amended from time to time pursuant to Section 11 (Amendments) of the CDS Clearing Supplement. The parties agree that any amendments made to the Mandatory Provisions in accordance with Section 11 (Amendments) of the CDS Clearing Supplement shall be deemed to apply automatically to the CCM Client Transaction(s) with effect from the date of such amendment to the Mandatory Provisions.

Formatted: Font: +Body (Calibri), 11 pt

File No. SR-LCH SA-2022-007 Page 133 of ...

PUBLIC

PART B CDS CLEARING SUPPLEMENT

FOR INDEX CLEARED TRANSACTIONS AND SINGLE NAME TRANSACTIONS INCORPORATING THE 2014 ISDA CREDIT DERIVATIVES DEFINITIONS

CONTENTS

SEC	TION	PAGE
1.	GENERAL PROVISIONS	134
2.	TERMS OF CLEARED TRANSACTIONS	146
3.	PAYMENTS AND DELIVERIES	155
4.	CREDIT EVENTS, SUCCESSION EVENTS AND RENAME EVENTS	156
5.	RESTRUCTURING	160
6.	PHYSICAL SETTLEMENT	162
7.	DELIVERY OF NOTICES AND FALLBACKS	175
8.	MATCHED PAIR DESIGNATIONS AND NOTICES	184
9.	SELF-REFERENCING TRANSACTIONS	188
10.	MANDATORY PROVISIONS FOR CCM CLIENT TRANSACTIONS	192
11.	AMENDMENTS	192
12.	FORMS OF NOTICES	192
13.	EXCLUSION OF LIABILITY	193
14.	DISPUTE RESOLUTION	193
15	GOVERNING I AW	193

1. GENERAL PROVISIONS

1.1 Incorporation of Defined Terms

Capitalised terms used in this CDS Clearing Supplement and not otherwise defined herein shall have the meaning given pursuant to the Index Cleared Transaction Confirmation, Single Name Cleared Transaction Confirmation, the 2014 ISDA Credit Derivatives Definitions or the CDS Clearing Rule Book, as applicable. In the case of any such terms defined in the CDS Clearing Rule Book, such terms shall be interpreted in accordance with the governing law specified therefore in the CDS Clearing Rule Book.

1.2 Terms defined in the CDS Clearing Supplement

For the purposes of the CDS Clearing Documentation, the following capitalised terms shall, unless otherwise specified, have the respective meanings set out below:

2003 ISDA Credit Derivatives Definitions: The 2003 ISDA Credit Derivatives Definitions published by ISDA as supplemented by the 2009 ISDA Credit Derivatives Determinations Committees, Auction Settlement and Restructuring Supplement to the 2003 ISDA Credit Derivatives Definitions, including the DC Rules and Credit Derivatives Auction Settlement Terms (each as defined therein).

2014 ISDA Credit Derivatives Definitions: The 2014 ISDA Credit Derivatives Definitions published by ISDA.

Affected Cleared Transaction: Following

- (i) a DC Credit Event Announcement or a publication by the DC Secretary of:
 - (a) a Successor Resolution; or
 - (b) a DC Resolution that a Substitute Reference Obligation has been determined in respect of a Non-Standard Reference Obligation; or
- (ii) publication of a revised SRO List,
- a Cleared Transaction (or the relevant Component Transaction(s) of such Cleared Transaction) that references the affected Reference Entity and also references a Transaction Type and/or a Reference Obligation affected by the event in (i) or (ii) above, as applicable.

Asset Package Cash Settlement Amount: In respect of an Asset Package, an amount in the Settlement Currency of the Physically Settled Cleared Transaction to which such Asset Package relates equal to the total amount of cash in such Settlement Currency (whether actual or deemed in accordance with Section 6.19(e) (Asset Package Delivery) comprising such Asset Package.

Asset Package Notice: A notice in the form set out in Appendix IVB.

Buy-in Effective Date: As set out at Section 6.8 (*Buy-in of Bonds – Matched Seller has entered into CCM Client Transaction*).

CCM Client Cleared Transaction: A Cleared Transaction between a CCM and LCH SA registered in a CCM Client Trade Account of a CCM.

CCM Client Transaction: A Transaction between a CCM and a CCM Client which is on the same economic terms as its corresponding CCM Client Cleared Transaction.

CCM Client Transaction Documents: The documentation entered into by a CCM and its CCM Client to document a CCM Client Transaction.

CDSClear Preferred Reference Obligation: This term shall have the meaning set out in Section 4 of the Procedures.

CDSClear Product Committee: A committee composed of representatives of LCH SA and representatives of Clearing Members for the purposes of carrying-out the tasks apportioned to it by the CDS Clearing Documentation as further described in the terms of reference for such committee, agreed in consultation with the Clearing Members and as may be amended from time to time in consultation with the Clearing Members.

CDS Clearing Rule Book: The document entitled "CDS Clearing Rule Book" published by LCH SA, as amended from time to time.

CDS Type: A class of Cleared Transactions that are identical as to their terms, except that they may differ as to:

- (a) the Trade Date;
- (b) in respect of Index Cleared Transactions, the Original Notional Amount;
- (c) in respect of Single Name Cleared Transactions, the Effective Date (provided that the current and future Fixed Rate Payer Calculation Periods for such Single Name Cleared Transactions are the same), the Floating Rate Payer Calculation Amount and the Reference Obligation (provided that LCH SA determines that the Reference Obligations of the Cleared Transactions are equivalent for the purposes of this CDS Clearing Supplement);
- (d) the Initial Payment Payer;
- (e) the Initial Payment Amount;
- (f) the Initial Payment Date; and
- (g) the identity of the relevant Buyer and Seller.

CEN Triggering Period: In relation to any Restructuring Cleared Transaction, the period during which the parties thereto may deliver a Credit Event Notice in respect of an M(M)R

Restructuring in relation to all or part of such Restructuring Cleared Transaction, such period starting at 9:00 a.m. on and including the earlier to occur of:

- (a) the Transaction Business Day following publication of the related Final List; and
- (b) the fifth calendar day following the No Auction Announcement Date,

and ending on and including the Exercise Cut-off Date.

Clearing Member Acknowledgement: As set out at Section 7.7 (*Clearing Member Acknowledgements*).

Clearing Member Communications Failure Event: As set out at Section 7.4 (*Notification of DTCC Failure and Resolution*).

Clearing Member Self Referencing Transaction: A Single Name Cleared Transaction which is registered in the House Account Structure of the Clearing Member and in respect of which, the Reference Entity is either the relevant Clearing Member or an Affiliate of such Clearing Member.

Client Self Referencing Transaction: A Single Name Cleared Transaction (a) that is registered in the Client Account Structure of a Clearing Member, and (b) in respect of which, the Reference Entity is either the relevant Client or an Affiliate of such Client.

Component Transaction: As defined in the Index Cleared Transaction Confirmation.

Compression Cut-off Date: The last date on which a Clearing Member may submit a request for any Cleared Transaction to be compressed pursuant to the ad hoc compression methodology and on which a daily automatic compression cycle will be run by LCH SA, in each case in accordance with Chapter 3 (*Compression*) of Title III (*Clearing Operations*) of the CDS Clearing Rule Book and Section 5 of the Procedures, being:

- in respect of any Index Cleared Transaction, the date falling one Transaction Business
 Day prior to the Novation Cut-off Date in respect of the relevant Eligible Index Version;
- (b) in respect of any Single Name Cleared Transaction and:
 - (i) an M(M)R Restructuring, the earlier of (A) the date of publication of the relevant Initial List (as defined in the DC Rules), (B) the date falling two Transaction Business Days prior to the relevant date on which the related RMP Notification Deadline falls and (C) such other date falling between the dates in (A) and (B), as determined by LCH SA in consultation with the CDSClear Product Committee and notified to the relevant Clearing Members prior to such date; and
 - (ii) a Failure to Pay Credit Event, a Governmental Intervention Credit Event, a Bankruptcy <u>Credit Event, an Obligation Acceleration Credit Event, a</u> <u>Repudiation/Moratorium</u> Credit Event or a Restructuring Credit Event (other than an M(M)R Restructuring), the date falling one Transaction Business Day

File No. SR-LCH SA-2022-007 Page 137 of 439

PUBLIC

prior to the calendar day following the related Auction Final Price Determination Date, Auction Cancellation Date or No Auction Announcement Date;

- (iii) a Succession Event, a date determined by LCH SA in consultation with the CDSClear Product Committee, which shall be not later that the Transaction Business Day before the date on which DTCC will amend its records in respect of such Single Name Cleared Transaction to take into account the occurrence of such Succession Event, and notified to the relevant Clearing Members prior to such date; and
- (iv) a Rename Event, the Transaction Business Day before the date on which DTCC will amend its records in respect of Single Name Cleared Transactions to take into account the occurrence of such Rename Event.

For the avoidance of doubt, where a Clearing Member has specified automatic compression on a weekly basis then the last such weekly automatic compression cycle performed by LCH SA will be performed on the Clearing Day falling on the Thursday on or before the relevant Compression Cut-off Date.

DC Restructuring Announcement Date: The date on which the DC Credit Event Announcement of an M(M)R Restructuring is made, *provided that* where such DC Credit Event Announcement is made after 6.30 p.m. on a Business Day or on a day which is not a Business Day, the DC Restructuring Announcement Date will be the first following Business Day.

Deemed Buy-in Period: As set out at Section 6.8(b)(i).

Dispute: This term shall have the meaning set out in the CDS Dispute Resolution Protocol.

DTCC Failure Event: As set out at Section 7.4 (Notification of DTCC Failure and Resolution).

DTCC Failure Event Time: As set out at Section 7.4 (Notification of DTCC Failure and Resolution).

DTCC Notice Facility: A facility made available pursuant to the DTCC Rules for the delivery of Credit Event Notices relating to M(M)R Restructurings or Notices to Exercise Movement Option.

DTCC Resolution Time: As set out at Section 7.4 (Notification of DTCC Failure and Resolution).

DTCC Rules: The "Operating Procedures", as published by DTCC and as amended from time to time.

Failed Amount: As set out at Section 6.10 (Failure to pay Physical Settlement Amount).

First Novation Date: In respect of:

(a) an Index Cleared Transaction, the first date on which LCH SA will accept Original Transactions referencing the relevant Eligible Index Version for clearing pursuant to

the CDS Clearing Documentation, being, in the case of any such version published pursuant to the occurrence of a Succession Event or Credit Event, the date on which credit default swap transactions referencing such version of such index are accepted for registration in the TIW in accordance with the DTCC Rules;

- (b) a Single Name Cleared Transaction in respect of which a Novation Cut-off Date has previously occurred pursuant to the occurrence of an M(M)R Restructuring which is applicable to the relevant combination of Reference Entity, Transaction Type and Reference Obligation, the calendar day following the Transaction Business Day following the latest possible Exercise Cut-off Date for the relevant M(M)R Restructuring or such other date on which LCH SA determines in consultation with the CDSClear Product Committee that LCH SA will begin to again accept Original Transactions referencing the relevant combination of Reference Entity, Transaction Type and Reference Obligation; and
- (c) a Single Name Cleared Transaction (other than in the circumstances set-out in (b) above), the first date on which LCH SA determines in consultation with the CDSClear Product Committee that LCH SA will accept or will begin to again accept (as applicable) Original Transactions referencing the relevant combination of Reference Entity, Transaction Type and Reference Obligation.

Index Cleared Transaction: A Cleared Transaction which references a portfolio of Reference Entities specified in a credit default swap index and consists of a Component Transaction in respect of each such Reference Entity, the terms of which are as evidenced by an Index Cleared Transaction Confirmation.

Index Cleared Transaction Confirmation: For

- (a) any Index Cleared Transaction which references (i) a Markit iTraxx® Australia Index Series [27] or above or (ii) a Markit iTraxx® Asia ex-Japan Index Series [27] or above, the form of confirmation which incorporates the iTraxx® Asia/Pacific Untranched Standard Terms Supplement, each being the lastest version in force, as published by Markit Group Limited;
- (b) any Index Cleared Transaction which references a Markit iTraxx® Europe Index Series 22 or above, the form of confirmation which incorporates the iTraxx® Europe Untranched Standard Terms Supplement, each being the lastest version in force, as published by Markit Group Limited;
- (c) any Index Cleared Transaction which references a Markit CDX™ Index Series 23 or above, the form of confirmation which incorporates the CDX Untranched Transactions Standard Terms Supplement, each being the latest version in force, as published by Markit North America, Inc.;
- (e)(d) any Index Cleared Transaction which references a Markit CDX™ Emerging Markets
 Index Series [27] or above, the form of confirmation which incorporates the CDX

Emerging Markets Untranched Transactions Standard Terms Supplement, each being the latest version in force, as published by Markit North America, Inc.;

- (d)(e) any Index Cleared Transaction which references a Markit iTraxx® Europe Index Series 21 or below, the form of confirmation which incorporates the iTraxx® Europe Legacy Untranched Standard Terms Supplement, each being the latest version in force, as published by Markit Group Limited; and
- (e)(f) any Index Cleared Transaction which references a Markit CDX™ Index Series 22 or below, the form of confirmation which incorporates the CDX Legacy Untranched Transactions Standard Terms Supplement, each being the latest version in force, as published by Markit North America, Inc.,

in each case as amended by this CDS Clearing Supplement and as completed by reference to the relevant Transaction Data- (or, in each case, such other form of confirmation as may be adopted in respect of any CDS Type in accordance with Section 1.2.2 (*Modification*) of the CDS Clearing Rule Book).

Index CCM Client Transaction: A CCM Client Transaction which references a portfolio of Reference Entities specified in a credit default swap index and consists of a Component Transaction in respect of each such Reference Entity and which is on the same economic terms as an Index Cleared Transaction registered in the CCM Client Trade Account in the name of such CCM for such CCM Client.

Initial Re-couponing Notice: As set-out in Section 4.4 (*Re-couponing of Restructuring Cleared Transactions*).

Initial Single Name Cleared Transaction: A Cleared Transaction entered into following the novation of an Original Transaction and which references a single Reference Entity, the terms of which are evidenced by a Single Name Cleared Transaction Confirmation.

Initial Single Name CCM Client Transaction: A CCM Client Transaction which references a single Reference Entity and which is on the same economic terms as an Initial Single Name Cleared Transaction registered in the CCM Client Trade Account in the name of such CCM for such CCM Client.

ISDA: The International Swaps and Derivatives Association, Inc. and any successor thereto.

Mandatory Provisions: As set-out in Appendix XIII (CCM Client Transaction Requirements).

Matched Buyer: A CDS Buyer comprised in a Matched Pair.

Matched Buyer Contract: A Cleared Transaction (or part thereof) between a Matched Buyer and LCH SA which is the subject of a Matched Pair having the Restructuring Matched Pair Amount or the Settlement Matched Pair Delivery Amount, as the case may be, relating to that Matched Pair.

Matched Contract: A Matched Seller Contract or a Matched Buyer Contract, as applicable.

Matched Pair: A Restructuring Matched Pair or Settlement Matched Pair, as applicable.

Matched Seller: A CDS Seller comprised in a Matched Pair.

Matched Seller Contract: A Cleared Transaction (or part thereof) between a Matched Seller and LCH SA which is the subject of a Matched Pair having the Restructuring Matched Pair Amount or the Settlement Matched Pair Delivery Amount, as the case may be, relating to that Matched Pair.

Matching Information Notification Deadline: In respect of a Restructuring Cleared Transaction, 9.00 a.m. on the first day of the CEN Triggering Period, *provided that* the Matching Information Notification Deadline shall fall no earlier than the fifth Transaction Business Day following the relevant DC Credit Event Announcement.

Matrix Re-versioning Date: As set-out in Section 2.5 (Physical Settlement Matrix Updates).

NEMO Triggering Period: In relation to any Restructuring Cleared Transaction, the period starting at 9:00 a.m. on the day falling on the relevant Movement Option Cut-off Date for the relevant Credit Event and ending at the last time for delivery of a valid Notice to Exercise Movement Option under the terms of the relevant Cleared Transaction.

No Physical Settlement Confirmation: As set out in Section 6.19 (*Miscellaneous Provisions relating to Physical Settlement*).

No Physical Settlement Confirmation Deadline: As set out in Section 6.19 (*Miscellaneous Provisions relating to Physical Settlement*).

Non-Deliverable Obligation: As set out at Section 6.12 (*Fallback to Cash Settlement in respect of Non-Deliverable Obligations*).

Non-DVP Asset Package Settlement Amount: As set out in Section 6.6(b) (*Delivery of Non-DVP Obligations*).

Non-DVP Obligation: In respect of any Physically Settled Cleared Transaction: (1) a Deliverable Obligation, validly specified in a Notice of Physical Settlement, which does not, in accordance with then current market standards, settle on a delivery-versus-payment basis through the books and records of a central securities depository or securities settlement system; or (2) any Asset forming part of an Asset Package validly specified in an Asset Package Notice given in accordance with the final paragraph of Section 8.2 (*Notice of Physical Settlement*) of the 2014 ISDA Credit Derivatives Definitions.

Notice Acknowledgement Deadline: As set out in Section 7.7 (*Clearing Member Acknowledgements*).

Notice Reconciliation Deadline: As set out in Section 7.7 (*Clearing Member Acknowledgements*).

Novation Cut-off Date: The date with effect from which LCH SA will no longer accept Original Transactions referencing an Eligible Index Version or Original Transactions referencing the

relevant combination of Reference Entity, Transaction Type and Reference Obligation for novation, being:

- (a) following the occurrence of an M(M)R Restructuring, the earlier of:
 - a date determined by LCH SA in consultation with the CDSClear Product Committee, which shall not be earlier than the DC Restructuring Announcement Date; and
 - (ii) the calendar day immediately following the No Auction Announcement Date,

provided that LCH SA in consultation with the CDSClear Product Committee may, but is not required to, determine that the Novation Cut-off Date for Index Cleared Transactions and the Novation Cut-off Date for Initial Single Name Cleared Transactions are to occur on different days;

- (b) following the occurrence of a Failure to Pay Credit Event, a Bankruptcy Credit Event, an Obligation Acceleration Credit Event, a Repudiation/Moratorium Credit Event, a Governmental Intervention Credit Event or a Restructuring Credit Event (other than an M(M)R Restructuring), the calendar day following the related Auction Final Price Determination Date, Auction Cancellation Date or No Auction Announcement Date, as applicable;
- (c) following the occurrence of a Succession Event:
 - (i) in respect of an Index Cleared Transaction, the date notified by DTCC to LCH SA as being the date on which DTCC will amend its records in respect of such Cleared Transaction to take into account a new version of such index published by the relevant index publisher taking into account the occurrence of such Succession Event; or
 - in respect of a Single Name Cleared Transaction, the day after the date on which the DC Secretary publicly announces a Succession Resolution in respect of such Succession Event;
- (d) following the occurrence of a Rename Event in respect of Single Name Cleared Transactions, the date determined by LCH SA in consultation with the CDSClear Product Committee, which such date shall not be later than the date on which DTCC amends its records in respect of Single Name Cleared Transactions to take into account the occurrence of such Rename Event; or
- (e) otherwise, as determined by LCH SA in consultation with the CDSClear Product Committee.

Partial Cash Settlement Terms: As set out in Section 9.6 (*Partial Cash Settlement Terms*) of the 2014 ISDA Credit Derivatives Definitions, as amended by this CDS Clearing Supplement.

Payer: As set out at Section 6.17 (Physical Settlement Costs).

File No. SR-LCH SA-2022-007 Page 142 of 439

PUBLIC

Physical Settlement Confirmation: As set out in Section 6.19 (*Miscellaneous Provisions relating to Physical Settlement*).

Physical Settlement Confirmation Deadline: As set out in Section 6.19 (*Miscellaneous Provisions relating to Physical Settlement*).

Physically Settled Cleared Transaction: As set out at Section 6.3 (*Physically Settled Cleared Transactions*).

Physical Settlement Matrix: The Credit Derivatives Physical Settlement Matrix (as defined in Section 13.2 (*Credit Derivatives Physical Settlement Matrix*) of the 2014 ISDA Credit Derivatives Definitions).

Recipient: As set out at Section 6.17 (*Physical Settlement Costs*).

Re-couponing Date: As set-out in Section 4.4 (*Re-couponing of Restructuring Cleared Transactions*).

Re-couponing Notice: As set-out in Section 4.4 (*Re-couponing of Restructuring Cleared Transactions*).

Relevant Physical Settlement Matrix: As set out in Section 4.3 (*Eligible Reference Entities*) of the Procedures.

Rename Event: As set out in Section 4.7 (Rename Events).

Restructuring Cleared Transaction: A Cleared Transaction which references a single Reference Entity and is created following the creation of Restructuring Matched Pairs pursuant to Section 5.2 (*Creation of Restructuring Cleared Transactions*), the terms of which are evidenced by a Single Name Cleared Transaction Confirmation.

Restructuring CCM Client Transaction: A CCM Client Transaction between a CCM and a CCM Client which references a single Reference Entity and which is on the same economic terms as a Restructuring Cleared Transaction registered in the CCM Client Trade Account in the name of such CCM for such CCM Client.

Restructuring Matched Pair: A pairing comprised of a Matched Buyer and a Matched Seller in respect of Restructuring Cleared Transactions created by LCH SA for the purposes of delivery of Credit Event Notices in respect of the relevant M(M)R Restructuring under Section 5.1 (*Creation and Notification of Restructuring Matched Pairs*).

Restructuring Matched Pair Amount: In respect of a Restructuring Matched Pair, the amount of the Floating Rate Payer Calculation Amount in respect of the relevant Restructuring Cleared Transactions allocated by LCH SA to such Restructuring Matched Pair under Section 8.1 (*Creation of Matched Pairs*).

Resulting Single Name Cleared Transaction: A Cleared Transaction which references a single Reference Entity and is created following the termination of a Restructuring Cleared

Transaction pursuant to Section 5.15 (*Recouponing*) of the Procedures, the terms of which are evidenced by a Single Name Cleared Transaction Confirmation.

RMP Notification Deadline: In respect of a notification by LCH SA to the relevant Clearing Members in relation to Restructuring Matched Pairs, 10.00 a.m. on the Transaction Business Day immediately prior to the first day of the CEN Triggering Period *provided that* the RMP Notification Deadline shall in no event fall prior to the second Transaction Business Day following the occurrence of the related DC Credit Event Announcement.

Self Referencing Transaction: A Clearing Member Self Referencing Transaction or a Client Self Referencing Transaction, as applicable.

Settlement Matched Pair: A pairing comprised of a Matched Buyer and a Matched Seller in respect of Physically Settled Cleared Transactions deemed to have been created by LCH SA under Section 6.2 (*Creation and Notification of Settlement Matched Pairs*).

Settlement Matched Pair Delivery Amount: In respect of a Settlement Matched Pair, the amount of the Floating Rate Payer Calculation Amount in respect of the relevant Physically Settled Cleared Transactions allocated by LCH SA to such Settlement Matched Pair under Section 8.1 (*Creation of Matched Pairs*).

Single Name Cleared Transaction: An Initial Single Name Cleared Transaction, a Spin-off Single Name Cleared Transaction, a Restructuring Cleared Transaction and a Resulting Single Name Cleared Transaction.

Single Name Cleared Transaction Confirmation: The form of confirmation for use with the Physical Settlement Matrix that incorporates the 2014 ISDA Credit Derivatives Definitions, as amended by this CDS Clearing Supplement and as completed by reference to:

- (a) in the case of an Initial Single Name Cleared Transaction, the relevant Transaction Data; or
- (b) in the case of any other type of Single Name Cleared Transaction, the relevant Cleared Transaction or Cleared Transactions that existed immediately prior to the relevant event that resulted in the creation of such Single Name Cleared Transaction (with such amendments as are required pursuant to the terms of the CDS Clearing Documentation).

or such other form of confirmation as may be adopted in respect of any CDS Type in accordance with Section 1.2.2 (*Modification*) of the CDS Clearing Rule Book.

Single Name CCM Client Transaction: A CCM Client Transaction between a CCM and a CCM Client which is on the same economic terms as a Single Name Cleared Transaction registered in the CCM Client Trade Account in the name of such CCM for such CCM Client.

SMP Notification Deadline: In respect of a notification by LCH SA to the relevant Clearing Members in relation to Settlement Matched Pairs, noon on the Transaction Business Day following the day on which the Fallback Settlement Method first becomes applicable.

Spin-off Single Name Cleared Transaction: The separate Single Name Cleared Transaction formed in respect of a Reference Entity from a Component Transaction of an Index Cleared Transaction following the occurrence of a DC Credit Event Announcement in respect of an M(M)R Restructuring which is applicable to the combination of Reference Entity, Transaction Type and Reference Obligation of such Component Transaction in accordance with the terms of such Index Cleared Transaction, the terms of which are evidenced by a Single Name Cleared Transaction Confirmation.

Succession Event: An entity (or entities) constitute a successor or successors in respect of a Reference Entity as determined pursuant to Section 2.2 (*Provisions for Determining a Successor*) of the 2014 ISDA Credit Derivative Definitions.

Successor Resolution: This term shall have the meaning set out in the DC Rules.

Transaction Business Day: A "Business Day", as defined in the Index Cleared Transaction Confirmation or the Single Name Cleared Transaction Confirmation, as applicable that if the relevant Index Cleared Transaction Confirmation or Single Name Cleared Transaction Confirmation defines such term differently depending upon its use, such distinction shall also apply to the use of the term Transaction Business Day herein.

Transaction Data: In respect of an Original Transaction to be novated pursuant to Title III (*Clearing Operations*) of the CDS Clearing Rule Book and cleared by LCH SA as an Index Cleared Transaction or Initial Single Name Cleared Transaction, the data provided by an Approved Trade Source System to LCH SA for such purposes, which includes, without limitation:

- in respect of an Index Cleared Transaction, the relevant index, including details of the index name, series and version, the annex date, the Original Notional Amount and the currency of the Original Notional Amount;
- (b) in respect of an Initial Single Name Cleared Transaction, the Reference Entity, Reference Obligation, applicable Transaction Type, Floating Rate Payer Calculation Amount, the currency of the Floating Rate Payer Calculation Amount and the Fixed Rate:
- (c) the Trade Date;
- (d) the Scheduled Termination Date;
- (e) the Floating Rate Payer;
- (f) the Fixed Rate Payer;
- (g) the Fixed Rate Payer Payment Dates;
- (h) the Initial Payment Payer;
- (i) the Initial Payment Amount; and

(j) whether the 2003 Credit Derivatives Definitions or the 2014 Credit Derivatives Definitions are applicable to such Original Transaction (or, in the case of an Index Cleared Transaction, each component of such Original Transaction).

1.3 Inconsistency

To the extent of any conflict between:

- any definition or provision contained in Appendix 1 (CDS Default Management Process) of the CDS Clearing Rule Book;
- (b) the remaining sections of the CDS Clearing Rule Book;
- (c) the CDS Admission Agreement;
- (d) this CDS Clearing Supplement;
- (e) an Index Cleared Transaction Confirmation or a Single Name Cleared Transaction Confirmation;
- (f) the Procedures; or
- (g) any Clearing Notices,

the first referenced document shall prevail except in relation to determining the existence and amount of any payment and delivery obligations under any Cleared Transactions, in respect of which this CDS Clearing Supplement, the Index Cleared Transaction Confirmation or the Single Name Cleared Transaction Confirmation, as applicable, shall prevail to the extent permitted by law.

1.4 Timing

Pursuant to Article 1.2.8 (*Time reference*) of the CDS Clearing Rule Book, any reference to a time of day herein shall be deemed to be a reference to Central European Time unless otherwise provided herein.

1.5 Third Party Rights

Unless otherwise provided in this CDS Clearing Supplement or in the CDS Clearing Rulebook, a person who is not a party to a Cleared Transaction does not have any right under the Contracts (Rights of Third Parties) Act 1999 to enforce any provision of such Cleared Transaction or this CDS Clearing Supplement as it relates to such Cleared Transaction.

1.6 Recording of Conversations

Each of LCH SA and each Clearing Member consents to the recording of telephone conversations between the trading, marketing and other relevant personnel of the parties in connection with any Cleared Transaction and agrees to obtain any necessary consent of, and give any necessary notice of such recording to, its relevant personnel and agrees, to the

extent permitted by applicable law, that such recordings may be submitted as evidence in any related court or arbitral proceedings.

1.7 Application to FCM/BD Clearing Members

- (a) Upon the taking of any action pursuant to this CDS Clearing Supplement by an FCM/BD Clearing Member in respect of an FCM/BD Cleared Transaction entered into as agent for the account of an FCM/BD Client (as described in FCM/BD CDS Clearing Regulation 1(c)), such FCM/BD Clearing Member shall be deemed to represent to LCH SA that it has the power and authority to, and has been duly authorised to, take such action for the account of such FCM/BD Client.
- (b) For purposes of this CDS Clearing Supplement, with regard to any Cleared Transaction entered into by an FCM/BD Clearing Member acting as agent for the account of an FCM/BD Client (as described in FCM/BD CDS Clearing Regulation 1(c)):
 - (i) references herein to "CDS Buyer" or "CDS Seller", as the case may be, shall be understood to be references to such FCM/BD Client (which shall not, for the avoidance of doubt, prejudice LCH SA's right to deal solely with the FCM/BD Clearing Member pursuant to Article 6.1.1.3(vi) of the CDS Clearing Rule Book); and
 - (ii) references herein to: (1) a Clearing Member entering into a Cleared Transaction with LCH SA; and (2) a Clearing Member forming part of a Matched Pair, shall each be understood as such FCM/BD Clearing Member acting as agent for the account of such FCM/BD Client (as described in FCM/BD CDS Clearing Regulation 1(c)).

2. TERMS OF CLEARED TRANSACTIONS

2.1 General Terms of Cleared Transactions

(a) Terms of Index Cleared Transactions and Initial Single Name Cleared Transactions

- (i) Upon novation of an Original Transaction at the Novation Time in accordance with Title III (Clearing Operations) of the CDS Clearing Rule Book, each resulting Index Cleared Transaction or Initial Single Name Cleared Transaction is entered into by LCH SA and the relevant Clearing Member on the terms of the related Index Cleared Transaction Confirmation or Single Name Cleared Transaction Confirmation (as applicable);
- (ii) Each Component Transaction of an Index Cleared Transaction which references a Markit iTraxx® Europe Index Series 21 or below or a Markit CDX™ Index Series 22 or below to which Section A of the relevant Index Cleared Transaction Confirmation applies shall be subject to the terms of Part A of this CDS Clearing Supplement; and

(iii) Each Component Transaction of an Index Cleared Transaction which references a Markit iTraxx® Europe Index Series 21 or below or a Markit CDX™ Index Series 22 or below to which Section B of the relevant Index Cleared Transaction Confirmation applies shall be subject to the terms of Part B of this CDS Clearing Supplement.

(b) Terms of Spin-off Single Name Cleared Transactions, Restructuring Cleared Transactions and Resulting Single Name Cleared Transactions

If any Spin-off Single Name Cleared Transaction is deemed to have been entered into by the parties in accordance with the terms of any Index Cleared Transaction or if any Restructuring Cleared Transaction is created pursuant to Section 5.2 (*Creation of Restructuring Cleared Transactions*) or if any Resulting Single Name Cleared Transaction is created pursuant to Section 5.15 (*Recouponing*) of the Procedures, such Spin-off Single Name Cleared Transaction, Restructuring Cleared Transaction or Resulting Single Name Cleared Transaction is entered into by LCH SA and the relevant Clearing Member on the terms of the related Single Name Cleared Transaction Confirmation.

(c) Trade Date of Cleared Transactions following Compression

Notwithstanding paragraphs (a) and (b) above, where Cleared Transactions are subject to compression in accordance with Chapter 3 (*Compression*) of Title III (*Clearing Operations*) of the CDS Clearing Rule Book, the Trade Date of any resulting Cleared Transaction(s) shall be, in respect of any Cleared Transaction subject to:

- (i) ad hoc compression (as described in Chapter 3 (Compression) of Title III (Clearing Operations) of the CDS Clearing Rule Book), the date on which the request for compression was effectively received and processed in accordance with Section 5 of the Procedures, which shall be:
 - (A) the Clearing Day on which such request is submitted and uploaded by the relevant Clearing Member provided that such request for compression was received by LCH SA before 7.00 p.m. on such Clearing Day (if such request is submitted and uploaded by the relevant Clearing member via any means of access specified in a Clearing Notice) or 5.00 p.m. (if such request is not submitted via any means of access specified in a Clearing Notice in the case of a disruption of the relevant means of access);
 - (B) the Clearing Day on which such request is submitted if such request is not submitted via any means of access specified in a Clearing Notice in the case of disruption of the relevant means of access and is submitted after 5.00 p.m. but LCH SA, in its sole discretion, processes such request on the Clearing Day on which such request is submitted; and

File No. SR-LCH SA-2022-007 Page 148 of 439

PUBLIC

- (C) unless the relevant Clearing Member instructs the Operations Department to withdraw such request, the Clearing Day following the Clearing Day on which such request is submitted if such request is not submitted via any means of access specified in a Clearing Notice in the case of disruption of the relevant means of access and is submitted after 5.00 p.m. and LCH SA, in its sole discretion, does not process such request on the Clearing Day on which such request is submitted; or
- (ii) automatic compression (as described in Chapter 3 (Compression) of Title III (Clearing Operations) of the CDS Clearing Rule Book), the Clearing Day on which such Cleared Transaction is automatically compressed by LCH SA in accordance with Section 5 of the Procedures.

2.2 Index Cleared Transaction Confirmation

The Index Cleared Transaction Confirmation is amended, supplemented and completed as follows:

- (a) (i) if the Index Cleared Transaction references a Markit iTraxx® Australia Index, a Markit iTraxx® Asia ex-Japan Index or a Markit iTraxx® Europe Index, by deleting the words "between [●] (Party A) and [●] (Party B)" in the third line of the first paragraph; and
 - (ii) if the Index Cleared Transaction references a Markit CDX™ Index <u>or a Markit CDX™</u> <u>Emerging Markets Index</u>, by deleting the words "between [Party A] ("<u>Party A</u>") and [counterparty's name] ("<u>Party B</u>")" in the third and fourth lines of the first paragraph,

and in each case replacing them with:

- "between LCH SA ("Party A") and Clearing Member, as identified in the relevant CDS Admission Agreement between such Clearing Member and Party A ("Party B")";
- (i) if the Index Cleared Transaction references a Markit iTraxx® Australia Index, a Markit iTraxx® Asia ex-Japan Index or a Markit iTraxx® Europe Index Series 22 or above, by deleting the fourth paragraph thereof (or, for Markit iTraxx® Europe Series 21 or below, the fifth paragraph thereof); and
 - (ii) if the Index Cleared Transaction references a Markit CDX™ Index <u>or a Markit CDX™</u> <u>Emerging Markets Index</u>, by deleting the third paragraph thereof,

and in each case replacing it with the following:

- "This Confirmation supplements, forms a part of, and is subject to, the CDS Clearing Documentation, as defined in the CDS Clearing Rule Book.";
- (c) if the Index Cleared Transaction references a Markit iTraxx® Australia Index, a Markit iTraxx® Asia ex-Japan Index or a Markit iTraxx® Europe Index Series 22 or above, by

Formatted: Font: Calibri

File No. SR-LCH SA-2022-007 Page 149 of 439

PUBLIC

deleting the fifth paragraph thereof (or, for Markit iTraxx® Europe Series 21 or below, the sixth paragraph thereof) and replacing it with the following:

"The terms of the iTraxx® Master Transaction, which is an Index Cleared Transaction, to which this Confirmation relates are as follows:";

(d) if the Index Cleared Transaction references a Markit CDX™ Index or a Markit CDX™ Emerging Markets Index, by deleting the fourth paragraph thereof and replacing it with the following:

"The terms of the Master Transaction, which is an Index Cleared Transaction, to which this Confirmation relates are as follows:";

- (e) by specifying that the "Calculation Agent" is Party A;
- (f) by inserting the following "Additional terms":

or above:

(i) "The applicable Physical Settlement Matrix is the version of the Physical Settlement Matrix which is in force on the Clearing Day on which the Index Cleared Transaction is registered by LCH SA";

i) if the Index Cleared Transaction references a Markit iTraxx® Australia Index, a Markit iTraxx® Asia ex-Japan Index or a Markit iTraxx® Europe Index Series 22

"The Standard Terms Supplement is amended for the purposes of this Transaction:

- (a) by deleting in its entirety the last paragraph of the definition of "Reference Obligation", beginning "If there is no Standard Reference Obligation and the Index Sponsor..." and deleting the words "and the following paragraph:" from the first paragraph of the definition of "Reference Obligation"; and
- (b) by deleting paragraph 5.5 (De Minimis Cash Settlement) in its entirety";
- (iii) if the Index Cleared Transaction references a Markit iTraxx® Europe Index Series 21 or below:

"The Legacy Standard Terms Supplement is amended for the purposes of this Transaction by deleting paragraph 5.4 (*De Minimis Cash Settlement*) of Section A and Section B in its entirety";

(ivi if the Index Cleared Transaction references a Markit CDX™ Index Series 23 or above:

149

"The CDX Untranched Terms is amended for the purposes of this Transaction by deleting paragraph 5.6 (*De Minimis Cash Settlement*) in its entirety and by

Formatted: Not Highlight

Formatted: Not Highlight

Formatted: Font: Calibri

deleting paragraph 5.2 (*Monoline Insurer as Reference Entity*) in its entirety and replacing it with the following:

The "Additional Provisions for Monoline Insurer Reference Entities", published on September 15, 2014, are incorporated by reference herein and shall be applicable to each Reference Entity that is identified as a "monoline" in the Index Annex";

(iv) if the Index Cleared Transaction references a Markit CDX™ Index Series 22 or below:

"The CDX Legacy Untranched Terms is amended for the purposes of this Transaction by deleting paragraph 5.3 (*De Minimis Cash Settlement*) of Section A and Section B in its entirety and by deleting paragraph 5.1 (*Monoline Insurer as Reference Entity*) of Section B in its entirety and replacing it with the following:

The "Additional Provisions for Monoline Insurer Reference Entities", published on September 15, 2014, are incorporated by reference herein and shall be applicable to each Reference Entity that is identified as a "monoline" in the Index Annex.";

(vi) if the Index Cleared Transaction references a Markit CDX™ Emerging Markets Index:

"The CDX Untranched Terms is amended for the purposes of this Transaction by deleting paragraph 5.6 (De Minimis Cash Settlement) in its entirety";

- (g) by deleting the contact details for notices and the account details; and
- (h) by deleting the signature blocks.

2.3 Single Name Cleared Transaction Confirmation

The Single Name Cleared Transaction Confirmation is amended, supplemented and completed as follows:

(a) by deleting the words "between us" from the first paragraph thereof and replacing them with:

"between LCH SA ("Party A") and Clearing Member, as identified in the relevant CDS Admission Agreement between Clearing Member and Party A ("Party B")";

(b) by deleting the third paragraph thereof and replacing it with the following:

"This Confirmation supplements, forms a part of and is subject to the CDS Clearing Documentation, as defined in the CDS Clearing Rule Book.";

(c) by specifying that the "Calculation Agent" is Party A;

- (d) notwithstanding the terms of the relevant Original Transaction, Standard Reference Obligation is applicable, provided that if there is no Standard Reference Obligation, by specifying that the Reference Obligation is the CDSClear Preferred Reference Obligation;
- by deleting references to the Initial Payment Payer and the Initial Payment Amount in part 2 in relation to Single Name Cleared Transactions in respect of which there is no Initial Payment Payer and Initial Payment Amount;
- (f) by deleting part 4 in its entirety unless the Transaction Type specified in respect of the Single Name Cleared Transaction is "North American Corporate" or "Standard North American Corporate", in which case part 4 shall be included and "Restructuring: Not Applicable" shall be specified in part 4;
- (g) by deleting parts 5, 6, 7 and 8 in their entirety unless the Reference Entity specified in respect of a Single Name Cleared Transaction is a monoline insurer, in which case part 6 shall be included and "Additional Provisions for Monoline Insurer Reference Entities (September 15, 2014): Applicable" shall be specified in part 6;
- (h) by inserting the following provision in the "Additional Terms" section of the Confirmation:
 - "Section 11.4 (Merger of Reference Entity and Seller) of the 2014 Definitions shall not apply."; and
- (i) by deleting the signature blocks.

LCH SA @ 202110219615309-v3

2.4 Amendments to 2014 ISDA Credit Derivatives Definitions

- (a) For the purposes of this CDS Clearing Supplement, Section 11.2(c)(iv) of the 2014 ISDA Credit Derivatives Definitions as incorporated in any Cleared Transaction shall be amended such that, where LCH SA is the designator in relation to any Restructuring Cleared Transaction or Physically Settled Cleared Transaction, as applicable, it is permitted to designate any relevant CDS Buyer or CDS Seller, as applicable, in accordance with Section 8 (Matched Pair Designations and Notices) as its designee, notwithstanding that such relevant CDS Buyer or CDS Seller, as applicable, is not its Affiliate and for the additional purposes of:
 - delivering or receiving any Credit Event Notice relating to an M(M)R Restructuring or Notice to Exercise Movement Option (in each case, to the extent not delivered via a DTCC Notice Facility);
 - (ii) delivering or receiving any Notice of Physical Settlement or NOPS Amendment Notice or any Asset Package Notice pursuant to Section 8.2 (Notice of Physical Settlement) of the 2014 ISDA Credit Derivatives Definitions;
 - (iii) making or receiving payment of any Physical Settlement Amount and any costs and expenses of Physical Settlement; and

151

- (iv) delivering or receiving any notice and otherwise exercising any rights or performing any obligations of LCH SA for the purposes of Sections 9.7 (Buy-in of Bonds Not Delivered), 9.8 (Alternative Procedures Relating to Loans Not Delivered) or 9.9 (Alternative Procedures Relating to Assets Not Delivered) of the 2014 ISDA Credit Derivatives Definitions.
- (b) Section 11.2(a) of the 2014 ISDA Credit Derivatives Definitions is amended by deleting the words "any master agreement governing the Credit Derivative Transaction" at the end thereof and replacing them with the words "the CDS Clearing Documentation".
- (c) Section 11.2(c)(ii) of the 2014 ISDA Credit Derivatives Definitions is amended by deleting the words "any master agreement governing the Credit Derivative Transaction" in the second paragraph thereof and replacing them with the words "the CDS Clearing Documentation".
- (d) Section 11.2(c)(iii) of the 2014 ISDA Credit Derivatives Definitions is amended by deleting both occurrences of the words "a master agreement governing the Credit Derivative Transaction" and replacing each with the words "the CDS Clearing Documentation".
- (e) The Protocol Effectiveness Condition (as defined in the 2019 ISDA Narrowly Tailored Credit Event Protocol published by ISDA on 27 August 2019 and as further supplemented from time to time (the "2019 NTCE Protocol")) being satisfied, then from and including the Implementation Date (as defined in the 2019 NTCE Protocol) the amendments set forth in Schedule 1 (Amendments) of the 2019 NTCE Protocol shall apply to each Cleared Transaction which incorporates the 2014 ISDA Credit Derivatives Definitions and each such Cleared Transaction shall be a Protocol Covered Transaction for the purposes of the 2019 NTCE Protocol provided that:
 - the First Novation Date in respect of such Cleared Transaction is prior to the Implementation Date;
 - (ii) such Cleared Transaction is not an Excluded Transaction (as defined in the 2019 NTCE Protocol) provided that if such Cleared Transaction relates to more than one Reference Entity and would otherwise be a Protocol Covered Transaction but for the inclusion of any Affected Portion (as defined in the 2019 NTCE Protocol) therein, such Affected Portion shall be excluded for the purposes of this Section 2.4(e) and the remaining portion of such Cleared Transaction shall be deemed to be a Protocol Covered Transaction;
 - (iii) the Scheduled Termination Date in respect of such Cleared Transaction is on or after the Implementation Date; and
 - (iv) no Event Determination Date has occurred in respect of such Cleared Transaction prior to the Implementation Date.

152

2.5 Physical Settlement Matrix Updates

(a) Publication of Revised Matrix

If ISDA publishes a new version (the "Revised Matrix") of the Physical Settlement Matrix that is a Relevant Physical Settlement Matrix in respect of a Reference Entity (the "Existing Matrix"), LCH SA will, in consultation with the CDSClear Product Committee, determine whether the Existing Matrix and the Revised Matrix are fungible for the purposes of Single Name Cleared Transactions referencing the relevant combination of Reference Entity, Transaction Type and Reference Obligation.

If there are any changes between the terms of the Revised Matrix and the Existing Matrix that would apply to Single Name Cleared Transactions referencing the relevant combination of Reference Entity, Transaction Type and Reference Obligation and the Existing Matrix were they to reference the Revised Matrix instead, then the Existing Matrix and the Revised Matrix shall be deemed not to be fungible and paragraph (c) below shall apply.

(b) Revised Matrix Fungible

If LCH SA determines in consultation with the CDSClear Product Committee that the Existing Matrix and the Revised Matrix are fungible for the purposes of Single Name Cleared Transactions referencing the relevant combination of Reference Entity, Transaction Type and Reference Obligation (such date of determination, a **Matrix Reversioning Date**), then:

- (i) LCH SA will promptly notify all Clearing Members of such determination; and
- (ii) each Single Name Cleared Transaction referencing the relevant combination of Reference Entity, Transaction Type and Reference Obligation and the Existing Matrix will, as of the close of business on the Matrix Re-versioning Date, be deemed to have been amended so as to reference the Revised Matrix, which shall become the Relevant Physical Settlement Matrix for the Reference Entity in respect of such combination of Transaction Type and Reference Obligation.

For the avoidance of doubt, following the publication of a Revised Matrix, there may be different Matrix Re-versioning Dates in respect of (A) Single Name Cleared Transactions referencing different Reference Entities and (B) Single Name Cleared Transactions referencing the same Reference Entity but referencing a different Transaction Type or Reference Obligation.

(c) Revised Matrix not Fungible

If it is determined that the Revised Matrix and the Existing Matrix are not fungible for the purposes of Single Name Cleared Transactions referencing the relevant combination of Reference Entity, Transaction Type and Reference Obligation, then existing Single Name Cleared Transactions referencing such combination of Reference File No. SR-LCH SA-2022-007 Page 154 of 439

PUBLIC

Entity, Transaction Type and Reference Obligation and the Existing Matrix shall continue to reference the Existing Matrix.

(d) Original Transactions submitted prior to Matrix Re-versioning Date

Any Original Transaction referencing a combination of Reference Entity, Transaction Type and Reference Obligation submitted for clearing after a Matrix Re-versioning Date in respect of the relevant combination of Reference Entity, Transaction Type and Reference Obligation and which references the Existing Matrix prior to such Matrix Re-versioning Date will, following novation, result in an Initial Single Name Cleared Transaction referencing the Revised Matrix.

2.6 Terms of STM Cleared Transaction

The following terms of a registered STM Cleared Transaction are designated as the "STM Cleared Terms".

If a Cleared Transaction is designated as an STM Cleared Transaction in accordance with Article 3.1.10.7 of the CDS Clearing Rule Book the terms of such Cleared Transaction will automatically, and without any further action by either party, include the STM Cleared Terms.

(a) Net present value

- (i) LCH SA shall, at least once per Cash Payment Day, determine the net present value of this STM Cleared Transaction in such manner and at such times as may be provided in Section 2 of the Procedures.
- (ii) Immediately upon LCH SA making such a determination of the net present value of this STM Cleared Transaction:
 - (A) An amount (determined in accordance with Section 2.14 (NPV Amount) of the Procedures) of cash may become due and payable by the Clearing Member or LCH SA (as applicable, and as determined in accordance with Section 2.14 (NPV Amount) of the Procedures) under these STM Cleared Terms; and
 - (B) the net present value of this STM Cleared Transaction shall for all purposes be reset to zero.
- (iii) The amount (if any) referred to in (ii)(A) above shall immediately become due and payable by the relevant party under the STM Cleared Terms applicable to this STM Cleared Transaction and shall be made in such manner and at such times as may be provided in Section 2 of the Procedures.
- (iv) In any circumstance preventing the NPV Payments or Variation Margin Collateral Transfers from being performed in US Dollar, LCH SA shall be entitled

Formatted: Indent: Left: 1.5 cm, Keep with next

to convert any amount denominated in US Dollar into Euro in accordance with Article 4.2.3.2 of the CDS Clearing Rule Book.

(b) Price Alignment Amount

- (i) LCH SA shall, at least once per Cash Payment Day, determine the Price Alignment Amount in respect of this STM Cleared Transaction in such manner and at such times as may be provided in Section 2 of the Procedures.
- (ii) Immediately upon LCH SA making such a determination of the Price Alignment Amount in respect of this STM Cleared Transaction, an amount (determined in accordance with Section 2.17 (Price Alignment Amount) of the Procedures) of cash may become due and payable by the Clearing Member or LCH SA (as applicable, and as determined in accordance with Section 2.17 (Price Alignment Amount) of the Procedures) under these STM Cleared Terms.
- (iii) The amount (if any) referred to in (ii) above shall immediately become due and payable by the relevant party under the STM Cleared Terms applicable to this STM Cleared Transaction and shall be made in such manner and at such times as may be provided in Section 2 of the Procedures.

3. PAYMENTS AND DELIVERIES

3.1 Obligation to pay and deliver

Each of LCH SA and each Clearing Member will make each payment or delivery specified under the terms of each Cleared Transaction to be made by it, subject to the other provisions of the CDS Clearing Documentation.

Payments under any Cleared Transaction will be made on the due date for value on that date in the place of the account specified for the relevant party in the CDS Admission Agreement (or such other account as may be designated by it from time to time for such purpose in accordance with the CDS Admissions Agreement and/or the Procedures, as applicable) and otherwise in accordance with the CDS Clearing Documentation, in freely transferable funds and in the manner customary for payments in the required currency.

Deliveries under any Cleared Transaction will be made for receipt on the due date in the manner customary for the relevant obligation unless otherwise specified in or pursuant to the CDS Clearing Documentation.

3.2 Payments under Original Transactions

(a) If any Initial Payment Amount or Fixed Amount is due and payable under the terms of an Original Transaction on or before the Clearing Day on which the related Cleared Transactions are created by novation pursuant to Title III (Clearing Operations) of the CDS Clearing Rule Book, such amount shall be payable under and in accordance with the terms of such Original Transaction. In such event, no corresponding Initial Payment Amount or Fixed Amount shall be payable in respect of such Cleared Transactions.

(b) If the Initial Payment Date of an Original Transaction is a date falling after the Clearing Day on which the Cleared Transactions related to such Original Transaction are created by novation pursuant to Title III (Clearing Operations) of the CDS Clearing Rule Book then the corresponding Initial Payment Date for the related Cleared Transactions shall occur on the Transaction Business Day which is also a Clearing Day immediately following the Clearing Day on which such related Cleared Transactions are created.

4. CREDIT EVENTS, SUCCESSION EVENTS AND RENAME EVENTS

4.1 Determination of Credit Events and Succession Events

Notwithstanding any provision of any Cleared Transaction to the contrary:

- (a) LCH SA (in its capacity as Calculation Agent with respect to such Cleared Transaction) shall not make any determination pursuant to Section 2.10 (Substitute Reference Obligation) of the 2014 ISDA Credit Derivatives Definitions or in respect of any matter which is or may be subject to resolution under Sections 3.5 (Successor Resolutions) or 3.6 (Substitute Reference Obligation Resolutions) of the DC Rules; and
- (b) neither LCH SA nor any Clearing Member shall be entitled to deliver a Successor Notice, a Repudiation/Moratorium Extension Notice or a Credit Event Notice (other than Credit Event Notices in relation to an M(M)R Restructuring in accordance with the terms of any Restructuring Cleared Transaction and, where applicable, the DTCC Rules and Section 7.8 (Failure to notify Matched Pairs)).

4.2 Credit Event Timeline

(a) Publication of Credit Event Timeline

Upon a DC Credit Event Announcement, LCH SA will publish and make available to Clearing Members a timeline in respect of the relevant Credit Event and related Affected Cleared Transactions, to notify, among other things, the relevant Novation Cut-off Date(s), Compression Cut-off Date(s) and First Novation Date(s).

Any such timeline shall (i) be published and made available on the date of such DC Credit Event Announcement or, if LCH SA determines that such publication on such date is not practicably possible, as soon as practicable thereafter (but in no event later than the Transaction Business Day before the earlier of the relevant Novation Cut-off Date and the relevant Compression Cut-off Date) and (ii) in all cases be without prejudice to and consistent with the terms of the CDS Clearing Documentation and any relevant DC Resolutions.

(b) Amendment of Credit Event Timeline

Any such timeline may be subject to subsequent amendment by LCH SA, by means of a Clearing Notice to Clearing Members, only to reflect subsequent DC Resolutions, timing provisions of any relevant Transaction Auction Settlement Terms and/or actions of DTCC, or in each case any subsequent amendments thereto. Any such

amendment shall be made by LCH SA as soon as reasonably practicable following the relevant event.

4.3 Novation and Compression following Credit Events

Any Restructuring Cleared Transaction (or portion thereof) in respect of which a valid Credit Event Notice is not delivered in respect of the relevant M(M)R Restructuring during the relevant CEN Triggering Period shall become eligible for compression in accordance with Chapter 3 (Compression) of Title III (Clearing Operations) of the CDS Clearing Rule Book on the Transaction Business Day following the related Exercise Cut-off Date applicable to the relevant Buyer.

By way of clarification to Chapter 3 (*Compression*) of Title III (*Clearing Operations*) of the CDS Clearing Rule Book, LCH SA shall effect compression of relevant Cleared Transactions if and to the extent requested by a Clearing Member.

4.4 Re-couponing of Restructuring Cleared Transactions

Following an M(M)R Restructuring, LCH SA may perform re-couponing in accordance with Section 5 of the Procedures on any Restructuring Cleared Transaction (or portion thereof) in respect of which a valid Credit Event Notice has not been delivered during the relevant CEN Triggering Period, provided that:

- (a) LCH SA notifies (such notice, the "Initial Re-couponing Notice") all of the relevant Clearing Members of its intention to perform such re-couponing and the intended date of such re-couponing (such date, the "Re-Couponing Date");
- (b) the Re-Couponing Date is no earlier than the later of (i) the end of the relevant CEN Trigger Period; and (ii) eight Transaction Business Days after the effective date of the Initial Re-couponing Notice;
- (c) all of the relevant Clearing Members have subsequently consented to the proposed re-couponing, subject to confirmation of the terms of such re-couponing, within five Transaction Business Days of the effective date of the Initial Re-couponing Notice;
- (d) after receiving consent from all relevant Clearing Members in accordance with paragraph (c) above, LCH SA notifies (such notice, the "Re-couponing Notice") the relevant Clearing Members, no later than three Transactions Business Days prior to the Re-Couponing Date, of the proposed terms of the Resulting Single Name Cleared Transactions determined by LCH SA in accordance with Section 5 of the Procedures; and
- (e) all of the relevant Clearing Members have subsequently consented to the proposed terms of such re-couponing process as set out in the Re-couponing Notice within two Transaction Business Days of the effective date of the Re-couponing Notice.

For the avoidance of doubt, if, in respect of a proposed re-couponing process, LCH SA does not obtain consent from all of the relevant Clearing Members in accordance with paragraphs

(c) or (e) above, then LCH SA may elect to propose a re-couponing process again in accordance with this Section 4.4 for so long as any relevant Restructuring Cleared Transactions remain outstanding.

4.5 Succession Events and Cleared Transactions

If LCH SA determines that any Original Transaction submitted for novation or any Cleared Transaction subject to compression would have been subject to a Succession Event but will no longer be subject to such Succession Event upon novation or compression (as applicable) because of the Trade Date that would be specified with respect to the relevant Cleared Transactions, LCH SA may take such action as it deems necessary to ensure that such Succession Event is given effect with respect to such Cleared Transactions, including, without limitation, specifying an alternate Trade Date for purposes of Section 2.1 (*Reference Entity*) of the 2014 ISDA Credit Derivatives Definitions with respect to each relevant Cleared Transaction or, where LCH SA determines that an alternative course of action is not practicable, declining to accept such Original Transaction for novation or Cleared Transaction for compression (as applicable).

4.6 Succession Event Timeline

(a) Publication of Succession Event Timeline

Following the publication of a Successor Resolution, LCH SA will publish and make available to Clearing Members a timeline in respect of the relevant Succession Event and related Affected Cleared Transactions, to notify, amongst other things, the relevant Novation Cut-off Date(s), Compression Cut-off Date(s) and First Novation Date(s)

Any such timeline shall be published and made available as soon as practicable (but in no event later than the Transaction Business Day before the earlier of the relevant Novation Cut-off Date and the relevant Compression Cut-off Date) and shall in all cases be without prejudice to and consistent with the terms of the CDS Clearing Documentation and any relevant DC Resolutions.

(b) Amendment of Succession Event Timeline

Any such timeline may be subject to subsequent amendment by LCH SA, by means of a Clearing Notice to Clearing Members, to reflect subsequent DC Resolutions and any relevant actions of DTCC, or in each case any subsequent amendments thereto. Any such amendment shall be made by LCH SA as soon as reasonably practicable following the relevant event.

4.7 Rename Events

In respect of Single Name Cleared Transactions, if a Reference Entity changes its name (a "Rename Event"), LCH SA will publish and make available to Clearing Members as soon as practicable upon becoming aware of such Rename Event a timeline in respect of the relevant Rename Event and related Affected Cleared Transactions, to notify, amongst other things,

the relevant Novation Cut-off Date(s), Compression Cut-off Date(s) and First Novation Date(s).

Any such timeline may be subject to subsequent amendment by LCH SA, by means of a Clearing Notice to Clearing Members, to reflect any relevant actions of DTCC. Any such amendment shall be made by LCH SA as soon as reasonably practicable following the relevant event.

4.8 Updating Eligible Index Versions

(a) Publication of Revised Index

Where the Index Publisher of an Eligible Index Version publishes a revised version of such index following:

- (i) a DC Credit Event Announcement;
- (ii) a Successor Resolution;
- (iii) the determination of a Substitute Reference Obligation in respect of a Non-Standard Relevant Obligation; or
- (iv) publication of a revised SRO List,

LCH SA will in consultation with the CDSClear Product Committee determine whether such revised index version is fungible with the Eligible Index Version after taking account of the relevant Credit Event, Succession Event or, as applicable, Substitute Reference Obligation.

(b) Index Version not Fungible

If LCH SA determines in consultation with the CDSClear Product Committee that such revised index version is not fungible with the Eligible Index Version after taking account of the relevant Credit Event, Succession Event or, as applicable, Substitute Reference Obligation, it shall notify DTCC accordingly so that DTCC does not automatically update the trade records for Index Cleared Transactions and Index CCM Client Transactions referencing the relevant Eligible Index Version in the TIW.

(c) Index Fungible

Unless LCH SA has notified DTCC in accordance with paragraph (b) above, following confirmation from DTCC that the trade records for Cleared Transactions referencing the relevant index have been updated in the TIW (such date a **DTCC Re-versioning Date**) so that such Cleared Transactions reference the revised index version (which such revised index version shall become the Eligible Index Version), LCH SA will update its corresponding records in the CDS Clearing System overnight following such DTCC Re-versioning Date. LCH SA will send the relevant reports referred to in Section 5 of the Procedures to the relevant Clearing Members on the Transaction Business Day

following such DTCC Re-versioning Date which will be updated so as to refer to Cleared Transactions referencing the revised index version.

4.9 Reversal of DC Credit Event Announcements and Margining

If a DC Credit Event Announcement is reversed then, subject to Section 10.2(a)(i) of the 2014 ISDA Credit Derivatives Definitions, LCH SA shall be obliged in accordance with Section 3 of the Procedures to calculate and shall be entitled to call for Margin and/or be obliged to return Margin with respect to each Clearing Member on the basis that the DC Credit Event Announcement that is reversed had not been made.

5. **RESTRUCTURING**

5.1 Creation and Notification of Restructuring Matched Pairs

Following the occurrence of an M(M)R Restructuring, LCH SA will create (on one or, if the Novation Cut-off Date in respect of Index Cleared Transactions is prior to the Novation Cut-off Date in respect of Initial Single Name Cleared Transactions, two or more occasions) Restructuring Matched Pairs in accordance with Section 8.1 (*Creation of Matched Pairs*). Each such Restructuring Matched Pair shall be composed of two Restructuring Cleared Transactions created at the same time as and as a result of the termination of the relevant Single Name Cleared Transactions pursuant to Section 5.2 (*Creation of Restructuring Cleared Transactions*) below.

LCH SA shall notify the relevant Matched Buyer and Matched Seller comprised within each Restructuring Matched Pair of:

- (a) the identity of the other Clearing Member (together with the address, fax number, telephone number, email address and other applicable notice details of such other Clearing Member) of such Matched Pair; and
- (b) the associated Restructuring Matched Pair Amount,

as soon as reasonably practicable following the related Compression Cut-off Date, but in any event, at or prior to the RMP Notification Deadline.

LCH SA will additionally provide to DTCC all relevant "Matching Information" (as defined in the DTCC Rules) on or prior to the Matching Information Notification Deadline.

5.2 Creation of Restructuring Cleared Transactions

Upon the notification to the relevant Clearing Members of Restructuring Matched Pairs, the relevant Initial Single Name Cleared Transactions, Restructuring Cleared Transactions (created following the occurrence of a previous M(M)R Restructuring), Spin-off Single Name Cleared Transactions and Resulting Single Name Cleared Transactions entered into by each Clearing Member with LCH SA will be deemed terminated and new Restructuring Cleared Transactions of the same CDS Type will be deemed to be entered into between each such Clearing Member and LCH SA, with each such Restructuring Cleared Transaction having a

Floating Rate Payer Calculation Amount corresponding to the Restructuring Matched Pair Amount of a Restructuring Matched Pair in which the relevant Clearing Member is comprised as a Matched Buyer or a Matched Seller, as applicable.

5.3 Triggering of Restructuring Cleared Transactions

Subject as set out in Section 7 (*Delivery of Notices and Fallbacks*), and notwithstanding anything to the contrary in the terms of any Cleared Transaction, Clearing Members may only deliver Credit Event Notices in relation to an M(M)R Restructuring during the CEN Triggering Period following notification of Restructuring Matched Pairs by LCH SA and subject always to the terms of the relevant Restructuring Cleared Transaction and, where applicable, the DTCC Rules.

Any Credit Event Notice in respect of an M(M)R Restructuring delivered in respect of a Restructuring Matched Pair for an amount which is greater than the related Floating Rate Payer Calculation Amount shall be ineffective as to such excess.

5.4 Notice to Exercise Movement Option

Subject as set out in Section 7 (*Delivery of Notices and Fallbacks*), Clearing Members may only deliver a Notice to Exercise Movement Option during the NEMO Triggering Period, subject always to the other terms of the relevant Restructuring Cleared Transaction and, where applicable, the DTCC Rules.

5.5 Reversal of DC Credit Event Announcements

If a DC Credit Event Announcement made in relation to an M(M)R Restructuring is reversed then, subject to Section 10.2(a)(i) of the 2014 ISDA Credit Derivatives Definitions:

- (a) LCH SA shall have no obligation to create Restructuring Matched Pairs in accordance with Section 5.1 (Creation and Notification of Restructuring Matched Pairs), any such Restructuring Matched Pairs so created shall be deemed not to have been created and any Credit Event Notices in respect of the relevant M(M)R Restructuring delivered in connection with such Restructuring Matched Pairs shall be deemed to be ineffective;
- (b) LCH SA shall, where applicable, make relevant registrations in the TIW in order to reflect such reversal which shall also automatically result in such registrations being made in respect of any related CCM Client Transactions; and
- (c) Section 4.9 (Reversal of DC Credit Event Announcements and Margining) shall apply.

5.6 Reports

Without prejudice to the notification requirements set out elsewhere in the CDS Clearing Documentation, LCH SA will communicate to the relevant Clearing Members, on the basis of information received from DTCC or, as applicable, from Clearing Members, amongst other things:

File No. SR-LCH SA-2022-007 Page 162 of 439

PUBLIC

- (a) the aggregate Floating Rate Payer Calculation Amounts of Restructuring Cleared Transactions to which they are a party and which are the subject of Credit Event Notices; and
- (b) the results of the exercise of any Movement Option in relation to Cleared Transactions to which they are a party,

at or around 7.00 p.m. on each day during each CEN Triggering Period and NEMO Triggering Period, as applicable, through the reports referred to in Section 5 of the Procedures.

For the avoidance of doubt, such communication shall not affect the validity or effectiveness of any Credit Event Notice or Notice to Exercise Movement Option which shall be subject to the terms of the relevant Restructuring Cleared Transaction.

6. PHYSICAL SETTLEMENT

6.1 General Terms relating to Physical Settlement

Where Physical Settlement is applicable as the Fallback Settlement Method pursuant to Section 6.1 (*Auction Settlement*) or 6.15 (*Movement Option*) of the 2014 ISDA Credit Derivatives Definitions, each Cleared Transaction will be subject to settlement in accordance with its terms and this Section 6 (*Physical Settlement*).

LCH SA has implemented a process, as set-out in this Section 6 (*Physical Settlement*), pursuant to which Clearing Members will manage the physical delivery process bilaterally in respect of any Cleared Transactions for which Physical Settlement is applicable.

Notwithstanding such process, LCH SA shall remain the legal counterparty for any such Cleared Transactions for which Physical Settlement is applicable and shall continue to be liable with respect to its obligations relating to such Physical Settlement, subject to its terms and this Section 6 (*Physical Settlement*).

6.2 Creation and Notification of Settlement Matched Pairs

LCH SA will create Settlement Matched Pairs in accordance with Section 8.1 (*Creation of Matched Pairs*) and notify Matched Buyer and Matched Seller comprised within each Settlement Matched Pair of:

- the identity of the other Clearing Member (together with the address, fax number, telephone number, email address and other applicable notice details of each such other Clearing Member); and
- (b) the associated Settlement Matched Pair Delivery Amount,

at or prior to the SMP Notification Deadline.

Notwithstanding the above, the Settlement Matched Pair for a Restructuring Cleared Transaction shall be the Restructuring Matched Pair previously created by LCH SA in

accordance with Section 5.1 (Creation and Notification of Restructuring Matched Pairs) in respect of such Restructuring Cleared Transaction.

6.3 **Physically Settled Cleared Transactions**

Following the actual or deemed creation of such Settlement Matched Pairs, the relevant Cleared Transactions in respect of which the Fallback Settlement Method applies will be construed as if they had been terminated and new Cleared Transactions of the same CDS Type will be deemed to have been entered into between each such Clearing Member and LCH SA (each such deemed Cleared Transaction being a "Physically Settled Cleared Transaction").

Each such Physically Settled Cleared Transaction shall have a Floating Rate Payer Calculation Amount corresponding to the Settlement Matched Pair Delivery Amount of a Settlement Matched Pair in which the relevant Clearing Member is comprised as a Matched Buyer or a Matched Seller, as applicable.

Matched Seller Calculation Agent 6.4

Notwithstanding any provision to the contrary in a Physically Settled Cleared Transaction (but subject as set out in Section 6.10 (Failure to pay Physical Settlement Amount), the "Calculation Agent" for the purposes of Article VIII (Terms relating to Physical Settlement) and Section 9.6 (Partial Cash Settlement Terms) of the 2014 ISDA Credit Derivatives Definitions shall be the relevant Matched Seller.

6.5 **Notices of Physical Settlement**

No Notices of Physical Settlement until Notification of Settlement Matched Pairs

Subject as set out at Section 7.8 (Failure to notify Matched Pairs) and notwithstanding anything to the contrary in the terms of any Cleared Transaction, Clearing Members may not deliver any Notices of Physical Settlement or Asset Package Notices in respect of any Affected Cleared Transactions until they have been notified of their Settlement Matched Pairs.

Notice of Physical Settlement and Asset Package Notice to contain Matched Buyer **Account Information**

Each Notice of Physical Settlement and Asset Package Notice delivered by Matched Buyer shall contain, in addition to the information required under the terms of the relevant Physically Settled Cleared Transaction, Matched Buyer's account information.

Details of Asset Package (c)

If Asset Package Delivery is applicable in respect of a Physically Settled Cleared Transaction, the Matched Buyer shall on the NOPS Effective Date, or as soon as reasonably practicable thereafter, deliver an Asset Package Notice to the Matched Seller containing a detailed description of the Asset Package that the Matched Buyer

163

intends to deliver to Matched Seller in lieu of the Prior Deliverable Obligation or Package Observable Bond that is specified in the Notice of Physical Settlement or NOPS Amendment Notice, as applicable.

(d) LCH SA not Liable

LCH SA shall have no liability with respect to any loss, cost, damage or expense suffered or incurred by a Matched Seller as a result of any error or inaccuracy in any Notice of Physical Settlement or any NOPS Amendment Notice or Asset Package Notice sent by a Matched Buyer and shall have no duty to verify any such notice or the contents thereof.

6.6 Delivery of Non-DVP Obligations

The following shall apply in relation to any Settlement Matched Pair where: (1) the relevant Matched Buyer specifies a Non-DVP Obligation in a Notice of Physical Settlement or a NOPS Amendment Notice, or (2) if Asset Package Delivery is applicable in respect of a Physically Settled Cleared Transaction:

- (a) The relevant Matched Buyer shall notify LCH SA that it is ready to Deliver the relevant Non-DVP Obligation (or if Asset Package Delivery is applicable, the Assets that are to be delivered) at or prior to 5.30 p.m. on the Transaction Business Day prior to the date on which Delivery is scheduled to occur. In such notice, the relevant Matched Buyer shall also specify the amount of any expenses payable to it under Section 11.2(c)(vi) of the 2014 ISDA Credit Derivatives Definitions.
- (b) LCH SA shall notify the relevant Matched Seller at or prior to 6.30 p.m. on the Transaction Business Day prior to the date on which Delivery is scheduled to occur that it is obligated to pay LCH SA the amount of the relevant Physical Settlement Amount corresponding to the Outstanding Amount of such Non-DVP Obligation or, if Asset Package Delivery is applicable in respect of such Settlement Matched Pair, the amount of such Physical Settlement Amount which corresponds to the Outstanding Amount of the Deliverable Obligation(s) in lieu of which the Asset Package is being Delivered less an amount equal to the Asset Package Cash Settlement Amount determined in respect of such Asset Package (the "Non-DVP Asset Package Settlement Amount") and in each case any amounts in respect of expenses notified by the relevant Matched Buyer before 9.00 a.m. on the following Transaction Business Day.
- (c) The relevant Matched Seller shall pay to LCH SA the amount(s) so requested on or prior to 9.00 a.m. on the Transaction Business Day immediately following the date on which the relevant Matched Buyer notified LCH SA of its readiness to Deliver provided that the request for payment has been made in a timely manner as set out above.
- (d) LCH SA shall notify the relevant Matched Buyer that it has received payment at or prior to 5.30 p.m. on the Transaction Business Day in which LCH SA receives payment,

provided that payment is received by LCH SA at or prior to 9.00 a.m. on such Transaction Business Day as set out above.

- (e) The relevant Matched Buyer shall Deliver the relevant non-DVP Obligations (or if Asset Package Delivery is applicable, the Assets that are to be delivered) to the relevant Matched Seller, provided that, if Asset Package Delivery is applicable the relevant Matched Buyer's obligation to deliver any Assets in the form of cash in the Settlement Currency forming part of the relevant Asset Package will be automatically satisfied and discharged.
- (f) The relevant Matched Seller shall notify LCH SA that Delivery has occurred by 5.30 p.m. on the Transaction Business Day on which Matched Seller receives Delivery of the relevant Non-DVP Obligations (or if Asset Package Delivery is applicable, the relevant Assets that are to be delivered pursuant to this Section 6.6 (*Delivery of Non-DVP Obligations*). Notices received after 3.30 p.m. will be deemed received on the next following Transaction Business Day, unless LCH SA agrees otherwise.
- (g) If and to the extent that LCH SA has received payment from the relevant Matched Seller of the Physical Settlement Amount or the Non-DVP Asset Package Settlement Amount, as applicable in full on a timely basis as set out above, LCH SA shall pay the Physical Settlement Amount or the Non-DVP Asset Package Settlement Amount, as applicable and any expenses due to the relevant Matched Buyer under Section 11.2(c)(vi) of the 2014 ISDA Credit Derivatives Definitions on or prior to 9.00 a.m. on the Transaction Business Day following the Transaction Business Day on which LCH SA receives the notice referred to in sub-paragraph (e) above from the relevant Matched Seller.
- (h) If and to the extent that Delivery of the relevant Non-DVP Obligations has not been effected by the relevant Matched Buyer in accordance with terms of the relevant Physically Settled Cleared Transaction as of the expiry of the third Transaction Business Day following delivery of the relevant Matched Buyer's notice referred to above, the relevant Matched Seller may request LCH SA to repay the Physical Settlement Amount or the Non-DVP Asset Package Settlement Amount, as applicable, not earlier than the first Transaction Business Day following the date on which such request is effectively delivered to LCH SA.

6.7 Alternative Delivery Procedure

A Matched Buyer and Matched Seller comprising a Settlement Matched Pair may elect to exercise their rights against and perform obligations to LCH SA in relation to the Settlement Matched Pair Delivery Amount (or any portion thereof) directly as between themselves. To exercise such option, the relevant Matched Buyer and Matched Seller must each notify LCH SA accordingly (in the form set out in Appendix V hereto) specifying the relevant Matched Contracts in respect of the related relevant Settlement Matched Pair Delivery Amounts and obtain the consent of LCH SA, which consent will not be unreasonably withheld or delayed by more than one Transaction Business Day following receipt of such notification by Matched Buyer and Matched Seller.

With effect from the time that LCH SA confirms its consent, the Settlement Matched Pair Delivery Amount will be reduced by the amount specified in the joint notice of the relevant Matched Buyer and Matched Seller, and the relevant Matched Buyer and Matched Seller shall each perform their obligations to each other and shall each acquire enforcement rights in respect of the other party's obligations to it pursuant to the Contracts (Rights of Third Parties) Act 1999 in respect of any such reduction as agreed between them.

6.8 Buy-in of Bonds – Matched Seller has entered into CCM Client Transaction

The following provisions shall solely be applicable to a Matched Seller that is a CCM in respect of the Matched Contracts of the Settlement Matched Pair if such Matched Seller notifies Matched Buyer and LCH SA that it has a CCM Client Transaction which corresponds to the Matched Seller Contract:

(a) Buy-in of Bonds – Settlement Matched Pair

Section 9.7 (*Buy-in of Bonds Not Delivered*) of the 2014 ISDA Credit Derivatives Definitions shall not apply in respect of the Matched Contracts of the Settlement Matched Pair

(b) Deemed Buy-in of Bonds resulting from CCM Client Transaction of Matched Seller that is a CCM

Provided that Physical Settlement has not already occurred in respect of the Matched Contracts of a Settlement Matched Pair, if:

- (i) the Matched Seller that is a CCM receives a Buy-in Notice from its CCM Client in respect of the CCM Client Transaction between such Matched Seller and its CCM Client which is validly delivered pursuant to the terms of such CCM Client Transaction (including the Mandatory Provisions), then such Matched Seller shall notify Matched Buyer and LCH.SA of its receipt of such Buy-in Notice and of the content thereof and Matched Buyer's right to Deliver the Relevant Bonds specified in the relevant Buy-in Notice shall be suspended until the fourth Business Day (inclusive) following the Buy-In Date specified in the relevant Buy-In Notice (the "Deemed Buy-in Period"); and
- (ii) such Matched Seller notifies Matched Buyer and LCH SA that it has been notified by its CCM Client pursuant to Section 9.7 (Buy-in of Bonds Not Delivered) of the 2014 ISDA Credit Derivatives Definitions that a Buy-in Price has been determined in respect of Relevant Bonds for the purposes of the CCM Client Transaction between such Matched Seller and its CCM Client, then on the third Business Day following receipt by Matched Seller of such notice from its CCM Client (which such date Matched Seller shall specify) (the "Buy-in Effective Date"):
 - (A) Matched Buyer will be deemed to have Delivered to such Matched Seller an Outstanding Principal Balance of the Deliverable Obligations specified in the Notice of Physical Settlement or any NOPS Amendment

Notice, as applicable, for which a Buy-in Price was determined by the CCM Client; and

(B) the Physical Settlement Amount to be paid by such Matched Seller to Matched Buyer in respect of the Matched Contracts of the Settlement Matched Pair shall be reduced (but not below zero) by an amount equal to the amount by which the Physical Settlement Amount to be paid to such Matched Seller by its CCM Client in respect of the CCM Client Transaction has been reduced pursuant to Section 9.7 (Buy-in of Bonds Not Delivered) of the 2014 ISDA Credit Derivatives Definitions.

Such Matched Seller shall notify Matched Buyer and LCH SA of the Outstanding Principal Balance of the Deliverable Obligations and the Physical Settlement Amount reduction determined in respect of its CCM Client Transaction for the purposes of (A) and (B) above.

(c) Buy-in of Bonds – Cap on Settlement

If "60 Business Day Cap on Settlement" is applicable to the Matched Contracts of the Settlement Matched Pair, then the delivery of a valid notice of Buy-in Price to Matched Seller by the CCM Client under the CCM Client Transaction between them that results in a Buy-in Effective Date pursuant to sub-paragraph (b)(ii) above shall be deemed to be the 'valid notice of Buy-in Price' referred to in Section 9.10 (Cap on Settlement) of the 2014 ISDA Credit Derivatives Definitions for the purposes of the Matched Contracts of the Settlement Matched Pair.

6.9 Alternative Procedures relating to Loans – Buyer Right to Deliver

The following provision shall solely be applicable in respect of the Matched Contracts of the Settlement Matched Pair if a Matched Buyer that is a CCM notifies Matched Seller and LCH SA that it has a CCM Client Transaction which corresponds to the Matched Buyer Contract:

If a Matched Buyer that is a CCM notifies Matched Seller and LCH SA that it has received a Bond or Loan from its CCM Client in respect of its CCM Client Transaction pursuant to Section 9.8(i) (Alternative Procedures Relating to Loans Not Delivered) of the 2014 ISDA Credit Derivatives Definitions, then such Bonds or Loans shall be deemed specified in a NOPS Amendment Notice for the purposes of the Matched Contracts of the Settlement Matched Pair and such NOPS Amendment Notice will be effective notwithstanding the fact that it is deemed specified after the Physical Settlement Date.

6.10 Failure to pay Physical Settlement Amount

If, in relation to any Settlement Matched Pair, a Matched Seller fails to pay all or part of the Physical Settlement Amount (the **Failed Amount**) to the related Matched Buyer, as designee for LCH SA, when due:

- (a) the relevant Matched Buyer may and the relevant Matched Seller shall, as soon as practicable, give notice in writing to LCH SA, giving all material details of the Settlement Matched Pair involved, the relevant failure to pay and the Failed Amount;
- (b) such failure to pay shall not constitute or be deemed to constitute a breach of contract, event of default or failure to pay by LCH SA under the CDS Clearing Documentation or otherwise (but such failure to pay may, for the avoidance of doubt, constitute a breach of the CDS Clearing Documentation and/or a Payment Failure for the purposes of Article 4.3.1.1 of the CDS Clearing Rule Book by or with respect to the relevant Clearing Member);
- (c) if the relevant Matched Buyer elects to notify LCH SA of such failure to pay as contemplated above, such Matched Buyer may give any such notice as soon as reasonably practicable after the occurrence of such failure to pay by the relevant Matched Seller;
- (d) upon notice being given to LCH SA by the relevant Matched Buyer, "Cash Settlement" between the relevant Matched Buyer and LCH SA and the relevant Matched Seller and LCH SA pursuant to the Partial Cash Settlement Terms (set out in Section 9.6 (Partial Cash Settlement Terms) of the 2014 ISDA Credit Derivatives Definitions, as amended by this CDS Clearing Supplement) shall be deemed to apply to the relevant Physically Settled Cleared Transactions of the Settlement Matched Pair with respect to the Deliverable Obligations corresponding to the Failed Amount as though:
 - (i) the Deliverable Obligations not Delivered were Undeliverable Obligations;
 - the Latest Permissible Physical Settlement Date were the date on which the relevant Matched Buyer gave the relevant notice to LCH SA;
 - (iii) Indicative Quotations were not applicable;
 - the relevant Matched Buyer were the Calculation Agent in respect of the Physically Settled Cleared Transaction of the Settlement Matched Pair to which it is a direct party; and
 - (v) the Cash Settlement Amount determined in respect of the Physically Settled Cleared Transaction between Matched Buyer and LCH SA is also the Cash Settlement Amount in respect of the corresponding Physically Settled Cleared Transaction of the Settlement Matched Pair; and
- (e) LCH SA and the relevant Matched Buyer will settle the relevant Physically Settled Cleared Transaction accordingly.

168

6.11 Partial Cash Settlement due to Impossibility or Illegality

Section 9.1 (Partial Cash Settlement Due to Impossibility or Illegality) of the 2014 ISDA Credit Derivatives Definitions shall apply to a Physically Settled Transaction in the circumstances contemplated therein, provided that Matched Buyer or Matched Seller notifies the other Clearing Member comprised in the relevant Settlement Matched Pair and LCH SA accordingly.

In such case, the related Physically Settled Cleared Transaction entered into between LCH SA and the other Clearing Member comprised in the relevant Settlement Matched Pair shall likewise be subject to "Cash Settlement" pursuant to the Partial Cash Settlement Terms and Section 6.14 (Consequences of Cash Settlement).

6.12 Fallback to Cash Settlement in respect of Non-Deliverable Obligations

- (a) If Matched Buyer is not permitted to Deliver one or more Deliverable Obligations (such Deliverable Obligations, the Non-Deliverable Obligations) specified in the relevant Notice of Physical Settlement or NOPS Amendment Notice to Matched Seller as designee for LCH SA in the relevant Matched Pair because:
 - the amount of such Deliverable Obligation to be Delivered is less than the relevant minimum denomination of such Deliverable Obligation; or
 - (ii) such Matched Seller is not a permitted transferee under such Deliverable Obligation (and, in the case of this sub-section (ii), such circumstance would not constitute an illegality or impossibility outside the control of a relevant party for the purposes of Section 9.1 (Partial Cash Settlement Due to Impossibility or Illegality) of the 2014 ISDA Credit Derivatives Definitions),

then it shall notify the relevant Matched Seller and LCH SA (in the form set out in Appendix VI hereto) accordingly describing in reasonable detail the relevant circumstances.

With effect from such notification, such occurrence shall be treated, in relation to each relevant Physically Settled Cleared Transaction, as an illegality or impossibility outside the control of a relevant party for the purpose of Section 9.1 (*Partial Cash Settlement Due to Impossibility or Illegality*) of the 2014 ISDA Credit Derivatives Definitions.

(b) Upon notice being given to LCH SA by Matched Buyer under sub-paragraph (a) of this Section 6.12 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations), "Cash Settlement" pursuant to the Partial Cash Settlement Terms shall be deemed to apply to the Matched Contracts in respect of the relevant Settlement Matched Pair with respect to the Non-Deliverable Obligations as though the Non-Deliverable Obligations were Undeliverable Obligations and the provisions set out in Section 6.14 (Consequences of Cash Settlement) of this CDS Clearing Supplement shall apply.

6.13 Cash Settlement Resulting from CCM Client Transaction of a Matched Buyer that is a CCM

The following provision shall solely be applicable in respect of the Matched Contracts of the Settlement Matched Pair if a Matched Buyer that is a CCM notifies Matched Seller and LCH SA that it has a CCM Client Transaction which corresponds to the Matched Buyer Contract:

If a Matched Buyer that is a CCM notifies Matched Seller and LCH SA that the corresponding CCM Client Transaction between such Matched Buyer and its CCM Client is to be settled (in whole or in part) by Cash Settlement pursuant to Section 9.1 (*Partial Cash Settlement Due to Impossibility or Illegality*) of the 2014 ISDA Credit Derivatives Definitions or Mandatory Provision 7.2 (*Fallback to Cash Settlement in respect of Non-Deliverable Obligations*), then:

- (a) "Cash Settlement" shall be deemed to apply (in whole or in part, as applicable) to the Matched Contracts of the Settlement Matched Pair pursuant to the Partial Cash Settlement Terms and Section 6.14 (Consequences of Cash Settlement);
- (b) the 'Undeliverable Obligations' for the purposes of the Partial Cash Settlement Terms shall be the Undeliverable Obligations or Non-Deliverable Obligations in respect of the corresponding CCM Client Transaction; and
- (c) the Cash Settlement Amount and the Cash Settlement Date in respect of the Matched Contracts of the Settlement Matched Pair shall be the same as the Cash Settlement Amount and the Cash Settlement Date determined in respect of the corresponding CCM Client Transaction.

6.14 Consequences of Cash Settlement

If the circumstances set out in either Section 6.11 (Partial Cash Settlement Due to Impossibility or Illegality) or paragraph (a) of Section 6.12 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) or Section 6.13 (Cash Settlement Resulting from CCM Client Transaction of a Matched Buyer that is a CCM) apply, then:

- (a) the Latest Permissible Physical Settlement Date in respect of the relevant Physically Settled Cleared Transaction will be deemed to be the first date on which the relevant Matched Buyer or Matched Seller effectively gave the relevant notice to both LCH SA and the other Clearing Member as referred to in Section 6.11 (Partial Cash Settlement Due to Impossibility or Illegality) or paragraph (a) of Section 6.12 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) or Section 6.13 (Cash Settlement Resulting from CCM Client Transaction of a Matched Buyer that is a CCM) above, as applicable, (and for these purposes, Section 9.5 (Latest Permissible Physical Settlement Date) of the 2014 ISDA Credit Derivatives Definitions shall not apply);
- (b) the relevant Matched Buyer will be deemed to be the Calculation Agent;
- (c) LCH SA and the relevant Matched Buyer will settle the applicable Matched Buyer Contract, and LCH SA and the relevant Matched Seller will settle the applicable Matched Seller Contract, accordingly; and

(d) where sub-paragraph (a)(ii) of Section 6.12 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) applies, Indicative Quotations shall not be applicable.

6.15 Amendments to Section 9.6(k) of 2014 ISDA Credit Derivatives Definitions

Solely for the purposes of Section 6.11 (Partial Cash Settlement due to Impossibility or Illegality), Section 6.12 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations), Section 6.13 (Cash Settlement Resulting from CCM Client Transaction of a Matched Buyer that is a CCM) and Section 6.14 (Consequences of Cash Settlement) of this CDS Clearing Supplement, Section 9.6(k) of the 2014 ISDA Credit Derivatives Definitions is amended by inserting the following at the beginning thereof:

- '(A) For the purposes hereof, in addition to the requirements of Section 7.10, each firm Quotation shall:
 - (1) be for a transaction with Matched Buyer (or its designee) (the Relevant Buyer) pursuant to which the Relevant Buyer agrees to Deliver the Deliverable Obligations to the Dealer submitting the firm quotation (the Quoting Dealer), which transaction shall be governed by documentation that is consistent with market practice applicable to the sale and purchase of Deliverable Obligations on the Valuation Date (which may be determined by the relevant Credit Derivatives Determinations Committee), including, without limitation, a representation that the Quoting Dealer has completed all "know your customer" or similar requirements under all applicable laws, regulations and internal compliance procedures relating to a transaction with the Relevant Buyer and in respect of the Reference Entity;
 - (2) be capable of acceptance by the Relevant Buyer (for such purposes, each firm Quotation must, inter alia, be obtained from a Dealer with whom the Relevant Buyer, in its sole and absolute discretion, determines that it is able, in accordance with all its internal compliance and policy requirements, to transact and to Deliver the Deliverable Obligations) and be open for acceptance by the relevant party for at least 30 minutes; and
 - (3) be obtained on the basis that if the Relevant Buyer agrees to Deliver the Deliverable Obligations to such Quoting Dealer on the terms set forth herein, such Quoting Dealer agrees to pay the settlement amount (calculated and payable for this purpose in accordance with the relevant market standard documentation and based on the price so quoted) that would be payable to the Relevant Buyer for such Deliverable Obligations.
- (B) Otherwise,".

6.16 Subsequent Determination by the Credit Derivatives Determinations Committee with respect to an Auction

If the Fallback Settlement Method applies in respect of a Cleared Transaction and a subsequent resolution of the Credit Derivatives Determinations Committee determines that

File No. SR-LCH SA-2022-007 Page 172 of 439

PUBLIC

Transaction Auction Settlement Terms will be published, LCH SA shall have no obligation to create Settlement Matched Pairs in accordance with Section 6.2 (*Creation and Notification of Settlement Matched Pairs*).

Unless settlement has occurred with respect to any Settlement Matched Pairs prior to such determination, any such Settlement Matched Pairs so created shall be deemed not to have been created and any notices delivered in connection with such Settlement Matched Pairs shall be deemed to be ineffective.

LCH SA shall effect any relevant registrations in the TIW in order to reflect the application of Auction Settlement as the Settlement Method in respect of the relevant Cleared Transactions.

6.17 Physical Settlement Costs

If the Fallback Settlement Method applies in respect of a Cleared Transaction and any amounts are payable in relation to any costs or expenses of Physical Settlement (including any costs or expenses relating to the delivery of any Assets forming part of an Asset Package), then:

- (a) where such amount would otherwise be payable by LCH SA to Matched Buyer or Matched Seller (the Recipient), such amount shall be payable to the Recipient by the other party to the Matched Contract who is not the Recipient, as designee to pay such amount on behalf of LCH SA; and
- (b) where such amount would otherwise be payable to LCH SA by Matched Buyer or Matched Seller (the **Payer**), such amount shall be payable by the Payer to the other party to the Matched Contract who is not the Payer, as designee to receive such amount on behalf of LCH SA.

Any Matched Seller or Matched Buyer who is required to make any payment as designee on behalf of LCH SA pursuant to this Section 6.17 shall not be entitled to any reimbursement from LCH SA in respect of such amount.

Any Matched Seller or Matched Buyer who receives any payment pursuant to this Section 6.17 shall not be obliged to remit to LCH SA any such amount so received (without prejudice to any rights of LCH SA where there is a failure to Deliver).

LCH SA shall not be liable to a Matched Buyer or a Matched Seller for any of the costs and expenses of Physical Settlement of any Cleared Transaction.

6.18 Representations and Agreements relating to Physical Settlement

(a) Claims by Matched Seller against LCH SA in respect of Obligations Delivered by Physical Settlement

File No. SR-LCH SA-2022-007 Page 173 of 439

PUBLIC

If a Matched Seller pursues a claim against LCH SA under Sections 11.2(a), (b), (c)(i) or (c)(iv) of the 2014 ISDA Credit Derivatives Definitions in respect of any obligations Delivered by way of Physical Settlement of any Matched Seller Contract, then:

- (i) notwithstanding any provision of the 2014 ISDA Credit Derivatives Definitions to the contrary, LCH SA shall only be liable to make payments to that Matched Seller in respect of that claim to the extent of amounts recovered, including, without limitation, any amounts recovered by way of set-off or netting, by LCH SA from or on behalf of the related Matched Buyer in respect of any corresponding claims under or in connection with the Matched Buyer Contract (including, without limitation, following the declaration of an Event of Default in respect of such Matched Buyer) and after deducting any costs and expenses reasonably incurred by LCH SA in pursuing such corresponding claims for recovery under or in connection with the Matched Buyer Contract;
- (ii) LCH SA will use reasonable efforts to pursue any claim it may have against the related Matched Buyer but, notwithstanding Section 6.18(a)(i) above, LCH SA will, in the pursuit of such claims, act as though its liability to Matched Seller under the Matched Seller Contract was not limited or restricted in any way; and
- (iii) references to indemnity provisions set out in Section 11.2(a), 11.2(b) and 11.2(c)(iv) of the 2014 ISDA Credit Derivatives Definitions shall, in connection with a Matched Buyer Contract, be interpreted without regard to the limitations to Matched Seller's recourse to LCH SA under the corresponding Matched Seller Contract imposed by the provisions of this Section 6.18(a) and such limitations shall not restrict a Matched Seller from making a claim or raising a Dispute.

(b) Right of Matched Seller to enforce against Matched Buyer

Without prejudice to paragraph (a) above, a Matched Seller shall be entitled to enforce Sections 11.2(a), (b), (c)(i) and (c)(iv) of the 2014 ISDA Credit Derivatives Definitions against the related Matched Buyer in respect of any obligations Delivered by way of Physical Settlement of any Physically Settled Cleared Transaction under the Contracts (Rights of Third Parties) Act 1999 as though Matched Seller were party to the relevant Matched Buyer Contract in place of LCH SA.

 Satisfaction of Claim by Matched Buyer discharges Liabilities owed to and by LCH← SA

For the avoidance of doubt, if a Matched Buyer satisfies a claim made by a Matched Seller directly against the relevant Matched Buyer under Sections 11.2(a), (b), (c)(i) or (c)(iv) of the 2014 ISDA Credit Derivatives Definitions pursuant to the rights granted to such Matched Seller under paragraph (b) above, such satisfaction shall also constitute complete satisfaction and discharge of the corresponding liability of such Matched Buyer to LCH SA and the corresponding liability of LCH SA to such Matched Seller in respect of such claim.

Formatted: Indent: Left: 1.5 cm, Keep with next

(d) Effect of Illegality or Tax or other Expense resulting from Designation through Creation of Matched Pairs

The last sentence of Section 11.2(c)(iv) of the 2014 ISDA Credit Derivatives Definitions shall not operate so as to prevent LCH SA from creating any Matched Pair in accordance with this CDS Clearing Supplement, and LCH SA shall have no obligation to grant any indemnity with respect to any Tax, loss or cost to any Matched Buyer or Matched Seller thereunder.

If the circumstances contemplated at Section 11.2(c)(iv)(A) or (B) of the 2014 ISDA Credit Derivatives Definitions apply in respect of any Physically Settled Cleared Transaction and any related Settlement Matched Pair (in the case of (B), as notified by the Clearing Member which is the non-designating party for such purpose to the other Clearing Member comprised in the relevant Settlement Matched Pair prior to the first Delivery Date), then an impossibility shall be deemed to have occurred with respect to Physical Settlement of the relevant Physically Settled Cleared Transactions, and Section 9.1 (Partial Cash Settlement Due to Impossibility or Illegality) of the 2014 ISDA Credit Derivatives Definitions (as amended by this CDS Clearing Supplement) shall apply.

6.19 Miscellaneous Provisions relating to Physical Settlement

(a) Margin

For the avoidance of doubt, LCH SA will continue to call all Margin and such Margin will remain due in relation to any Cleared Transaction to which Physical Settlement applies until:

- LCH SA has received a Notice of Physical Settlement from each Clearing Member;
- LCH SA has received a No Physical Settlement Confirmation from each Clearing Member; or
- (iii) in each case, any related dispute is referred to and finally resolved by arbitration or by litigation, as applicable, in accordance with the CDS Dispute Resolution Protocol, subject to the provisions of Sections 8 and 9 of the Procedures.

(b) Notification of Completion of Physical Settlement

Upon completion of Physical Settlement by the relevant Matched Pair of a Physically Settled Cleared Transaction, Matched Buyer and Matched Seller shall inform LCH SA as soon as reasonably practicable and in any event before the date falling two Transaction Business Days following such completion (the **Physical Settlement Confirmation**) by notice (a **Physical Settlement Confirmation**) in the form set out at Appendix VII hereto.

(c) Notification that Physical Settlement will not occur

If no Notice of Physical Settlement has been delivered within the relevant time period permitted for such delivery in accordance with the terms of the relevant Physically Settled Cleared Transaction and, accordingly, Physical Settlement will not, under the terms of the relevant Physically Settled Cleared Transaction, occur, Matched Buyer and Matched Seller shall inform LCH SA as soon as reasonably practicable thereafter and in any event before the date falling two Transaction Business Days following the relevant date after which delivery of a Notice of Physical Settlement is no longer permitted in accordance with the terms of the relevant Physically Settled Cleared Transaction (the No Physical Settlement Confirmation Deadline) by notice (a No Physical Settlement Confirmation) in the form set out at Appendix VIII hereto.

(d) Dispute regarding Physical Settlement

If LCH SA receives a Physical Settlement Confirmation or No Physical Settlement Confirmation from one relevant Matched Buyer or Matched Seller only at or prior to the relevant Physical Settlement Confirmation Deadline or No Physical Settlement Confirmation Deadline, as the case may be, there shall be deemed to be a Dispute with respect to the Physically Settled Cleared Transactions between LCH SA and each relevant Clearing Member.

(e) Asset Package Delivery

If Asset Package Delivery is applicable in respect of a Physically Settled Cleared Transaction and an Asset to be delivered is a Non-Transferable Instrument or Non-Financial Instrument, then the Asset shall be deemed to be an amount of cash equal to the value determined by the CDSClear Product Committee (which, notwithstanding anything to the contrary, shall be the "Calculation Agent" for the purposes of Section 8.15 (Asset Market Value) of the 2014 ISDA Credit Derivatives Definitions).

7. DELIVERY OF NOTICES AND FALLBACKS

7.1 General Rules relating to Notices

(a) Methods of Delivery and deemed Time of Delivery

Subject to Section 7.2 (*Oral Notices*) and without prejudice to Section 1.38 (*Requirements Regarding* Notices) of the 2014 ISDA Credit Derivatives Definitions and the remaining provisions of the 2014 ISDA Credit Derivatives Definitions (including, for the avoidance of doubt, in relation to notices permitted to be given orally), any notice or other communication in respect of any Cleared Transaction may be given in any manner described below and will be deemed effective as indicated:

- (i) if delivered in person or by courier, on the date and at the time it is delivered;
- (ii) if sent by facsimile transmission, on the date it is received by a responsible employee of the recipient in legible form (it being agreed that the burden of

proving receipt will be on the sender and will not be met by a transmission report generated by the sender's facsimile machine);

- (iii) if sent by certified or registered mail (airmail, if overseas) or the equivalent (return receipt requested), on the date it is delivered or its delivery is attempted; or
- (iv) if sent by electronic messaging system (including e-mail or any other electronic access solution established by LCH SA for such purpose), on the date it is received.

unless the date of that delivery (or attempted delivery) or that receipt, as applicable, is not a Transaction Business Day or that communication is delivered (or attempted) or received, as applicable, pursuant to the above after 6.00 p.m. in the location of the recipient on a Transaction Business Day, in which case that communication will be deemed given and effective on the first following day that is a Transaction Business Day.

(b) Notices from or to LCH SA

Any such notice or communication given by LCH SA to a Clearing Member or vice versa shall (except where delivered via a relevant DTCC Notice Facility) be given to the address or number previously specified in or previously notified for the relevant purpose in accordance with the CDS Admissions Agreement or the Procedures.

(c) Manual Notices between Clearing Members

Notices given by a Clearing Member to another Clearing Member comprised in a relevant Matched Pair and which are not delivered via a relevant DTCC Notice Facility shall be given to the address or number notified by LCH SA to the deliveror pursuant to Sections 5.1 (*Creation and Notification of Restructuring Matched Pairs*) or 6.2 (*Creation and Notification of Settlement Matched Pairs*).

Such notices shall only be deemed to be delivered effectively by LCH SA through the relevant Clearing Member as its designee as against the recipient where the address or number so notified by LCH SA corresponds in all material respects to the address or number, as applicable, specified by such recipient in or pursuant to the CDS Admissions Agreement.

(d) No Obligation on LCH SA to verify Signatories

LCH SA shall have no obligation to verify the authority of any signatory of any notice delivered by any Clearing Member directly pursuant to this Section 7 (*Delivery of Notices and Fallbacks*).

7.2 Oral Notices

Notwithstanding the provisions of the 2014 ISDA Credit Derivatives Definitions, where, by way of exception as contemplated in this Section 7, Credit Event Notices and Notices to Exercise Movement Option are to be delivered directly to LCH SA (and not via a relevant DTCC Notice Facility), such notices may not be delivered by telephone.

7.3 Credit Event Notices and NEMOs

(a) Credit Event Notices and NEMOs to be given via DTCC

Credit Event Notices and Notices to Exercise Movement Option shall be delivered by way of the relevant DTCC Notice Facility, save if and as expressly stated to the contrary in this Section 7 or expressly agreed by LCH SA. The deemed time of delivery of any such notices shall be as set out in the DTCC Rules from time to time.

(b) Credit Event Notices and NEMOs delivered in respect of CCM Client Transaction

If a Matched Buyer or a Matched Seller that is a CCM receives a valid Credit Event Notice or Notice to Exercise Movement Option from its CCM Client in respect of a CCM Client Transaction by way of the relevant DTCC Notice Facility, such notice shall be deemed also to be a Credit Event Notice or Notice to Exercise Movement Option (as applicable) for the purposes of the corresponding Matched Contracts of the Restructuring Matched Pair.

7.4 Notification of DTCC Failure and Resolution

(a) LCH SA to notify Clearing Members of DTCC Failure Event

If DTCC notifies LCH SA or LCH SA otherwise becomes aware that the relevant DTCC Notice Facility is or will be unavailable to process all or substantially all Credit Event Notices or Notices to Exercise Movement Option, as applicable, with respect to an M(M)R Restructuring in a timely manner (a **DTCC Failure Event**), then LCH SA will, as soon as reasonably practicable and in any event within one hour of such notification or of LCH SA becoming aware of such non-availability, as applicable, notify all relevant Clearing Members of such occurrence, including the time at which such failure occurred (or, if LCH SA is not notified of such time by or on behalf of DTCC, the time at which LCH SA received the relevant notification from or on behalf of DTCC or becomes aware of the relevant non-availability) (the **DTCC Failure Event Time**). LCH SA shall also publish such information on its Website as soon as reasonably practicable after becoming aware of a DTCC Failure Event.

(b) LCH SA to notify Clearing Members of Resolution of DTCC Failure Event

177

If, subsequent to a DTCC Failure Event, DTCC (or a third party designated under the DTCC Rules from time to time) notifies LCH SA that a DTCC Failure Event previously notified to Clearing Members is no longer in effect, LCH SA will as soon as reasonably practicable notify all relevant Clearing Members accordingly, including the time (the

Formatted: Indent: Left: 1.5 cm, Keep with next

DTCC Resolution Time) (which must be no earlier than 30 minutes following the time of such notification) at which such DTCC Failure Event is deemed to have been resolved and following which the relevant DTCC Notice Facility is operative for the purposes of delivery of relevant notices relating to Restructuring Cleared Transactions (including transactions in respect of which notices may have been delivered directly pursuant to Section 7.5 (*Consequences of DTCC failure*). LCH SA shall also publish such information on its Website as soon as reasonably practicable after notifying Clearing Members that the DTCC Failure Event is no longer in effect and of the DTCC Resolution Time.

(c) Notices given prior to DTCC Resolution Time to be confirmed to DTCC

LCH SA and, to the extent so requested by LCH SA, each Clearing Member shall, as soon as reasonably practicable and to the extent permitted by DTCC, provide or confirm to DTCC details of any relevant notices (in the case of a Clearing Member, being any relevant notices delivered directly by such Clearing Member) given in respect of Cleared Transactions prior to the DTCC Resolution Time, so as to permit delivery of subsequent notices in respect of such Cleared Transactions via the relevant DTCC Notice Facility.

7.5 Consequences of DTCC Failure

From (and including) the DTCC Failure Event Time to (but excluding) the DTCC Resolution Time:

- Section 7.3(a) (Credit Event Notices and NEMOs to be given via DTCC) shall not apply and accordingly Credit Event Notices and Notices to Exercise Movement Option shall be delivered directly (and not via the relevant DTCC Notice Facility);
- (b) any notice delivered via the relevant DTCC Notice Facility prior to the DTCC Failure Event Time will be valid and will not be affected by such DTCC Failure Event; and
- (c) any notice delivered or purported to be delivered via the relevant DTCC Notice Facility at or following the DTCC Failure Event Time but prior to the DTCC Resolution Time will not be valid and effective.

Section 7.3(a) (*Credit Event Notices and NEMOs to be given via DTCC*) shall apply with effect from the DTCC Resolution Time and, accordingly, any notice thereafter delivered or purported to be delivered directly (and not via the relevant DTCC Notice Facility) will not be valid and effective.

7.6 Clearing Member Communications Failure Event

(a) Right to deliver Notices manually following Clearing Member Communications
Failure Event

If a Clearing Member is affected by a significant communications or information technology failure resulting in it being impossible or impractical for such Clearing

Member to deliver any Credit Event Notice in relation to an M(M)R Restructuring or any Notice to Exercise Movement Option via a relevant DTCC Notice Facility (a **Clearing Member Communications Failure Event**) it may, notwithstanding Section 7.3(a) (*Credit Event Notices and NEMOs to be given via DTCC*) deliver Credit Event Notices and Notices to Exercise Movement Option directly (and not via the relevant DTCC Notice Facility).

(b) Clearing Member to notify LCH SA of Occurrence of Clearing Member Communications Failure Event

Following the occurrence of a Clearing Member Communications Failure Event, the affected Clearing Member shall, within one hour of delivering any Credit Event Notice or Notice to Exercise Movement Option directly, deliver to LCH SA a notice (in the form set out at Appendix IX hereto) signed by a senior officer (such as a managing director or equivalent) of such Clearing Member certifying that it is affected by a Clearing Member Communications Failure Event (or, if such Clearing Member is unable to deliver such notice in writing, orally by telephone).

LCH SA shall notify all Clearing Members accordingly as soon as reasonably practicable and in any event within one hour of receipt of any such notification.

(c) Notices to Clearing Member affected by Clearing Member Communications Failure Event

For the avoidance of doubt, Section 7.3(a) (*Credit Event Notices and NEMOs to be given via DTCC*) shall continue to apply in respect of notices given to the affected Clearing Member by Clearing Members comprised in any Matched Pair in respect of which the affected Clearing Member is a Matched Buyer or Matched Seller.

(d) Notification of Resolution of Clearing Member Communications Failure Event

As soon as reasonably practicable upon a Clearing Member ceasing to be subject to a Clearing Member Communications Failure Event, it shall notify LCH SA accordingly (in the form set out at Appendix X hereto) and thereupon Section 7.3(a) (*Credit Event Notices and NEMOs to be given via DTCC*) shall apply and, accordingly, any Credit Event Notice or Notice to Exercise Movement Option thereafter delivered or purported to be delivered directly (and not via the relevant DTCC Notice Facility) will not be valid and effective.

(e) Clearing Member Duty to Mitigate

A Clearing Member which is subject to a Clearing Member Communications Failure Event shall use reasonable endeavours to mitigate the operational impact on other Clearing Members and LCH SA of any Clearing Member Communications Failure Event, to cure such Clearing Member Communications Failure Event as soon as possible and to ensure that the circumstances giving rise to the relevant Clearing Member Communications Failure Event do not recur.

(f) Breach by Clearing Member does not Invalidate Valid Notices

Without prejudice to any other rights or remedy of LCH SA, any breach by a Clearing Member of the provisions of this Section 7.6 shall not cause any Credit Event Notice or Notice to Exercise Movement Option delivered otherwise than in accordance with the relevant Restructuring Cleared Transaction, which would otherwise be valid and effective, to be invalid or ineffective.

7.7 Clearing Member Acknowledgements

(a) Duty to deliver Clearing Member Acknowledgements

Each Clearing Member shall notify LCH SA or deliver a copy to LCH SA of any notice delivered or received by such Clearing Member to or from another Clearing Member comprised in a Matched Pair, including, without limitation, any Credit Event Notice or Notice to Exercise Movement Option which was delivered or received directly (and not via a DTCC Notice Facility) pursuant to Sections 7.5 (Consequences of DTCC Failure) or 7.6 (Clearing Member Communications Failure Event), and which such Clearing Member asserts or acknowledges was effective for the purposes of this CDS Clearing Supplement and the relevant Cleared Transactions (such notification, or delivery of such copy, in respect of any relevant notice, a Clearing Member Acknowledgement) by no later than 6.00 p.m. on:

- in the case of a Notice of Physical Settlement or NOPS Amendment Notice or Asset Package Notice, the date falling two Transaction Business Days following the date on which such notice was sent; or
- (ii) in the case of any other notice, on the last date on which such notice could validly be sent,

(in each case, the Notice Acknowledgement Deadline).

(b) Clearing Member Acknowledgement received from both Clearing Members

Where LCH SA receives a Clearing Member Acknowledgement in respect of any notice from both relevant Clearing Members comprised in a Matched Pair at or prior to the Notice Acknowledgement Deadline, LCH SA shall perform its obligations in respect of the relevant Cleared Transactions in accordance with and subject to the remaining provisions of the CDS Clearing Documentation.

(c) Clearing Member Acknowledgement received from one Clearing Member

Where LCH SA receives a Clearing Member Acknowledgement in respect of any notice from one relevant Clearing Member only at or prior to the Notice Acknowledgement Deadline, the provisions of Section 7.11 (*Disputes as to Notices*) shall apply and LCH SA and each relevant Clearing Member shall perform their obligations in respect of the relevant Cleared Transactions in accordance with and subject to the remaining provisions of the CDS Clearing Documentation and the terms of any final resolution of

the relevant dispute, as agreed between the relevant parties or as determined by arbitration or by litigation, as applicable, in accordance with the CDS Dispute Resolution Protocol, subject to the provisions of Sections 8 and 9 of the Procedures.

In such case, LCH SA shall notify the Clearing Member from which it has not received a Clearing Member Acknowledgement of the asserted delivery or, as applicable, receipt of the relevant notice (in the case of a Credit Event Notice or Notice to Exercise Movement Option, through the reports referred to at Section 5.6 (*Reports*)).

(d) No Clearing Member Acknowledgement received

Where LCH SA does not receive Clearing Member Acknowledgement or confirmation of valid delivery in respect of any notice from either Clearing Member in the relevant Matched Pair on or prior to the relevant Notice Acknowledgement Deadline, the rights and obligations of LCH SA as against each relevant Clearing Member, and vice versa, shall be construed as if no such notice had been given.

(e) Consequences of no Clearing Member Acknowledgement being received

Where sub-section (d) above is applicable, the following provisions shall apply:

 Notices other than Notices of Physical Settlement, NOPS Amendment Notices and Asset Package Notices

Save in the case of a Notice of Physical Settlement, a NOPS Amendment Notice or an Asset Package Notice, an amount shall be payable between the Clearing Members equal to the difference between the value of the Matched Buyer Contract had Clearing Member Acknowledgement been given to LCH SA on a timely basis and the value of such contract in the absence of such acknowledgement; such difference in value shall be determined as of the earlier of the day on which notice is given by any relevant Clearing Member that such amount is due and payable and, in the case of a Credit Event Notice or Notice to Exercise Movement Option, the eighth Transaction Business Day following the Auction Settlement Date, no Auction Announcement Date or Auction Cancellation Date, as applicable or otherwise the eighth Transaction Business Day following the last day on which such notice would validly have been delivered, by reference to the relevant Auction Settlement Amount or end of day contributed prices, in each case if available.

(ii) Notices of Physical Settlement, NOPS Amendment Notices and Asset Package Notices

Where the relevant notice is a Notice of Physical Settlement, a NOPS Amendment Notice or an Asset Package Notice, the relevant Clearing Members shall acquire rights as against the other as though party to a bilateral credit default swap transaction on the terms of the related Matched Buyer Contract (including, without limitation, as to the occurrence of an Event Determination Date) and the Notice of Physical Settlement, NOPS Amendment Notice or Asset

Package Notice, as applicable shall be deemed to have been given in respect of such transaction. Any resulting payment shall be due and payable two Transaction Business Days following the giving of a notice that such amount is due and payable.

In each case, the relevant Clearing Members shall have enforcement rights as against each other pursuant to the Contracts (Rights of Third Parties) Act 1994 in respect of any resulting payments and deliveries; LCH SA shall have no liability in respect thereof.

7.8 Failure to notify Matched Pairs

Notwithstanding Section 5.3 (*Triggering of Restructuring Cleared Transactions*) and 6.5 (*Notices of Physical Settlement*), if LCH SA does not notify the relevant Clearing Members of Restructuring Matched Pairs created pursuant to Section 5.1 (*Creation and Notification of Restructuring Matched Pairs*) on or prior to the RMP Notification Deadline or Settlement Matched Pairs and related information specified in Section 6.2 (*Creation and Notification of Settlement Matched Pairs*) by the SMP Notification Deadline, as applicable:

- the relevant Clearing Members may deliver Credit Event Notices, Notices to Exercise Movement Option, Notices of Physical Settlement or NOPS Amendment Notices to LCH SA, and vice versa;
- (b) the relevant Cleared Transactions shall be subject to Physical Settlement in accordance with their terms; and
- (c) the provisions of Sections 2.4 (Amendments to 2014 ISDA Credit Derivatives Definitions), 6 (Physical Settlement) and 8 (Matched Pair Designations and Notices) shall not apply and the terms of this CDS Clearing Supplement shall be construed accordingly.

For such purpose, Section 7.3 (*Credit Event Notices and NEMOs*) shall not apply in respect of notices given by the affected Clearing Members and accordingly Credit Event Notices and Notices to Exercise Movement Option shall be delivered directly (and not via the relevant DTCC Notice Facility).

7.9 Failure to notify Matching Information

If LCH SA notifies relevant Clearing Members of Restructuring Matched Pairs created pursuant to Section 5.1 (*Creation and Notification of Restructuring Matched Pairs*) on or prior to the RMP Notification Deadline, but does not notify DTCC of relevant Matching Information on or prior to the RMP Notification Deadline, then Section 7.3(a) (*Credit Event Notices and NEMOs to be given via DTCC*) shall not apply in respect of notices to be delivered by affected Clearing Members and accordingly Credit Event Notices and Notices to Exercise Movement Option shall be delivered directly as between Clearing Members (as designees of LCH SA) (and not via the relevant DTCC Notice Facility).

7.10 Uncertain Delivery

(a) Manual Notice permitted if Delivery of Notice in DTCC uncertain

Notwithstanding Section 7.3(a) (*Credit Event Notices and NEMOs to be given via DTCC*), where such notices are permitted to be delivered by means other than the relevant DTCC Notice Facility pursuant to this Section 7, and a Clearing Member is uncertain as to whether or not a Credit Event Notice or Notice to Exercise Movement Option (as applicable) it (or, in the case of a CCM, its CCM Client) attempted to deliver via a DTCC Notice Facility has:

- (i) actually been delivered; or
- (ii) was delivered prior to the DTCC Failure Time,

that Clearing Member shall be entitled to deliver such a notice directly to any Clearing Member comprised in a relevant Matched Pair (as designee of LCH SA) specifying that such notice is only to be effective to the extent that the other purported notice is not effective.

(b) Details to be provided of Uncertain Notice

If a Clearing Member delivers a manual notice pursuant to sub-section (a) (Manual Notice permitted if Delivery of Notice in DTCC uncertain) above, such Clearing Member shall be required to provide (together with such notice) sufficient details of the notice attempted to be given by way of the relevant DTCC Notice Facility so as to allow the other Clearing Member and LCH SA to identify the communication concerned.

(c) DTCC Notice delivered successfully

If the first Credit Event Notice or Notice to Exercise Movement Option (as applicable) to which the manual notice delivered pursuant to sub-section (a) (Manual Notice permitted if Delivery of Notice in DTCC uncertain) above relates was actually delivered successfully, any subsequent Credit Event Notice or Notice to Exercise Movement Option delivered shall be deemed not to have been delivered.

7.11 Disputes as to Notices

If any Clearing Member comprised in a Matched Pair where one such party is acting as designee of LCH SA disputes the effective delivery in accordance with the terms of the relevant Cleared Transactions of any notice delivered directly (and not via a relevant DTCC Notice Facility) in accordance with this Section 7 (and for such purposes, a dispute between the relevant Clearing Member and LCH SA shall be deemed to have arisen if LCH SA receives a Clearing Member Acknowledgement from one relevant Clearing Member only in respect of any such notice as contemplated at Section 7.7(c) (Clearing Member Acknowledgement received from one Clearing Member)):

- (a) LCH SA shall be entitled in accordance with the Procedures to calculate and call for Margin with respect to each such Clearing Member on the basis of the maximum requirement for such Clearing Member that could result from any foreseeable outcome of such dispute;
- (b) following final resolution of such dispute by arbitration or by litigation, as applicable, in accordance with the CDS Dispute Resolution Protocol, subject to the provisions of Sections 8 and 9 of the Procedures, the Clearing Members comprised in the relevant Matched Pair shall take such actions with respect to the Cleared Transactions the subject of such dispute as LCH SA determines appropriate to give effect to any relevant binding resolution; and
- (c) without prejudice to its obligations upon final resolution of the dispute, pending final resolution of any such dispute, LCH SA shall not be obliged to take any step pursuant to the terms of the relevant Cleared Transactions which would be required to have been taken by it had the relevant notice been validly delivered.

Each relevant Clearing Member must promptly notify LCH SA of any such disputes (in the form set out at Appendix XI hereto).

8. MATCHED PAIR DESIGNATIONS AND NOTICES

8.1 Creation of Matched Pairs

LCH SA will create Matched Pairs where required to do so pursuant to Section 5.1 (*Creation and Notification of Restructuring Matched Pairs*) or 6.2 (*Creation and Notification of Settlement Matched Pairs*) using a matching procedure that matches CDS Sellers with CDS Buyers pursuant to an algorithm incorporating the following principles:

- (a) the procedure seeks to create Matched Pairs between the same Clearing Member to the extent it is possible to do so before creating Matched Pairs between different Clearing Members and, for this purpose, in the context of CCMs, the procedure will create Matched Pairs separately for CCMs and their CCM Clients (individually where the CCM Client is allocated to a single CCM Client Margin Account or together where a set of CCM Clients is allocated to a single CCM Client Margin Account) and Clearing Member will be construed accordingly;
- (b) the procedure seeks to minimise the number of Matched Pairs (and accordingly, largest positions will be matched first);
- (c) each Matched Pair will, to the extent possible, have an aggregate Restructuring Matched Pair Amount or, as applicable, Settlement Matched Pair Delivery Amount, which meets the criteria as set out in a Clearing Notice; and
- (d) LCH SA will allocate a Restructuring Matched Pair Amount or, as applicable, Settlement Matched Pair Delivery Amount to each Matched Pair such that:

- (i) the sum of all Restructuring Matched Pair Amounts or, as applicable, Settlement Matched Pair Delivery Amounts, of each CDS Buyer is equal to the aggregate Floating Rate Payer Calculation Amounts in respect of all (A) Initial Single Name Cleared Transactions, Restructuring Cleared Transactions (created following the occurrence of a previous M(M)R Restructuring), Spinoff Single Name Cleared Transactions and Resulting Single Name Cleared Transactions or (B) Cleared Transactions in respect of which the Fallback Settlement Method applies to the CDS Type for such Cleared Transaction, as applicable, to which such CDS Buyer is a party; and
- (ii) the sum of all Restructuring Matched Pair Amounts or, as applicable, Settlement Matched Pair Delivery Amounts, of each CDS Seller is equal to the aggregate Floating Rate Payer Calculation Amounts in respect of (A) Initial Single Name Cleared Transactions, Restructuring Cleared Transactions (created following the occurrence of a previous M(M)R Restructuring), Spinoff Single Name Cleared Transactions and Resulting Single Name Cleared Transactions or (B) Cleared Transactions in respect of which the Fallback Settlement Method applies to the CDS Type for such Cleared Transaction, as applicable, to which such CDS Seller is a party.

Notwithstanding the above, if the Fallback Settlement Method applies in relation to a Cleared Transaction and an M(M)R Restructuring, the Restructuring Matched Pairs previously created pursuant to Section 5.1 (*Creation of Restructuring Matched Pairs*) and this Section 8.1 shall be deemed to be Settlement Matched Pairs created in accordance with Section 6.2 (*Creation and Notification of Settlement Matched Pairs*) and LCH SA shall have no obligation to create Settlement Matched Pairs in respect of such Cleared Transaction.

8.2 Registration of new Cleared Transactions and Removal of original Cleared Transactions

To the extent that any Cleared Transaction created pursuant to Section 5.2 (*Creation of Restructuring Cleared Transactions*) or Section 6.3 (*Physically Settled Cleared Transactions*) is not automatically registered in accordance with the DTCC Rules, LCH SA shall register such new Cleared Transaction in the TIW in accordance with the DTCC Rules prior to 6.00 p.m. on the date on which the RMP Notification Deadline or SMP Notification Deadline (as applicable) falls. In respect of CCMs and CCM Clients only, such registration by LCH SA shall also result in the automatic registration of any amendments made to the corresponding CCM Client Transactions.

In addition, LCH SA will, on behalf of the relevant Clearing Member, send an "Exit" message to the TIW in accordance with the DTCC Rules to terminate and remove the corresponding original Cleared Transaction(s) from the TIW prior to 6.00 p.m. on the date on which the RMP Notification Deadline or SMP Notification Deadline (as applicable) falls.

8.3 Matched Buyer Contracts

In respect of each Matched Buyer Contract which is the subject of a Matched Pair, LCH SA, pursuant to Section 11.2(c)(iv) of the 2014 ISDA Credit Derivatives Definitions (amended as

set out at Section 2.4 (Amendments to 2014 ISDA Credit Derivatives Definitions)), as designator, shall be deemed to have designated Matched Seller in such Matched Pair as its designee:

- to receive on its behalf from, and to deliver on its behalf to, Matched Buyer of the Matched Pair any applicable notices or certifications in accordance with the terms of the applicable Cleared Transaction (other than notices required to be delivered via a DTCC Notice Facility);
- (b) other than in respect of the Physical Settlement Amount relating to the settlement of Non-DVP Obligations as referred to in Section 6.6 (*Delivery of Non-DVP Obligations*), to pay on behalf of LCH SA any applicable Physical Settlement Amount in accordance with the terms of the applicable Physically Settled Cleared Transaction, and to pay to, and to receive from, Matched Buyer of the Matched Pair, in each case, on behalf of LCH SA, any other amounts due and payable (including costs and expenses of settlement due under the applicable Matched Buyer Contract); and
- (c) to take Delivery, on behalf of LCH SA, of Deliverable Obligations from Matched Buyer of the Settlement Matched Pair.

The relevant Matched Seller shall be deemed to have accepted such designation upon notification of the relevant Matched Pair created and notified in accordance with the provisions of this CDS Clearing Supplement.

8.4 Matched Seller Contracts

In respect of each Matched Seller Contract which is the subject of a Matched Pair, LCH SA, pursuant to Section 11.2(c)(iv) of the 2014 ISDA Credit Derivatives Definitions (as amended pursuant to Section 2.4 (*Amendments to 2014 ISDA Credit Derivatives Definitions*) above), as designator, shall be deemed to have designated Matched Buyer in such Matched Pair as its designee:

- to receive on its behalf from, and to deliver on its behalf to, Matched Seller of the Matched Pair any applicable notices or certifications in accordance with the terms of the applicable Cleared Transaction (other than notices required to be delivered via a DTCC Notice Facility);
- (b) other than in respect of any Physical Settlement Amount relating to the settlement of Non-DVP Obligations as referred to in Section 6.6 (*Delivery of Non-DVP Obligations*), to receive on behalf of LCH SA the applicable Physical Settlement Amount in accordance with the terms of any applicable Physically Settled Cleared Transaction, and to pay to, and to receive from, Matched Seller of the Matched Pair, in each case, on behalf of LCH SA, any other amounts due and payable (including costs and expenses of settlement due under the Matched Seller Contract); and
- (c) to Deliver, on behalf of LCH SA, the relevant Deliverable Obligations to Matched Seller of the Settlement Matched Pair.

The relevant Matched Buyer shall be deemed to have accepted such designation upon notification of the relevant Matched Pair created and notified in accordance with the provisions of this CDS Clearing Supplement.

8.5 Exercise of Rights

In relation to each Matched Pair:

- (a) the exercise of any rights by Matched Buyer against LCH SA under a Matched Buyer Contract (other than the right to give any notice via DTCC Notice Facility) shall be deemed to constitute the exercise of equal and simultaneous rights by LCH SA against Matched Seller under the Matched Seller Contract of the relevant Matched Pair; and
- (b) the exercise of any rights by Matched Seller against LCH SA under a Matched Seller Contract (other than a right to give notice under a DTCC Notice Facility) shall be deemed to constitute the exercise of equal and simultaneous rights by LCH SA against Matched Buyer under the Matched Buyer Contract of the relevant Matched Pair.

8.6 Clearing Member matched with Itself

a) Notices

In the event that:

- Matched Buyer and Matched Seller of a Matched Pair pursuant to this Section 8 (Matched Pair Designations and Notices) is the same Clearing Member; and
- (ii) notwithstanding Section 7.3(a) (*Credit Event Notices and NEMOs to be given via DTCC*), a notice or certification is permitted to be delivered in respect of one of the Cleared Transactions forming part of such Matched Pair by means other than the relevant DTCC Notice Facility pursuant to Section 7 (*Delivery of Notices and Fallbacks*),

such notice shall be deemed to be given upon such Clearing Member sending a Clearing Member Acknowledgement to LCH SA pursuant to Section 7.7(a) (*Duty to deliver Clearing Member Acknowledgements*) above in respect of such notice and Section 7.7(b) (*Clearing Member Acknowledgement received from both Clearing Members*) shall apply.

(b) Payments and Deliveries

In the event that:

- Matched Buyer and Matched Seller of a Matched Pair pursuant to this Section 8 (Matched Pair Designations and Notices) is the same Clearing Member; and
- such Clearing Member is required to make a payment or delivery pursuant to the terms of one of the Cleared Transactions forming part of such Matched Pair as designate of LCH SA,

such payment or delivery shall be deemed to have been made upon such Clearing Member giving notice to LCH SA in accordance with Section 7.1 (*General Rules relating to Notices*) that such payment or delivery should be deemed to have been made for the purposes of such Cleared Transaction.

8.7 Notices

In relation to each Matched Pair:

- (a) where Matched Buyer validly delivers or serves any notice to Matched Seller as designee of LCH SA in accordance with the terms of a relevant Matched Buyer Contract, such notice shall additionally be effective as a notice given by such Matched Buyer as designee of LCH SA to Matched Seller for the purposes of the relevant Matched Seller Contract; and
- (b) where Matched Seller validly delivers or serves any notice to Matched Buyer as designee of LCH SA in accordance with the terms of a relevant Matched Seller Contract, such notice shall additionally be effective as a notice given by such Matched Seller as designee of LCH SA to Matched Buyer for the purposes of the relevant Matched Buyer Contract.

9. SELF-REFERENCING TRANSACTIONS

9.1 Occurrence of Clearing Member Self Referencing Transaction

(a) Duty to notify

In respect of a Single Name Cleared Transaction that is registered in the House Account Structure of a Clearing Member, the relevant Clearing Member shall, unless prohibited from so doing by applicable law or its internal policies, notify LCH SA as soon as reasonably practicable if:

- such Clearing Member is or consolidates or amalgamates with, or merges into, or transfers all or substantially all of its assets to, the Reference Entity in relation to such Single Name Cleared Transaction, or enters into any agreement in respect of any of the foregoing;
- such Clearing Member and the Reference Entity in relation to such Single Name Cleared Transaction are or become Affiliates; or
- (iii) in respect of a Restructuring Cleared Transaction, such Clearing Member is or becomes the Reference Entity in relation to such Restructuring Cleared Transaction as a result of the occurrence of the relevant M(M)R Restructuring.

(b) Auction of Affected Transactions

Following receipt of any such notification or occurrence of a Clearing Member Self Referencing Transaction due to a registration of Transfer Positions under Clause 6 of the CDS Default Management Process or a transfer of Client Cleared Transactions to

the House Trade Account under TITLE V, Chapter 4 or TITLE VI, Chapter 4 of the CDS Clearing Rule Book, LCH SA will, unless otherwise agreed in consultation with the CDS Default Management Group and the affected Clearing Member, and with the assistance of the CDS Default Management Group, conduct an auction process to liquidate the relevant Single Name Cleared Transaction and enter into an equivalent Single Name Cleared Transaction with a Clearing Member other than the affected Clearing Member.

(c) Alternative Action

If LCH SA, after consultation with the CDS Default Management Group and the affected Clearing Member, believes that the circumstances are such that an auction may be inappropriate, LCH SA may take such other action in consultation with the Risk Committee as it considers reasonably necessary to achieve its primary aim in these circumstances of addressing the risks resulting from a Clearing Member being party to a Clearing Member Self Referencing Transaction, while endeavouring, as far as is reasonably practicable in the circumstances without prejudicing the achievement of the primary aim, to avoid materially and adversely affecting the relevant Clearing Member.

(d) Compression of Affected Transactions prior to Auction

Prior to determining the Single Name Cleared Transactions to be subject to any auction pursuant to paragraph (b) above, where the affected Clearing Member acts as CDS Buyer and CDS Seller in respect of fungible Single Name Cleared Transactions, LCH SA shall, in consultation with the CDS Default Management Committee as to the transaction sizes of resulting Single Name Cleared Transactions to be auctioned, compress such Single Name Cleared Transactions up to the extent that, following such compression, Single Name Cleared Transactions representing in aggregate the Open Position of the affected Clearing Member in respect of such fungible Single Name Cleared Transactions are recognised.

For these purposes, LCH SA will provide the affected Clearing Member with a report detailing the Single Name Cleared Transactions to be subject to such compression.

The affected Clearing Member will be deemed to have submitted a request to LCH SA prior to 5.00 p.m. on the day on which LCH SA carries out the compression for ad hoc compression of such Single Name Cleared Transactions in accordance with Section 5 of the Procedures and such compression shall be carried out in accordance with Section 5 of the Procedures on the basis of such deemed request for ad hoc compression.

(e) Auction Terms

LCH SA shall determine the timing and other particular characteristics of each such auction in consultation with the CDS Default Management Committee, including determining the size of the bid/offer spread and/or of the Single Name Cleared

Transactions to be auctioned, whether one or more such auctions are to be held and the timing and structure of such auctions (including the frequency at which firm bid and firm offer quotations will be requested and the transaction size (that is, the Floating Rate Payer Calculation Amount)).

Clearing Members (excluding the affected Clearing Member) may be requested, and will not be required, to submit actionable quotations in such an auction.

(f) Creation of New Transactions and Termination of Existing Transactions

LCH SA will enter into Single Name Cleared Transactions with Clearing Members, other than the affected Clearing Member, in the amount and at the prices determined pursuant to such auction.

At the time of entering into such Single Name Cleared Transactions, the corresponding Single Name Cleared Transactions of the affected Clearing Member shall be terminated by reference to the prices at which LCH SA enters into such new Single Name Cleared Transactions.

The affected Clearing Member, LCH SA and the other Clearing Members, as applicable, shall submit such information as is required in accordance with the DTCC Rules so as to reflect the terms of any reduction to, termination of or entry into of any Single Name Cleared Transaction as a result of any such auction(s).

(g) Costs of LCH SA

The affected Clearing Member will bear the cost of the associated bid/offer spread and any reasonable out-of-pocket costs and expenses of LCH SA in connection with such auction(s) and its entering into such new Single Name Cleared Transactions.

(h) LCH SA to determine Amounts Owed and Payable

Amounts owed by the affected Clearing Member to, or receivable by it from, LCH SA in connection with any such auction shall be determined by LCH SA.

In addition, any amounts payable (and the dates of settlement with respect thereto) relating to any Single Name Cleared Transactions created, reduced or terminated pursuant to any such auction shall be determined by LCH SA.

9.2 Occurrence of Client Self Referencing Transactions

(a) Notification

In respect of a Single Name Cleared Transaction registered in the Client Account Structure of a Clearing Member, the relevant Clearing Member, as applicable, shall, unless prohibited from so doing by applicable law or its internal policies, notify LCH SA as soon as reasonably practicable if it receives a notice from the Client that:

- the Client is or has consolidated or amalgamated with, or merged into, or has transferred all or substantially all of its assets to, the Reference Entity in relation to such Single Name Cleared Transaction or entered into any agreement in respect of any of the foregoing;
- (ii) the Client and the Reference Entity in relation to such Single Name Cleared Transaction are or have become Affiliates; or
- (iii) in respect of a Restructuring Cleared Transaction, the Client is or has become the Reference Entity in relation to such Restructuring Cleared Transaction as a result of the occurrence of the relevant M(M)R Restructuring.

(b) Auction Process

Following the giving of any such notification:

- (i) LCH SA will, unless otherwise agreed in consultation with the CDS Default Management Group and the relevant Clearing Member, conduct an auction process in consultation with the CDS Default Management Group and the relevant Clearing Member, to liquidate the relevant Single Name Cleared Transaction and enter into an equivalent Single Name Cleared Transaction with another Clearing Member; and
- (ii) the provisions of Section 9.1. (Occurrence of Clearing Member Self Referencing Transaction) will apply mutatis mutandis, provided that:
 - (A) Section 9.1(a) (Duty to Notify) and 9.1(b) (Auction of Affected Transactions) shall not apply;
 - (B) Section 9.1(d) (Compression of Affected Transactions prior to Auction) is amended by deleting the words "to paragraph (b) above" and replacing them with the words "as a result of Section 9.2 (Occurrence of Client Self Referencing Transactions)";
 - (C) Section 9.1(d) (Compression of Affected Transactions prior to Auction) is amended in the case of CCM Clients only by inserting the words "that have a corresponding CCM Client Transaction with the affected CCM Client" immediately after the words "in respect of fungible Single Name Cleared Transactions";
 - (D) Section 9.1(e) (Auction Terms) is amended by deleting the words "(excluding the affected Clearing Member)" in the final paragraph thereof;
 - (E) Section 9.1(f) (Creation of New Transactions and Termination of Existing Transactions) is amended by deleting the words "other than the affected Clearing Member" in the first paragraph thereof; and

(F) Each reference to "affected Clearing Member" in Section 9.1 (Occurrence of Clearing Member Self Referencing Transaction) is deleted and replaced with a reference to "relevant Clearing Member".

(c) Costs of LCH SA

The cost of the associated bid/offer spread and any reasonable out-of-pocket costs and expenses of LCH SA in connection with such auction(s) or any alternative action shall be allocated to the CCM Client Collateral Account or the FCM/BD Client Collateral Account as applicable.

10. MANDATORY PROVISIONS FOR CCM CLIENT TRANSACTIONS

In Appendix XIII, certain provisions are set-out (the "Mandatory Provisions") for incorporation into a CCM Client Transaction between a CCM and its CCM Client that corresponds to a CCM Client Cleared Transaction registered in the CCM Client Trade Account in the name of such CCM for such CCM Client. The CDS Clearing Supplement and these Mandatory Provisions have been drafted so as to complement each other.

LCH SA shall not be responsible for any loss suffered or expense incurred by a CCM or any CCM Client as a result of the inclusion or non-inclusion of the Mandatory Provisions in the CCM Client Transaction Documents.

11. AMENDMENTS

LCH SA may amend the provisions of this CDS Clearing Supplement (including, without limitation, the Mandatory Provisions) from time to time so as to comply with any legal or regulatory developments or any recommendations adopted by the industry in respect of CDS or Cleared Transactions or CCM Client Transactions, as applicable, or so as to reflect any technological advancements, in each case in accordance with the provisions of Section 1.2.2 (*Modification*) of Chapter 2 (*General Provisions*) of the CDS Clearing Rule Book.

12. FORMS OF NOTICES

A form of Credit Event Notice, Notice to Exercise Movement Option, Notice of Physical Settlement, NOPS Amendment Notice, Asset Package Notice, notice to exercise alternative delivery procedure pursuant to Section 6.7 (Alternative Delivery Procedure), notice to fallback to Cash Settlement in respect of Non-Deliverable Obligations pursuant to Section 6.12 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations), Physical Settlement Confirmation, No Physical Settlement Confirmation, notice of Clearing Member Communications Failure Event and where no longer subject to a Clearing Member Communications Failure Event, in each case pursuant to Section 7.6 (Clearing Member Communications Failure Event), notice of dispute relating to any Matched Pair as contemplated by Section 7.11 (Dispute as to Notices) and notice relating to Self-Referencing Transactions as contemplated by Section 9 (Self-Referencing Transactions) is set out in Appendix I, II, III, IVA, IVB, V, VI, VIII, III, IX, X, XI and XII respectively hereto.

Formatted: Font: +Body (Calibri), 11 pt

Formatted: Font: +Body (Calibri), 11 pt

Formatted: Font: +Body (Calibri), 11 pt

Any of the above referenced notices shall be delivered in substantially the form appended hereto, provided, for the avoidance of doubt, that such notices may refer to multiple transactions and may have certain firm-specific variations.

For the avoidance of doubt, the above referenced notices shall be governed by and construed in accordance with English law.

13. EXCLUSION OF LIABILITY

Without prejudice to the provisions of Article 1.2.10.3 of the CDS Clearing Rule Book:

(a) No liability for Failure of Designee to perform in respect of Matched Pair

Without prejudice to its obligations under or in respect of a Cleared Transaction, LCH SA shall not be liable for any loss or cost arising out of any failure of any Clearing Member comprised in a Matched Pair to perform its obligations as designee of LCH SA against a related Matched Buyer or Matched Seller, as applicable.

(b) No liability for Fault of Third Party or Force Majeure

LCH SA shall have no liability to any person where Restructuring Cleared Transactions are not or are improperly created, Restructuring Cleared Transactions are not or are improperly terminated or the Movement Option process is not or is improperly implemented, in each case for the purposes of the DTCC Rules, because of a third party's fault or a force majeure event. In particular, LCH SA shall not incur any liability arising as a result of any action or omission of DTCC.

(c) No Obligation to verify Notices received

LCH SA shall have no responsibility to verify the contents of any notice received by it from any Clearing Member under the terms of any Cleared Transaction.

14. **DISPUTE RESOLUTION**

For the avoidance of doubt, all Disputes shall be referred to and finally resolved by arbitration or by litigation, as applicable, in accordance with the CDS Dispute Resolution Protocol, subject to the provisions of Sections 8 and 9 of the Procedures.

15. GOVERNING LAW

For the avoidance of doubt, the governing law applicable to this CDS Clearing Supplement (excluding the Mandatory Provisions to the extent that such terms are incorporated by reference in the CCM Client Transaction Documents entered into between a CCM and its CCM Client in respect of a CCM Client Transaction), the 2014 ISDA Credit Derivatives Definitions and any Cleared Transactions (and any related definitions or Clearing Notices issued in respect of the CDS Clearing Supplement, the 2014 ISDA Credit Derivatives Definitions or any Cleared Transactions) and any non-contractual obligations arising out of, relating to or having any connection with them shall be as set out in Section 1.2.14 (*Governing Law*) of the CDS Clearing Rule Book.

File No. SR-LCH SA-2022-007 Page 194 of ...

PUBLIC

APPENDIX I: FORM OF CREDIT EVENT NOTICE

To: [Restructuring Matched Pair Counterparty Address and Contact Information]

[To/Copy to:]

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

CREDIT EVENT NOTICE:

Credit Derivative Transaction Details: As set out in the Schedule hereto².

Reference is made to the Credit Derivative Transaction[s] described in the Schedule hereto (the **Transaction[s]**). Capitalised terms used and not otherwise defined in this letter shall have the meanings given them in the confirmation of the relevant Transaction.

This letter is our Credit Event Notice to you that a [insert type] Credit Event occurred with respect to [insert name of Reference Entity] on or about [insert date], when [describe Credit Event].

Nothing in this letter shall be construed of a waiver of any rights we may have with respect to the Transaction.

[Clearing Member]

Sincerely

Name: Title:

<u>10219615309-v3</u> <u>70-41024903</u>

Formatted: Line spacing: single

² A single Credit Event Notice may be submitted for multiple trades in respect of the same Counterparty

File No. SR-LCH SA-2022-007 Page 195 of 439

PUBLIC

SCHEDULE

Credit Derivative Transaction Details

[Clearing Member] acting as Seller/Buyer	Restructuring Matched Pair ID	Trade ID	Reference Entity	Trade Date	Effective Date	Exercise Amount ³
[Seller] [Buyer]	[●]	[•]	[●]	[●]	[●]	[•]

70-41024903

Formatted: Line spacing: single

10219615309-v3

 $^{^{\}rm 3}$ Where different to the outstanding Floating Rate Payer Calculation Amount

APPENDIX II: FORM OF NOTICE TO EXERCISE MOVEMENT OPTION

To: [Restructuring Matched Pair Counterparty Address and Contact Information]

[To/Copy to:]

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

Dear Sir/Madam

Notice to Exercise Movement Option

Credit Derivative Transaction Details: As set out in the Schedule hereto4.

Reference is made to: (a) the Credit Derivative Transaction[s] described in the Schedule hereto (the **Transaction[s]**) between [], as Seller, and [], as Buyer; (b) the Credit Event Notice dated insert date], previously delivered to you on [insert date]; and (c) the occurrence of the No Auction Announcement Date on [insert date] pursuant to Section 6.11(b) or Section 6.11(c)(ii) of the 2014 ISDA Credit Derivatives Definitions (the **Definitions**).

This letter constitutes a Notice to Exercise Movement Option. Any capitalised term not otherwise defined in this letter will have the meaning, if any, assigned to such term in the confirmation of the relevant Transaction or, if no meaning is specified therein, in the Definitions.

We hereby exercise the Movement Option, confirm that each Transaction will be settled in accordance with the relevant Credit Derivatives Auction Settlement Terms specified in the column entitled "Auction Settlement Terms" corresponding to such Transaction in the Schedule hereto and require performance by you in accordance therewith.

Yours faithfully,		
[Matched Buyer/Matched Seller]		
Name:		
Title:		

Formatted: Line spacing: single

⁴ A single Notice to Exercise Movement Option may be submitted for multiple trades in respect of the same Counterparty

File No. SR-LCH SA-2022-007 Page 197 of ...

PUBLIC

SCHEDULE

Credit Derivative Transaction Details

Restructuring Matched Pair ID	Trade ID	Reference Entity	Trade Date	Effective Date	Auction Settlement Terms
[●]	[●]	[●]	[●]	[●]	[•]

APPENDIX III: FORM OF NOTICE OF PHYSICAL SETTLEMENT

To: Settlement Matched Pair Matched Seller Address and Contact Information

Copy to:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

Notice of Physical Settlement

Credit Derivative Transaction Details: As set out in the Schedule hereto⁵.

Reference is made to: (a) the Credit Derivative Transaction[s] described in the Schedule hereto (the **Transaction[s]**) between [], as Seller, and [], as Buyer. Reference is also made to the Credit Event Notice [and Notice of Publicly Available Information] dated [insert date], previously delivered to you on [insert date].

This letter constitutes a Notice of Physical Settlement. Any capitalised term not otherwise defined in this letter will have the meaning, if any, assigned to such term in the confirmation of the relevant Transaction (the **Relevant Confirmation**) or, if no meaning is specified therein, in the 2014 ISDA Credit Derivatives Definitions (the **Definitions**).

We hereby confirm that we will settle the Transaction[s] and require performance by you in accordance with the provisions of the Relevant Confirmation and the Definitions relating to Physical Settlement. Subject to the terms of the relevant Transaction, we will Deliver to you on or before the Physical Settlement Date, an amount of the Deliverable Obligation(s) described in the column entitled "Deliverable Obligation(s)" in the Schedule hereto, corresponding to such Transaction:

Yours faithfully,		
[Matched Buyer]		
Name:		
Title:		
Title.		

Formatted: Line spacing: single

⁵ A single Notice of Physical Settlement may be submitted for multiple trades in respect of the same Counterparty

File No. SR-LCH SA-2022-007 Page 199 of ...

PUBLIC

SCHEDULE

Credit Derivative Transaction Details

Settlement Matched Pair ID	Trade ID	Reference Entity	Trade Date	Effective Date	Deliverable Obligation(s)₅
[●]	[●]	[●]	[●]	[●]	[Currency][Due and Payable Amount]][●]

Formatted: Line spacing: single

⁶ Describe the Deliverable Obligation(s) to be Delivered, including the currency and Outstanding Principal Balance or Due and Payable Amount for each such Deliverable Obligation and, if available and applicable, the CUSIP or ISIN number (or, if such identifying number is not available, the rate and tenor of the Deliverable Obligation)

File No. SR-LCH SA-2022-007 Page 200 of 439

PUBLIC

APPENDIX IVA: FORM OF NOPS AMENDMENT NOTICE

To: Settlement Matched Pair Matched Seller Address and Contact Information

Copy to:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

Settlement Matched Pair ID: [●]

Trade ID: [●]

Yours faithfully,

NOPS Amendment Notice

Credit Derivative Transaction Details: [Trade Date], [Effective Date], [Reference Entity]

Reference is made to the Credit Derivative Transaction described above (the **Transaction**) between [], as Seller, and [], as Buyer. Reference is also made to the Notice of Physical Settlement] NOPS Amendment Notice] dated [insert date], previously delivered to you on [insert date] dated [insert date].

This letter constitutes a NOPS Amendment Notice. Any capitalised term not otherwise defined in this letter will have the meaning, if any, assigned to such term in the Confirmation of the Transaction or, if no meaning is specified therein, in the CDS Clearing Supplement.

We hereby notify you that we are replacing the following Deliverable Obligation(s) specified in the Notice of Physical Settlement] NOPS Amendment Notice] specified above with the following Replacement Deliverable Obligation(s):

[describe the Deliverable Obligation(s) to be replaced, including the Replaced Deliverable Obligation Outstanding Amount for each such Deliverable Obligation and, if available and applicable, the CUSIP or ISIN number (or, if such identifying number is not available, the rate and tenor of the Deliverable Obligation) and the Replacement Deliverable Obligation(s) for each Replaced Deliverable Obligation Outstanding Amount so specified and, if available and applicable, the CUSIP or ISIN number (or, if such identifying number is not available, the rate and tenor of the Replacement Deliverable Obligation)].

[Matched Buyer]	
Name:	
Title:	
10219615309-v3	70-41024903

File No. SR-LCH SA-2022-007 Page 201 of 439

PUBLIC

APPENDIX IVB: FORM OF ASSET PACKAGE NOTICE

To: Settlement Matched Pair Matched Seller Address and Contact Information

Copy to:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

Settlement Matched Pair ID: [●]

Trade ID: [●]

Asset Package Notice

Credit Derivative Transaction Details: [Trade Date], [Effective Date], [Reference Entity]

Reference is made to the Credit Derivative Transaction described above (the **Transaction**) between [], as Seller, and [], as Buyer. Reference is also made to the Notice of Physical Settlement] [NOPS Amendment Notice] dated [insert date], previously delivered to you on [insert date] dated [insert date].

This letter constitutes an Asset Package Notice. Any capitalised term not otherwise defined in this letter will have the meaning, if any, assigned to such term in the Confirmation of the Transaction or, if no meaning is specified therein, in the CDS Clearing Supplement.

We hereby notify you that our obligation to Deliver the following Deliverable Obligation(s) specified in the Notice of Physical Settlement] NOPS Amendment Notice] specified above shall be satisfied by Delivery of the following Asset Package:

[describe the Deliverable Obligation(s) to which the Asset Package relates, including the Deliverable Obligation Amount for each such Deliverable Obligation and, if available and applicable, the CUSIP or ISIN number (or, if such identifying number is not available, the rate and tenor of the Deliverable Obligation) and the Assets comprising the Asset Package for each such Deliverable Obligation(s) and, if available and applicable, the CUSIP or ISIN number (or, if such identifying number is not available, the rate and tenor of the Asset, if applicable)].

Yours faithfully,	
[Matched Buyer]	
Name:	
Title:	
10219615309-v3	70-41024903

APPENDIX V: FORM OF NOTICE TO EXERCISE ALTERNATIVE DELIVERY PROCEDURE PURSUANT TO SECTION 6.7 (ALTERNATIVE DELIVERY PROCEDURE)

To:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

[date]

Dear Sir/Madam

Notice to Exercise Alternative Delivery Procedure⁷

Reference is made to: the Settlement Matched Pairs described in the Schedule hereto, being the Settlement Matched Pairs to which this notice relates and Section 6.7 (*Alternative Delivery Procedure*) of the CDS Clearing Supplement. Defined terms shall have the meanings assigned to them in the CDS Clearing Supplement, unless otherwise defined herein.

The Settlement Matched Pair Delivery Amount relating to each Settlement Matched Pair is that amount specified in the column entitled "Settlement Matched Pair Delivery Amount" in the Schedule hereto, corresponding to such Settlement Matched Pair.

In respect of each Settlement Matched Pair and in accordance with Section 6.7 (*Alternative Delivery Procedure*) we hereby elect to exercise our rights against and perform obligations to you in accordance with the alternative delivery procedure in relation to such percentage and amount of the Settlement Matched Pair Delivery Amount as set out in the column entitled "Percentage and Amount of Settlement Matched Pair Delivery Amount" in the Schedule hereto corresponding to such Settlement Matched Pair.

 $This \ notice \ may \ be \ executed \ in \ any \ number \ of \ counterparts \ which \ together \ shall \ constitute \ one \ notice.$

By countersigning this notice, you are deemed to have given your consent to the above in satisfaction of the requirement to obtain your consent contained in Section 6.7 (*Alternative Delivery Procedure*) of the CDS Clearing Supplement.

This notice (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this agreement or its formation) shall be governed by and construed in accordance with English law.

Yours faithfully,

[Matched Buyer/ Matched Seller]

⁷ A single Notice to Exercise Alternative Delivery Procedure may be submitted for multiple trades in respect of the same Counterparty Formatted: Line spacing: single

File No. SR-LCH SA-2022-007 Page 203 of 439

PUBLIC

Name:
ïtle:
Countersigned by LCH SA
,
lame:
itle:

File No. SR-LCH SA-2022-007 Page 204 of 439

PUBLIC

SCHEDULE

Credit Derivative Transaction Details

Settlement Matched Pair ID	Trade ID	Reference Entity	Trade Date	Effective Date	Settlement Matched Pair Delivery Amount	Percentage and amount of Settlement Matched Pair Delivery Amount
[●]	[●]	[●]	[●]	[●]	[●]	[●]

10219615309-v3 70-41024903

APPENDIX VI: FORM OF NOTICE OF FALLBACK TO CASH SETTLEMENT OF NON-DELIVERABLE OBLIGATIONS PURSUANT TO SECTION 6.12 (FALLBACK TO CASH SETTLEMENT IN RESPECT OF NONDELIVERABLE OBLIGATIONS)

To: Settlement Matched Pair Matched Seller Address and Contact Information

To:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

Settlement Matched Pair ID: [●]

Trade ID: [●]

Dear Sir/Madam

Notice of fallback to Cash Settlement in respect of Non-Deliverable Obligations pursuant to Section 6.12 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) of the CDS Clearing Supplement

Reference is made to Section 6.12 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) of the CDS Clearing Supplement and [insert details of the relevant Settlement Matched Pair(s)], being the Settlement Matched Pair[s] to which this notice relates and to the [Notice of Physical Settlement][NOPS Amendment Notice] previously delivered to you on [insert date] dated [insert date]. Defined terms shall have the meanings assigned to them in the CDS Clearing Supplement, unless otherwise defined herein.

In accordance with Section 6.12 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) of the CDS Clearing Supplement we hereby notify you we are not permitted to Deliver the Deliverable Obligations specified below (such Deliverable Obligations, the Non-Deliverable Obligations) as specified in the [Notice of Physical Settlement][NOPS Amendment Notice] specified above for the following reasons:

[Insert details of the relevant Non-Deliverable Obligations and reasonable detail of the relevant circumstances, as described paragraphs (a)(i) and (a)(ii) of Section 6.12 of the CDS Clearing Supplement].

This notice (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this agreement or its formation) shall be governed by and construed in accordance with English law.

Yours faithfully,

File No. SR-LCH SA-2022-007 Page 206 of 439

IRI	

[Matched Buyer]		
Name:		
Title:		

File No. SR-LCH SA-2022-007 Page 207 of 439

PUBLIC

APPENDIX VII: FORM OF PHYSICAL SETTLEMENT CONFIRMATION AS CONTEMPLATED BY SECTION 6.19 (MISCELLANEOUS PROVISIONS RELATING TO PHYSICAL SETTLEMENT)

To:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

[date]

Dear Sir/Madam

Physical Settlement Confirmation

Reference is made to Section 6.19(b) (Notification of Completion of Physical Settlement) of the CDS Clearing Supplement and the Settlement Matched Pair[s] described in the Schedule hereto, being the Settlement Matched Pair[s] to which this notice relates. Defined terms shall have the meanings assigned to them in the CDS Clearing Supplement, unless otherwise defined herein.

In accordance with Section 6.19(b) (*Notification of Completion of Physical Settlement*) of the CDS Clearing Supplement, we hereby notify you that we have completed Physical Settlement with respect to such Settlement Matched Pair[s].

This notice is delivered in satisfaction of the requirement in Section 6.19(b) (*Notification of Completion of Physical Settlement*) of the CDS Clearing Supplement to inform you of such completion.

This notice (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this agreement or its formation) shall be governed by and construed in accordance with English law.

Yours faithfully,	
[Matched Buyer]/[Matched Seller]	
	
Name:	
Title:	

File No. SR-LCH SA-2022-007 Page 208 of 439

PUBLIC

SCHEDULE

Settlement Matched Pair Details

Settlement Matched Pair ID	Trade ID	Reference Entity	Trade Date	Effective Date
[•]	[•]	[•]	[•]	[•]

File No. SR-LCH SA-2022-007 Page 209 of 439

PUBLIC

APPENDIX VIII: FORM OF NO PHYSICAL SETTLEMENT CONFIRMATION AS CONTEMPLATED BY SECTION 6.19 (MISCELLANEOUS PROVISIONS RELATING TO PHYSICAL SETTLEMENT)

To:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

[date]

Settlement Matched Pair ID: [●]

Trade ID: [●]

Dear Sir/Madam

No Physical Settlement Confirmation

Reference is made to Section 6.19(c) (*Notification that Physical Settlement will not occur*) of the CDS Clearing Supplement and the Settlement Matched Pair[s] described in the Schedule hereto, being the Settlement Matched Pair[s] to which this notice relates. Defined terms shall have the meanings assigned to them in the CDS Clearing Supplement, unless otherwise defined herein.

In accordance with Section 6.19(c) (Notification that Physical Settlement will not occur) of the CDS Clearing Supplement, we hereby notify you that no Notice of Physical Settlement has been delivered within the relevant time period permitted for such delivery in accordance with the terms of the relevant Physically Settled Cleared Transactions and, accordingly, that Physical Settlement will not, under the terms of such Physically Settled Cleared Transactions, occur.

This notice is delivered in satisfaction of the requirement in Section 6.19(c) (*Notification that Physical Settlement will not occur*) of the CDS Clearing Supplement to inform you of such completion.

This notice (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this agreement or its formation) shall be governed by and construed in accordance with English law

with English law.	
Yours faithfully,	
[Matched Buyer]/[Matched Seller]	
Name:	
Title:	

File No. SR-LCH SA-2022-007 Page 210 of 439

PUBLIC

SCHEDULE

Settlement Matched Pair Details

Settlement Matched Pair ID	Trade ID	Reference Entity	Trade Date	Effective Date
[•]	[•]	[•]	[●]	[●]

File No. SR-LCH SA-2022-007 Page 211 of 439

PUBLIC

APPENDIX IX: FORM OF NOTICE OF CLEARING MEMBER COMMUNICATIONS FAILURE EVENT PURSUANT TO SECTION 7.6 (CLEARING MEMBER COMMUNICATIONS FAILURE EVENT)

To:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

[date]

Dear Sir/Madam

Notice certifying occurrence of a Clearing Member Communications Failure Event

Reference is made to Section 7.6(b) (Clearing Member to notify LCH SA of Occurrence of Clearing Member Communications Failure Event) of the CDS Clearing Supplement. Defined terms shall have the meanings assigned to them in the CDS Clearing Supplement, unless otherwise defined herein.

In accordance with Section 7.6(b) (Clearing Member to notify LCH SA of Occurrence of Clearing Member Communications Failure Event) of the CDS Clearing Supplement, notice is hereby given that we are affected by a Clearing Member Communications Failure Event [insert details of such failure].

This notice (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this agreement or its formation) shall be governed by and construed in accordance with English law.

tti Ligisi iaw.
ours faithfully,
Clearing Member]

ame:
tle:
igned by a senior officer (such as a managing director or equivalent) on behalf of the Clearing lember]

File No. SR-LCH SA-2022-007 Page 212 of 439

PUBLIC

APPENDIX X: FORM OF NOTICE FOR CEASING TO BE SUBJECT TO A CLEARING MEMBER COMMUNICATIONS FAILURE EVENT PURSUANT TO SECTION 7.6 (CLEARING MEMBER COMMUNICATIONS FAILURE EVENT)

To:

LCH SA
18, rue du Quatre Septembre
75002 Paris
France
[Contact details]

[date]

Dear Sir/Madam

Notice that a Clearing Member is no longer subject to Clearing Member Communications Failure Event

Reference is made to Section 7.6(d) (Notification of Resolution of Clearing Member Communications Failure Event) of the CDS Clearing Supplement and the notice certifying the occurrence of a Clearing Member Communications Failure Event delivered by us to LCH SA on [●] (the Notice of Clearing Member Communications Failure Event). Defined terms shall have the meanings assigned to them in the CDS Clearing Supplement, unless otherwise defined herein.

In accordance with Section 7.6(d) (*Notification of Resolution of Clearing Member Communications Failure Event*) of the CDS Clearing Supplement, notice is hereby given that we are no longer subject to the relevant Clearing Member Communications Failure Event described in the Notice of Clearing Member Communications Failure Event.

This notice (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this agreement or its formation) shall be governed by and construed in accordance with English law.

-		

File No. SR-LCH SA-2022-007 Page 213 of 439

PUBLIC

APPENDIX XI: FORM OF NOTICE OF DISPUTE RELATING TO ANY MATCHED PAIR

To:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]	
[date]	
[Restructuring][Settlement] Matched Pair ID: [●]	
Trade ID: [●]	
Dear Sir/Madam	
Notice of dispute relating to [insert details of the relevant Matched F	Pairs subject to a dispute]
Reference is made to Section 7.11 (<i>Disputes as to Notices</i>) of the CDS terms shall have the meanings assigned to them in the CDS Clearing defined herein.	
In accordance with Section 7.11 (<i>Disputes as to Notices</i>) of the CDS hereby given of the following dispute(s):	Clearing Supplement, notice i
[insert details of Matched Pair(s) affected and the relevant dispute].	
This notice (and any dispute, controversy, proceedings or claim of wha any way relating to this agreement or its formation) shall be governed by with English law.	-
Yours faithfully,	
[Matched Buyer]/[Matched Seller]	
Name:	
Title:	

File No. SR-LCH SA-2022-007 Page 214 of 439

Formatted: Right: 10.25 cm

Formatted: Right: 7.04 cm

APPENDIX XII: FORM OF NOTICE RELATING TO SELF-REFERENCING TRANSACTIONS

To:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

[date]

Dear Sir/Madam

Notice relating to Self-Referencing Transactions

Credit Derivative Transaction Details: As set out in the Schedule hereto⁸.

Reference is made to the Credit Derivative Transaction[s] described in the Schedule hereto (the **Transaction[s]**) between $[\bullet]$, as Seller and $[\bullet]$, as Buyer and to Section 9.1(a) (*Duty to notify*) of the CDS Clearing Supplement. Defined terms shall have the meanings assigned to them in the CDS Clearing Supplement, unless otherwise defined herein.

In accordance with Section 9.1(a) (*Duty to notify*) of the CDS Clearing Supplement, notice is hereby given of the following: [insert details of one or more of the relevant events, as set out in paragraphs (i), (ii) and (iii) of Section 9.1(a) (Duty to notify) of the CDS Clearing Supplement].

This notice (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this agreement or its formation) shall be governed by and construed in accordance with English law.

· ·		
Yours faithfully,		
[Clearing Member]		
Name:		
Title:		

70-41024903

Formatted: Line spacing: single

10219615309-v3

⁸ A single Credit Event Notice may be submitted for multiple trades in respect of the same Counterparty

File No. SR-LCH SA-2022-007 Page 215 of 439

PUBLIC

SCHEDULE

Credit Derivative Transaction Details

[Restructuring] Matched Pair ID	Trade ID	Reference Entity	Trade Date	Effective Date
[●]	[●]	[●]	[●]	[●]

APPENDIX XIII: CCM CLIENT TRANSACTION REQUIREMENTS

The following provisions (the "Mandatory Provisions") are to be incorporated into a CCM Client Transaction between a CCM and its CCM Client that corresponds to a CCM Client Cleared Transaction registered in the CCM Client Trade Account in the name of such CCM for such CCM Client. The terms of the corresponding CCM Client Cleared Transaction will be governed by the CDS Clearing Supplement. The CDS Clearing Supplement and these Mandatory Provisions have been drafted so as to complement each other.

LCH SA shall not be responsible for any loss suffered or expense incurred by a CCM or any CCM Client as a result of the inclusion in the CCM Client Transaction Documents of the requirements set-out in this Appendix XIII.

The Mandatory Provisions, when they are incorporated into any CCM Client Transaction Documents, shall be governed by and construed in accordance with the governing law applicable to such CCM Client Transaction Documents of which they form part, or if different and applicable, in accordance with such CCM Client Transaction Documents, the governing law applicable to transactions entered into under such CCM Client Transaction Documents. The Mandatory Provisions shall be subject to such dispute resolution mechanisms and procedures and such courts or other forum for hearing disputes as are applicable in respect of such CCM Client Transaction Documents of which they form part. Each CCM and its CCM Client to which the Mandatory Provisions apply will waive any right to object to any such choice of law or proceedings on the basis of *forum non conveniens*, that the governing law or forum is not specified on the face of this document or otherwise.

In this Appendix XIII:

"CCM Client Buyer" means a CCM Client that is party to a CCM Client Transaction as protection buyer;

"CCM Client Seller" means a CCM Client that is party to a CCM Client Transaction as protection seller;

"CCM Buyer/Matched Seller" means a CCM that is party to a CCM Client Transaction as protection buyer and to the corresponding CCM Client Cleared Transaction as protection seller; and

"CCM Seller/Matched Buyer" means a CCM that is party to a CCM Client Transaction as protection seller and to the corresponding CCM Client Cleared Transaction as protection buyer.

1. Defined Terms

Terms used in the Mandatory Provisions and not otherwise defined herein or in the 2014 ISDA Credit Derivatives Definitions shall have the meanings given to them in the CDS Clearing Supplement.

2. Terms of CCM Client Transactions

2.1 2014 ISDA Credit Derivatives Definitions

The definitions and provisions contained in the 2014 ISDA Credit Derivatives Definitions published by the International Swaps and Derivatives Association ("ISDA") (the "2014 ISDA Credit Derivatives Definitions"), are incorporated into each CCM Client Transaction.

2.2 Single Name CCM Client Transactions - Reference Obligation

With effect from the date on which the CCM Client Cleared Transaction corresponding to a Single Name CCM Client Transaction is registered in the TIW, such Single Name CCM Client Transaction shall be deemed to reference the Standard Reference Obligation, provided that if there is no Standard Reference Obligation, such Single Name CCM Client Transaction shall be deemed to reference the CDSClear Preferred Reference Obligation.

2.3 Single Name CCM Client Transactions - Updating Physical Settlement Matrix

With effect from the date on which the CCM Client Cleared Transaction corresponding to a Single Name CCM Client Transaction is registered in the TIW, such Single Name CCM Client Transaction shall be deemed to reference the Relevant Physical Settlement Matrix.

With effect from the close of business on any Matrix Re-versioning Date, any Single Name CCM Client Transaction referencing the Existing Matrix will be deemed to have been amended so as to reference the Revised Matrix.

2.4 Index CCM Client Transactions - Updating Eligible Index Versions

Upon the occurrence of a DTCC Re-versioning Date, any Index CCM Client Transaction shall be automatically amended so as to reference the portfolio of Reference Entities specified in the revised version of the index published by the Index Publisher and referenced in the TIW with effect from such DTCC Re-versioning Date.

2.5 Initial Payment Date

Notwithstanding anything to the contrary in the 2014 ISDA Credit Derivatives Definitions, if the Initial Payment Date specified in the CCM Client Transaction Documents in respect of any CCM Client Transaction is a date falling after the Clearing Day on which the Cleared Transactions related to such CCM Client Transaction are created by novation pursuant to Title III (*Clearing Operations*) of the CDS Clearing Rule Book, the Initial Payment Date in respect of such CCM Client Transaction shall be deemed to be the Transaction Business Day immediately following the Clearing Day on which the Cleared Transactions relating to such CCM Client Transaction are created.

2.6 ISDA 2019 NTCE Protocol

The Protocol Effectiveness Condition (as defined in the ISDA 2019 NTCE Protocol published by ISDA on 27 August 2019 and as further supplemented from time to time (the "2019 NTCE Protocol")) being satisfied, then from and including the Implementation Date (as defined in the 2019 Protocol) the amendments set forth in Schedule 1 (*Amendments*) of the 2019 NTCE Protocol shall apply to each Index CCM Client Transaction and each Single Name CCM Client Transaction which incorporates the 2014 ISDA Credit Derivatives Definitions and each such CCM Client Transaction shall be a Protocol Covered Transaction for the purposes of the 2019 NTCE Protocol provided that:

 the date on which the CCM Client Cleared Transaction corresponding to such CCM Client Transaction is registered in the TIW is prior to the Implementation Date;

- (ii) such CCM Client Transaction is not an Excluded Transaction (as defined in the 2019 NTCE Protocol) provided that if such CCM Client Transaction relates to more than one Reference Entity and would otherwise be a Protocol Covered Transaction but for the inclusion of any Affected Portion (as defined in the 2019 NTCE Protocol) therein, such Affected Portion shall be excluded for the purposes of this Mandatory Provision 2.6 and the remaining portion of such CCM Client Transaction shall be deemed to be a Protocol Covered Transaction;
- the Scheduled Termination Date in respect of such CCM Client Transaction is on or after the Implementation Date; and
- (iv) no Event Determination Date has occurred in respect of such CCM Client Transaction prior to the Implementation Date.

3. Additional CCM Client Transactions, Compression and Succession Events

3.1 Creation of Additional CCM Client Transactions

Immediately following:

- (a) the creation of Matched Pairs by LCH SA pursuant to Section 8.1 (*Creation of Matched Pairs*) of the CDS Clearing Supplement; or
- (b) the creation of Resulting Single Name Cleared Transactions pursuant to Section 4.4 (Re-couponing of Restructuring Cleared Transactions) of the CDS Clearing Supplement,

if a CCM Client Transaction has been specified to have been split into or replaced by two or more separate CCM Client Transactions in the TIW as a result of the creation of such Matched Pairs or Resulting Single Name Cleared Transactions, such CCM Client Transaction shall be split into or terminated and replaced by two or more (as applicable) corresponding CCM Client Transactions. The Floating Rate Payer Calculation Amount and Fixed Rate of each such CCM Client Transaction shall correspond to the Floating Rate Payer Calculation Amount and Fixed Rate specified in TIW for such CCM Client Transaction. In respect of CCM Client Transactions created as a result of the creation of Resulting Single Name Cleared Transactions, the Trade Date of such new CCM Client Transactions shall be the same as the Trade Date of the equivalent Resulting Single Name Cleared Transactions. Otherwise, each new CCM Client Transaction shall have the same terms as the original CCM Client Transaction.

3.2 Reversal of Creation of Additional CCM Client Transactions

If a CCM Client Transaction has been split into two or more CCM Client Transactions pursuant to Mandatory Provision 3.1 (*Creation of Additional CCM Client Transactions*) above and the relevant DC Credit Event Announcement that led to the creation of the Matched Pairs is reversed such that Section 5.5 (*Reversal of DC Credit Event Announcements*) of the CDS Clearing Supplement applies, then, subject to Section 10.2(a)(i) of the 2014 ISDA Credit Derivatives Definitions, any additional CCM Client Transactions created pursuant to Mandatory Provision 3.1 (*Creation of Additional CCM Client Transactions*) above shall be

deemed not to have been created and any Credit Event Notices delivered in connection with such CCM Client Transactions shall be deemed to be ineffective.

3.3 Compression of CCM Client Transactions

If two or more CCM Client Transactions are specified in TIW to have been compressed into a single CCM Client Transaction pursuant to Chapter 3 (*Compression*) of Title III (*Clearing Operations*) of the CDS Clearing Rule Book, such CCM Client Transactions shall be compressed into a single CCM Client Transaction with a Floating Rate Payer Calculation Amount equal to the aggregate Floating Rate Payer Calculation Amounts of the original CCM Client Transactions.

3.4 Succession Events and Cleared Transactions

If LCH SA takes any action with respect to a CCM Client Cleared Transaction pursuant to Section 4.5 (*Succession Events and Cleared Transactions*) of the CDS Clearing Supplement so as to give effect to a Succession Event, such action shall also be deemed to have been taken with respect to the corresponding CCM Client Transaction.

4. Notices

4.1 Validity of Notices

Save if and as expressly stated to the contrary in the Mandatory Provisions, any notice delivered by a CCM Client to its CCM in respect of a CCM Client Transaction (including, without limitation, a Credit Event Notice, Notice of Physical Settlement, Notice to Exercise Movement Option or NOPS Amendment Notice) at a time or in a manner in which the CCM would not be permitted to deliver such a notice to LCH SA (or to a relevant Matched Buyer or Matched Seller as designee of LCH SA (as applicable)) in respect of the corresponding CCM Client Cleared Transaction pursuant to the terms of the CDS Clearing Supplement shall be deemed not to have been delivered.

4.2 Credit Event Notices and NEMOs given via DTCC

(a) Credit Event Notices and NEMOs to be given via DTCC

Credit Event Notices and Notices to Exercise Movement Option shall be delivered by way of the relevant DTCC Notice Facility, save if and as expressly stated to the contrary in the Mandatory Provisions or otherwise agreed between the parties to the CCM Client Transaction. The deemed time of delivery of any such notices shall be as set out in the DTCC Rules from time to time.

(b) Credit Event Notices and NEMOs delivered in respect of corresponding CCM Client Cleared Transaction

In respect of a CCM Client Transaction, if:

 CCM Seller/Matched Buyer or CCM Buyer/Matched Seller delivers a valid Credit Event Notice or Notice to Exercise Movement Option in respect of the

corresponding CCM Client Cleared Transaction by way of the relevant DTCC Notice Facility; or

(ii) a Credit Event Notice or Notice to Exercise Movement Option is deemed to have been delivered in respect of the corresponding CCM Client Cleared Transaction pursuant to Section 7.3(b) (Credit Event Notices and NEMOs delivered in respect of CCM Client Transaction) of the CDS Clearing Supplement as a result of the receipt of a valid Credit Event Notice or Notice to Exercise Movement Option (as applicable) by way of the relevant DTCC Notice Facility in respect of the CCM Client Transaction between the other CCM of a Matched Pair and its CCM Client,

and a Credit Event Notice or Notice to Exercise Movement Option (as applicable) has not already been given in respect of such CCM Client Transaction in accordance with Mandatory Provision 4.4 (Communications Failure Event) such notice (or deemed notice) shall be deemed also to be a Credit Event Notice or Notice to Exercise Movement Option (as applicable) for the purposes of such CCM Client Transaction.

4.3 Consequences of DTCC Failure

If a DTCC Failure Event occurs, from (and including) the DTCC Failure Event Time to (but excluding) the DTCC Resolution Time:

- (a) Mandatory Provision 4.2(a) (Credit Event Notices and NEMOs to be given via DTCC) shall not apply and accordingly Credit Event Notices and Notices to Exercise Movement Option shall be delivered directly (and not via the relevant DTCC Notice Facility);
- (b) any notice delivered via the relevant DTCC Notice Facility prior to the DTCC Failure Event Time will be valid and will not be affected by such DTCC Failure Event; and
- (c) any notice delivered or purported to be delivered via the relevant DTCC Notice Facility at or following the DTCC Failure Event Time but prior to the DTCC Resolution Time will not be valid and effective.

Mandatory Provision 4.2(a) (*Credit Event Notices and NEMOs to be given via DTCC*) shall apply with effect from the DTCC Resolution Time and, accordingly, any notice thereafter delivered or purported to be delivered directly (and not via the relevant DTCC Notice Facility) will not be valid and effective.

4.4 Communications Failure Event

(a) Right to deliver Notices manually following Communications Failure Event

If a party is affected by a significant communications or information technology failure resulting in it being impossible or impractical for such party to deliver any Credit Event Notice in relation to an M(M)R Restructuring or any Notice to Exercise Movement Option via a relevant DTCC Notice Facility (a "Communications Failure Event") it may, notwithstanding Mandatory Provision 4.2(a) (Credit Event Notices and NEMOs to be

given via DTCC), deliver Credit Event Notices and Notices to Exercise Movement Option directly (and not via the relevant DTCC Notice Facility).

Such party shall deliver, together with any Credit Event Notice or Notice to Exercise Movement Option delivered by it directly, a notice signed by a senior officer (such as a managing director or equivalent) of such party certifying that it is affected by a Communications Failure Event (or, if such party is unable to deliver such notice in writing, orally by telephone).

(b) Notices to party affected by Communications Failure Event

For the avoidance of doubt, Mandatory Provision 4.2(a) (Credit Event Notices and NEMOs to be given via DTCC) shall continue to apply in respect of notices given by the party not affected by the Communications Failure Event to the party affected by the Communications Failure Event.

(c) Notification of Resolution of Communications Failure Event

As soon as reasonably practicable upon a party ceasing to be subject to a Communications Failure Event, it shall notify the other party accordingly and thereupon Mandatory Provision 4.2(a) (Credit Event Notices and NEMOs to be given via DTCC) shall apply and, accordingly, any notice thereafter delivered or purported to be delivered directly (and not via the relevant DTCC Notice Facility) will not be valid and effective.

(d) Duty to Mitigate

A party which is subject to a Communications Failure Event shall use reasonable endeavours to mitigate the operational impact on the other party of any Communications Failure Event, to cure such Communications Failure Event as soon as possible and to ensure that the circumstances giving rise to the relevant Communications Failure Event do not recur.

(e) Breach does not Invalidate Valid Notices

Without prejudice to any other rights or remedies of the parties, any breach by a party of the provisions of this Mandatory Provision 4.4 shall not cause any Credit Event Notice or Notice to Exercise Movement Option delivered otherwise than in accordance with the terms of the relevant CCM Client Transaction, which would otherwise be valid and effective, to be invalid or ineffective.

4.5 Uncertain Delivery

(a) Manual Notice permitted if Delivery of Notice in DTCC uncertain

Notwithstanding Mandatory Provision 4.2(a) (*Credit Event Notices and NEMOs to be given via DTCC*), where such notices are permitted to be delivered by means other than the relevant DTCC Notice Facility pursuant to this Mandatory Provision 4 (*Notices*), and a party is uncertain as to whether or not a Credit Event Notice or Notice

to Exercise Movement Option (as applicable) it attempted to deliver via a DTCC Notice Facility has:

- (i) actually been delivered; or
- (ii) was delivered prior to the DTCC Failure Time,

that party shall be entitled to deliver such a notice directly to the other party specifying that such notice is only to be effective to the extent that the other purported notice is not effective.

(b) Details to be provided of Uncertain Notice

If a party delivers a manual notice pursuant to Mandatory Provision 4.5(a) (*Manual Notice permitted if Delivery of Notice in DTCC uncertain*) above, such party shall be required to provide (together with such notice) sufficient details of the notice attempted to be given by way of the relevant DTCC Notice Facility so as to allow the other party to identify the communication concerned.

(c) DTCC Notice delivered successfully

If the first Credit Event Notice or Notice to Exercise Movement Option (as applicable) to which the manual notice delivered pursuant to Mandatory Provision 4.5(a) (*Manual Notice permitted if Delivery of Notice in DTCC uncertain*) above related was actually delivered successfully, any subsequent Credit Event Notice or Notice to Exercise Movement Option delivered shall be deemed not to have been delivered.

5. Determination of Credit Events and Succession Events

Notwithstanding any provision to the contrary:

- (a) the Calculation Agent shall not make any determination in respect of any matter which is or may be subject to resolution under Sections 3.5 (Successor Resolutions) or 3.6 (Substitute Reference Obligation Resolutions) of the DC Rules; and
- (b) neither party shall be entitled to deliver a Successor Notice. or, a Repudiation/Moratorium Extension Notice (other than Credit Event Notices in relation to an M(M)R Restructuring in accordance with the terms of any Restructuring CCM Client Transaction (including the Mandatory Provisions) and, where applicable, the DTCC Rules).

6. Timings for the Delivery of Manual Notices

The following provisions shall solely be applicable in respect of a CCM Client Transaction between a CCM Client Buyer and its CCM Seller/Matched Buyer:

6.1 Delivery of Manual Notices by CCM Client Buyer

For the purposes of the delivery by CCM Client Buyer of any notice in respect of a CCM Client Transaction which is permitted pursuant to the terms of such CCM Client Transaction (including the Mandatory Provisions) to be delivered manually (rather than via the relevant

DTCC Notice Facility), Section 1.38 (*Requirements Regarding Notices*) of the 2014 ISDA Credit Derivatives Definitions shall be amended so as to provide that, solely in respect of the final day on which such manual notice could validly be delivered pursuant to the terms of such CCM Client Transaction (including the Mandatory Provisions), any such notice shall be required to be delivered on or prior to 2:00 p.m. (Calculation Agent City time) in order to be effective.

A manual notice (including, without limitation, a Credit Event Notice or a Notice to Exercise Movement Option) delivered after 2:00 p.m. (Calculation Agent City time) on the final day on which such notice could validly be delivered pursuant to the terms of the relevant CCM Client Transaction (including the Mandatory Provisions) shall be deemed not to have been delivered.

6.2 Onward Delivery of Certain Notices by CCM Seller/Matched Buyer to Matched Seller

Any Credit Event Notice, Notice to Exercise Movement Option, Notice of Physical Settlement, NOPS Amendment Notice, any notice given pursuant to Section 8.2 (*Notice of Physical Settlement*) of the 2014 ISDA Credit Derivatives Definitions in respect of any Asset or Asset Package, any notice given pursuant to Section 9.7 (*Buy-in of Bonds Not Delivered*) of the 2014 ISDA Credit Derivatives Definitions or any notice given pursuant to Section 9.8 (*Alternative Procedures Relating to Loans Not Delivered*) of the 2014 ISDA Credit Derivatives Definitions which is permitted to be and is delivered manually by CCM Client Buyer to CCM Seller/Matched Buyer in respect of and pursuant to the terms of a CCM Client Transaction (including the Mandatory Provisions) shall not be effective unless and until CCM Seller/Matched Buyer effectively delivers the relevant equivalent notice to the relevant Matched Seller in respect of and pursuant to the terms of the corresponding Restructuring Cleared Transaction or Physically Settled Cleared Transaction, as applicable.

CCM Seller/Matched Buyer undertakes to deliver such a notice to the relevant Matched Seller within two hours of its receipt of the equivalent notice from CCM Client Buyer if such notice is received between 9:00 a.m. (Calculation Agent City time) and 4:00 p.m. (Calculation Agent City time) on a Calculation Agent City Business Day. Any such notice received by CCM Seller/Matched Buyer after 4:00 p.m. (Calculation Agent City time) on a Calculation Agent City Business Day shall be deemed to have been received at 9:00 a.m. (Calculation Agent City time) on the following Calculation Agent City Business Day. Any such notice received by CCM Seller/Matched Buyer before 9:00 a.m. (Calculation Agent City time) on a Calculation Agent City Business Day shall be deemed to have been received at 9:00 a.m. (Calculation Agent City time) on such Calculation Agent City Business Day. Any such notice delivered on a day that is not a Calculation Agent City Business Day shall be deemed to have been delivered at 9:00 a.m. (Calculation Agent City time) on the following Calculation Agent City Business Day.

The following provision shall solely be applicable in respect of a CCM Client Transaction between a CCM Client Seller and its CCM Buyer/Matched Seller:

6.3 Receipt of Certain Notices by CCM Buyer/Matched Seller deemed to be Receipt by CCM Client Seller

Any Credit Event Notice, Notice to Exercise Movement Option, Notice of Physical Settlement, NOPS Amendment Notice, any notice given pursuant to Section 8.2 (Notice of Physical

Settlement) of the 2014 ISDA Credit Derivatives Definitions in respect of any Asset or Asset Package, any notice given pursuant to Section 9.7 (Buy-in of Bonds Not Delivered) of the 2014 ISDA Credit Derivatives Definitions or any notice given pursuant to Section 9.8 (Alternative Procedures Relating to Loans Not Delivered) of the 2014 ISDA Credit Derivatives Definitions which is permitted to be and is delivered manually by the relevant Matched Buyer to CCM Buyer/Matched Seller in respect of and pursuant to the terms of a Restructuring Cleared Transaction or Physically Settled Cleared Transaction (as applicable) relating to a CCM Client Transaction between such CCM Buyer/Matched Seller and CCM Client shall be deemed to constitute simultaneous delivery by CCM Buyer/Matched Seller to CCM Client Seller of such notice in respect of such CCM Client Transaction.

CCM Buyer/Matched Seller undertakes to deliver such a notice to CCM Client Seller within two hours of its receipt of the equivalent notice from the relevant Matched Buyer if such notice is received between 9:00 a.m. (Calculation Agent City time) and 4:00 p.m. (Calculation Agent City time) on a Calculation Agent City Business Day. Any such notice received by CCM Buyer/Matched Seller after 4:00 p.m. (Calculation Agent City time) on a Calculation Agent City Business Day shall be deemed to have been received at 9:00 a.m. (Calculation Agent City time) on the following Calculation Agent City Business Day. Any such notice received by CCM Buyer/Matched Seller before 9:00 a.m. (Calculation Agent City time) on a Calculation Agent City Business Day shall be deemed to have been received at 9:00 a.m. (Calculation Agent City time) on such Calculation Agent City Business Day. Any such notice delivered on a day that is not a Calculation Agent City Business Day shall be deemed to have been delivered at 9:00 a.m. (Calculation Agent City time) on the following Calculation Agent City Business Day.

7. Physical Settlement

7.1 Fallback to Cash Settlement resulting from corresponding CCM Client Cleared Transaction

If a CCM notifies its CCM Client that the Physically Settled Cleared Transaction corresponding to their CCM Client Transaction is to be settled (in whole or in part) by Cash Settlement pursuant to Section 6 (*Physical Settlement*) of the CDS Clearing Supplement and such CCM Client Transaction has not already been settled by Physical Settlement, such CCM Client Transaction shall also be settled (in whole or in part, as applicable) by Cash Settlement and the Cash Settlement Amount and the Cash Settlement Date shall be the same as the Cash Settlement Amount and the Cash Settlement Date determined in respect of the corresponding Physically Settled Cleared Transaction.

In respect of the CCM Client Transaction between CCM Client Buyer and CCM Seller/Matched Buyer, if CCM Client Buyer has already Delivered the Deliverable Obligations to CCM Seller/Matched Buyer, CCM Seller/Matched Buyer shall redeliver equivalent Deliverable Obligations to CCM Client Buyer in whole (if Cash Settlement applies) or in part (if Partial Cash Settlement applies in which case CCM Seller/Matched Buyer shall retain a proportion of the Deliverable Obligations equal to the proportion of the Transaction to be settled by Physical Settlement).

7.2 Fallback to Cash Settlement in respect of Non-Deliverable Obligations

If, in respect of a CCM Client Transaction, Buyer is not permitted to Deliver one or more Deliverable Obligations (such Deliverable Obligations, the Non-Deliverable Obligations)

specified in the relevant Notice of Physical Settlement or NOPS Amendment Notice to Seller because:

- (a) the amount of such Deliverable Obligation is less than the relevant minimum denomination of such Deliverable Obligation; or
- (b) Seller is not a permitted transferee under such Deliverable Obligation (and, in the case of this sub-section (ii), such circumstance would not constitute an illegality or impossibility outside the control of a relevant party for the purposes of Section 9.1 (Partial Cash Settlement Due to Impossibility or Illegality) of the 2014 ISDA Credit Derivatives Definitions).

then it shall notify Seller accordingly describing in reasonable detail the relevant circumstances.

With effect from such notification, such occurrence shall be treated, in relation to such CCM Client Transaction, as an illegality or impossibility outside the control of a relevant party for the purpose of Section 9.1 (*Partial Cash Settlement Due to Impossibility or Illegality*) of the 2014 ISDA Credit Derivatives Definitions and "Cash Settlement" pursuant to the Partial Cash Settlement Terms shall be deemed to apply to such CCM Client Transaction with respect to the Non-Deliverable Obligations as though the Non-Deliverable Obligations were Undeliverable Obligations and the provisions set out in Mandatory Provision 7.3 (*Consequences of Cash Settlement*) below shall apply.

7.3 Consequences of Cash Settlement

If the circumstances set out in either Section 9.1 (*Partial Cash Settlement Due to Impossibility or Illegality*) of the 2014 ISDA Credit Derivatives Definitions or Mandatory Provision 7.2 (*Fallback to Cash Settlement in respect of Non-Deliverable Obligations*) apply to a CCM Client Transaction, then:

- (a) the Latest Permissible Physical Settlement Date in respect of such CCM Client Transaction will be deemed to be the first date on which the relevant Buyer or Seller effectively gave the relevant notice to the other pursuant to either Section 9.1 (Partial Cash Settlement Due to Impossibility or Illegality) of the 2014 ISDA Credit Derivatives Definitions or Mandatory Provision 7.2 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) above, as applicable, (and for these purposes, Section 9.5 (Latest Permissible Physical Settlement Date) of the 2014 ISDA Credit Derivatives Definitions shall not apply); and
- (b) where sub-paragraph (ii) of Mandatory Provision 7.2 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) applies, Indicative Quotations shall not be applicable.

7.4 Asset Package Delivery

If Asset Package Delivery is applicable in respect of a Physically Settled Cleared Transaction and an Asset to be delivered is a Non-Transferable Instrument or Non-Financial Instrument, then the Asset shall be deemed to be an amount of cash equal to the value determined by the

CDSClear Product Committee (which, notwithstanding anything to the contrary shall be the "Calculation Agent" for the purposes of Section 8.15 (Asset Market Value) of the 2014 IDSA Credit Derivatives Definitions).

If a CCM notifies its CCM Client that Asset Package Delivery is applicable in respect of the Physically Settled Cleared Transaction corresponding to their CCM Client Transaction and such CCM Client Transaction has not already been settled by Physical Settlement and the relevant Asset Package comprises Assets in the form of cash in the Settlement Currency (whether pursuant to Section 8.12(b)(v) of the 2014 ISDA Credit Derivatives Definitions or otherwise) then Asset Package Delivery shall also apply in respect of such CCM Client Transaction and

- (i) the Physical Settlement Amount in respect of such CCM Client Transaction shall be an amount equal to the Physical Settlement Amount minus the Asset Package Cash Settlement Amount determined in respect of the corresponding Physically Settled Cleared Transaction; and
- (ii) the only Assets to be Delivered in respect of such CCM Client Transaction shall be such Assets as are Delivered in respect of the corresponding Physically Settled Cleared Transaction

The following provisions shall solely be applicable in respect of a CCM Client Transaction between a CCM Client Buyer and its CCM Seller/Matched Buyer:

7.5 Delivery of Deliverable Obligations by CCM Client Buyer to CCM Seller/Matched Buyer

This Mandatory Provision 7.5 shall be applicable unless the CCM Client Buyer and CCM Seller/Matched Buyer agree that it shall not apply in respect of a specific CCM Client Transaction.

In respect of a CCM Client Transaction, the Delivery of any Deliverable Obligations (or if Asset Package Delivery is applicable, the Assets) to be Delivered by CCM Client Buyer to CCM Seller/Matched Buyer shall be deemed not to have occurred for the purposes of such CCM Client Transaction unless and until CCM Seller/Matched Buyer Delivers equivalent Deliverable Obligations or Assets to Matched Seller pursuant to the Physical Settlement of the corresponding Physically Settled Cleared Transaction.

Unless in respect of the corresponding Physically Settled Cleared Transactions:

- (a) Cash Settlement is applicable (in whole or in part);
- a Buy-in Period is applicable or Buy-in Price has been determined pursuant to Section
 9.7 (Buy-in of Bonds Not Delivered) of the 2014 ISDA Credit Derivatives Definitions;
- (c) a Deemed Buy-in Period is applicable pursuant to Section 6.8(b)(i) of the CDS Clearing Supplement or Section 6.8(b)(ii) of the CDS Clearing Supplement is applicable;
- (d) Section 9.8(i) of the 2014 ISDA Credit Derivatives Definitions is applicable or Matched Seller has required Matched Buyer to Deliver a Bond or Loan pursuant to Section 9.8(ii) of the 2014 ISDA Credit Derivatives Definitions; or

(e) Matched Seller has required Matched Buyer to Deliver a Deliverable Obligation pursuant to Section 9.9 (Alternative Procedures Relating to Assets Not Delivered) of the 2014 ISDA Credit Derivatives Definitions,

(each such event, a "Non-delivery Event"), CCM Seller/Matched Buyer undertakes to Deliver the Deliverable Obligations (or such portion of the Deliverable Obligations not affected by a Non-delivery Event) or, if Asset Package Delivery is applicable, the Assets, to Matched Seller not later than the first Business Day after the day on which a trade in such Deliverable Obligation(s) or Assets would, if effected on the day on which CCM Seller/Matched Buyer received the Deliverable Obligations or Assets from CCM Client Buyer or on which the relevant Non-delivery Event ceased to apply, as applicable, (or if such day is not a Business Day, the following Business Day), be settled in accordance with then current market practice of such Deliverable Obligations or Assets, as the case may be, as determined by the Calculation Agent after consultation with the parties.

For the purposes of Article VIII (*Terms relating to Physical Settlement*) of the 2014 ISDA Credit Derivatives Definitions, the Physical Settlement Amount shall not be payable by CCM Seller/Matched Buyer to CCM Client Buyer in respect of the CCM Client Transaction until the Physical Settlement of the corresponding Physically Settled Cleared Transaction has occurred.

7.6 Buy-in of Bonds not Applicable

Section 9.7 (Buy-in of Bonds Not Delivered) of the 2014 ISDA Credit Derivatives Definitions shall not apply.

7.7 Buyer's Right to Deliver suspended during Buy-in Period

If CCM Seller/Matched Buyer notifies CCM Client Buyer in respect of a CCM Client Transaction that (i) it has received a Buy-in Notice from Matched Seller in respect of the Matched Contracts of the related Settlement Matched Pair or (ii) it has been notified by Matched Seller in respect of the Matched Contracts of the related Settlement Matched Pair pursuant to Section 6.8(b) (Deemed Buy-in of Bonds resulting from CCM Client Transaction of Matched Seller) of the CDS Clearing Supplement that such Matched Seller has received a Buy-in Notice from its CCM Client in respect of the CCM Client Transaction between such Matched Seller and its CCM Client, such notice from CCM Seller/Matched Buyer specifying:

- i) the Buy-in Date;
- ii) the Relevant Bonds; and
- iii) the Outstanding Principal Balance thereof sought to be bought-in,

then CCM Client Buyer's right to Deliver the specified Relevant Bonds shall be suspended until the fourth Business Day (inclusive) following such Buy-in Date.

7.8 Buy-in of Bonds in respect of Matched Contracts of the Settlement Matched Pair

Provided that Physical Settlement has not already occurred in respect of a CCM Client Transaction, if CCM Seller/Matched Buyer notifies CCM Client Buyer that a Buy-in Price has

been (i) determined or (ii) deemed to be determined pursuant to Section 6.8(b) (*Deemed Buyin of Bonds resulting from CCM Client Transaction of Matched Seller*) of the CDS Clearing Supplement, in respect of Relevant Bonds for the purposes of the Matched Contracts of the related Settlement Matched Pair, then on the Buy-in Effective Date:

- i) CCM Client Buyer will be deemed to have Delivered to CCM Seller/Matched Buyer an Outstanding Principal Balance of the Deliverable Obligations equal to the Outstanding Principal Balance of the Deliverable Obligations Delivered or deemed to be Delivered by CCM Seller/Matched Buyer to Matched Seller in respect of the Matched Contracts; and
- ii) the Physical Settlement Amount to be paid by CCM Seller/Matched Buyer to CCM Client Buyer in respect of this CCM Client Transaction shall be reduced (but not below zero) by an amount equal to the amount by which the Physical Settlement Amount to be paid to CCM Seller/Matched Buyer by Matched Seller in respect of the Matched Contracts is to be reduced.

CCM Seller/Matched Buyer shall notify CCM Client Buyer of such Outstanding Principal Balance of the Deliverable Obligations and such Physical Settlement Amount reduction for the purposes of i) and ii) above and of the Buy-in Effective Date.

If CCM Client Buyer has already Delivered Deliverable Obligations to CCM Seller/Matched Buyer, CCM Seller/Matched Buyer shall redeliver equivalent Deliverable Obligations to CCM Client Buyer and such equivalent Deliverable Obligations shall have an Outstanding Principal Balance equal to the Outstanding Principal Balance of the Deliverable Obligation(s) specified by CCM Seller/Matched Buyer in the above notice and deemed to have been Delivered by CCM Client Buyer to CCM Seller/Matched Buyer pursuant to this Mandatory Provision 7.8.

If "60 Business Day Cap on Settlement" is applicable to the CCM Client Transaction and the Termination Date of the Affected Portion of the Matched Contracts of the Settlement Matched Pair that corresponds to such CCM Client Transaction is deferred pursuant to Section 9.10(a) (Cap on Settlement) of the 2014 ISDA Credit Derivatives Definitions and Section 6.8(c) of the CDS Clearing Supplement then the Termination Date of an equal portion of the CCM Client Transaction shall be deferred until the same date.

7.9 Alternative Procedures relating to Loans – Seller Right to Select

Section 9.8(ii) of the 2014 ISDA Credit Derivatives Definitions shall not apply.

7.10 Alternative Procedures relating to Loans – Seller designates alternative Loan or Bond

In respect of a CCM Client Transaction, CCM Seller/Matched Buyer shall notify CCM Client Buyer if it has purchased Bond(s) and/or Loan(s) pursuant to Section 9.8(ii) of the 2014 ISDA Credit Derivatives Definitions on the instructions of the Matched Seller in respect of the corresponding CCM Client Cleared Transaction.

Following such notification, such Bond(s) and/or Loan(s) shall be deemed to have been Delivered by CCM Client Buyer to CCM Seller/Matched Buyer in Physical Settlement of the CCM Client Transaction and the Physical Settlement Amount payable by CCM Seller/Matched

Buyer to CCM Client Buyer shall be reduced (but not below zero) by an amount equal to the price at which such Bond(s) and or Loan(s) were purchased.

CCM Seller/Matched Buyer shall provide CCM Client Buyer with details of such Bond(s) and/or Loan(s) and the related purchase price(s) in such notice.

If CCM Client Buyer has already Delivered Deliverable Obligations to CCM Seller/Matched Buyer, CCM Seller/Matched Buyer shall redeliver equivalent Deliverable Obligations to CCM Client Buyer and such equivalent Deliverable Obligations shall have an Outstanding Principal Balance equal to the Outstanding Principal Balance of the Bond(s) and/or Loan(s) specified by CCM Seller/Matched Buyer in the above notice and deemed to have been Delivered by CCM Client Buyer to CCM Seller/Matched Buyer pursuant to this Mandatory Provision 7.10.

7.11 Alternative Procedures relating to Assets Not Delivered

Section 9.9 (*Alternative Procedures Relating to Assets Not Delivered*) of the 2014 ISDA Credit Derivatives Definitions shall not apply.

7.12 Alternative Procedures relating to Assets Not Delivered – Seller designates alternative Deliverable Obligation(s)

In respect of a CCM Client Transaction, CCM Seller/Matched Buyer shall notify CCM Client Buyer if it has purchased one or more Deliverable Obligations pursuant to Section 9.9 (*Alternative Procedures Relating to Assets Not Delivered*) of the 2014 ISDA Credit Derivatives on the instructions of the Matched Seller in respect of the corresponding CCM Client Cleared Transaction.

Following such notification, such Deliverable Obligations shall be deemed to have been Delivered by CCM Client Buyer to CCM Seller/Matched Buyer in Physical Settlement of the CCM Client Transaction and the Physical Settlement Amount payable by CCM Seller/Matched Buyer to CCM Client Buyer shall be reduced (but not below zero) by an amount equal to the price at which such Deliverable Obligations were purchased.

CCM Seller/Matched Buyer shall provide CCM Client Buyer with details of such Deliverable Obligation(s) and the related purchase price(s) in such notice.

If CCM Client Buyer has already Delivered Deliverable Obligations to CCM Seller/Matched Buyer, CCM Seller/Matched Buyer shall redeliver equivalent Deliverable Obligations to CCM Client Buyer and such equivalent Deliverable Obligations shall have an Outstanding Principal Balance equal to the Outstanding Principal Balance of the Deliverable Obligation(s) specified by CCM Seller/Matched Buyer in the above notice and deemed to have been Delivered by CCM Client Buyer to CCM Seller/Matched Buyer pursuant to this Mandatory Provision 7.12.

The following provisions shall solely be applicable in respect of a CCM Client Transaction between CCM Client Seller and CCM Buyer/Matched Seller:

7.13 Delivery of Deliverable Obligations to CCM Client Seller

This Mandatory Provision 7.13 shall be applicable unless the CCM Client Seller and CCM Buyer/Matched Seller agree that it shall not apply in respect of a specific CCM Client Transaction.

Subject to the proviso below, in respect of a CCM Client Transaction, the Delivery of any Deliverable Obligations (or if Asset Package Delivery is applicable, the Assets) to be Delivered by CCM Buyer/Matched Seller to CCM Client Seller shall be deemed to have occurred for the purposes of such CCM Client Transaction upon receipt by CCM Buyer/Matched Seller of the Deliverable Obligations or Assets in respect of the Physical Settlement of the related Physically Settled Cleared Transaction; provided, however, that if the CCM Client Transaction is to be settled by Cash Settlement (in whole or in part) pursuant to Section 9.1 (Partial Cash Settlement Due to Impossibility or Illegality) of the 2014 ISDA Credit Derivatives Definitions or Mandatory Provision 7.2 (Fallback to Cash Settlement in respect of Non-Deliverable Obligations) then such portion of the CCM Client Transaction which is to be settled by Cash Settlement shall not be deemed to be settled until such Cash Settlement occurs.

Provided that Cash Settlement is not applicable, CCM Buyer/Matched Seller undertakes to Deliver the Deliverable Obligations or Assets to CCM Client Seller not later than the first Business Day after the day on which a trade in such Deliverable Obligations or Assets would, if effected on the day on which CCM Buyer/Matched Seller received the Deliverable Obligations or Assets from Matched Buyer (or if such day is not a Business Day, the following Business Day), be settled in accordance with then current market practice of such Deliverable Obligations or Assets, as the case may be, as determined by the Calculation Agent after consultation with the parties.

7.14 Alternative Procedures relating to Loans – Seller Right to Select

Section 9.8(ii) of the 2014 ISDA Credit Derivatives Definitions shall not apply.

7.15 Alternative Procedures relating to Loans in respect of Matched Contracts

In respect of a CCM Client Transaction, if CCM Buyer/Matched Seller notifies CCM Client Seller that a Bond or Loan has been deemed specified in a NOPS Amendment Notice in respect of the Matched Contracts of the Settlement Matched Pair that corresponds to such CCM Client Transaction pursuant to (i) Sections 9.8(i) or (ii) of the 2014 ISDA Credit Derivatives Definitions or (ii) Section 6.9 (Alternative Procedures Relating to Loans Not Delivered – Buyer Right to Deliver) of the CDS Clearing Supplement, then for the purposes of the Matched Contracts of the related Settlement Matched Pair—such Bond or Loan shall be deemed to have been specified in a NOPS Amendment Notice in respect of such CCM Client Transaction and such NOPS Amendment Notice will be effective notwithstanding the fact that it is deemed specified after the Physical Settlement Date.

Formatted: Font: Not Italic

7.16 Alternative Procedures relating to Assets Not Delivered

Section 9.9 (*Alternative Procedures relating to Assets Not Delivered*) of the 2014 ISDA Credit Derivatives Definitions shall not apply.

7.17 Alternative Procedures relating to Assets Not Delivered

In respect of a CCM Client Transaction, if CCM Buyer / Matched Seller notifies CCM Client Seller that a Deliverable Obligation has been deemed specified in a NOPS Amendment Notice in respect of the Matched Contracts of the Settlement Matched Pair that corresponds to such CCM Client Transaction pursuant to Section 9.9 (Alternative Procedures Relating to Assets Not Delivered) of the 2014 ISDA Credit Derivatives Definitions then for the purposes of the Matched Contracts of the related Settlement Matched Pair such Deliverable Obligation shall be deemed to have been specified in a NOPS Amendment Notice in respect of such CCM Client Transaction and such NOPS Amendment Notice will be effective notwithstanding the fact that it is deemed specified after the Physical Settlement Date.

7.18 Alternative Procedures and Cap on Settlement

If "60 Business Day Cap on Settlement" is applicable to the CCM Client Transaction and the Termination Date of the Affected Portion of each of the Matched Contracts of the Settlement Matched Pair that corresponds to such CCM Client Transaction is deferred pursuant to Section 9.10(b) (Cap on Settlement) of the 2014 ISDA Credit Derivatives Definitions then the Termination Date of an equal portion of the CCM Client Transaction shall be deferred until the same date.

8. Self Referencing Transactions

8.1 Section 11.4 (Merger of Reference Entity and Seller) of the 2014 ISDA Credit Derivatives
Definitions

Section 11.4 (Merger of Reference Entity and Seller) of the 2014 ISDA Credit Derivatives Definitions shall not apply.

8.2 Notification of Self Referencing Transactions

In respect of any Single Name CCM Client Transaction, the CCM Client shall, unless prohibited from so doing by applicable law, notify the CCM as soon as reasonably practicable if:

- the CCM Client is or consolidates or amalgamates with, or merges into, or transfers all or substantially all of its assets to, the Reference Entity in relation to such Single Name CCM Client Transaction or enters into any agreement in respect of any of the foregoing;
- ii) the CCM Client and the Reference Entity in relation to such Single Name CCM Client Transaction are or become Affiliates; or
- iii) in respect of a Restructuring CCM Client Transaction, the CCM Client is or becomes the Reference Entity in relation to such Restructuring CCM Client Transaction as a result of the occurrence of the relevant M(M)R Restructuring.

8.3 Termination of Self Referencing Transactions

A CCM Client Transaction shall be terminated (unless it has already been terminated) at the same time as the termination of the corresponding Single Name Cleared Transaction pursuant to Section 9.2 (*Occurrence of Client Self Referencing Transactions*) of the CDS Clearing Supplement and by reference to the price at which such Single Name Cleared Transaction is terminated and an amount will be payable:

- (a) if the CCM receives an amount from LCH SA in relation to such Single Name Cleared Transaction pursuant to Section 9.2 (Occurrence of Client Self Referencing Transactions), by the CCM to the CCM Client equal to such amount and on the Business Day following receipt by the CCM of such amount from LCH SA; and
- (b) if the CCM is obliged to pay an amount to LCH SA in relation to such Single Name Cleared Transaction pursuant to Section 9.2 (Occurrence of Client Self Referencing Transactions), by the CCM Client to the CCM equal to such amount and on the later of (I) the Business Day prior to the day on which the CCM is obliged to pay such amount to LCH SA and (II) the Business Day following the Business Day on which the CCM gives notices to the CCM Client of the relevant amount.

8.4 Costs of Terminating Self Referencing Transactions

Without prejudice to any other indemnity agreed between the CCM and the CCM Client in relation to CCM Client Transactions, the CCM Client agrees to indemnify and hold harmless the CCM from and against all costs and expenses that the CCM is obliged to bear pursuant to Section 9.2 (Occurrence of Client Self Referencing Transactions) of the CDS Clearing Supplement.

8.5 Compression of Self Referencing Transactions

Where the CCM acts as Matched Buyer and Matched Seller in respect of fungible Single Name Cleared Transactions that have a corresponding CCM Client Transaction in respect of which CCM Client has given notice to CCM pursuant to Mandatory Provision 8.2 (Notification of Self Referencing Transactions) and the relevant Single Name Cleared Transactions are compressed pursuant to Section 9.2 (Occurrence of Client Self Referencing Transactions) of the CDS Clearing Supplement, the CCM Client will be deemed to have submitted to CCM a request to compress the corresponding CCM Client Transactions.

9. Calculation Agent

9.1 Appointment of Calculation Agent

The Calculation Agent in respect of any CCM Client Transaction shall be the CCM.

9.2 Calculations and Determinations of Calculation Agent

In the event that the Calculation Agent is entitled or required to make any calculation or determination in respect of a CCM Client Transaction in respect of a matter that has already been or will be determined in respect of and pursuant to the terms of the corresponding CCM Client Cleared Transaction, the Calculation Agent in respect of the CCM Client Transaction

File No. SR-LCH SA-2022-007 Page 233 of 439

PUBLIC

shall be obliged to make the same calculation or determination in respect of such CCM Client Transaction as the determination in respect of the corresponding CCM Client Cleared Transaction (including, without limitation, any determination of any Cash Settlement Amount payable in respect of the CCM Client Transaction).

10. Amendments

The Mandatory Provisions may be amended from time to time pursuant to Section 11 (Amendments) of the CDS Clearing Supplement. The parties agree that any amendments made to the Mandatory Provisions in accordance with Section 11 (Amendments) of the CDS Clearing Supplement shall be deemed to apply automatically to the CCM Client Transaction(s) with effect from the date of such amendment to the Mandatory Provisions.

Formatted: Font: +Body (Calibri), 11 pt

File No. SR-LCH SA-2022-007 Page 234 of 439

Formatted: Header, Tab stops: Not at 2.54 cm + 7.7 cm + 15.4 cm

PART C

CDS CLEARING SUPPLEMENT

FOR CREDIT INDEX SWAPTIONS

CONTENTS

SECTION PAGE			
1.	GENERAL PROVISIONS	235	
2.	TERMS OF CLEARED TRANSACTIONS	246	
3.	PAYMENTS	250	
4.	CREDIT EVENTS AND SUCCESSION EVENTS	251	
5.	RESTRUCTURING	252	
6.	EXERCISE MATCHED PAIRS	258	
7.	SETTLEMENT	259	
8.	DELIVERY OF NOTICES	271	
9.	MATCHED PAIR DESIGNATIONS	271	
10.	MANDATORY PROVISIONS FOR CCM CLIENT TRANSACTIONS	276	
11.	AMENDMENTS	277	
12.	FORMS OF NOTICES	277	
13.	EXCLUSION OF LIABILITY	277	
14.	DISPUTE RESOLUTION	277	Field Code Changed

Formatted: Header, Tab stops: Not at 2.54 cm + 7.7 cm + 15.4 cm

GENERAL PROVISIONS

1.1 Incorporation of Defined Terms

1.

Capitalised terms used in this CDS Clearing Supplement and not otherwise defined herein shall have the meaning given pursuant to the Index Swaption Cleared Transaction Confirmation, the 2014 ISDA Credit Derivatives Definitions, the 2006 Definitions or the CDS Clearing Rule Book, as applicable. In the case of any such terms defined in the CDS Clearing Rule Book, such terms shall be interpreted in accordance with the governing law specified therefor in the CDS Clearing Rule Book.

1.2 Terms defined in the CDS Clearing Supplement

For the purposes of the CDS Clearing Documentation, the following capitalised terms shall, unless otherwise specified, have the respective meanings set out below:

2006 Definitions: The 2006 ISDA Definitions published by ISDA.

2014 ISDA Credit Derivatives Definitions: The 2014 ISDA Credit Derivatives Definitions published by ISDA.

Abandon: The Abandonment of an Exercise Cleared Transaction.

Abandonment: The delivery of a valid Abandonment Notice by Swaption Buyer (or deemed delivery pursuant to Section 6.3 (*Exercise and Abandonment by way of EEP*) or Section 6.4 (*Delegation by Clearing Members to Clients*) below) in respect of the Exercise Cleared Transactions of an Exercise Matched Pair.

Abandonment Notice: The notice of abandonment given by Swaption Buyer to Swaption Seller in accordance with Section 6.6 (*Abandonment of Exercise Cleared Transactions*).

CCM Client Cleared Transaction: A Cleared Transaction between a CCM and LCH SA registered in a CCM Client Trade Account of a CCM.

CCM Client Communications Failure Event: As set out at Mandatory Provision 5.7(b) (*CCM Client Communications Failure Event*).

CCM Client Transaction: A Transaction between a CCM and a CCM Client which is on the same economic terms as its corresponding CCM Client Cleared Transaction.

CCM Client Transaction Documents: The documentation entered into by a CCM and its CCM Client to document a CCM Client Transaction.

CDS Buyer: The Fixed Rate Payer under the relevant Underlying Index Transaction.

CDSClear Preferred Reference Obligation: This term shall have the meaning set out in Section 4 of the Procedures.

CDSClear Product Committee: A committee composed of representatives of LCH SA and representatives of Clearing Members for the purposes of carrying-out the tasks apportioned to it by the CDS Clearing Documentation as further described in the terms of reference for

File No. SR-LCH SA-2022-007 Page 236 of 439

such committee, agreed in consultation with the Clearing Members and as may be amended from time to time in consultation with the Clearing Members.

CDS Clearing Rule Book: The document entitled "CDS Clearing Rule Book" published by LCH SA, as amended from time to time.

CDS Seller: The Floating Rate Payer under the relevant Underlying Index Transaction.

CDX Swaption Standard Terms Supplement: The CDX Untranched Transactions Swaption Standard Terms Supplement as published on 22 September 2014 by Markit North America, Inc. and as amended by this CDS Clearing Supplement.

CEN Triggering Period: In relation to any M(M)R Restructuring, the period during which the parties to the Swaption Restructuring Cleared Transaction of a Swaption Restructuring Matched Pair may deliver a Credit Event Notice in relation to the relevant M(M)R Restructuring, such period starting at 9:00 a.m. on and including the earlier to occur of:

- (a) the Relevant City Business Day following publication of the related Final List; and
- (b) the fifth calendar day following the No Auction Announcement Date,

and ending on and including the Exercise Cut-off Date applicable to CDS Buyer or CDS Seller (as applicable).

Clearing Member Communications Failure Event: As set out at 6.10 (Clearing Member Communications Failure Event).

Clearing Member Notice: A Swaption Clearing Member Notice or a Swaption Restructuring Clearing Member Notice.

Clearing Member Portal Account: The account of a Clearing Member established in the LCH Portal for, amongst other things, the purposes of the Exercise and Abandonment of Exercise Cleared Transactions.

Client Portal Account: The account of a Client established in the LCH Portal for the purposes of, amongst other things, the Exercise and Abandonment of Exercise Cleared Transactions.

Client Portal Account Number: The unique account number assigned by LCH SA to a Client Portal Account.

Component Transaction: As defined in the Index Cleared Transaction Confirmation.

Compression Cut-off Date: The last date on which a Clearing Member may submit a request for an Index Swaption Cleared Transaction to be compressed pursuant to the ad hoc compression methodology and on which a daily automatic compression cycle will be run by LCH SA, in each case in accordance with Chapter 3 (Compression) of Title III (Clearing Operations) of the CDS Clearing Rule Book and Section 5 of the Procedures, being:

the date falling one Transaction Business Day prior to the Novation Cut-off Date in respect of the relevant Eligible Underlying Index Transaction Version; or

File No. SR-LCH SA-2022-007 Page 237 of 439

(b) following the occurrence of DC Credit Event Announcement of an M(M)R Restructuring (and prior to the creation of Swaption Restructuring Matched Pairs pursuant to Section 5.1 (Creation and Notification of Swaption Restructuring Matched Pairs)), the earlier of (A) the date of publication of the relevant Initial List (as defined in the DC Rules), (B) the date falling two Transaction Business Days prior to the relevant date on which the related SRMP Notification Deadline falls and (C) such other date falling between the dates in (A) and (B), as determined by LCH SA in consultation with the CDSClear Product Committee and notified to the relevant Clearing Members prior to such date.

For the avoidance of doubt, where a Clearing Member has specified automatic compression on a weekly basis then the last such weekly automatic compression cycle performed by LCH SA will be performed on the Clearing Day falling on the Thursday on or before the relevant Compression Cut-off Date.

DC Restructuring Announcement Date: The date on which the DC Credit Event Announcement of an M(M)R Restructuring is made, provided that where such DC Credit Event Announcement is made after 6.30 p.m. on a Transaction Business Day or on a day which is not a Transaction Business Day, the DC Restructuring Announcement Date will be the first following Transaction Business Day.

Dispute: This term shall have the meaning set out in the CDS Dispute Resolution Protocol.

DTCC Rules: The "Operating Procedures", as published by DTCC and as amended from time to time.

EEP Controls: As set out at Section 6.3 (Exercise and Abandonment by way of EEP).

EEP Failure Event: As set out at Section 6.5 (Notification of EEP Failure and Resolution).

EEP Failure Event Time: As set out at Section 6.5 (Notification of EEP Failure and Resolution).

EEP Resolution Time: As set out at Section 6.5 (*Notification of EEP Failure and Resolution*).

Electronic Exercise Platform or EEP: A platform made available by LCH SA for the Exercise and Abandonment of Exercise Cleared Transactions through the submission of Option Intents.

Eligible Underlying Index Transaction Version: This term shall have the meaning set out in Section 4 of the Procedures.

EMP Creation Period: The period from (and including) the final Transaction Business Day of the calendar week immediately preceding the week in which the Expiration Date falls to (but excluding) the Transaction Business Day immediately preceding the Expiration Date.

EMP Notification Deadline: In respect of a notification by LCH SA to the relevant Clearing Members in relation to Exercise Matched Pairs, 10.00 p.m. on the Transaction Business Day immediately prior to the relevant Expiration Date.

File No. SR-LCH SA-2022-007 Page 238 of 439

Exercise: The delivery of a valid Exercise Notice by Swaption Buyer (or deemed delivery pursuant to Section 6.3 (Exercise and Abandonment by way of EEP) or Section 6.4 (Delegation by Clearing Members to Clients) below) in respect of the Exercise Cleared Transactions of an Exercise Matched Pair.

Exercise Cleared Transaction: Each Index Swaption Cleared Transaction (including each Swaption Restructuring Cleared Transaction, as applicable) forming part of an Exercise Matched Pair pursuant to Section 6.2 (Creation of Exercise Cleared Transactions).

Exercise Delegation Beneficiary: The Client of a Clearing Member designated by such Clearing Member pursuant to Section 6.4 (Delegation by Clearing Members to Clients) below as being entitled to Exercise and Abandon Exercise Cleared Transactions on such Clearing Member's behalf.

Exercise Matched Pair: A pairing comprised of a Matched Buyer and a Matched Seller in respect of Exercise Cleared Transactions created by LCH SA under Section 6.1 (Creation and Notification of Exercise Matched Pairs) for the purposes of the exercise (or non-exercise) by Swaption Buyer of the relevant Exercise Cleared Transactions.

Exercise Matched Pair Amount: In respect of an Exercise Matched Pair, the amount of the Swaption Notional Amount(s) of the relevant Index Swaption Cleared Transactions (or Swaption Restructuring Cleared Transactions) from which the Exercise Matched Pair (and the Exercise Cleared Transactions of such Exercise Matched Pair) are created which is allocated by LCH SA to such Exercise Matched Pair under Section 6.1 (Creation and Notification of Exercise Matched Pairs).

Exercise Notice: The notice of exercise (in whole or in part) given by Swaption Buyer to Swaption Seller in accordance with Section 13.2 (Procedure for Exercise) of the 2006

Exercise Percentage: The amount (expressed as a percentage) of the Swaption Notional Amount of an Exercise Cleared Transaction that is exercised on the Expiration Date of an Exercise Cleared Transaction.

First Novation Date: In respect of:

- an Original Transaction of a Swaption Type for which a Novation Cut-off Date has (a) previously occurred as a result of a DC Credit Event Announcement of an M(M)R Restructuring which is applicable to a Component Transaction of the relevant Underlying Index Transaction for such Swaption Type, the Transaction Business Day following the latest possible Exercise Cut-off Date for the relevant M(M)R Restructuring or such other date on which LCH SA determines in consultation with the CDSClear Product Committee that LCH SA will begin to again accept Original Transactions in respect of which the Underlying Index Transaction references the relevant Eligible Underlying Index Transaction Version for clearing pursuant to the CDS Clearing Documentation: and
- an Original Transaction (other than in the circumstances set-out in (a) above), the first date on which LCH SA will accept Original Transactions in respect of which the

File No. SR-LCH SA-2022-007 Page 239 of 439

Underlying Index Transaction references the relevant Eligible Underlying Index Transaction Version for clearing pursuant to the CDS Clearing Documentation.

First Restructuring: As set out at Section 5.1 (Creation and Notification of Swaption Restructuring Matched Pairs).

Force Submission: This term shall have the meaning set out in Section 5 of the Procedures.

Index Cleared Transaction: As set out in Part B of the CDS Clearing Supplement.

Index Cleared Transaction Confirmation: As set out in Part B of the CDS Clearing Supplement.

Index Swaption Cleared Transaction: A Cleared Transaction, the terms of which are as evidenced by an Index Swaption Cleared Transaction Confirmation, which gives Swaption Buyer the right to enter into a specified Underlying Index Transaction with Swaption Seller.

Index Swaption Cleared Transaction Confirmation: For:

- any Index Swaption Cleared Transaction in respect of which the Underlying Index Transaction references a Markit iTraxx® Europe Index Series, the form of confirmation which incorporates the iTraxx® Swaption Standard Terms Supplement; and
- (b) any Index Swaption Cleared Transaction in respect of which the Underlying Index Transaction references a Markit CDX™ Index Series, the form of confirmation which incorporates the CDX Swaption Standard Terms Supplement,

as completed by reference to the relevant Transaction Data (or, in each case, such other form of confirmation as may be adopted in respect of any Swaption Type in accordance with Section 1.2.2 (Modification) of the CDS Clearing Rule Book).

ISDA: The International Swaps and Derivatives Association, Inc. and any successor thereto.

iTraxx® Swaption Standard Terms Supplement: The iTraxx® Europe Untranched Transactions Swaption Standard Terms Supplement, as published on 20 March 2017 by Markit Indices Limited and as amended by this CDS Clearing Supplement.

LCH Portal: As set out in Section 5 of the Procedures.

Mandatory Provisions: As set-out in Appendix VIII (CCM Client Transaction Requirements).

Matched Buyer: A Swaption Buyer comprised in a Swaption Restructuring Matched Pair or an Exercise Matched Pair (as applicable).

Matched Buyer Contract: A Swaption Restructuring Cleared Transaction or an Exercise Cleared Transaction (as applicable) between a Matched Buyer and LCH SA which is the subject of a Matched Pair having the Swaption Restructuring Matched Pair Amount or the Exercise Matched Pair Amount, as the case may be, relating to that Matched Pair.

Matched Contract: A Matched Seller Contract or a Matched Buyer Contract, as applicable.

File No. SR-LCH SA-2022-007 Page 240 of 439

Matched Pair: An Exercise Matched Pair or a Swaption Restructuring Matched Pair (as applicable).

Matched Pair Amount: An Exercise Matched Pair Amount or a Swaption Restructuring Matched Pair Amount (as applicable).

Matched Seller: A Swaption Seller comprised in a Swaption Restructuring Matched Pair or an Exercise Matched Pair (as applicable).

Matched Seller Contract: A Swaption Restructuring Cleared Transaction or an Exercise Cleared Transaction between a Matched Seller and LCH SA which is the subject of a Matched Pair having the Swaption Restructuring Matched Pair Amount or the Exercise Matched Pair Amount, as the case may be, relating to that Matched Pair.

NEMO Triggering Period: The period starting at 9:00 a.m. on the date that is the Movement Option Cut-off Date for the relevant M(M)R Restructuring and ending at the last time for delivery of a valid Notice to Exercise Movement Option under the terms of the 2014 ISDA Credit Derivatives Definitions.

Notification Deadline: In respect of an Exercise Matched Pair, the EMP Notification Deadline and in respect of a Swaption Restructuring Matched Pair, the SRMP Notification Deadline.

Novation Cut-off Date: The date with effect from which LCH SA will no longer accept for novation Original Transactions in respect of which the Underlying Index Transaction references a particular Eligible Underlying Index Transaction Version, being:

- following the occurrence of a DC Credit Event Announcement in respect of an M(M)R Restructuring which would be applicable to a Component Transaction of such Underlying Index Transaction, the earlier of:
 - a date determined by LCH SA in consultation with the CDSClear Product Committee, which shall not be earlier than the DC Restructuring Announcement Date: and
 - the calendar day immediately following the No Auction Announcement Date; (ii)
- following the occurrence of a DC Credit Event Announcement in respect of a Credit Event (other than an M(M)R Restructuring) which would be applicable to a Component Transaction of such Underlying Index Transaction, the calendar day following the DC Announcement Coverage Cut-off Date;
- (c) the date falling 1 (one) Transaction Business Day prior to the Expiration Date for the Original Transaction; or
- (d) otherwise, as determined by LCH SA in consultation with the CDSClear Product Committee.

Option Intent: The election of Matched Buyer (or its Exercise Delegation Beneficiary if applicable) in the EEP to Exercise (in whole or in part) or Abandon an Exercise Cleared Transaction.

File No. SR-LCH SA-2022-007 Page 241 of 439

Protected Exercise Matched Pair Report: As set out in Section 6.1 (*Creation and Notification of Exercise Matched Pairs*).

SRMP Notification Deadline: In respect of a notification by LCH SA to the relevant Clearing Members in relation to Swaption Restructuring Matched Pairs, 10.00 a.m. on the Relevant City Business Day immediately prior to the first day of the CEN Triggering Period provided that the SRMP Notification Deadline shall in no event fall prior to the second Relevant City Business Day following the occurrence of the related DC Credit Event Announcement.

SRMP Triggerable Amount: An amount equal to the Swaption Notional Amount of the Swaption Restructuring Cleared Transactions of a Swaption Restructuring Matched Pair *multiplied by* the 'Reference Entity Weighting' (under the Underlying Index Transaction referenced by such Swaption Restructuring Cleared Transaction) of the Reference Entity in respect of which an M(M)R Restructuring has occurred.

Submission Deadline: In respect of:

- (a) an Index Swaption Cleared Transaction in respect of which the Underlying Index Transaction references a Markit iTraxx® Europe Index Series: the Latest Exercice Time as defined in the iTraxx® Swaption Standard Terms Supplement; or
- (b) an Index Swaption Cleared Transaction in respect of which the Underlying Index Transaction references a Markit CDX™ Index Series: the Expiration Time as defined in the CDX Swaption Standard Terms Supplement.

Submission Time: As set out at Section 6.3 (Exercise and Abandonment by way of EEP).

Subsequent Restructuring: As set out at Section 5.1 (*Creation and Notification of Swaption Restructuring Matched Pairs*).

Succession Event: An entity (or entities) constitute a successor or successors in respect of a Reference Entity as determined pursuant to Section 2.2 (*Provisions for Determining a Successor*) of the 2014 ISDA Credit Derivatives Definitions.

 $\textbf{Successor Resolution:} \ \textbf{This term shall have the meaning set out in the DC Rules}.$

Swaption CCM Client Notice: As set out at Mandatory Provision 5.5 (*Duty to Deliver Swaption CCM Client Notice*).

Swaption CCM Client Notice Deadline: As set out at Mandatory Provision 5.5 (*Duty to Deliver Swaption CCM Client Notice*).

Swaption Clearing Member Notice: As set out in Section 6.5(c) (*Duty to deliver Swaption Clearing Member Notices*).

Swaption Clearing Member Notice Deadline: As set out in Section 6.5(c) (*Duty to deliver Swaption Clearing Member Notices*).

Swaption Notice: An Exercise Notice or Abandonment Notice.

File No. SR-LCH SA-2022-007 Page 242 of 439

Swaption Restructuring Cleared Transaction: Each Index Swaption Cleared Transaction forming part of a Swaption Restructuring Matched Pair pursuant to Section 5.2 (*Creation of Swaption Restructuring Cleared Transactions*).

Swaption Restructuring Clearing Member Notice: As set out at Section 5.7 (*Swaption Restructuring Clearing Member Notices*).

Swaption Restructuring Clearing Member Notice Deadline: As set out in Section 5.7 (*Swaption Restructuring Clearing Member Notices*).

Swaption Restructuring Matched Pair: A pairing comprised of a Matched Buyer and a Matched Seller in respect of Swaption Restructuring Cleared Transactions created by LCH SA under Section 5.1 (*Creation and Notification of Swaption Restructuring Matched Pairs*) or Section 5.4 (*Partial triggering of a Swaption Restructuring Cleared Transaction*) for the purposes of the delivery of Credit Event Notices and Notices to Exercise Movement Option as applicable.

Swaption Restructuring Matched Pair Amount: In respect of a Swaption Restructuring Matched Pair, the amount of the Swaption Notional Amount(s) of the relevant Index Swaption Cleared Transactions from which the Swaption Restructuring Matched Pair (and the Swaption Restructuring Cleared Transactions of such Swaption Restructuring Matched Pair) are created which is allocated by LCH SA to such Swaption Restructuring Matched Pair under Section 5.1 (*Creation and Notification of Swaption Restructuring Matched Pairs*) or Section 5.4 (*Partial triggering of a Swaption Restructuring Cleared Transaction*).

Swaption Type: A class of Index Swaption Cleared Transactions that are identical as to their terms (including, without limitation, as to the terms of the Underlying Index Transaction to which such Index Swaption Cleared Transactions relate), except that they may differ as to:

- (a) the Swaption Trade Date;
- (b) the Swaption Notional Amount;
- (c) the Premium;
- (d) the Premium Payment Date; and
- (e) the identity of the relevant Swaption Buyer and Swaption Seller.

Transaction Business Day: A day determined as follows:

- (a) in respect of any date to be determined or time period to be determined which is referenced in the 2006 Definitions:
 - a day that would be a 'Business Day' (where this would be the applicable term under the 2006 Definitions for the purpose of the determination of such date or time period); or

File No. SR-LCH SA-2022-007 Page 243 of 439

(ii) a day that would be an 'Exercise Business Day' (where this would be the applicable term under the 2006 Definitions for the purpose of the determination of such date or time period),

pursuant to the business day elections made in the Index Swaption Cleared Transaction Confirmation; or

- in respect of any date to be determined or time period to be determined which is referenced in the 2014 ISDA Credit Derivatives Definitions:
 - a day that would be a 'Business Day' (where this would be the applicable term under the 2014 ISDA Credit Derivatives Definitions for the purpose of the determination of such date or time period); or
 - (ii) a day that would be a 'Calculation Agent City Business Day', 'Grace Period Business Day' or 'Relevant City Business Day', as applicable (where this would be the applicable term under the 2014 ISDA Credit Derivatives Definitions for the purpose of the determination of such date or time period); or
- (c) for any other purpose, a day that would be a 'Business Day' as defined in the CDS Clearing Rule Book.

Transaction Data: In respect of an Original Transaction to be novated pursuant to Title III (Clearing Operations) of the CDS Clearing Rule Book and cleared by LCH SA as an Index Swaption Cleared Transaction, the data provided by an Approved Trade Source System to LCH SA for such purposes, which includes, without limitation:

- in respect of the Index Swaption Cleared Transaction:
 - the Swaption Trade Date;
 - (ii) the Swaption Seller;
 - (iii) the Swaption Buyer;
 - the Option Type (if applicable); (iv)
 - (v) the Quoting Style (if applicable);
 - the Premium;
 - (vii) the Strike Price; and
 - (viii) the Expiration Date; and
- (b) in respect of the Underlying Index Transaction to which such Index Swaption Cleared Transaction relates:

243

the relevant index, including details of the index name, series and version, the annex date:

File No. SR-LCH SA-2022-007 Page 244 of 439

- (ii) the Original Notional Amount and the currency of the Original Notional Amount:
- (iii) the Scheduled Termination Date;
- (iv) the Floating Rate Payer;
- (v) the Fixed Rate Payer; and
- (vi) the Fixed Rate Payer Payment Dates.

Triggered Amount: This term shall have the meaning given to the term "Exercise Amount" in the 2014 ISDA Credit Derivatives Definitions.

Underlying Index Transaction: means, in respect of an Index Swaption, the Index Cleared Transaction which Swaption Buyer has the right to enter into with Swaption Seller, the terms of which are specified in the Index Swaption Cleared Transaction Confirmation and which references a portfolio of Reference Entities specified in a credit default swap index.

1.3 Inconsistency

To the extent of any conflict between:

- (a) any definition or provision contained in Appendix 1 (CDS Default Management Process) of the CDS Clearing Rule Book;
- (b) the remaining sections of the CDS Clearing Rule Book;
- (c) the CDS Admission Agreement;
- (d) this CDS Clearing Supplement;
- (e) an Index Swaption Cleared Transaction Confirmation;
- (f) the Procedures; or
- (g) any Clearing Notices,

the first referenced document shall prevail except in relation to determining the existence and amount of any payment and delivery obligations under any Cleared Transactions, in respect of which this CDS Clearing Supplement or the Index Swaption Cleared Transaction Confirmation, as applicable, shall prevail to the extent permitted by law.

1.4 Timing

Pursuant to Article 1.2.8 (*Time reference*) of the CDS Clearing Rule Book, any reference to a time of day herein shall be deemed to be a reference to Central European Time unless otherwise provided herein.

Formatted: Header, Tab stops: Not at 2.54 cm + 7.7 cm +

15.4 cm

1.5 **Third Party Rights**

Unless otherwise provided in this CDS Clearing Supplement or in the CDS Clearing Rulebook, a person who is not a party to a Cleared Transaction does not have any right under the Contracts (Rights of Third Parties) Act 1999 to enforce any provision of such Cleared Transaction or this CDS Clearing Supplement as it relates to such Cleared Transaction.

1.6 **Recording of Conversations**

Each of LCH SA and each Clearing Member consents to the recording of telephone conversations between the trading, marketing and other relevant personnel of the parties in connection with any Cleared Transaction and agrees to obtain any necessary consent of, and give any necessary notice of such recording to, its relevant personnel and agrees, to the extent permitted by applicable law, that such recordings may be submitted as evidence in any related court or arbitral proceedings.

1.7 **Application to FCM/BD Clearing Members**

- Upon the taking of any action pursuant to this CDS Clearing Supplement by an FCM/BD Clearing Member in respect of an FCM/BD Cleared Transaction entered into as agent for the account of an FCM/BD Client (as described in FCM/BD CDS Clearing Regulation 1(c)), such FCM/BD Clearing Member shall be deemed to represent to LCH SA that it has the power and authority to, and has been duly authorised to, take such action for the account of such FCM/BD Client.
- For purposes of this CDS Clearing Supplement, with regard to any Cleared Transaction (b) entered into by an FCM/BD Clearing Member acting as agent for the account of an FCM/BD Client (as described in FCM/BD CDS Clearing Regulation 1(c)):
 - references herein to "Swaption Buyer", "Swaption Seller", "CDS Buyer" or "CDS Seller", as the case may be, shall be understood to be references to such FCM/BD Client (which shall not, for the avoidance of doubt, prejudice LCH SA's right to deal solely with the FCM/BD Clearing Member pursuant to Article 6.1.1.3(vi) of the CDS Clearing Rule Book); and
 - references herein to: (1) a Clearing Member entering into a Cleared Transaction (ii) with LCH SA; and (2) a Clearing Member forming part of a Matched Pair, shall each be understood as such FCM/BD Clearing Member acting as agent for the account of such FCM/BD Client (as described in FCM/BD CDS Clearing Regulation 1(c)).
- (c) Notwithstanding and without prejudice to sub-paragraph (b)(i) above, an FCM/BD Clearing Member (acting as agent for the account of its FCM/BD Client) shall designate such FCM/BD Client to Exercise and Abandon Exercise Cleared Transactions on its behalf as its Exercise Delegation Beneficiary (including, without limitation, by means of submitting Option Intents in the EEP) in accordance with Section 6.4 (Delegation by Clearing Members to Clients) of this CDS Clearing Supplement.

Formatted: Header, Tab stops: Not at 2.54 cm + 7.7 cm + 15.4 cm

TERMS OF CLEARED TRANSACTIONS

2.1 General Terms of Cleared Transactions

2.

(a) Terms of Index Swaption Cleared Transactions, Swaption Restructuring Cleared Transactions and Exercise Cleared Transactions

Upon novation of an Original Transaction which is an Index Swaption at the Novation Time in accordance with Title III (Clearing Operations) of the CDS Clearing Rule Book or the creation of a Swaption Restructuring Cleared Transaction in accordance with Section 5.2 (Creation of Swaption Restructuring Cleared Transactions) or an Exercise Cleared Transaction in accordance with Section 6.2 (Creation of Exercise Cleared Transactions), each resulting Index Swaption Cleared Transaction and each such Swaption Restructuring Cleared Transaction and Exercise Cleared Transaction is entered into by LCH SA and the relevant Clearing Member on the terms of the related Index Swaption Cleared Transaction Confirmation.

(b) Swaption Trade Date of Index Swaption Cleared Transactions following Compression

Notwithstanding paragraph (a) above and subject to paragraph (c) below, where Index Swaption Cleared Transactions are subject to compression in accordance with Chapter 3 (*Compression*) of Title III (*Clearing Operations*) of the CDS Clearing Rule Book, the Swaption Trade Date of any resulting Index Swaption Cleared Transaction(s) shall be, in respect of any Index Swaption Cleared Transaction subject to:

- (i) ad hoc compression (as described in Chapter 3 (Compression) of Title III (Clearing Operations) of the CDS Clearing Rule Book), the date on which the request for compression was effectively received and processed in accordance with Section 5 of the Procedures, which shall be:
 - (A) the Clearing Day on which such request is submitted and uploaded by the relevant Clearing Member provided that such request for compression was received by LCH SA before 7.00 p.m. on such Clearing Day (if such request is submitted and uploaded by the relevant Clearing Member via any means of access specified in a Clearing Notice) or 5.00 p.m. (if such request is not submitted via any means of access specified in a Clearing Notice in the case of a disruption of the relevant means of access);
 - (B) the Clearing Day on which such request is submitted if such request is not submitted via any means of access specified in a Clearing Notice in the case of disruption of the relevant means of access and is submitted after 5.00 p.m. but LCH SA, in its sole discretion, processes such request on the Clearing Day on which such request is submitted; and
 - (C) unless the relevant Clearing Member instructs the Operations
 Department to withdraw such request, the Clearing Day following the
 Clearing Day on which such request is submitted if such request is not

File No. SR-LCH SA-2022-007 Page 247 of 439

submitted via any means of access specified in a Clearing Notice in the case of disruption of the relevant means of access and is submitted after 5.00 p.m. and LCH SA, in its sole discretion, does not process such request on the Clearing Day on which such request is submitted; or

(ii) automatic compression (as described in Chapter 3 (Compression) of Title III (Clearing Operations) of the CDS Clearing Rule Book), the Clearing Day on which such Index Swaption Cleared Transaction is automatically compressed by LCH SA in accordance with Section 5 of the Procedures.

(c) Resetting of Swaption Trade Date following Compression

If LCH SA determines that the Swaption Trade Date for a Cleared Transaction following compression would be after a DC Announcement Coverage Cut-off Date whereas the Swaption Trade Date for the Cleared Transactions from which it was created would have been prior to such DC Announcement Coverage Cut-off Date had the Cleared Transaction not been compressed, LCH SA shall take such action as it deems necessary to ensure that such DC Credit Event Announcement is applicable to such Cleared Transaction, including, without limitation, specifying an earlier Swaption Trade Date for such Cleared Transaction.

(d) Compression Cut-off Date

An Index Swaption Cleared Transaction (or a Swaption Restructuring Cleared Transaction or Exercise Cleared Transaction created from such an Index Swaption Cleared Transaction) may not be subject to compression (in accordance with Chapter 3 (*Compression*) of Title III (*Clearing Operations*) of the CDS Clearing Rule Book and Section 5 of the Procedures) after the Compression Cut-off Date applicable to such Index Swaption Cleared Transaction.

2.2 Index Swaption Cleared Transaction Confirmation

The Index Swaption Cleared Transaction Confirmation is amended, supplemented and completed as follows:

- (a) by deleting the words "between [●] (Party A) and [●] (Party B)" in the second and third line of the first paragraph and replacing them with:
 - "between LCH SA ("Party A") and Clearing Member, as identified in the relevant CDS Admission Agreement between such Clearing Member and Party A ("Party B")";
- (b) by deleting the third paragraph thereof and replacing it with the following:
 - "This Confirmation supplements, forms a part of, and is subject to, the CDS Clearing Documentation, as defined in the CDS Clearing Rule Book.";
- (c) by deleting the fourth paragraph thereof and replacing it with the following:
 - "The terms of the Swaption Transaction, which is an Index Swaption Cleared Transaction, to which this Confirmation relates are as follows:";

File No. SR-LCH SA-2022-007 Page 248 of 439

- (d) by specifying that "Standard Terms Date" is:
 - if the Index Swaption Cleared Transaction in respect of which the Underlying Index Transaction references a Markit iTraxx® Europe Index Series:
 - "the most recent iTraxx® Europe Untranched Standard Terms Supplement published prior to the Swaption Trade Date";
 - (ii) if the Index Swaption Cleared Transaction in respect of which the Underlying Index Transaction references a Markit CDX™ Index Series:
 - "the most recent CDX Untranched Transactions Standard Terms Supplement published prior to the Swaption Trade Date";
- (e) by specifying that the "Calculation Agent" is Party A;
- (f) by inserting the following under "Additional terms":

"On the Expiration Date, Swaption Buyer may notify Swaption Seller (which such notification may be given orally, including by telephone) that the Swaption Transaction specified in such notice shall be terminated in whole and that no further amounts shall become due and payable by Swaption Buyer to Swaption Seller or *vice versa* in respect of such Swaption Transaction and that notice will be deemed to be irrevocable. Swaption Buyer will execute and deliver to Swaption Seller a written confirmation confirming the substance of any telephonic notice within one Exercise Business Day of that notice. Failure to provide that written confirmation will not affect the validity of the telephonic notice.";

- (g) by deleting the contact details for notices and the account details;
- (h) by specifying that the "Specified Derivatives Clearing Organization" is LCH SA (subject to the terms of Part C of the CDS Clearing Supplement); and
- (i) by deleting the signature blocks.

2.3 Amendments to 2014 ISDA Credit Derivatives Definitions

- (a) For the purposes of this Part C of the CDS Clearing Supplement, Section 11.2(c)(iv) of the 2014 ISDA Credit Derivatives Definitions as incorporated in any Index Cleared Transaction shall be deemed to be amended so as to enable LCH SA to designate a designee for the additional purposes of delivering or receiving any Credit Event Notice or Notice to Exercise Movement Option relating to an M(M)R Restructuring and such that, where LCH SA is the designator in relation to any Swaption Restructuring Cleared Transaction or Exercise Cleared Transaction, as applicable, it is permitted to designate any relevant Matched Buyer or Matched Seller, as applicable, in accordance with Section 9 (Matched Pair Designations) as its designee, notwithstanding that such relevant Matched Buyer or Matched Seller, as applicable, is not its Affiliate.
- (b) The Protocol Effectiveness Condition (as defined in the 2019 ISDA Narrowly Tailored Credit Event Protocol published by ISDA on 27 August 2019 and as further

File No. SR-LCH SA-2022-007 Page 249 of 439

supplemented from time to time (the "2019 NTCE Protocol")) being satisfied, then from and including the Implementation Date (as defined in the 2019 NTCE Protocol) the amendments set forth in Schedule 1 (*Amendments*) of the 2019 NTCE Protocol shall apply to each Index Swaption Cleared Transaction which incorporates the 2014 ISDA Credit Derivatives Definitions and each such Index Swaption Cleared Transaction shall be a Protocol Covered Transaction for the purposes of the 2019 NTCE Protocol provided that:

- the First Novation Date in respect of such Index Swaption Cleared Transaction is prior to the Implementation Date;
- (ii) such Index Swaption Cleared Transaction is not an Excluded Transaction (as defined in the 2019 NTCE Protocol) provided that if the Underlying Index Transaction of such Index Swaption Cleared Transaction relates to more than one Reference Entity and would otherwise be a Protocol Covered Transaction but for the inclusion of any Affected Portion (as defined in the 2019 NTCE Protocol) therein, such Affected Portion shall be excluded for the purposes of this Section 2.3(b) and the remaining portion of such Index Swaption Cleared Transaction shall be deemed to be a Protocol Covered Transaction;
- the Expiration Date in respect of such Index Swaption Cleared Transaction is on or after the Implementation Date; and
- (iv) no Event Determination Date has occurred in respect of the Underlying Index Transaction of such Index Swaption Cleared Transaction prior to the Implementation Date.

2.4 Terms of STM Cleared Transaction

The following terms of a registered STM Cleared Transaction are designated as the "STM Cleared Terms".

If a Cleared Transaction is designated as an STM Cleared Transaction in accordance with Article 3.1.10.7 of the CDS Clearing Rule Book the terms of such Cleared Transaction will automatically, and without any further action by either party, include the STM Cleared Terms.

(a) Net present value

- (i) LCH SA shall, at least once per Cash Payment Day, determine the net present value of this STM Cleared Transaction in such manner and at such times as may be provided in Section 2 of the Procedures.
- (ii) Immediately upon LCH SA making such a determination of the net present value of this STM Cleared Transaction:
 - (A) An amount (determined in accordance with Section 2.14 (NPV Amount) of the Procedures) of cash may become due and payable by the Clearing Member or LCH SA (as applicable, and as determined in accordance with

File No. SR-LCH SA-2022-007 Page 250 of 439

Section 2.14 (NPV Amount) of the Procedures) under these STM Cleared Terms; and

15.4 cm

Formatted: Header, Tab stops: Not at 2.54 cm + 7.7 cm +

- the net present value of this STM Cleared Transaction shall for all (B) purposes be reset to zero.
- The amount (if any) referred to in (ii)(A) above shall immediately become due and payable by the relevant party under the STM Cleared Terms applicable to this STM Cleared Transaction and shall be made in such manner and at such times as may be provided in Section 2 of the Procedures.
- In any circumstance preventing the NPV Payments or Variation Margin Collateral Transfers from being performed in US Dollar, LCH SA shall be entitled to convert any amount denominated in US Dollar into Euro in accordance with Article 4.2.3.2 of the CDS Clearing Rule Book.

(b) **Price Alignment Amount**

- LCH SA shall, at least once per Cash Payment Day, determine the Price Alignment Amount in respect of this STM Cleared Transaction in such manner and at such times as may be provided in Section 2 of the Procedures.
- Immediately upon LCH SA making such a determination of the Price Alignment (ii) Amount in respect of this STM Cleared Transaction, an amount (determined in accordance with Section 2.17 (Price Alignment Amount) of the Procedures) of cash may become due and payable by the Clearing Member or LCH SA (as applicable, and as determined in accordance with Section 2.17 (Price Alignment Amount) of the Procedures) under these STM Cleared Terms.
- The amount (if any) referred to in (ii) above shall immediately become due and payable by the relevant party under the STM Cleared Terms applicable to this STM Cleared Transaction and shall be made in such manner and at such times as may be provided in Section 2 of the Procedures.

PAYMENTS 3.

3.1 Obligation to pay

Each of LCH SA and each Clearing Member will make each payment specified under the terms of each Cleared Transaction to be made by it, subject to the other provisions of the CDS Clearing Documentation.

Payments under any Cleared Transaction will be made on the due date for value on that date in the place of the account specified for the relevant party in the CDS Admission Agreement (or such other account as may be designated by it from time to time for such purpose in accordance with the CDS Admissions Agreement and/or the Procedures, as applicable) and otherwise in accordance with the CDS Clearing Documentation, in freely transferable funds and in the manner customary for payments in the required currency.

3.2 Payment of Premium under Original Transactions

- (a) If the Premium is due and payable under the terms of an Original Transaction on or before the Clearing Day on which the related Index Swaption Cleared Transactions are created by novation pursuant to Title III (Clearing Operations) of the CDS Clearing Rule Book, such amount shall be payable under and in accordance with the terms of such Original Transaction. In such event, no corresponding Premium shall be payable in respect of such Index Swaption Cleared Transactions.
- (b) If the Premium Payment Date of an Original Transaction would be a date falling after the Clearing Day on which the Index Swaption Cleared Transactions related to such Original Transaction are created by novation pursuant to Title III (Clearing Operations) of the CDS Clearing Rule Book then the corresponding Premium Payment Date for the related Index Swaption Cleared Transactions shall occur on the Transaction Business Day which is also a Clearing Day immediately following the Clearing Day on which such related Index Swaption Cleared Transactions are created and the Index Swaption Cleared Transaction Confirmation shall be deemed to have been amended accordingly.

4. CREDIT EVENTS AND SUCCESSION EVENTS

4.1 Determination of Credit Events and Succession Events

Notwithstanding any provision of any Cleared Transaction to the contrary:

- (a) LCH SA (in its capacity as Calculation Agent with respect to such Cleared Transaction) shall not make any determination pursuant to Section 2.10 (Substitute Reference Obligation) of the 2014 ISDA Credit Derivatives Definitions or in respect of any matter which is or may be subject to resolution under Sections 3.5 (Successor Resolutions) or 3.6 (Substitute Reference Obligation Resolutions) of the DC Rules; and
- (b) neither LCH SA nor any Clearing Member shall be entitled to deliver a Successor Notice or a Credit Event Notice (other than Credit Event Notices in relation to an M(M)R Restructuring in accordance with Section 5.3 (*Triggering of Swaption Restructuring Cleared Transactions*)).

4.2 M(M)R Restructuring Timeline

(a) Publication of Credit Event Timeline

Upon a DC Credit Event Announcement of an M(M)R Restructuring, LCH SA will publish and make available to Clearing Members a timeline in respect of the relevant M(M)R Restructuring and related Cleared Transactions for which a Component Transaction of the Underlying Index Transaction references the relevant combination of Reference Entity, Transaction Type and Reference Obligation to which such M(M)R Restructuring applies, to notify, among other things, the relevant Novation Cut-off Date, Compression Cut-off Date and First Novation Date.

Any such timeline shall (i) be published and made available on the date of such DC Credit Event Announcement or, if LCH SA determines that such publication on such

Formatted: Header, Tab stops: Not at 2.54 cm + 7.7 cm +

15.4 cm

File No. SR-LCH SA-2022-007 Page 252 of 439

date is not practicably possible, as soon as practicable thereafter (but in no event later than the Transaction Business Day before the earlier of the relevant Novation Cut-off Date and the relevant Compression Cut-off Date) and (ii) in all cases be without prejudice to and consistent with the terms of the CDS Clearing Documentation and any relevant DC Resolutions.

(b) Amendment of Credit Event Timeline

Any such timeline may be subject to subsequent amendment by LCH SA, by means of a Clearing Notice to Clearing Members, only to reflect subsequent DC Resolutions, timing provisions of any relevant Transaction Auction Settlement Terms, or in each case any subsequent amendments thereto. Any such amendment shall be made by LCH SA as soon as reasonably practicable following the relevant event.

4.3 Reversal of DC Credit Event Announcements and Margining

If a DC Credit Event Announcement is reversed then, subject to Section 10.2(a)(i) of the 2014 ISDA Credit Derivatives Definitions, LCH SA shall be obliged in accordance with Section 3 of the Procedures to calculate and shall be entitled to call for Margin and/or be obliged to return Margin with respect to each Clearing Member on the basis that the DC Credit Event Announcement that is reversed had not been made.

5. **RESTRUCTURING**

5.1 Creation and Notification of Swaption Restructuring Matched Pairs

Following the occurrence of a DC Credit Event Announcement in respect of an M(M)R Restructuring which would be applicable to a Component Transaction of the Underlying Index Transaction to which a set of Index Swaption Cleared Transactions of the same Swaption Type relates, LCH SA will create (on one or more occasions) Swaption Restructuring Matched Pairs in accordance with Section 9.1 (*Creation of Matched Pairs*). Each such Swaption Restructuring Matched Pair shall be composed of two Swaption Restructuring Cleared Transactions.

If a DC Credit Event Announcement occurs in respect of an M(M)R Restructuring in respect of a further Reference Entity (a "Subsequent Restructuring") which would be applicable to another Component Transaction of the Underlying Index Transaction of a Swaption Restructuring Cleared Transaction prior to the expiry of the CEN Triggering Period for the first Reference Entity in respect of which an M(M)R Restructuring occurred (the "First Restructuring") and the CEN Triggering Period for such Subsequent Restructuring commences prior to the Expiration Date, any Swaption Restructuring Matched Pairs for the First Restructuring shall also be Swaption Restructuring Matched Pairs in respect of the Subsequent Restructuring. The same shall apply to any further relevant DC Credit Event Announcements in respect of M(M)R Restructurings in respect of further Reference Entities during this period and the terms 'First Restructuring' and 'Subsequent Restructuring' in this Section 5 shall be interpreted accordingly.

LCH SA shall notify the relevant Matched Buyer and Matched Seller comprised within each Swaption Restructuring Matched Pair of:

File No. SR-LCH SA-2022-007 Page 253 of 439

- (a) the identity of the other Clearing Member (together with the address, fax number, telephone number, contact email address and other applicable notice details of such other Clearing Member) of such Matched Pair;
- (b) the associated Swaption Restructuring Matched Pair Amount; and
- the associated SRMP Triggerable Amount, (c)

as soon as reasonably practicable following the related Compression Cut-off Date, but in any event, at or prior to the SRMP Notification Deadline.

5.2 **Creation of Swaption Restructuring Cleared Transactions**

Upon the notification to the relevant Clearing Members of Swaption Restructuring Matched Pairs, where two or more Index Swaption Cleared Transactions have been combined into a single transaction as part of the matching process and/or where any Index Swaption Cleared Transaction has been split into two or more separate transactions as part of the matching process, the relevant original Index Swaption Cleared Transactions entered into by each Clearing Member with LCH SA will be deemed terminated and new Swaption Restructuring Cleared Transactions of the same Swaption Type will be deemed to be entered into between each such Clearing Member and LCH SA, with each such Swaption Restructuring Cleared Transaction having a Swaption Notional Amount (and with the Underlying Index Transaction in respect of each such Swaption Restructuring Cleared Transaction having an Original Notional Amount) corresponding to the Swaption Restructuring Matched Pair Amount of the Swaption Restructuring Matched Pair in which the relevant Clearing Member is comprised as a Matched Buyer or a Matched Seller, as applicable. Subject to Section 9.3 (Resetting of Swaption Trade Date) below, the Swaption Trade Date of such Swaption Restructuring Cleared Transactions shall be the date of such notification.

With effect from the notification of the Swaption Restructuring Matched Pairs, any Index Swaption Cleared Transaction forming part of a Swaption Restructuring Matched Pair shall henceforth constitute a Swaption Restructuring Cleared Transaction for the purposes of this CDS Clearing Supplement and the Swaption Notional Amount of such Swaption Restructuring Cleared Transactions shall be the Swaption Restructuring Matched Pair Amount.

5.3 **Triggering of Swaption Restructuring Cleared Transactions**

Subject as set out in Section 8 (Delivery of Notices), and notwithstanding anything to the contrary in the terms of any Cleared Transaction, Clearing Members may only deliver Credit Event Notices (as CDS Buyer or CDS Seller) in relation to an M(M)R Restructuring during the CEN Triggering Period following notification of Swaption Restructuring Matched Pairs by LCH SA and subject always to the terms of the relevant Swaption Restructuring Cleared Transaction.

For the purposes of the delivery of Credit Event Notices in respect of an M(M)R Restructuring and the Swaption Restructuring Cleared Transactions of a Swaption Restructuring Matched Pair, the Floating Rate Payer Calculation Amount by reference to which a Credit Event Notice may be delivered shall be an amount equal to the applicable SRMP Triggerable Amount.

File No. SR-LCH SA-2022-007 Page 254 of 439

Any Credit Event Notice delivered in respect of a Swaption Restructuring Matched Pair for an amount which is greater than the relevant SRMP Triggerable Amount shall be ineffective as to such excess.

Formatted: Header, Tab stops: Not at 2.54 cm + 7.7 cm + 15.4 cm

5.4 Partial triggering of a Swaption Restructuring Cleared Transaction

If a Notifying Party has delivered a valid Credit Event Notice during the CEN Triggering Period applicable to it that specifies a Triggered Amount that is less than the SRMP Triggerable Amount, following the expiry of the CEN Triggering Period, each Swaption Restructuring Cleared Transaction (an "Original Swaption Restructuring Cleared Transaction") of the relevant Swaption Restructuring Matched Pair shall be split into two separate Swaption Restructuring Cleared Transactions:

- (a) one of such Swaption Restructuring Cleared Transactions:
 - (i) shall have an SRMP Triggerable Amount equal to the Triggered Amount; and
 - (ii) shall have a Swaption Notional Amount equal to (x) the Swaption Notional Amount of the Original Swaption Restructuring Cleared Transaction divided by (y) the SRMP Triggerable Amount of the Original Swaption Restructuring Cleared Transaction multiplied by (z) the Triggered Amount specified in the Credit Event Notice.

and will be deemed to have been triggered in whole by the relevant Credit Event Notice (a "Triggered Cleared Transaction"); and

- (b) the other of such Swaption Restructuring Cleared Transactions (the "Remaining Untriggered Cleared Transaction"):
 - shall have an SRMP Triggerable Amount equal to the SRMP Triggerable Amount of the Original Swaption Restructuring Cleared Transaction *minus* the Triggered Amount; and
 - (ii) shall have a Swaption Notional Amount equal to (x) the Swaption Notional Amount of the Original Swaption Restructuring Cleared Transaction less (y) the Swaption Notional Amount determined in accordance with sub-paragraph (a)(ii) above,

and no Credit Event Notice will be deemed to have been delivered in respect of such Remaining Untriggered Cleared Transaction.

In the event that more than one Credit Event Notice is delivered during the CEN Triggering Period, the process set-out above shall be applied to the Credit Event Notice which was validly delivered first; provided, however, that for this purpose any Credit Event Notice(s) delivered by CDS Seller on or prior to the Exercise Cut-off Date applicable to CDS Seller shall be deemed to have been delivered prior to any Credit Event Notice(s) delivered by CDS Buyer on or prior to the Exercise Cut-off Date applicable to CDS Buyer. Each subsequent Credit Event Notice shall then be applied to the Remaining Untriggered Cleared Transaction (and each subsequent Remaining Untriggered Cleared Transaction) until either no further Credit Event Notices remain or the SRMP Triggerable Amount has been reduced to zero.

Section 1.33(b) of the 2014 ISDA Credit Derivatives Definitions shall not apply.

In the event that a Credit Event Notice has been validly delivered in respect of a Subsequent Restructuring prior to the expiry of the CEN Triggering Period for the First Restructuring, such Credit Event Notice shall be split between each Triggered Cleared Transaction and Remaining Untriggered Cleared Transaction created pursuant to this Section 5.4 on a proportionate basis and a Credit Event Notice shall consequently be deemed to have been delivered in respect of each such Triggered Cleared Transaction and Remaining Untriggered Cleared Transaction.

5.5 Notice to Exercise Movement Option

Subject as set out in Section 8 (*Delivery of Notices*), Clearing Members may only deliver a Notice to Exercise Movement Option during the NEMO Triggering Period.

5.6 Delivery of Credit Event Notices and Notices to Exercise Movement Option

Credit Event Notices and Notices to Exercise Movement Option shall be delivered between Matched Buyer and Matched Seller of a Swaption Restructuring Matched Pair in accordance with Section 8.1 (*General Rules relating to Notices*) below and the terms of the Swaption Restructuring Cleared Transactions, in each case acting in its own name in respect of the Cleared Transaction of the Swaption Restructuring Matched Pair to which it is a party and as designee of LCH SA in respect of the other Swaption Restructuring Cleared Transaction of the Swaption Restructuring Matched Pair.

5.7 Swaption Restructuring Clearing Member Notices

(a) Duty to deliver Swaption Restructuring Clearing Member Notices

Each Clearing Member shall notify LCH SA or deliver a copy to LCH SA of any notice delivered or received by such Clearing Member to or from another Clearing Member comprised in a Swaption Restructuring Matched Pair, including, without limitation, any Credit Event Notice or Notice to Exercise Movement Option which was delivered in accordance with Section 5.6 (*Delivery of Credit Event Notices and Notices to Exercise Movement Option*) above and which such Clearing Member asserts or acknowledges was effective for the purposes of this CDS Clearing Supplement and the relevant Swaption Restructuring Cleared Transactions (such notification, or delivery of such copy, in respect of any relevant notice, a **Swaption Restructuring Clearing Member Notice**) by no later than 5.00 p.m. on the last date on which such notice could validly be sent (in each case, the **Swaption Restructuring Clearing Member Notice Deadline**).

(b) No Swaption Restructuring Clearing Member Notice received

Where LCH SA does not receive a Swaption Restructuring Clearing Member Notice in respect of a Credit Event Notice or a Notice to Exercise Movement Option from both Clearing Members in the relevant Swaption Restructuring Matched Pair on or prior to the relevant Swaption Restructuring Clearing Member Notice Deadline, LCH SA shall take no action in respect of the relevant Swaption Restructuring Matched Pair.

(c) Consequences of no Swaption Restructuring Clearing Member Notice being received by LCH SA

If LCH SA determines in its sole discretion that, notwithstanding the fact that no Swaption Restructuring Clearing Member Notice has been received by LCH SA in respect of any Credit Event Notice or Notice to Exercise Movement Option by the relevant Swaption Restructuring Clearing Member Notice Deadline, such notice was in fact delivered or received directly by a Clearing Member and would have been effective for the purposes of this CDS Clearing Supplement, LCH SA shall use commercially reasonable efforts to give effect to the terms of such Credit Event Notice or Notice to Exercise Movement Option, as the case may be. If LCH SA gives effect to the terms of such notice pursuant to this Section 5.7 (c) then the provisions of this Part C of the CDS Clearing Supplement shall apply as if LCH SA had received a Swaption Restructuring Clearing Member Notice in respect of the relevant notice prior to the Swaption Restructuring Clearing Member Notice Deadline.

If LCH SA determines in its sole discretion that it is not possible to give effect to the terms of any such Credit Event Notice or Notice to Exercise Movement Option in respect of which no Swaption Restructuring Clearing Member Notice was received by LCH SA by the relevant Swaption Restructuring Clearing Member Notice Deadline then, following Exercise, an amount shall be payable between the Clearing Members equal to the difference between the value of the Matched Buyer Contract had the Swaption Restructuring Clearing Member Notice been given to LCH SA prior to the Swaption Restructuring Clearing Member Notice Deadline and the value of such contract in the absence of such Swaption Restructuring Clearing Member Notice having been given. Such difference in value shall be determined as of the earlier of the day on which notice is given by any relevant Clearing Member that such amount is due and payable and the eighth Transaction Business Day following the Auction Settlement Date, No Auction Announcement Date or Auction Cancellation Date, as applicable, by reference to the relevant Auction Settlement Amount or end of day contributed prices, in each case if available. In each case, the relevant Clearing Members shall have enforcement rights as against each other pursuant to the Contracts (Rights of Third Parties) Act 1999 in respect of any resulting payments and deliveries; LCH SA shall have no liability in respect thereof.

5.8 Effect of Credit Event Notices and Notices to Exercise Movement Option

Notwithstanding any other provision hereof or of the 2014 ISDA Credit Derivatives Definitions, Matched Buyer and Matched Seller shall have no payment or delivery obligations in respect of the M(M)R Restructuring as a result of the delivery of a Credit Event Notice or Notice to Exercise Movement Option. Such payment and delivery obligations shall instead arise under the Restructuring Cleared Transactions created pursuant to Section 7.3 (Creation of Restructuring Cleared Transactions for Triggering and/or Settlement purposes) following exercise (if applicable).

5.9 Reversal of DC Credit Event Announcements

Formatted: Header, Tab stops: Not at 2.54 cm + 7.7 cm + 15.4 cm

If a DC Credit Event Announcement made in respect of an M(M)R Restructuring is reversed then, subject to Section 10.2(a)(i) of the 2014 ISDA Credit Derivatives Definitions:

- LCH SA shall have no obligation to create Swaption Restructuring Matched Pairs in accordance with Section 5.1 (Creation and Notification of Swaption Restructuring Matched Pairs);
- (b) provided that no Subsequent Restructuring has occurred, any Swaption Restructuring Matched Pairs so created (or created pursuant to Section 5.4 (Partial triggering of a Swaption Restructuring Cleared Transaction)) shall be deemed not to have been created;
- any Credit Event Notices delivered in connection with such M(M)R Restructuring shall be deemed to be ineffective;
- (d) LCH SA shall, where applicable, make relevant registrations in the TIW in order to reflect such reversal which shall also automatically result in such registrations being made in respect of any related CCM Client Transactions; and
- (e) Section 4.3 (Reversal of DC Credit Event Announcements and Margining) shall apply.

5.10 Reports

Without prejudice to the notification requirements set out elsewhere in the CDS Clearing Documentation, LCH SA will communicate to the relevant Clearing Members, on the basis of information received from Clearing Members, amongst other things:

- (a) the aggregate Triggered Amounts of Swaption Restructuring Cleared Transactions to which they are a party; and
- (b) the results of the exercise of any Movement Option in relation to Swaption Restructuring Cleared Transactions to which they are a party,

at or around 7.00 p.m. on each day during each CEN Triggering Period and NEMO Triggering Period, as applicable, through the reports referred to in Section 5 of the Procedures.

For the avoidance of doubt, such communication shall not affect the validity or effectiveness of any Credit Event Notice or Notice to Exercise Movement Option which shall be subject to the terms of the relevant Swaption Restructuring Cleared Transaction.

5.11 Expiry of CEN Triggering Period

Upon the expiry of the CEN Triggering Period (or, if a Subsequent Restructuring has occurred, after the expiry of the CEN Triggering Period for such Subsequent Restructuring) and, if applicable, subsequent to the application of any valid Credit Event Notices delivered during the CEN Triggering Period pursuant to Section 5.4 (*Partial triggering of a Swaption Restructuring Cleared Transaction*), the Swaption Restructuring Cleared Transactions of any Swaption Restructuring Matched Pairs (including, without limitation any Swaption

File No. SR-LCH SA-2022-007 Page 258 of 439

Restructuring Matched Pairs composed of Remaining Untriggered Cleared Transactions created pursuant to Section 5.4 (*Partial triggering of a Swaption Restructuring Cleared Transaction*)) in respect of which no valid Credit Event Notice in respect of an M(M)R Restructuring has been delivered shall cease to be paired together and shall instead constitute Index Swaption Cleared Transactions for the purposes of this CDS Clearing Supplement and the Swaption Restructuring Matched Pair Amount of such Swaption Restructuring Cleared Transactions shall become the Swaption Notional Amount of such Index Swaption Cleared Transactions.

6. EXERCISE MATCHED PAIRS

6.1 Creation and Notification of Exercise Matched Pairs

Subject to Section 5.11 (Expiry of CEN Triggering Period) above, if Swaption Restructuring Matched Pairs have previously been created pursuant to Section 5.1 (Creation and Notification of Swaption Restructuring Matched Pairs) or Section 5.4 (Partial triggering of a Swaption Restructuring Cleared Transaction), such Swaption Restructuring Matched Pairs and the Swaption Restructuring Cleared Transactions from which they are formed shall also automatically constitute Exercise Matched Pairs and Exercise Cleared Transactions (in addition to being Swaption Restructuring Matched Pairs and Swaption Restructuring Cleared Transactions) for the purposes of this CDS Clearing Supplement.

Notwithstanding the above, if the CEN Triggering Period for an M(M)R Restructuring ends on a date falling during the EMP Creation Period, the relevant Swaption Restructuring Matched Pairs created in respect of such M(M)R Restructuring shall not become Exercise Matched Pairs pursuant to the above until after the expiry of the CEN Triggering Period and subject to the prior application of Sections 5.4 (Partial triggering of a Swaption Restructuring Cleared Transaction) and 5.11 (Expiry of CEN Triggering Period) above, if applicable.

On each Transaction Business Day during the EMP Creation Period, LCH SA will create Exercise Matched Pairs in accordance with Section 9.1 (*Creation of Matched Pairs*) for a set of Index Swaption Cleared Transactions of the same Swaption Type. Each such Exercise Matched Pair shall be composed of two Exercise Cleared Transactions.

Upon the creation of an Exercise Matched Pair, LCH SA shall:

- (a) notify the relevant Matched Buyer and Matched Seller comprised within each Exercise Matched Pair (and any Exercise Delegation Beneficiary thereof) of the creation of such Exercise Matched Pair, without however providing any detail in respect of the identity of the other Clearing Member (or its Exercise Delegation Beneficiary if applicable); and
- (b) provide the relevant Matched Buyer and Matched Seller comprised within each Exercise Matched Pair (and any Exercise Delegation Beneficiary thereof) with a report the access to which will be restricted (a "Protected Exercise Matched Pair Report") and which contains information on:
 - the identity of the other Clearing Member (or its Exercise Delegation Beneficiary if applicable) together with the address, fax number, telephone

File No. SR-LCH SA-2022-007 Page 259 of 439

number, email address and other applicable notice details of such other Clearing Member (or its Exercise Delegation Beneficiary if applicable) of the Exercise Matched Pair; and

(ii) the associated Exercise Matched Pair Amount,

as soon as reasonably practicable following the related Compression Cut-off Date, but in any event, at or prior to the EMP Notification Deadline.

LCH will provide the Protected Exercise Matched Pair Report in a manner such that the relevant Matched Buyer and Matched Seller or, if applicable, any relevant Exercise Delegation Beneficiaries, may access the information contained in the Protected Exercise Matched Pair Reports only upon occurrence of an EEP Failure Event, in accordance with Section 6.5(a)(ii)below.

6.2 Creation of Exercise Cleared Transactions

Upon the creation of Exercise Matched Pairs, where two or more Index Swaption Cleared Transactions have been combined into a single transaction as part of the matching process and/or where any Index Swaption Cleared Transaction has been split into two or more separate transactions as part of the matching process, the relevant original Index Swaption Cleared Transactions entered into by each Clearing Member with LCH SA will be deemed terminated and new Exercise Cleared Transactions of the same Swaption Type will be deemed to be entered into between each such Clearing Member and LCH SA, with each such Exercise Cleared Transaction having a Swaption Notional Amount (and with the Underlying Index Transaction in respect of each such Exercise Cleared Transaction having an Original Notional Amount) corresponding to the Exercise Matched Pair Amount of the Exercise Matched Pair in which the relevant Clearing Member is comprised as a Matched Buyer or a Matched Seller, as applicable. Subject to Section 9.3 (Resetting of Swaption Trade Date) below, the Swaption Trade Date of such Exercise Cleared Transactions shall be the date of such creation.

With effect from the creation of Exercised Matched Pairs, any Index Swaption Cleared Transaction forming part of an Exercised Matched Pair shall henceforth constitute an Exercise Cleared Transaction for the purposes of this CDS Clearing Supplement and the Swaption Notional Amount of such Index Swaption Cleared Transactions shall be the Exercise Matched Pair Amount

6.3 Exercise and Abandonment by way of EEP

The Exercise (in whole or in part) and Abandonment of Exercise Cleared Transactions shall be conducted by way of the submission of Options Intents in the EEP only, save in the circumstances specified in Section 6.5(b) (Consequences of EEP Failure) below.

An Option Intent submitted by Matched Buyer (or its Exercise Delegation Beneficiary on its behalf, if applicable) via the EEP shall be deemed to constitute the delivery by Matched Buyer of a valid Swaption Notice for the purposes of each of the Exercise Cleared Transactions of the Exercise Matched Pair (in its own name in respect of the Exercise Cleared Transaction of the Exercise Matched Pair to which it is a party and as designee of LCH SA in respect of the

File No. SR-LCH SA-2022-007 Page 260 of 439

other Exercise Cleared Transaction of the Exercise Matched Pair) if (a) the Submission Time for such Option Intent is prior to the Submission Deadline and (b) LCH SA has completed those steps necessary to make such Option Intent available for viewing in the EEP, including validation of the EEP Controls.

As used herein, (i) the "Submission Time" for an Option Intent will be the time, as recorded by LCH SA, as of which such Option Intent is submitted via the EEP by the relevant Matched Buyer (or its Exercise Delegation Beneficiary on its behalf, if applicable) and (ii) "EEP Controls" means the controls specified in Section 5 of the Procedures performed by LCH SA immediately following the Submission Time.

An Option Intent shall become irrevocable by the Swaption Buyer as from the Submission Time.

For the avoidance of doubt, a Matched Buyer (or its Exercise Delegation Beneficiary if applicable) shall not be able to require a Force Submission after the Submission Deadline, including if the relevant Option Intent is deemed illogical further to EEP Controls carried out at or after the Submission Deadline.

The deemed time of delivery of the Swaption Notice in respect of the relevant Exercise Cleared Transaction shall be the Submission Time specified by the EEP in respect of such Option Intent.

The Exercise (in whole or in part) or Abandonment of the Exercise Cleared Transaction of the Exercise Matched Pair to which Matched Seller is a party will become accessible in the Clearing Member Portal Account of Matched Seller (and in the Client Portal Account of its Exercise Delegation Beneficiary if applicable) as soon as reasonably practicable following the registration of such Exercise or Abandonment by Matched Buyer (or its Exercise Delegation Beneficiary) in the EEP.

6.4 Delegation by Clearing Members to Clients

Solely in respect of the Exercise and Abandonment of the Exercise Cleared Transactions of an Exercise Matched Pair which are Client Cleared Transactions, Matched Buyer or Matched Seller shall designate its relevant Client to act on its behalf in accordance with this Section 6. Any such designation will take effect as soon as reasonably practicable (but not later than five Business Days) following receipt by LCH SA of a duly completed and signed Exercise Delegation Form (as defined in Section 5 of the Procedures). Such designation cannot be revoked.

Where Matched Buyer designates its Client in accordance with the above, any submission of an Option Intent by the Client of Matched Buyer via its Client Portal Account in the EEP prior to the Submission Deadline on the Expiration Date shall be deemed to constitute the delivery by Matched Buyer of a valid Exercise Notice or Abandonment Notice (as applicable) for the purposes of the terms of each of the Exercise Cleared Transactions of the Exercise Matched Pair in accordance with Section 6.3 (Exercise and Abandonment by way of EEP) above.

Where a Client has been designated by its Clearing Member pursuant to this Section 6.4 for purposes of the Exercise and Abandonment of Exercise Cleared Transactions, any reference

Formatted: Header, Tab stops: Not at 2.54 cm + 7.7 cm + 15.4 cm

260

File No. SR-LCH SA-2022-007 Page 261 of 439

in this CDS Clearing Supplement to a Swaption Notice being delivered to or by a Clearing Member, to the submission of Option Intents or the Exercise or Abandonment of an Exercise Cleared Transaction by a Clearing Member shall be interpreted accordingly.

Formatted: Header, Tab stops: Not at 2.54 cm + 7.7 cm + 15.4 cm

6.5 EEP failure and resolution

(a) Notification of EEP Failure and Resolution

 LCH SA to notify Clearing Members and Exercise Delegation Beneficiaries of EEP Failure Event

If LCH SA becomes aware that the EEP is or will be unavailable for the submission or receipt of Option Intents with respect to an Expiration Date in a timely manner (an **EEP Failure Event**), then LCH SA will, as soon as reasonably practicable and in any case within one hour after the occurrence of the EEP Failure Event, notify all relevant Clearing Members (and any Exercise Delegation Beneficiary thereof) of such occurrence, including the time at which such failure occurred (the **EEP Failure Event Time**).

(ii) Access to the Protected Exercise Matched Pair Report

As soon as possible following the occurrence of an EEP Failure Event, LCH will authorise Clearing Members or their Exercise Delegation Beneficiaries, as applicable (by any communication method including, without limitation, phone, email or any electronic messaging system) to access the information contained in the Protected Exercise Matched Pair Report.

(iii) LCH SA to notify Clearing Members of Resolution of EEP Failure Event

If, subsequent to an EEP Failure Event, LCH SA determines that an EEP Failure Event previously notified to Clearing Members is no longer in effect, LCH SA will as soon as reasonably practicable notify all relevant Clearing Members and Exercise Delegation Beneficiaries accordingly, including the time (the **EEP Resolution Time**) at which such EEP Failure Event is deemed to have been resolved and following which the EEP is operative for the purposes of the submission of Option Intents in respect of Exercise Cleared Transactions.

(b) Consequences of EEP Failure

From (and including) the EEP Failure Event Time to (but excluding) the EEP Resolution Time:

- Section 6.3 (Exercise and Abandonment by way of EEP) shall not apply and accordingly Swaption Notices shall be delivered in accordance with the provisions for the delivery of notices at Section 8 (Delivery of Notices) hereof and the terms of the relevant Exercise Cleared Transaction (and not via the EEP);
- (ii) if Matched Buyer has designated its Client as its Exercise Delegation Beneficiary pursuant to Section 6.4 (Delegation by Clearing Members to Clients) above,

File No. SR-LCH SA-2022-007 Page 262 of 439

such Exercise Delegation Beneficiary shall be entitled to send a Swaption Notice to Matched Seller as designee of Matched Buyer in accordance with Section 8 (*Delivery of Notices*) hereof using the notice details provided by LCH SA in the Protected Exercise Matched Pair Report and such Swaption Notice shall be copied to Matched Buyer (provided that failure to provide such copy shall not affect the validity of the relevant Swaption Notice);

- (iii) if Matched Seller has designated its Client as its Exercise Delegation Beneficiary, Swaption Notices shall be sent by Matched Buyer (or its Client as its Exercise Delegation Beneficiary if applicable) to the Client of Matched Seller as its Exercise Delegation Beneficiary in accordance with Section 8 (*Delivery of Notices*) hereof using the notices details provided by LCH SA in the Protected Exercise Matched Pair Report and shall be copied to Matched Seller and, where the Swaption Notice is sent by the Client of Matched Buyer as its Exercise Delegation Beneficiary, shall be copied to Matched Buyer (provided in each case that failure to provide such copy shall not affect the validity of the relevant Swaption Notice);
- (iv) subject to Section 8.3 (Oral Notices) below in respect of Abandonment Notices delivered to LCH SA, an Abandonment Notice given in accordance with this Section 6.5(b) may be given orally, including by telephone. Matched Buyer (or its Exercise Delegation Beneficiary if such Exercise Delegation Beneficiary gives an Abandonment Notice orally) will execute and deliver to Matched Seller (and the Exercise Delegation Beneficiary of Matched Seller if applicable) a written confirmation confirming the substance of any telephonic notice within one Transaction Business Day of the telephonic notice. Without prejudice to Section 6.8 (Consequences of no Swaption Clearing Member Notice being received by LCH SA) below, failure to provide that written confirmation will not affect the validity of the telephonic notice;
- any Exercise Notice delivered in respect of an Exercise Matched Pair for an amount which is greater than the related Exercise Matched Pair Notional Amount shall be ineffective as to such excess;
- (vi) any Swaption Notice delivered via the EEP prior to the EEP Failure Event Time will be valid and will not be affected by such EEP Failure Event; and
- (vii) any Swaption Notice delivered or purported to be delivered via the EEP at or following the EEP Failure Event Time but prior to the EEP Resolution Time will not be valid and effective.

Section 6.3 (Exercise and Abandonment by way of EEP) shall apply with effect from the EEP Resolution Time and, accordingly, any Swaption Notice thereafter delivered or

File No. SR-LCH SA-2022-007 Page 263 of 439

purported to be delivered directly pursuant to this Section 6.5(b) will not be valid and effective.

Formatted: Header, Tab stops: Not at 2.54 cm + 7.7 cm + 15.4 cm

(c) Duty to deliver Swaption Clearing Member Notices

Each Clearing Member shall notify LCH SA or deliver a copy to LCH SA of any Swaption Notice delivered by such Clearing Member to another Clearing Member comprised in an Exercise Matched Pair pursuant to Section 6.5(b) (Consequences of EEP Failure) above which such Clearing Member asserts or acknowledges was effective for the purposes of this CDS Clearing Supplement and the relevant Exercise Cleared Transactions (such notification, or delivery of such copy, in respect of any relevant Swaption Notice, a Swaption Clearing Member Notice) by no later than the Submission Deadline on the Expiration Date (the Swaption Clearing Member Notice Deadline).

If a Clearing Member has designated its Client as being entitled to send and receive Swaption Notices on its behalf pursuant to Section 6.4 (*Delegation by Clearing Members to Clients*) above, a Swaption Clearing Member Notice in respect of a Swaption Notice delivered pursuant to Section 6.5(b) (*Consequences of EEP Failure*) may be sent on its behalf by its Client as its Exercise Delegation Beneficiary.

6.6 Abandonment of Exercise Cleared Transactions

If, on the Expiration Date, Swaption Buyer elects to Abandon the Exercise Cleared Transactions of the Exercise Matched Pair, each Exercise Cleared Transaction shall be terminated in whole.

6.7 Termination of Exercise Cleared Transactions

If:

- (a) no Option Intent is submitted pursuant to Section 6.3 (Exercise and Abandonment by way of EEP) or Section 6.4 (Delegation by Clearing Members to Clients) above in respect of the Exercise Cleared Transactions of an Exercise Matched Pair; and
- (b) where Section 6.5(b) (Consequences of EEP Failure) (or, in respect of a CCM Client Cleared Transaction, Mandatory Provision 5.4 (Consequences of EEP Failure)) is applicable for some or all of the Exercise Period, LCH SA does not receive a Swaption Clearing Member Notice (or Swaption CCM Client Notice) in respect of an Exercise Notice or an Abandonment Notice from Swaption Buyer (or its Exercise Delegation Beneficiary, as applicable) of the relevant Exercised Matched Pair on or prior to the Swaption Clearing Member Notice Deadline (or Swaption CCM Client Notice Deadline, as applicable),

then, subject to Section 6.8 (Consequences of no Swaption Clearing Member Notice being received by LCH SA) below (or, in the case of a CCM Client Cleared Transaction, Mandatory Provision 5.5 (Duty to Deliver Swaption CCM Client Notice), LCH SA shall terminate the Exercise Cleared Transactions of the relevant Exercised Matched Pair.

6.8 Consequences of no Swaption Clearing Member Notice or Swaption CCM Client Notice being received by LCH SA

If Section 6.5(b) (Consequences of EEP Failure) is applicable for some or all of the Exercise Period and LCH SA determines in its sole discretion that, notwithstanding the fact that no Swaption Clearing Member Notice has been received by LCH SA in respect of an Exercise Notice by the Swaption Clearing Member Notice Deadline (or, in the case of a CCM Client Cleared Transaction, that no Swaption CCM Client Notice has been received by LCH SA in respect of an Exercise Notice by the Swaption CCM Client Notice Deadline), an Exercise Notice was in fact delivered by a Clearing Member (or its Exercise Delegation Beneficiary) pursuant to Section 6.5(b) (Consequences of EEP Failure) (or, in the case of a CCM Client Cleared Transaction, pursuant to Mandatory Provision 5.4(a) (Consequences of EEP Failure)) and would have been effective for the purposes of this CDS Clearing Supplement, LCH SA shall use commercially reasonable efforts to give effect to the terms of such Exercise Notice. If LCH SA gives effect to the terms of such Exercise Notice pursuant to this Section 6.8 then the provisions of this Part C of the CDS Clearing Supplement shall apply as if LCH SA had received a Swaption Clearing Member Notice in respect of the relevant Exercise Notice by the Swaption Clearing Member Notice Deadline (or, in the case of a CCM Client Cleared Transaction, had received a Swaption CCM Client Notice in respect of the relevant Exercise Notice by the Swaption CCM Client Notice Deadline).

If LCH SA determines in its sole discretion that it is not possible to give effect to the terms of any such Exercise Notice in respect of which no Swaption Clearing Member Notice was received by LCH SA by the Swaption Clearing Member Notice Deadline (or. in the case of a CCM Client Cleared Transaction, no Swaption CCM Client Notice was received by LCH SA by the Swaption CCM Client Notice Deadline), the relevant Clearing Members (or their Exercise Delegation Beneficiaries, as applicable), as identified in the Protected Exercise Matched Pair Report, shall acquire rights as against each other as though party to a bilateral credit default swap transaction on the terms of the relevant Underlying Index Transaction. The Settlement Payment shall be due and payable two Transaction Business Days following the giving of a notice that such amount is due and payable. The relevant Clearing Members (or their Exercise Delegation Beneficiaries, as applicable) shall have enforcement rights as against each other pursuant to the Contracts (Rights of Third Parties) Act 1999 in respect of any resulting payments and deliveries; LCH SA shall have no liability in respect thereof.

If the Clearing Members have entered into such a transaction following the delivery of an Exercise Notice and those Clearing Members wish to clear such transaction through LCH SA, those Clearing Members will each need to submit such transaction for clearing as an Index Cleared Transaction pursuant to Part B of this CDS Clearing Supplement.

6.9 **Cleared Transaction Exercise Reports**

Without prejudice to the notification requirements set out elsewhere in the CDS Clearing Documentation, LCH SA will communicate to the relevant Clearing Members, on the basis of information received from Clearing Members the aggregate Swaption Notional Amounts of Exercise Cleared Transactions to which they are a party as Swaption Buyer in respect of which Exercise Notices and Abandonment Notices have been delivered and the aggregate Swaption Notional Amounts of Exercise Cleared Transactions to which they are a party as Swaption

File No. SR-LCH SA-2022-007 Page 265 of 439

Seller in respect of which Exercise Notices and Abandonment Notices have been delivered, in each case on an ongoing basis on the Expiration Date, through the reports referred to in Section 5 of the Procedures with the final report being published as soon as practicable after LCH SA has confirmed with the relevant Clearing Members that all Exercise Notices and Abandonment Notices have been identified.

For the avoidance of doubt, such communication shall not affect the validity or effectiveness of any Exercise or Abandonment which shall be subject to the terms of the relevant Exercise Cleared Transaction and the 2006 Definitions.

6.10 Clearing Member Communications Failure Event and CCM Client Communications Failure **Event**

Right to deliver Swaption Notices manually following Clearing Member (a) **Communications Failure Event**

If a Clearing Member or its Exercise Delegation Beneficiary is affected by a significant communications or information technology failure resulting in it being impossible or impractical for such Clearing Member or Exercise Delegation Beneficiary to deliver or receive any Exercise Notice or Abandonment Notice in relation to an Expiration Date via the EEP (a Clearing Member Communications Failure Event) it shall, notwithstanding Section 6.3 (Exercise and Abandonment by way of EEP), deliver and receive Swaption Notices to and from LCH SA in accordance with the provisions for the delivery of notices at Section 8 (Delivery of Notices) hereof and the terms of the relevant Exercise Cleared Transaction (and, in the case of a CCM Client Cleared Transaction, in accordance with Mandatory Provision 5.7 (CCM Client Communications Failure Event)) and not via the EEP.

- Processing of Swaption Notices by LCH SA in case of Clearing Member **Communications Failure Event or CCM Client Communications Failure Event**
 - (i) In case of Clearing Member Communications Failure Event or CCM Client Communications Failure Event affecting a Matched Buyer

Upon receipt of a Swaption Notice delivered by a Clearing Member (or its Exercise Delegation Beneficiary) in accordance with Section 6.10(a) above or (in respect of a CCM Client Cleared Transaction) a Swaption Notice delivered by a CCM Client in accordance with Mandatory Provision 5.7 (CCM Client Communications Failure Event), LCH SA shall, as soon as reasonably practicable, submit via the EEP, on behalf of such Clearing Member or Exercise Delegation Beneficiary, an Option Intent conforming to such Swaption Notice. Provided that (A) the Submission Time for such Option Intent is prior to the Submission Deadline and (B) LCH SA has completed those steps necessary to make such Option Intent available for viewing in the EEP, including validation of the EEP Controls, such Option Intent shall be deemed to constitute the delivery by the relevant Clearing Member or Exercise Delegation Beneficiary (as applicable) of a valid Exercise Notice or Abandonment Notice (as applicable). The deemed

File No. SR-LCH SA-2022-007 Page 266 of 439

time of delivery of the relevant Swaption Notice shall be the Submission Time specified by the EEP in respect of such Option Intent.

Notwithstanding the above, in respect of any Swaption Notice delivered pursuant to Section 6.10(a) above or (in the case of a CCM Client Cleared Transaction) pursuant to Mandatory Provision 5.7 (*CCM Client Communications Failure Event*), LCH SA may determine in its sole discretion that it is not able to submit the relevant Option Intent in the relevant system prior to the Submission Deadline on the Expiration Date. In such case, LCH SA will inform the relevant Clearing Member or Exercise Delegation Beneficiary and such Clearing Member or Exercise Delegation Beneficiary will be deemed not to have submitted an Option Intent in respect of the relevant Exercise Cleared Transaction and the provisions of Sections 6.7 and 6.8 shall apply.

In respect of any Swaption Notice delivered pursuant to Section 6.10(a) above or (in the case of a CCM Client Cleared Transaction) pursuant to Mandatory Provision 5.7 (*CCM Client Communications Failure Event*), LCH SA may, in its sole discretion, elect to register the Exercise or Abandonment of an Exercise Cleared Transaction in an alternative internal system of LCH SA in which case the terms 'EEP', 'Option Intent', 'Submission Time' and 'EEP Controls' shall be interpreted accordingly *mutatis mutandis*.

(ii) In case of Clearing Member Communications Failure Event or CCM Client Communications Failure Event affecting a Matched Seller

If a Clearing Member (or its FCM/BD Client as Exercise Delegation Beneficiary) is subject to a Clearing Member Communications Failure Event duly notified to LCH SA in accordance with Section 6.10(c) below, or (in respect of a CCM Client Cleared Transaction) CCM Client as Exercise Delegation Beneficiary is subject to a CCM Client Communications Failure Event duly notified to LCH SA in accordance with Mandatory Provision 5.7(b) (Client to notify LCH SA of occurrence of CCM Client Communications Failure Event), any Exercise (in whole or in part) or Abandonment of an Exercise Cleared Transaction of an Exercise Matched Pair to which such Clearing Member is a party as Matched Seller shall be notified to such Matched Seller (and its Exercise Delegation Beneficiary if applicable) by LCH SA as soon as reasonably practicable in accordance with the provisions for the delivery of notices at Section 8 (Delivery of Notices) hereof.

(c) Clearing Member or Exercise Delegation Beneficiary to notify LCH SA of Occurrence of Clearing Member Communications Failure Event

Following the occurrence of a Clearing Member Communications Failure Event (or, in respect of a CCM Client Cleared Transaction, a CCM Client Communications Failure Event), the affected Clearing Member or Exercise Delegation Beneficiary shall, as soon as reasonably practicable and, in any event, prior to or at the same time as delivering any Swaption Notice to LCH SA in accordance with Section 6.10(a) above, deliver to LCH SA a notice (in the form set out at Appendix VI hereto) signed by an authorised

File No. SR-LCH SA-2022-007 Page 267 of 439

signatory of such Clearing Member (or its Exercise Delegation Beneficiary) certifying that such Clearing Member (or Exercise Delegation Beneficiary) is affected by a Clearing Member Communications Failure Event or CCM Client Communications Failure Event (or, if such Clearing Member or Exercise Delegation Beneficiary is unable to deliver such notice in writing, orally by telephone).

(d) Notification of Resolution of Clearing Member Communications Failure Event

As soon as reasonably practicable upon a Clearing Member (or its FCM/BD Client as Exercise Delegation Beneficiary, if applicable) ceasing to be subject to a Clearing Member Communications Failure Event (or, in respect of a CCM Client as Exercise Delegation Beneficiary, a CCM Client Communications Failure Event), it shall notify LCH SA accordingly in the form set out at Appendix VII hereto, such notice to be signed by an authorised signatory (or, if such Clearing Member or Exercise Delegation Beneficiary is unable to deliver such notice in writing, orally by telephone) and thereupon Section 6.3 (Exercise and Abandonment by way of EEP) shall apply and, accordingly, any Swaption Notice thereafter delivered or purported to be delivered by such Clearing Member (or its Exercise Delegation Beneficiary) directly (and not via the EEP) will not be valid and effective.

(e) Clearing Member Duty to Mitigate

A Clearing Member which is subject to a Clearing Member Communications Failure Event shall use reasonable endeavours to mitigate the operational impact on other Clearing Members and LCH SA of any Clearing Member Communications Failure Event, to cure such Clearing Member Communications Failure Event as soon as reasonably practicable and to ensure that the circumstances giving rise to the relevant Clearing Member Communications Failure Event do not recur.

7. **SETTLEMENT**

7.1 Creation of Index Cleared Transactions

Notwithstanding and in addition to any Initial Single Name Cleared Transaction(s) or Restructuring Cleared Transaction(s) created pursuant to Section 7.2 (*Creation of Initial Single Name Cleared Transactions for Settlement purposes in respect of Credit Events other than M(M)R Restructuring*), Section 7.3 (*Creation of Restructuring Cleared Transactions for Triggering and/or Settlement purposes*) or Section 7.4 (*Creation of Initial Single Name Cleared Transactions in respect of untriggered M(M)R Restructuring*) below, following Exercise an Index Cleared Transaction shall be deemed to have been entered into between each Clearing Member and LCH SA on the terms of the Underlying Index Transaction to which the relevant Exercise Cleared Transactions relate. Such Index Cleared Transaction shall reference the currently Eligible Underlying Index Transaction Version as of the Expiration Date of the index referenced by the relevant Underlying Index Transaction. Subject to the below, such Index Cleared Transaction shall be entered into by LCH SA and the relevant Clearing Member on the terms of the Index Cleared Transaction Confirmation (as defined in Part B of the CDS Clearing Supplement).

Each such Index Cleared Transaction shall have a Floating Rate Payer Calculation Amount equal to:

(a) subject to (b) below, the Swaption Notional Amount of the Exercise Cleared Transaction from which such Index Cleared Transaction resulted; or

(b) if the Exercise Cleared Transaction from which such Index Cleared Transaction resulted was exercised in part, the portion of the Swaption Notional Amount of the Exercise Cleared Transaction that was exercised.

The provisions of this Part C of the CDS Clearing Supplement shall cease to apply to such Index Cleared Transaction upon its creation and such Index Cleared Transaction shall instead be subject to and cleared in accordance with Part B of this CDS Clearing Supplement; *provided, however, that*:

- (i) an Initial Payment Amount shall be payable under such Index Cleared Transaction which shall be an amount equal to, and payable by the party that would be required to pay, the Settlement Payment (or the absolute value thereof, as applicable) that would be determined in respect of the Exercise Cleared Transaction from which such Index Cleared Transaction is created in accordance with the provisions of the Exercise Cleared Transaction (as amended by Section 2.2 (Index Swaption Cleared Transaction Confirmation) hereof). The Initial Payment Date in respect of such Initial Payment Amount shall be the Transaction Business Day (determined pursuant to paragraph (a)(i) of the definition of Transaction Business Day as if such payment were the Settlement Payment) immediately following the Expiration Date; and
- (ii) the paragraph entitled 'Successors' of the iTraxx® Swaption Standard Terms Supplement shall continue to apply to such Index Cleared Transaction.

Following the creation of such Index Cleared Transaction and any Initial Single Name Cleared Transaction(s) and Restructuring Cleared Transaction(s) created pursuant to Section 7.2 (Creation of Initial Single Name Cleared Transactions for Settlement purposes in respect of Credit Events other than M(M)R Restructuring), Section 7.3 (Creation of Restructuring Cleared Transactions for Triggering and/or Settlement purposes) and Section 7.4 (Creation of Initial Single Name Cleared Transactions in respect of untriggered M(M)R Restructuring) below, the Exercise Cleared Transaction from which it was created shall be terminated.

7.2 Creation of Initial Single Name Cleared Transactions for Settlement purposes in respect of Credit Events other than M(M)R Restructuring

If a DC Credit Event Announcement has been made on or after the Credit Event Backstop Date which would be applicable to a Component Transaction of the Underlying Index Transaction of an Exercise Cleared Transaction (other than in respect of an M(M)R Restructuring and subject to the operation of the paragraph entitled 'Operation of each Underlying Swap Transaction' of the iTraxx® Swaption Standard Terms Supplement or, as the case may be, of the paragraph entitled 'Operation of Underlying Swap Transaction' of the CDX Swaption Standard Terms Supplement) and the Auction Final Price Determination Date that would have been applicable to such Credit Event fell prior to the Expiration Date, then following Exercise an Initial Single Name Cleared Transaction (as defined in Part B of this CDS Clearing

File No. SR-LCH SA-2022-007 Page 269 of 439

Supplement) shall be deemed to have been entered into on the Expiration Date between each Clearing Member and LCH SA on the terms of the Component Transaction (subject as set out below) in respect of the relevant Reference Entity of the Underlying Index Transaction. Subject to the below, such Initial Single Name Cleared Transaction shall be entered into by LCH SA and the relevant Clearing Member on the terms of the Single Name Cleared Transaction Confirmation (as defined in Part B of the CDS Clearing Supplement).

Each such Initial Single Name Cleared Transaction shall have a Floating Rate Payer Calculation Amount equal to:

- (a) the Swaption Notional Amount of the Exercise Cleared Transaction from which such Initial Single Name Cleared Transaction resulted multiplied by the 'Reference Entity Weighting' (under the Underlying Index Transaction referenced by such Exercise Cleared Transaction) of the relevant Reference Entity; multiplied by
- (b) the Exercise Percentage.

The provisions of this Part C of the CDS Clearing Supplement shall cease to apply to such Initial Single Name Cleared Transaction upon its creation and such Initial Single Name Cleared Transaction shall instead be subject to and cleared in accordance with Part B of this CDS Clearing Supplement; provided, however, that:

- (i) the paragraph entitled 'Operation of each Underlying Swap Transaction' and the paragraph entitled 'Successors' of the iTraxx® Swaption Standard Terms Supplement or, as the case may be, the paragraph entitled 'Operation of Underlying Swap Transaction' of the CDX Swaption Standard Terms Supplement shall continue to apply to such Initial Single Name Cleared Transaction; and
- (ii) notwithstanding (i) above, the Auction Settlement Date in respect of such Initial Single Name Cleared Transaction shall be the later of (x) the Auction Settlement Date that would be determined in accordance with Section 6.3 of the 2014 ISDA Credit Derivatives Definitions and (y) the first Transaction Business Day following the Expiration Date.

7.3 Creation of Restructuring Cleared Transactions for Triggering and/or Settlement purposes

If:

- (a) a DC Credit Event Announcement has been made on or after the Credit Event Backstop Date in respect of an M(M)R Restructuring which would be applicable to a Component Transaction of the Underlying Index Transaction of an Exercise Cleared Transaction (subject to the operation of the paragraph entitled 'Operation of each Underlying Swap Transaction' of the iTraxx® Swaption Standard Terms Supplement); and
- either the CEN Triggering Period has not expired or, if the CEN Triggering Period has expired, a valid Credit Event Notice was delivered in respect of such M(M)R Restructuring prior to its expiry,

then following Exercise a Restructuring Cleared Transaction (as defined in Part B of this CDS Clearing Supplement) shall be deemed to have been entered into on the Expiration Date

File No. SR-LCH SA-2022-007 Page 270 of 439

between each Clearing Member and LCH SA on the terms of the Component Transaction (subject as set out below) in respect of the relevant Reference Entity of the Underlying Index Transaction and any Credit Event Notice or Notice to Exercise Movement Option validly delivered in respect of the Swaption Restructuring Cleared Transaction from which such Restructuring Cleared Transaction is created shall be deemed to have been validly delivered in respect of such Restructuring Cleared Transaction by CDS Buyer and/or CDS Seller (as applicable). Subject to the below, such Restructuring Cleared Transaction shall be entered into by LCH SA and the relevant Clearing Member on the terms of the Single Name Cleared Transaction Confirmation (as defined in Part B of the CDS Clearing Supplement).

Such Restructuring Cleared Transaction shall have a Floating Rate Payer Calculation Amount equal to (x) the SRMP Triggerable Amount (for the relevant M(M)R Restructuring) of the Swaption Restructuring Cleared Transaction from which such Restructuring Cleared Transaction is created *multiplied by* (y) the Exercise Percentage.

The provisions of this Part C of the CDS Clearing Supplement shall cease to apply to such Restructuring Cleared Transaction upon its creation and such Restructuring Cleared Transaction shall instead be subject to and cleared in accordance with Part B of this CDS Clearing Supplement; provided, however, that:

- (i) the paragraph entitled 'Operation of each Underlying Swap Transaction' and the paragraph entitled 'Successors' of the iTraxx® Swaption Standard Terms Supplement shall continue to apply to such Restructuring Cleared Transaction; and
- (ii) notwithstanding (i) above, the Auction Settlement Date in respect of such Restructuring Cleared Transaction (if a valid Credit Event Notice is delivered or deemed to be delivered in respect of such Restructuring Cleared Transaction) shall be the later of (x) the Auction Settlement Date applicable to such Restructuring Cleared Transaction pursuant to Part B of the CDS Clearing Supplement and (y) the first Transaction Business Day following the Expiration Date.

7.4 Creation of Initial Single Name Cleared Transactions in respect of untriggered M(M)R Restructuring

If:

- (a) a DC Credit Event Announcement has been made that an M(M)R Restructuring Credit Event occurred on or after the Credit Event Backstop Date which would be applicable to a Component Transaction of the Underlying Index Transaction of an Exercise Cleared Transaction (subject to the operation of the paragraph entitled 'Operation of each Underlying Swap Transaction' of the iTraxx® Swaption Standard Terms Supplement); and
- (b) no valid Credit Event Notice was delivered in respect of such M(M)R Restructuring prior to the expiry of the applicable CEN Triggering Period,

then following Exercise, an Initial Single Name Cleared Transaction (as defined in Part B of this CDS Clearing Supplement) shall be deemed to have been entered into on the Expiration Date between each Clearing Member and LCH SA on the terms of the Component Transaction

File No. SR-LCH SA-2022-007 Page 271 of 439

(subject as set out below) in respect of the relevant Reference Entity of the Underlying Index Transaction. Subject to the below, such Initial Single Name Cleared Transaction shall be entered into by LCH SA and the relevant Clearing Member on the terms of the Single Name Cleared Transaction Confirmation (as defined in Part B of the CDS Clearing Supplement).

Each such Initial Single Name Cleared Transaction shall have a Floating Rate Payer Calculation Amount equal to:

- (a) the Swaption Notional Amount of the Exercise Cleared Transaction from which such Initial Single Name Cleared Transaction resulted multiplied by the 'Reference Entity Weighting' (under the Underlying Index Transaction referenced by such Exercise Cleared Transaction) of the relevant Reference Entity; multiplied by
- (b) the Exercise Percentage.

The provisions of this Part C of the CDS Clearing Supplement shall cease to apply to such Initial Single Name Cleared Transaction upon its creation and such Initial Single Name Cleared Transaction shall instead be subject to and cleared in accordance with Part B of this CDS Clearing Supplement; provided, however, that the paragraph entitled 'Successors' of the iTraxx® Swaption Standard Terms Supplement shall continue to apply to such Initial Single Name Cleared Transaction.

8. **DELIVERY OF NOTICES**

8.1 General Rules relating to Notices

(a) Methods of Delivery and deemed Time of Delivery

Subject to Sections 6.3 (Exercise and Abandonment by way of EEP) and 6.5(b)(b)(iv) and without prejudice to Section 1.38 (Requirements Regarding Notices) of the 2014 ISDA Credit Derivatives Definitions and the remaining provisions of the 2014 ISDA Credit Derivatives Definitions and of the 2006 Definitions (including, for the avoidance of doubt, in relation to notices permitted to be given orally), any notice or other communication in respect of any Cleared Transaction may be given in any manner described below and will be deemed effective as indicated:

- (i) if delivered in person or by courier, on the date and at the time it is delivered;
- (ii) if sent by facsimile transmission, on the date it is received by a responsible employee of the recipient in legible form (it being agreed that the burden of proving receipt will be on the sender and will not be met by a transmission report generated by the sender's facsimile machine);
- (iii) if sent by certified or registered mail (airmail, if overseas) or the equivalent (return receipt requested), on the date it is delivered or its delivery is attempted; or
- if sent by electronic messaging system (including e-mail or any other electronic access solution established by LCH SA for such purpose), on the date it is received,

File No. SR-LCH SA-2022-007 Page 272 of 439

unless the date of that delivery (or attempted delivery) or that receipt, as applicable, is not a Transaction Business Day or that communication is delivered (or attempted) or received, as applicable, pursuant to the above after 6.00 p.m. in the location of the recipient on a Transaction Business Day, in which case that communication will be deemed given and effective on the first following day that is a Transaction Business Day.

(b) Notices from or to LCH SA

Any such notice or communication given by LCH SA to a Clearing Member or *vice versa* shall (except where given via the EEP in accordance with Section 6.3 (*Exercise and Abandonment by way of EEP*)) be given to the address or number previously specified in or previously notified for the relevant purpose in accordance with the CDS Admission Agreement or the Procedures.

In the case of the occurrence of an EEP Failure Event, any notice or communication given by LCH SA to a Client of a Clearing Member as its Exercise Delegation Beneficiary or *vice versa* shall be given to the address or number provided by such Client or its Clearing Member to LCH SA and *vice versa* when the relevant Clearing Member registered such Client for the clearing of Index Swaption Cleared Transactions, or any other address or number duly notified thereafter by the such Client to LCH SA or *vice versa*.

(c) Notices between Clearing Members

Notices given by a Clearing Member to another Clearing Member (or their respective Exercise Delegation Beneficiaries) comprised in a relevant Matched Pair shall be given to the address or number notified by LCH SA to the deliver or pursuant to Section 5.1 (*Creation and Notification of Swaption Restructuring Matched Pairs*) or in a Protected Exercise Matched Pair Report pursuant to Section 6.1 (*Creation and Notification of Exercise Matched Pairs*) and Section 6.5(a)(ii) (*Access to the Protected Exercise Matched Pair Report*), as applicable.

Such notices shall only be deemed to be delivered effectively by LCH SA through the relevant Clearing Member as its designee (acting itself or through such Clearing Member's Exercise Delegation Beneficiary, if applicable) as against the recipient where the address or number so notified by LCH SA corresponds in all material respects to the address or number, as applicable, specified by such recipient in or pursuant to the CDS Admissions Agreement.

(d) No Obligation on LCH SA to verify Signatories

LCH SA shall have no obligation to verify the authority of any signatory of any notice delivered by any Clearing Member pursuant to this Section 8 (*Delivery of Notices*).

8.2 Failure to notify Matched Pairs

Notwithstanding Section 5.3 (*Triggering of Swaption Restructuring Cleared Transactions*) or Section 6.5(b) (*Consequences of EEP Failure*), if LCH SA does not notify the relevant Clearing

File No. SR-LCH SA-2022-007 Page 273 of 439

Members of Swaption Restructuring Matched Pairs and related information specified in Section 5.1 (*Creation and Notification of Swaption Restructuring Matched Pairs*) by the SRMP Notification Deadline or provide the Protected Exercise Matched Pair Report by the EMP Notification Deadline (as applicable) (or, where Section 6.5(b) (*Consequences of EEP Failure*) is applicable as a result of the occurrence of an EEP Failure Event that occurs after the EMP Notification Deadline, from the time of such EEP Failure Event until such time as LCH provides the Protected Exercise Matched Pair Report):

- the relevant Clearing Members may deliver Credit Event Notices and Notices to Exercise Movement Options in respect of their Swaption Restructuring Cleared Transaction to LCH SA and vice versa;
- (b) if Section 6.5(b) (Consequences of EEP Failure) is applicable as a result of the occurrence of an EEP Failure Event, the relevant Clearing Members (or their Exercise Delegation Beneficiaries as their designees if applicable) may deliver Swaption Notices in respect of their Exercise Cleared Transaction to LCH SA, and vice versa; and
- (c) the provisions of Section 9 (Matched Pair Designations) shall not apply and the terms of this CDS Clearing Supplement shall be construed accordingly.

8.3 Oral Notices

Notwithstanding the provisions of the 2006 Definitions and of the 2014 ISDA Credit Derivatives Definitions and any other provision hereof, where Swaption Notices, Credit Event Notices and Notices to Exercise Movement Option are to be delivered directly to LCH SA, such notices may not be delivered orally or by telephone.

8.4 Disputes as to Notices

If any Clearing Member comprised in a Matched Pair where one such party is acting as designee of LCH SA disputes the effective delivery in accordance with the terms of the relevant Cleared Transactions of any notice delivered directly between Clearing Members pursuant to Section 5 (*Restructuring*) or Section 6.5(b) (*Consequences of EEP Failure*):

- (a) LCH SA shall be entitled in accordance with the Procedures to calculate and call for Margin with respect to each such Clearing Member on the basis of the maximum requirement for such Clearing Member that could result from any foreseeable outcome of such dispute;
- (b) following final resolution of such dispute by arbitration or by litigation, as applicable, in accordance with the CDS Dispute Resolution Protocol, subject to the provisions of Sections 8 and 9 of the Procedures, the Clearing Members comprised in the relevant Matched Pair shall take such actions with respect to the affected Exercise Cleared Transactions or Swaption Restructuring Cleared Transactions, as the case may be, the subject of such dispute as LCH SA determines appropriate to give effect to any relevant binding resolution; and
- (c) without prejudice to its obligations upon final resolution of the dispute, pending final resolution of any such dispute, LCH SA shall not be obliged to take any step pursuant

File No. SR-LCH SA-2022-007 Page 274 of 439

to the terms of the relevant Exercise Cleared Transactions or Swaption Restructuring Cleared Transactions which would be required to have been taken by it had the relevant notice been validly delivered.

Each relevant Clearing Member must promptly notify LCH SA of any such disputes (in the form set out at Appendix V hereto).

9. MATCHED PAIR DESIGNATIONS

Creation of Matched Pairs 9.1

LCH SA will create Matched Pairs where required to do so pursuant to Section 5.1 (Creation and Notification of Swaption Restructuring Matched Pairs) and Section 6.1 (Creation and Notification of Exercise Matched Pairs) using a matching procedure that matches Swaption Sellers with Swaption Buyers pursuant to an algorithm incorporating the following principles:

- the procedure seeks to create Matched Pairs between the same Clearing Member to the extent it is possible to do so before creating Matched Pairs between different Clearing Members and, for this purpose, in the context of CCMs, the procedure will create Matched Pairs separately for CCMs and their CCM Clients (individually where the CCM Client is allocated to a single CCM Client Margin Account or together where a set of CCM Clients is allocated to a single CCM Client Margin Account) and Clearing Member will be construed accordingly;
- (b) the procedure seeks to minimise the number of Matched Pairs (and accordingly, largest positions will be matched first);
- (c) each Matched Pair will, to the extent possible, have an aggregate Swaption Restructuring Matched Pair Amount or Exercise Matched Pair Amount (as applicable), which meets the criteria as set out in a Clearing Notice: and
- (d) LCH SA will allocate a Matched Pair Amount to each Matched Pair such that:
 - the sum of all Swaption Restructuring Matched Pair Amounts or Exercise Matched Pair Amounts (as applicable) of each Swaption Buyer under the Swaption Restructuring Cleared Transactions or Exercise Cleared Transactions matched on the relevant date pursuant to this Section 9.1 is equal to the aggregate of the Swaption Notional Amounts of the Index Swaption Cleared Transactions to which such Swaption Buyer is a party which are to be matched on such date pursuant to this Section 9.1; and
 - (ii) the sum of all Swaption Restructuring Matched Pair Amounts or Exercise Matched Pair Amounts (as applicable) of each Swaption Seller under the Swaption Restructuring Cleared Transactions or Exercise Cleared Transactions matched on the relevant date pursuant to this Section 9.1 is equal to the aggregate of the Swaption Notional Amounts of the Index Swaption Cleared Transactions to which such Swaption Seller is a party which are to be matched on such date pursuant to this Section 9.1.

9.2 Registration of new Swaption Restructuring Cleared Transactions and Exercise Cleared Transactions and Removal of original Index Swaption Cleared Transactions

To the extent that any Swaption Restructuring Cleared Transaction created pursuant to Section 5.2 (*Creation of Swaption Restructuring Cleared Transaction*) or any Exercise Cleared Transaction created pursuant to Section 6.2 (*Creation of Exercise Cleared Transactions*) is not automatically registered in accordance with the DTCC Rules, LCH SA shall register such new Swaption Restructuring Cleared Transaction or Exercise Cleared Transaction (as applicable) in the TIW in accordance with the DTCC Rules prior to 6.00 p.m. on the date on which the SRMP Notification Deadline or EMP Notification Deadline (as applicable) falls. In respect of CCMs and CCM Clients only, such registration by LCH SA shall also result in the automatic registration of any amendments made to the corresponding CCM Client Transactions.

In addition, LCH SA will, on behalf of the relevant Clearing Member, send an "Exit" message to the TIW in accordance with the DTCC Rules to terminate and remove the corresponding original Index Swaption Cleared Transactions from which such Swaption Restructuring Cleared Transactions or Exercise Cleared Transactions were created from the TIW prior to 10.00 p.m. on the date on which the SRMP Notification Deadline or EMP Notification Deadline (as applicable) falls.

9.3 Resetting of Swaption Trade Date

If LCH SA determines in respect of any Swaption Restructuring Cleared Transaction created pursuant to Section 5.2 (*Creation of Swaption Restructuring Cleared Transactions*) or any Exercise Cleared Transaction created pursuant to Section 6.2 (*Creation of Exercise Cleared Transactions*) that the Swaption Trade Date for such Swaption Restructuring Cleared Transaction or Exercise Cleared Transaction would be after a DC Announcement Coverage Cut-off Date whereas the Swaption Trade Date for the Cleared Transaction(s) from which such Swaption Restructuring Cleared Transaction or Exercise Cleared Transaction was created would have been prior to such DC Announcement Coverage Cut-off Date, LCH SA shall take such action as it deems necessary to ensure that such DC Credit Event Announcement is applicable to such Swaption Restructuring Cleared Transaction or Exercise Cleared Transaction, including, without limitation, specifying an earlier Swaption Trade Date for such Swaption Restructuring Cleared Transaction.

9.4 Matched Buyer and Seller Contracts

- (a) In respect of each Matched Buyer Contract which is the subject of a Matched Pair, LCH SA, pursuant to Section 11.2(c)(iv) of the 2014 ISDA Credit Derivatives Definitions (amended as set out at Section 2.3 (Amendments to 2014 ISDA Credit Derivatives Definitions)), as designator, shall be deemed to have designated Matched Seller in such Matched Pair as its designee to receive on its behalf from, and to deliver on its behalf to, Matched Buyer of the Matched Pair any applicable notices or certifications) in accordance with the terms of the applicable Cleared Transaction (including, without limitation, Exercise Notices and Abandonment Notices).
- (b) In respect of each Matched Seller Contract which is the subject of a Matched Pair, LCH SA, pursuant to Section 11.2(c)(iv) of the 2014 ISDA Credit Derivatives Definitions (as amended pursuant to Section 2.3 (Amendments to 2014 ISDA Credit Derivatives)

File No. SR-LCH SA-2022-007 Page 276 of 439

Definitions) above), as designator, shall be deemed to have designated Matched Buyer in such Matched Pair as its designee to receive on its behalf from, and to deliver on its behalf to, Matched Seller of the Matched Pair any applicable notices or certifications) in accordance with the terms of the applicable Cleared Transaction (including, without limitation, Exercise Notices and Abandonment Notices).

Formatted: Header, Tab stops: Not at 2.54 cm + 7.7 cm + 15.4 cm

9.5 Exercise of Rights

In relation to each Matched Pair:

- (a) the exercise of any rights by Matched Buyer against LCH SA under a Matched Buyer Contract shall be deemed to constitute the exercise of equal and simultaneous rights by LCH SA against Matched Seller under the Matched Seller Contract of the relevant Matched Pair; and
- (b) the exercise of any rights by Matched Seller against LCH SA under a Matched Seller Contract shall be deemed to constitute the exercise of equal and simultaneous rights by LCH SA against Matched Buyer under the Matched Buyer Contract of the relevant Matched Pair.

9.6 Clearing Member matched with Itself

In the event that Matched Buyer and Matched Seller of a Matched Pair pursuant to this Section 9 (Matched Pair Designations) is the same Clearing Member, such Clearing Member shall be deemed to have sent a notice from itself in its role as Matched Buyer to itself in its role as Matched Seller (and vice versa) upon such Clearing Member sending a Clearing Member Notice to LCH SA pursuant to Section 5.7 (a) (*Duty to deliver Swaption Restructuring Clearing Member Notices*) or Section 6.5(c) (*Duty to deliver Swaption Clearing Member Notices*) above in respect of such notice.

9.7 Notices

In relation to each Matched Pair:

- (a) where Matched Buyer validly delivers or serves any notice to Matched Seller as designee of LCH SA in accordance with the terms of a relevant Matched Buyer Contract, such notice shall additionally be effective as a notice given by such Matched Buyer as designee of LCH SA to Matched Seller for the purposes of the relevant Matched Seller Contract; and
- (b) where Matched Seller validly delivers or serves any notice to Matched Buyer as designee of LCH SA in accordance with the terms of a relevant Matched Seller Contract, such notice shall additionally be effective as a notice given by such Matched Seller as designee of LCH SA to Matched Buyer for the purposes of the relevant Matched Buyer Contract.

10. MANDATORY PROVISIONS FOR CCM CLIENT TRANSACTIONS

In Appendix VIII, certain provisions are set-out (the "Mandatory Provisions") for incorporation into a CCM Client Transaction between a CCM and its CCM Client that

File No. SR-LCH SA-2022-007 Page 277 of 439

corresponds to a CCM Client Cleared Transaction registered in the CCM Client Trade Account in the name of such CCM for such CCM Client. The CDS Clearing Supplement and these Mandatory Provisions have been drafted so as to complement each other.

LCH SA shall not be responsible for any loss suffered or expense incurred by a CCM or any CCM Client as a result of the inclusion or non-inclusion of the Mandatory Provisions in the CCM Client Transaction Documents.

11. AMENDMENTS

LCH SA may amend the provisions of this CDS Clearing Supplement (including, without limitation, the Mandatory Provisions) from time to time so as to comply with any legal or regulatory developments or any recommendations adopted by the industry in respect of CDS, Swaptions or Cleared Transactions or CCM Client Transactions, as applicable, or so as to reflect any technological advancements, in each case in accordance with the provisions of Section 1.2.2 (Modification) of Chapter 2 (General Provisions) of the CDS Clearing Rule Book.

12. FORMS OF NOTICES

A form of Exercise Notice, Abandonment Notice, Credit Event Notice, Notice to Exercise Movement Option, notice of dispute relating to any Matched Pair as contemplated by Section 8.4 (*Disputes as to Notices*) is set out in Appendix I, II, III, IV and V respectively hereto.

Any of the above referenced notices shall be delivered in substantially the form appended hereto, provided, for the avoidance of doubt, that such notices may refer to multiple transactions and may have certain firm-specific variations.

For the avoidance of doubt, the above referenced notices shall be governed by and construed in accordance with English law.

13. EXCLUSION OF LIABILITY

Without prejudice to the provisions of Article 1.2.10.3 of the CDS Clearing Rule Book:

(a) No liability for Failure of Designee to perform in respect of Matched Pair

Without prejudice to its obligations under or in respect of a Cleared Transaction, LCH SA shall not be liable for any loss or cost arising out of any failure of any Clearing Member comprised in a Matched Pair to perform its obligations as designee of LCH SA against a related Matched Buyer or Matched Seller, as applicable.

(b) No liability for Exercise or Abandonment by Exercise Delegation Beneficiary

LCH SA shall have no liability to a Clearing Member which has delegated to an Exercise Delegation Beneficiary its power to Exercise or Abandon Exercise Cleared Transactions in accordance with Section 6.4 (*Delegation by Clearing Members to Clients*) above for any loss, cost or expense arising out of any failure of such Exercise Delegation Beneficiary to perform its obligations in relation to such delegation or in connection with or arising from the Exercise or Abandonment (or attempted Exercise or

File No. SR-LCH SA-2022-007 Page 278 of 439

Abandonment) of an Exercise Cleared Transaction by such Exercise Delegation Beneficiary.

Formatted: Header, Tab stops: Not at 2.54 cm + 7.7 cm + 15.4 cm

(c) No liability for Fault of Third Party or Force Majeure

LCH SA shall have no liability to any person where Swaption Restructuring Cleared Transactions or Exercise Cleared Transactions are not or are improperly created, Index Swaption Cleared Transactions, Swaption Restructuring Cleared Transactions or Exercise Cleared Transactions are not or are improperly terminated or where the Exercise process or the Movement Option process is not or is improperly implemented, because of a third party's fault or a force majeure event. In particular, LCH SA shall not incur any liability arising as a result of any action or omission of DTCC.

(d) No Obligation to verify Notices received

LCH SA shall have no responsibility to verify the contents of any notice received by it from any Clearing Member (or from an Exercise Delegation Beneficiary of a Clearing Member) under the terms of any Cleared Transaction.

14. **DISPUTE RESOLUTION**

For the avoidance of doubt, all Disputes shall be referred to and finally resolved by arbitration or by litigation, as applicable, in accordance with the CDS Dispute Resolution Protocol, subject to the provisions of Sections 8 and 9 of the Procedures.

15. GOVERNING LAW

For the avoidance of doubt, the governing law applicable to this CDS Clearing Supplement (excluding the Mandatory Provisions to the extent that such terms are incorporated by reference in the CCM Client Transaction Documents entered into between a CCM and its CCM Client in respect of a CCM Client Transaction), the 2014 ISDA Credit Derivatives Definitions, the 2006 Definitions and any Cleared Transactions (and any related definitions or Clearing Notices issued in respect of the CDS Clearing Supplement, the 2014 ISDA Credit Derivatives Definitions, the 2006 Definitions or any Cleared Transactions) and any non-contractual obligations arising out of, relating to or having any connection with them shall be as set out in Section 1.2.14 (Governing Law) of the CDS Clearing Rule Book.

File No. SR-LCH SA-2022-007 Page 279 of 439

PUBLIC

APPENDIX I: FORM OF EXERCISE NOTICE

To: [Exercise Matched Pair Counterparty Address and Contact Information]

[To/Copy to:]

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details] [Date]	
EXERCISE NOTICE:	
Exercise Cleared Transaction Details: As set out in the Schedule hereto ¹ .	
Reference is made to the Exercise Cleared Transaction[s] described in the Schedule hereto (the Transaction[s]). Capitalised terms used and not otherwise defined in this letter shall have the meanings given them in the confirmation of the relevant Transaction and in the CDS Clearing Documentation (as defined in such confirmation).	
This letter is our Exercise Notice to you to confirm [our telephonic notice to you today] that we have elected to exercise our rights under the Transaction[s] as set out in the Schedule hereto.	
Sincerely	
[Clearing Member]	
Name:	
Title:	

Formatted: Line spacing: single

 $^{^{\}rm 1}$ A single Exercise Notice may be submitted for multiple trades in respect of the same Counterparty

File No. SR-LCH SA-2022-007 Page 280 of 439

PUBLIC

SCHEDULE

Exercise Cleared Transaction Details

Exercise Matched Pair ID	Underlying Index	Swaption Type		Expiration Date	Swaption Notional Amount / Original Notional Amount	Amount of Original Notional Amount exercised
[●]	iTraxx® Europe [index name] Series [●] Version [●]	[Payer Receiver]	/	[●]	[●]	[●]
[•]	CDX [index name] Series [•] Version [•]	[Payer Receiver]	/	[●]	[●]	[●]

The Underlying Index, Swaption Type and Swaption Notional Amount/Original Notional Amount for the Transaction[s] have been included in this Exercise Notice for identification purposes only. Any error in such items shall not invalidate this Exercise Notice and the Transaction[s] shall be exercised in the amount specified above notwithstanding any such error.

File No. SR-LCH SA-2022-007 Page 281 of 439

PUBLIC

APPENDIX II: FORM OF ABANDONMENT NOTICE

 ${\tt To:} \ [\textit{Exercise Matched Pair Counterparty Address and Contact Information}]$

[To/Copy to:]

Name: Title:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]	
ABANDONMENT NOTICE:	
Exercise Cleared Transaction Details: As set out in the Schedule hereto ² .	
Reference is made to the Exercise Cleared Transaction[s] described in the Schedule hereto (the Transaction[s]). Capitalised terms used and not otherwise defined in this letter shall have the meanings given them in the confirmation of the relevant Transaction and in the CDS Clearing Documentation (as defined in such confirmation).	
This letter is our Abandonment Notice to you to confirm [our telephonic notice to you today] that we have elected not to exercise our rights under the Transaction[s] and that the Transaction[s] shall instead be terminated in whole.	
Sincerely	
[Clearing Member]	

Formatted: Line spacing: single

 $^{^{2}}$ A single Abandonment Notice may be submitted for multiple trades in respect of the same Counterparty

PUBLIC

SCHEDULE

Exercise Cleared Transaction Details

Exercise Matched Pair ID	Underlying Index	Swaption Type	Expiration Date	Swaption Notional Amount / Original Notional Amount	
[●]	iTraxx® Europe [index name] Series [●] Version [●]	[Payer / Receiver]	[●]	[●]	
[n s	CDX index name] series [Version]	[Payer / Receiver]	•]	·] [·	•]

The Underlying Index, Swaption Type and Swaption Notional Amount/Original Notional Amount for the Transaction[s] have been included in this Exercise Notice for identification purposes only. Any error in such items shall not invalidate this Abandonment Notice and the Transaction[s] shall be terminated pursuant to this Abandonment Notice notwithstanding any such error.

File No. SR-LCH SA-2022-007 Page 283 of 439

PUBLIC

APPENDIX III: FORM OF CREDIT EVENT NOTICE

To: [Swaption Restructuring Matched Pair Counterparty Address and Contact Information]
[To/Copy to:]
LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]
[Date]

CREDIT EVENT NOTICE:

Swaption Restructuring Cleared Transaction Details: As set out in the Schedule hereto³.

Reference is made to the Swaption Restructuring Cleared Transaction[s] described in the Schedule hereto (the **Transaction[s]**). Capitalised terms used and not otherwise defined in this letter shall have the meanings given them in the confirmation of the relevant Transaction and in the CDS Clearing Documentation (as defined in such confirmation).

This letter is our Credit Event Notice to you that an M(M)R Restructuring occurred with respect to [insert name of Reference Entity] on or about [insert date], when [describe Credit Event].

Nothing in this letter shall be construed of a waiver of any rights we may have with respect to the

Transaction.		
Sincerely		
[Clearing Member]		
Name:		
Title:		

<u>10219615309-v3</u> <u>70-41024903</u>

Formatted: Line spacing: single

 $^{^{3}}$ A single Credit Event Notice may be submitted for multiple trades in respect of the same Counterparty

File No. SR-LCH SA-2022-007 Page 284 of 439

PUBLIC

SCHEDULE

Swaption Restructuring Cleared Transaction Details

[Clearing Member] acting as CDS Seller/Buyer	Swaption Restructuring Matched Pair ID	Reference Entity	Swaption Trade Date	Triggered Amount⁴
[Seller] [Buyer]	[●]	[●]	[●]	[●]

Formatted: Line spacing: single

 $^{^{\}rm 4}$ Where different to the relevant SRMP Triggerable Amount

File No. SR-LCH SA-2022-007 Page 285 of 439

PUBLIC

APPENDIX IV: FORM OF NOTICE TO EXERCISE MOVEMENT OPTION

To: [Swaption Restructuring Matched Pair Counterparty Address and Contact Information]
[To/Copy to:]

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

[Date]

Dear Sir/Madam

Notice to Exercise Movement Option

Swaption Restructuring Cleared Transaction Details: As set out in the Schedule hereto⁵.

Reference is made to: (a) the Swaption Restructuring Cleared Transaction[s] described in the Schedule hereto (the **Transaction[s]**) between [], as Matched Seller, and [], as Matched Buyer; (b) the Credit Event Notice previously delivered to you on [insert date]; and (c) the occurrence of the No Auction Announcement Date on [insert date] pursuant to Section 6.11(b) or Section 6.11(c)(ii) of the 2014 ISDA Credit Derivatives Definitions (the **Definitions**).

This letter constitutes a Notice to Exercise Movement Option. Any capitalised term not otherwise defined in this letter will have the meaning, if any, assigned to such term in the confirmation of the relevant Transaction and in the CDS Clearing Documentation (as defined in such confirmation) or, if no meaning is specified therein, in the Definitions.

We hereby exercise the Movement Option, confirm that the relevant transaction created from [each / the] Transaction[s] pursuant to Section 7 (Settlement) of Part C of the CDS Clearing Supplement following exercise of the Transaction will be settled in accordance with the relevant Credit Derivatives Auction Settlement Terms specified in the column entitled "Auction Settlement Terms" corresponding to such Transaction in the Schedule hereto and require performance by you in accordance therewith.

ours faithfully,		
Matched Buyer/Matched Seller]		
Name: Fitle:		

Formatted: Space After: 0 pt

Formatted: Line spacing: single

⁵ A single Notice to Exercise Movement Option may be submitted for multiple trades in respect of the same Counterparty

File No. SR-LCH SA-2022-007 Page 286 of ...

PUBLIC

SCHEDULE

Swaption Restructuring Cleared Transaction Details

Swaption Restructuring Matched Pair ID	Reference Entity	Swaption Trade Date	Auction Settlement Terms
[●]	[●]	[●]	[●]

File No. SR-LCH SA-2022-007 Page 287 of ...

PUBLIC

APPENDIX V: FORM OF NOTICE OF DISPUTE RELATING TO ANY SWAPTION RESTRUCTURING / EXERCISE MATCHED PAIR

To:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]
[Date]
[Exercise Matched Pair ID: [●]] / [Swaption Restructuring Matched Pair ID: [●]]
Trade ID: [●]
Dear Sir/Madam
Notice of dispute relating to [insert details of the relevant Exercise Matched Pairs / Swaption Restructuring Matched Pairs subject to a dispute]
Reference is made to Section 8.4 (<i>Disputes as to Notices</i>) of Part C of the CDS Clearing Supplement. Defined terms shall have the meanings assigned to them in Part C of the CDS Clearing Supplement, unless otherwise defined herein.
In accordance with Section 8.4 (<i>Disputes as to Notices</i>) of Part C of the CDS Clearing Supplement, notice is hereby given of the following dispute(s):
[insert details of Exercise Matched Pair(s) / Swaption Restructuring Matched Pair(s) affected and the relevant dispute].
This notice (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this agreement or its formation) shall be governed by and construed in accordance with English law.
Yours faithfully,
[Matched Buyer]/[Matched Seller]
Title:
nue.

File No. SR-LCH SA-2022-007 Page 288 of 439

PUBLIC

APPENDIX VI: FORM OF NOTICE OF CLEARING MEMBER COMMUNICATIONS FAILURE EVENT PURSUANT TO SECTION 6.10 (Clearing Member Communications Failure Event) OR CCM CLIENT COMMUNICATIONS FAILURE EVENT PURSUANT TO MANDATORY PROVISION 5.7 (CCM Client Communications Failure Event)

To:

LCH SA 18, rue du Quatre Septembre 75002 Paris France [Contact details]

[date]

Dear Sir/Madam

Notice certifying occurrence of a [Clearing Member Communications Failure Event] [CCM Client Communications Failure Event]

Reference is made to [Section 6.10(c) (Clearing Member or Exercise Delegation Beneficiary to notify LCH SA of Occurrence of Clearing Member Communications Failure Event) of the CDS Clearing Supplement] [Mandatory Provision 5.7(b) (Client to notify LCH SA of occurrence of CCM Client Communications Failure Event) in respect of the CDS Clearing Supplement]. Defined terms shall have the meanings assigned to them in the CDS Clearing Supplement, unless otherwise defined herein.

In accordance with Section 6.10(c) (Clearing Member or Exercise Delegation Beneficiary to notify LCH SA of Occurrence of Clearing Member Communications Failure Event) of the CDS Clearing Supplement, notice is hereby given that we are affected by a Clearing Member Communications Failure Event [insert details of such failure].]

[In accordance with Mandatory Provision 5.7(b) (Client to notify LCH SA of occurrence of CCM Client Communications Failure Event) in respect of the CDS Clearing Supplement, notice is hereby given that we are affected by a CCM Client Communications Failure Event [insert details of such failure].]

This notice (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this agreement or its formation) shall be governed by and construed in accordance with English law.

Yours faithfully,

[Clearing Member / Exercise Delegation Beneficiary]

File No.	SR-LCH	SA	-202	22-(007
	Pa	ge	289	of 4	439

DI	IRI	10

Name:

Title:

[Signed by a senior officer (such as a managing director or equivalent) on behalf of the Clearing Member / Exercise Delegation Beneficiary]

File No. SR-LCH SA-2022-007 Page 290 of 439

PUBLIC

Formatted: Header, Right

APPENDIX VII: FORM OF NOTICE FOR CEASING TO BE SUBJECT TO A CLEARING MEMBER COMMUNICATIONS FAILURE EVENT PURSUANT TO SECTION 6.10 (Clearing Member Communications Failure Event) OR CCM CLIENT COMMUNICATIONS FAILURE EVENT PURSUANT TO MANDATORY PROVISION 5.7 (CCM Client Communications Failure Event)

To:

LCH SA
18, rue du Quatre Septembre
75002 Paris
France
[Contact details]

[date]

Dear Sir/Madam

Notice of cessation of [Clearing Member Communications Failure Event] [CCM Client Communications Failure Event]

Reference is made to Section 6.10(d) (Notification of Resolution of Clearing Member Communications Failure Event) of the CDS Clearing Supplement and the notice certifying the occurrence of a Clearing Member Communications Failure Event delivered by us to LCH SA on [·] (the Notice of Clearing Member Communications Failure Event).] [Reference is made to Mandatory Provison 5.7(c) (Notification of Resolution of CCM Client Communications Failure Event) in respect of the CDS Clearing Supplement and the notice certifying the occurrence of a CCM Client Communications Failure Event delivered by us to LCH SA on [·] (the Notice of CCM Client Communications Failure Event). Defined terms shall have the meanings assigned to them in the CDS Clearing Supplement, unless otherwise defined herein.

In accordance with Section 6.10(d) (Notification of Resolution of Clearing Member Communications Failure Event) of the CDS Clearing Supplement, notice is hereby given that we are no longer subject to the relevant Clearing Member Communications Failure Event described in the Notice of Clearing Member Communications Failure Event.

[In accordance with Mandatory Provision 5.7(c) (Notification of Resolution of CCM Client Communications Failure Event) in respect of the CDS Clearing Supplement, notice is hereby given that we are no longer subject to the relevant CCM Client Communications Failure Event described in the Notice of CCM Client Communications Failure Event.]

This notice (and any dispute, controversy, proceedings or claim of whatever nature arising out of or in any way relating to this agreement or its formation) shall be governed by and construed in accordance with English law.

Yours faithfully,

10219615309-v3 70-41024903

File No. SR-LCH SA-2022-007 Page 291 of 439

PUBLIC

ſ	
Formatted: Header, Right	
i ormattea. ricader, Right	

[Clearing Member / Exercise Delegation Beneficiary]						
Name						
Name:						
Title						

Formatted: Header, Right

APPENDIX VIII: CCM CLIENT TRANSACTION REQUIREMENTS

The following provisions (the "Mandatory Provisions") are to be incorporated into a CCM Client Transaction between a CCM and its CCM Client that corresponds to a CCM Client Cleared Transaction (which is an Index Swaption Cleared Transaction, a Swaption Restructuring Cleared Transaction or an Exercise Cleared Transaction) registered in the CCM Client Trade Account in the name of such CCM for such CCM Client. The terms of the corresponding CCM Client Cleared Transaction will be governed by Part C of the CDS Clearing Supplement. Part C of the CDS Clearing Supplement and these Mandatory Provisions have been drafted so as to complement each other.

LCH SA shall not be responsible for any loss suffered or expense incurred by a CCM or any CCM Client as a result of the inclusion in the CCM Client Transaction Documents of the requirements set-out in this Appendix VIII.

The Mandatory Provisions, when they are incorporated into any CCM Client Transaction Documents, shall be governed by and construed in accordance with the governing law applicable to such CCM Client Transaction Documents of which they form part, or if different and applicable, in accordance with such CCM Client Transaction Documents, the governing law applicable to transactions entered into under such CCM Client Transaction Documents. The Mandatory Provisions shall be subject to such dispute resolution mechanisms and procedures and such courts or other forum for hearing disputes as are applicable in respect of such CCM Client Transaction Documents of which they form part. Each CCM and its CCM Client to which the Mandatory Provisions apply will waive any right to object to any such choice of law or proceedings on the basis of *forum non conveniens*, that the governing law or forum is not specified on the face of this document or otherwise.

1. Defined Terms

Terms used in the Mandatory Provisions and not otherwise defined herein or in the iTraxx® Swaption Standard Terms Supplement or, as the case may be, the CDX Swaption Standard Terms Supplement, the 2006 ISDA Definitions or in the 2014 ISDA Credit Derivatives Definitions shall have the meanings given to them in Part C of the CDS Clearing Supplement.

2. Terms of CCM Client Transactions

2.1 2006 Definitions and 2014 ISDA Credit Derivatives Definitions

The definitions and provisions contained in the 2006 Definitions and in the 2014 ISDA Credit Derivatives Definitions are incorporated into each CCM Client Transaction.

2.2 Premium Payment Date

Notwithstanding anything to the contrary in the 2006 Definitions, if the Premium Payment Date specified in the CCM Client Transaction Documents in respect of any CCM Client Transaction is a date falling after the Clearing Day on which the Cleared Transactions related to such CCM Client Transaction are created by novation pursuant to Title III (Clearing Operations) of the CDS Clearing Rule Book, the Premium Payment Date in respect of such CCM Client Transaction shall be deemed to be the Transaction Business Day which is also a

10219615309-v3 70-41024903

File No. SR-LCH SA-2022-007 Page 293 of 439

PUBLIC

Formatted: Header, Right

Clearing Day immediately following the Clearing Day on which the Cleared Transactions relating to such CCM Client Transaction are created.

2.3 Abandonment Notices

On the Expiration Date, Swaption Buyer may notify Swaption Seller (which such notification may be given orally, including by telephone, if notices may generally be given orally pursuant to the terms of the CCM Client Transaction) that the CCM Client Transaction specified in such notice shall be terminated in whole and that no further amounts shall become due and payable by Swaption Buyer to Swaption Seller or *vice versa* in respect of such CCM Client Transaction and that notice will be deemed to be irrevocable (such notice, an "Abandonment Notice"). If applicable, Swaption Buyer will execute and deliver to Swaption Seller a written confirmation confirming the substance of any telephonic notice within one Exercise Business Day of that notice. Failure to provide that written confirmation will not affect the validity of the telephonic notice.

2.4 ISDA 2019 NTCE Protocol

The Protocol Effectiveness Condition (as defined in the 2019 NTCE Protocol) being satisfied, then from and including the Implementation Date (as defined in the 2019 NTCE Protocol) the amendments set forth in Schedule 1 (*Amendments*) of the 2019 NTCE Protocol shall apply to each CCM Client Transaction which incorporates the 2014 ISDA Credit Derivatives Definitions and each such CCM Client Transaction shall be a Protocol Covered Transaction for the purposes of the 2019 NTCE Protocol provided that:

- (a) the date on which the Index Swaption CCM Client Cleared Transaction corresponding to such CCM Client Transaction is registered in the TIW is prior to the Implementation Date:
- (b) such CCM Client Transaction is not an Excluded Transaction (as defined in the 2019 NTCE Protocol) provided that if the Underlying Swap Transaction of such CCM Client Transaction relates to more than one Reference Entity and would otherwise be a Protocol Covered Transaction but for the inclusion of any Affected Portion (as defined in the 2019 NTCE Protocol) therein, such Affected Portion shall be exclude for the purposes of this Mandatory Provision 2.4 and the remaining portion of such CCM Client Transaction shall be deemed to be a Protocol Covered Transaction;
- (c) the Expiration Date in respect of such CCM Client Transaction is on or after the Implementation Date; and
- (d) no Event Determination Date has occurred in respect of the Underlying Swap Transaction of such CCM Client Transaction prior to the Implementation Date.

3. Additional CCM Client Transactions, Compression and Succession Events

3.1 Creation of Additional CCM Client Transactions

Immediately following the creation of Matched Pairs by LCH SA pursuant to Section 9.1 (*Creation of Matched Pairs*) of the CDS Clearing Supplement, if a CCM Client Transaction

Formatted: Header, Right

has been specified to have been split into or replaced by two or more separate CCM Client Transactions in the TIW as a result of the creation of such Matched Pairs, such CCM Client Transaction shall be split into or terminated and replaced by two or more (as applicable) corresponding CCM Client Transactions. The Swaption Notional Amount of each such CCM Client Transaction (and the Original Notional Amount for the relevant Underlying Swap Transaction) shall correspond to the Swaption Notional Amount (and the Original Notional Amount) specified in TIW for such CCM Client Transaction.

3.2 Reversal of Creation of Additional CCM Client Transactions

If a CCM Client Transaction has been split into two or more CCM Client Transactions pursuant to Mandatory Provision 3.1 (*Creation of Additional CCM Client Transactions*) above and the relevant DC Credit Event Announcement in respect of a Restructuring Credit Event that led to the creation of the Matched Pairs is reversed such that Section 5.9 (*Reversal of DC Credit Event Announcements*) of Part C of the CDS Clearing Supplement applies, then, subject to Section 10.2(a)(i) of the 2014 ISDA Credit Derivatives Definitions and provided that no Subsequent Restructuring has occurred, any additional CCM Client Transactions created pursuant to Mandatory Provision 3.1 (*Creation of Additional CCM Client Transactions*) above shall be deemed not to have been created. Any Credit Event Notices delivered in connection with such CCM Client Transactions in relation to such Restructuring Credit Event shall be deemed to be ineffective.

3.3 Compression of CCM Client Transactions

If two or more CCM Client Transactions are specified in TIW to have been compressed into a single CCM Client Transaction pursuant to Chapter 3 (*Compression*) of Title III (*Clearing Operations*) of the CDS Clearing Rule Book, such CCM Client Transactions shall be compressed into a single CCM Client Transaction with a Swaption Notional Amount (and an Original Notional Amount for the relevant Underlying Swap Transaction) equal to the aggregate Swaption Notional Amounts of the original CCM Client Transactions.

4. Validity of Notices

Save if and as expressly stated to the contrary in the Mandatory Provisions, any notice delivered by a CCM Client to its CCM in respect of a CCM Client Transaction (including, without limitation, a Credit Event Notice or Notice to Exercise Movement Option) at a time or in a manner in which the CCM would not be permitted to deliver such a notice to LCH SA (or to a relevant Matched Buyer or Matched Seller as designee of LCH SA (as applicable)) in respect of the corresponding CCM Client Cleared Transaction pursuant to the terms of Part C of the CDS Clearing Supplement shall be deemed not to have been delivered.

5. Designation of CCM Client by CCM for Exercise or Abandonment of Exercise Cleared Transactions

5.1 Designation by CCM

CCM and CCM Client hereby agree that CCM shall designate the CCM Client as its Exercise Delegation Beneficiary in respect of the CCM Client Cleared Transaction(s) relating to the

10219615309-v3 70-41024903

Formatted: Header, Right

CCM Client Transaction(s) between CCM and CCM Client pursuant to Section 6.4 (*Delegation by Clearing Members to CCM Clients*) of Part C of the CDS Clearing Supplement for the purpose of the Exercise (in whole or in part) or Abandonment of such CCM Client Cleared Transactions(s) and receipt of Swaption Notices on its behalf.

5.2 Exercise Notices and Abandonment Notices delivered in respect of CCM Client Cleared Transaction

Neither the CCM nor CCM Client may deliver Swaption Notices in relation to the CCM Client Transaction corresponding to a CCM Client Cleared Transaction in respect of which CCM Client has been designated by the CCM as its Exercise Delegation Beneficiary in accordance with Mandatory Provision 5.1 (*Designation by CCM*) above. Instead, if CCM Client as Exercise Delegation Beneficiary of the CCM delivers or receives (or is deemed to deliver or receive) a valid Swaption Notice in respect of the CCM Client Cleared Transaction corresponding to such CCM Client Transaction, such notice shall be deemed also to be a valid Swaption Notice for the purposes of such CCM Client Transaction.

5.3 Exercise and Abandonment by way of EEP

Subject to Mandatory Provision 5.4 (Consequences of EEP Failure) and Mandatory Provision 5.7 (CCM Client Communications Failure Event) below, any submission of an Option Intent by CCM Client in respect of a CCM Client Cleared Transaction in respect of which such CCM Client has been designated as Exercise Delegation Beneficiary shall be made via its Client Portal Account in the EEP. If (i) the CCM Client submits such an Option Intent via its Client Portal Account (ii) the Submission Time of such Option Intent is prior to the Submission Deadline on the Expiration Date and (iii) LCH SA has completed those steps necessary to make such Option Intent available for viewing in the EEP, including validation of the EEP Controls, such submission shall be deemed to constitute the delivery by the CCM (as Matched Buyer) of a valid Exercise Notice or Abandonment Notice (as applicable) in respect of the CCM Client Cleared Transactions pursuant to Section 6.4 (Delegation by Clearing Members to CCM Clients) of Part C of the CDS Clearing Supplement. An Option Intent shall become irrevocable by CCM Client as from the Submission Time. For the avoidance of doubt, CCM Client shall not be able to require a Force Submission in accordance with Section 5 of the Procedures after the Submission Deadline, including if the relevant Option Intent is deemed illogical further to EEP Controls carried out at or after the Submission Deadline. The deemed time of delivery of such Swaption Notice shall be the Submission Time specified by the EEP in respect of such Option Intent.

5.4 Consequences of EEP Failure

If an EEP Failure Event occurs, from (and including) the EEP Failure Event Time to (but excluding) the EEP Resolution Time:

(a) Mandatory Provision 5.3 (Exercise and Abandonment by way EEP) shall not apply and CCM Client shall deliver Swaption Notices directly to the Matched Seller or its relevant Exercise Delegation Beneficiary (with a copy to the Matched Seller), as applicable, in accordance with Section 6.5(b) (Consequences of EEP Failure) and Section 8 (Delivery of Notices) of Part C of the CDS Clearing Supplement using the notice details provided

Formatted: Header, Right

by LCH SA in the Protected Exercise Matched Pair Report pursuant to Section 6.1 (*Creation and Notification of Exercise Matched Pairs*) and Section 6.5(a)(ii) (*Access to the Protected Exercise Matched Pair Report*) of Part C of the CDS Clearing Supplement;

- (b) If LCH SA does not provide the Protected Exercise Matched Pair Report by the EMP Notification Deadline (or, where an EEP Failure Event occurs after the EMP Notification Deadline, from the time of such EEP Failure Event until such time as LCH provides the Protected Exercise Matched Pair Report), Mandatory Provision 5.2 (Exercise Notices and Abandonment Notices delivered in respect of CCM Client Cleared Transaction) and Mandatory Provision 5.3 (Exercise and Abandonment by way EEP) and paragraph (a) above shall not apply and accordingly CCM Client shall deliver Swaption Notices to LCH SA on behalf of the CCM;
- (c) any notice delivered via the EEP prior to the EEP Failure Event Time will be valid and will not be affected by such EEP Failure Event; and
- (d) any notice delivered or purported to be delivered via the EEP at or following the EEP Failure Event Time but prior to the EEP Resolution Time will not be valid and effective.

Mandatory Provision 5.2 (Exercise Notices and Abandonment Notices delivered in respect of CCM Client Cleared Transaction) and Mandatory Provision 5.3 (Exercise and Abandonment by way EEP) shall apply with effect from the EEP Resolution Time.

5.5 Duty to Deliver Swaption CCM Client Notice

CCM Client shall notify LCH SA and its CCM or deliver a copy to LCH SA and its CCM of any Swaption Notice delivered by it (as Exercise Delegation Beneficiary of Matched Buyer) to Matched Seller (or to the Client of Matched Seller as Exercise Delegation Beneficiary of Matched Seller on its behalf) in accordance with Mandatory Provision 5.4 (Consequences of EEP Failure) which such CCM Client asserts was effective (such notification, or delivery of such copy, in respect of any relevant notice, a Swaption CCM Client Notice) by no later than the Submission Deadline on the Expiration Date (the Swaption CCM Client Notice Deadline). If no Swaption CCM Client Notice is delivered by CCM Client prior to the Swaption CCM Client Notice Deadline, any Exercise Notice sent by CCM Client pursuant to Mandatory Provision 5.4 (Consequences of EEP Failure) shall, subject to Section 6.8 (Consequences of no Swaption Clearing Member Notice or Swaption CCM Client Notice being received by LCH SA) of Part C of the CDS Clearing Supplement, be deemed to be invalid for the purposes of Mandatory Provision 5.2 (Exercise Notices and Abandonment Notices delivered in respect of CCM Client Cleared Transaction).

Notwithstanding the above:

(a) if LCH SA elects to give effect to an Exercise Notice in respect of a Swaption CCM Client Notice that it determines to have been delivered pursuant to Section 6.8 (Consequences of no Swaption Clearing Member Notice or Swaption CCM Client Notice being received by LCH SA) of Part C of the CDS Clearing Supplement, then Part C of the CDS Clearing Supplement shall apply as if LCH SA had received a Swaption

10219615309-v3 70-41024903

File No. SR-LCH SA-2022-007 Page 297 of 439

PUBLIC

Formatted: Header, Right

- CCM Client Notice in respect of the relevant Exercise Notice by the Swaption CCM Client Notice Deadline; and
- (b) if LCH SA determines that it is not possible to give effect to the terms of any such Exercise Notice in respect of which no Swaption CCM Client Notice was received by LCH SA by the Swaption CCM Client Notice Deadline, the relevant Clearing Members (or their Exercise Delegation Beneficiaries, as applicable), as identified in the Protected Exercise Matched Pair Report, shall acquire rights as against each other as though party to a bilateral credit default swap transaction on the terms of the relevant Underlying Index Transaction. The Settlement Payment shall be due and payable two Transaction Business Days following the giving of a notice that such amount is due and payable. The relevant Clearing Members (or their Exercise Delegation Beneficiaries, as applicable) shall have enforcement rights as against each other pursuant to the Contracts (Rights of Third Parties) Act 1999 in respect of any resulting payments and deliveries; LCH SA shall have no liability in respect thereof.
- 5.6 Delivery of notices to and from LCH SA in case of EEP Failure Event

In the case of the occurrence of an EEP Failure Event, any notice or communication given by LCH SA to the CCM Client or *vice versa* shall be given to the address or number provided by the CCM Client to LCH SA and *vice versa* when the relevant CCM registered such CCM Client for the clearing of Index Swaption Cleared Transactions, or any other address or number duly notified thereafter by the such CCM Client to LCH SA or *vice versa*.

- 5.7 CCM Client Communications Failure Event
 - (a) Right to deliver Notices manually following CCM Client Communications Failure Event

If CCM Client is affected by a significant communications or information technology failure resulting in it being impossible or impractical for it to deliver a Swaption Notice via the EEP (a "CCM Client Communications Failure Event") it may, notwithstanding Mandatory Provision 5.3 (Exercise and Abandonment by way of EEP) above, deliver Swaption Notices in respect of the CCM Client Cleared Transaction to LCH SA in accordance with Section 6.10(a) of Part C of the CDS Clearing Supplement and any Option Intent submitted by LCH SA in the EEP (or an alternative internal system of LCH SA) in respect of such Swaption Notice pursuant to Section 6.10(b) of Part C of the CDS Clearing Supplement with a Submission Time prior to the Submission Deadline shall be deemed to have been submitted by CCM Client for the purposes of Mandatory Provision 5.3 (Exercise and Abandonment by way of EEP) above.

In respect of any Swaption Notice delivered pursuant to this Mandatory Provision 5.7, LCH SA may determine in its sole discretion that it is not able to submit the relevant Option Intent in the relevant system with a Submission Time prior to the Submission Deadline on the Expiration Date. In such case, LCH SA will inform the CCM Client and, subject to Mandatory Provision 5.5 above, such CCM Client will be deemed not to have submitted an Option Intent in respect of the relevant Exercise Cleared Transaction.

(b) Client to notify LCH SA of occurrence of CCM Client Communications Failure Event

Formatted: Header, Right

Following the occurrence of a CCM Client Communications Failure Event, the CCM Client shall, prior to or at the same time as delivering any Swaption Notice to LCH SA pursuant to sub-paragraph (a) above, deliver to LCH SA a notice (in the form set out at Appendix VI to Part C of the CDS Clearing Supplement) signed by an authorised signatory certifying that it is affected by a CCM Client Communications Failure Event (or, if CCM Client is unable to deliver such notice in writing, orally by telephone).

(c) Notification of Resolution of CCM Client Communications Failure Event

As soon as reasonably practicable upon CCM Client ceasing to be subject to a CCM Client Communications Failure Event, it shall notify LCH SA accordingly (in the form set out at Appendix VII to Part C of the CDS Clearing Supplement) and thereupon subparagraph (a) above shall cease to apply.

(d) Duty to Mitigate

If CCM Client is subject to a CCM Client Communications Failure Event, it shall use reasonable endeavours to mitigate the operational impact of any CCM Client Communications Failure Event, to cure such CCM Client Communications Failure Event as soon as reasonably practicable and to ensure that the circumstances giving rise to the relevant CCM Client Communications Failure Event do not recur.

5.8 Confidentiality Waiver

CCM Client hereby agrees and consents to the disclosure of its address, fax number, telephone number, contact email address (and any other applicable notice details provided by it) by CCM to LCH SA and by LCH SA in any Protected Exercise Matched Pair Report in accordance with Section 6.1 (*Creation and Notification of Exercise Matched Pairs*) and Section 6.5(a)(ii) (*Access to the Protected Exercise Matched Pair Report*) of Part C of the CDS Clearing Supplement.

6. Determination of Credit Events and Succession Events

Notwithstanding any provision to the contrary:

- the Calculation Agent shall not make any determination in respect of any matter which is or may be subject to resolution under Sections 3.5 (Successor Resolutions) or 3.6 (Substitute Reference Obligation Resolutions) of the DC Rules; and
- (b) neither party shall be entitled to deliver a Successor Notice or a Credit Event Notice (other than Credit Event Notices in relation to a Restructuring Credit Event in accordance with the terms of any CCM Client Transaction (including the Mandatory Provisions)).

Formatted: Header, Right

7. Timings for the Delivery of Notices for CCM Client Transactions

In this Mandatory Provision 7:

"Swaption Restructuring CCM Client Buyer" means a CCM Client that is party to a CCM Client Transaction and is protection buyer under the Underlying Swap Transaction for such CCM Client Transaction;

"Swaption Restructuring CCM Client Seller" means a CCM Client that is party to a CCM Client Transaction and is protection seller under the Underlying Swap Transaction for such CCM Client Transaction:

"Swaption Restructuring CCM Buyer/Matched Seller" means a CCM that is party to (a) a CCM Client Transaction and is protection buyer under the Underlying Swap Transaction for such CCM Client Transaction; and (b) a corresponding CCM Client Cleared Transaction and is protection seller under the Underlying Swap Transaction for such CCM Client Cleared Transaction; and

"Swaption Restructuring CCM Seller/Matched Buyer" means a CCM that is party to (a) a CCM Client Transaction and is protection seller under the Underlying Swap Transaction for such CCM Client Transaction; and (b) a corresponding CCM Client Cleared Transaction and is protection buyer under the Underlying Swap Transaction for such CCM Client Cleared Transaction.

The following provisions shall solely be applicable in respect of a CCM Client Transaction between a Swaption Restructuring CCM Client Buyer and its Swaption Restructuring CCM Seller/Matched Buyer:

7.1 Delivery of Notices by Swaption Restructuring CCM Client Buyer

For the purposes of the delivery by Swaption Restructuring CCM Client Buyer of any notice in respect of a CCM Client Transaction (other than a Swaption Notice to which this Mandatory Provision 7 shall not apply), Section 1.38 (*Requirements Regarding Notices*) of the 2014 ISDA Credit Derivatives Definitions shall be amended so as to provide that, solely in respect of the final day on which such notice could validly be delivered pursuant to the terms of such CCM Client Transaction (including the Mandatory Provisions), any such notice shall be required to be delivered on or prior to 2:00 p.m. (Calculation Agent City time) in order to be effective.

A notice (including, without limitation, a Credit Event Notice or a Notice to Exercise Movement Option) delivered by Swaption Restructuring CCM Client Buyer after 2:00 p.m. (Calculation Agent City time) on the final day on which such notice could validly be delivered pursuant to the terms of the relevant CCM Client Transaction (including the Mandatory Provisions) shall be deemed not to have been delivered.

7.2 Onward Delivery of Credit Event Notices and Notices to Exercise Movement Option by Swaption Restructuring CCM Seller/Matched Buyer to Matched Seller

Any Credit Event Notice or Notice to Exercise Movement Option delivered by Swaption Restructuring CCM Client Buyer shall not be effective unless and until Swaption Restructuring

File No. SR-LCH SA-2022-007 Page 300 of 439

PUBLIC

Formatted: Header, Right

CCM Seller/Matched Buyer effectively delivers the relevant equivalent notice to the relevant Matched Seller in respect of and pursuant to the terms of the corresponding Swaption Restructuring Cleared Transaction.

Swaption Restructuring CCM Seller/Matched Buyer undertakes to deliver such a notice to the relevant Matched Seller within two hours of its receipt of the equivalent notice from Swaption Restructuring CCM Client Buyer if such notice is received between 9:00 a.m. (Calculation Agent City time) and 4:00 p.m. (Calculation Agent City time) on a Calculation Agent City Business Day. Any such notice received by Swaption Restructuring CCM Seller/Matched Buyer after 4:00 p.m. (Calculation Agent City time) on a Calculation Agent City Business Day shall be deemed to have been received at 9:00 a.m. (Calculation Agent City time) on the following Calculation Agent City Business Day. Any such notice received by Swaption Restructuring CCM Seller/Matched Buyer before 9:00 a.m. (Calculation Agent City time) on a Calculation Agent City Business Day shall be deemed to have been received at 9:00 a.m. (Calculation Agent City time) on such Calculation Agent City Business Day. Any such notice delivered on a day that is not a Calculation Agent City Business Day shall be deemed to have been delivered at 9:00 a.m. (Calculation Agent City Business Day shall be deemed to have been delivered at 9:00 a.m. (Calculation Agent City Business Day shall be deemed to have been delivered at 9:00 a.m. (Calculation Agent City Business Day.

7.3 Receipt of Credit Event Notices and Notices to Exercise Movement Option by Swaption Restructuring CCM Seller/Matched Buyer deemed to be Receipt by Swaption Restructuring CCM Client Buyer

Any Credit Event Notice or Notice to Exercise Movement Option delivered by Matched Seller to Matched Buyer in respect of and pursuant to the terms of a Swaption Restructuring Cleared Transaction relating to a CCM Client Transaction between such Swaption Restructuring CCM Seller/Matched Buyer and Swaption Restructuring CCM Client Buyer shall be deemed to constitute simultaneous delivery by Swaption Restructuring CCM Seller/Matched Buyer to Swaption Restructuring CCM Client Buyer of such notice in respect of such CCM Client Transaction.

Swaption Restructuring CCM Seller/Matched Buyer undertakes to deliver such a notice to Swaption Restructuring CCM Client Buyer within two hours of its receipt of the equivalent notice from the relevant Matched Seller if such notice is received between 9:00 a.m. (Calculation Agent City time) and 4:00 p.m. (Calculation Agent City time) on a Calculation Agent City Business Day. Any such notice received by Swaption Restructuring CCM Seller/Matched Buyer after 4:00 p.m. (Calculation Agent City time) on a Calculation Agent City Business Day shall be deemed to have been received at 9:00 a.m. (Calculation Agent City time) on the following Calculation Agent City Business Day. Any such notice received by Swaption Restructuring CCM Seller/Matched Buyer before 9:00 a.m. (Calculation Agent City time) on a Calculation Agent City Business Day shall be deemed to have been received at 9:00 a.m. (Calculation Agent City time) on such Calculation Agent City Business Day. Any such notice delivered on a day that is not a Calculation Agent City Business Day shall be deemed to have been delivered at 9:00 a.m. (Calculation Agent City Business Day shall be deemed to have been delivered at 9:00 a.m. (Calculation Agent City Business Day shall be deemed to have been delivered at 9:00 a.m. (Calculation Agent City Business Day.

10219615309-v3 70-41024903

Formatted: Header, Right

The following provisions shall solely be applicable in respect of a CCM Client Transaction between a Swaption Restructuring CCM Client Seller and its Swaption Restructuring CCM Buyer/Matched Seller:

7.4 Delivery of Credit Event Notices and Notices to Exercise Movement Option by Swaption Restructuring CCM Client Seller

For the purposes of the delivery by Swaption Restructuring CCM Client Seller of any Credit Event Notice or any Notice to Exercise Movement Option in respect of a CCM Client Transaction, Section 1.38 (*Requirements Regarding Notices*) of the 2014 ISDA Credit Derivatives Definitions shall be amended so as to provide that, solely in respect of the final day on which such notice could validly be delivered pursuant to the terms of such CCM Client Transaction (including the Mandatory Provisions), any such notice shall be required to be delivered on or prior to 2:00 p.m. (Calculation Agent City time) in order to be effective.

A Credit Event Notice or Notice to Exercise Movement Option delivered by Swaption Restructuring CCM Client Seller after 2:00 p.m. (Calculation Agent City time) on the final day on which such notice could validly be delivered pursuant to the terms of the relevant CCM Client Transaction (including the Mandatory Provisions) shall be deemed not to have been delivered.

7.5 Onward Delivery of Credit Event Notices and Notices to Exercise Movement Option by Swaption Restructuring CCM Buyer/Matched Seller to Matched Buyer

Any Credit Event Notice or Notice to Exercise Movement Option delivered by Swaption Restructuring CCM Client Seller shall not be effective unless and until Swaption Restructuring CCM Buyer/Matched Seller effectively delivers the relevant equivalent notice to the relevant Matched Buyer in respect of and pursuant to the terms of the corresponding Swaption Restructuring Cleared Transaction.

Swaption Restructuring CCM Buyer/Matched Seller undertakes to deliver such a notice to the relevant Matched Buyer within two hours of its receipt of the equivalent notice from Swaption Restructuring CCM Client Seller if such notice is received between 9:00 a.m. (Calculation Agent City time) and 4:00 p.m. (Calculation Agent City time) on a Calculation Agent City Business Day. Any such notice received by Swaption Restructuring CCM Buyer/Matched Seller after 4:00 p.m. (Calculation Agent City time) on a Calculation Agent City Business Day shall be deemed to have been received at 9:00 a.m. (Calculation Agent City time) on the following Calculation Agent City Business Day. Any such notice received by Swaption Restructuring CCM Buyer/Matched Seller before 9:00 a.m. (Calculation Agent City time) on a Calculation Agent City Business Day shall be deemed to have been received at 9:00 a.m. (Calculation Agent City time) on such Calculation Agent City Business Day. Any such notice delivered on a day that is not a Calculation Agent City Business Day shall be deemed to have been delivered at 9:00 a.m. (Calculation Agent City Business Day shall be deemed to have been delivered at 9:00 a.m. (Calculation Agent City Business Day shall be deemed to have been delivered at 9:00 a.m. (Calculation Agent City Business Day shall be deemed

10219615309-v3 70-41024903

Formatted: Header, Right

7.6 Receipt of Credit Event Notices and Notices to Exercise Movement Option by Swaption Restructuring CCM Buyer/Matched Seller deemed to be Receipt by Swaption Restructuring CCM Client Seller

Any Credit Event Notice or Notice to Exercise Movement Option which is delivered by Matched Buyer in respect of and pursuant to the terms of a Swaption Restructuring Cleared Transaction relating to a CCM Client Transaction between such Swaption Restructuring CCM Buyer/Matched Seller and Swaption Restructuring CCM Client Seller shall be deemed to constitute simultaneous delivery by Swaption Restructuring CCM Buyer/Matched Seller to Swaption Restructuring CCM Client Seller of such notice in respect of such CCM Client Transaction.

Swaption Restructuring CCM Buyer/Matched Seller undertakes to deliver such a notice to Swaption Restructuring CCM Client Seller within two hours of its receipt of the equivalent notice from the relevant Matched Buyer if such notice is received between 9:00 a.m. (Calculation Agent City time) and 4:00 p.m. (Calculation Agent City time) on a Calculation Agent City Business Day. Any such notice received by Swaption Restructuring CCM Buyer/Matched Seller after 4:00 p.m. (Calculation Agent City time) on a Calculation Agent City Business Day shall be deemed to have been received at 9:00 a.m. (Calculation Agent City time) on the following Calculation Agent City Business Day. Any such notice received by Swaption Restructuring CCM Buyer/Matched Seller before 9:00 a.m. (Calculation Agent City time) on a Calculation Agent City Business Day shall be deemed to have been received at 9:00 a.m. (Calculation Agent City time) on such Calculation Agent City Business Day shall be deemed to have been delivered at 9:00 a.m. (Calculation Agent City Business Day shall be deemed to have been delivered at 9:00 a.m. (Calculation Agent City Business Day shall be deemed to have been delivered at 9:00 a.m. (Calculation Agent City Business Day shall be deemed to have been delivered at 9:00 a.m. (Calculation Agent City Business Day.

8. Settlement

8.1 Creation of Underlying Swap Transaction

Following exercise of the CCM Client Transaction (and the related CCM Client Cleared Transaction) and upon creation of the Underlying Swap Transaction, such Underlying Swap Transaction shall constitute a CCM Client Transaction between the CCM and its CCM Client for the purposes of Part B of the CDS Clearing Supplement corresponding to the CCM Client Cleared Transaction registered in the CCM Client Trade Account in the name of the CCM for the CCM Client created pursuant to Section 7.1 (*Creation of Index Cleared Transactions*) of Part C of the CDS Clearing Supplement and governed following creation by Part B of the CDS Clearing Supplement. The provisions of Appendix XIII (*CCM Client Transaction Requirements*) of Part B of the CDS Clearing Supplement shall be deemed to be incorporated into the new CCM Client Transaction.

8.2 Creation of Restructuring Single Name Transaction

Following exercise of the CCM Client Transaction (and the related CCM Client Cleared Transaction), if any 'New Trade' is created pursuant to the provisions of sub-paragraph 5.2(b) (*Transfer and termination of Component Transactions*) of the Relevant Standard Terms Supplement (as defined in the iTraxx® Swaption Standard Terms Supplement) as a result of

Formatted: Header, Right

the occurrence of an M(M)R Restructuring, such 'New Trade' shall constitute a CCM Client Transaction for the purposes of Part B of the CDS Clearing Supplement corresponding to the CCM Client Cleared Transaction registered in the CCM Client Trade Account in the name of the CCM for the CCM Client created pursuant to Section 7.3 (*Creation of Restructuring Cleared Transactions for Triggering and/or Settlement purposes*) or Section 7.4 (*Creation of Initial Single Name Cleared Transactions in respect of untriggered M(M)R Restructuring*) of Part C of the CDS Clearing Supplement and governed following creation by Part B of the CDS Clearing Supplement. The provisions of Appendix XIII (*CCM Client Transaction Requirements*) of Part B of the CDS Clearing Supplement shall be deemed to be incorporated into the new CCM Client Transaction.

8.3 Auction Settlement Date

If an Event Determination Date has occurred in respect of a Credit Event for a Reference Entity referenced by the Underlying Swap Transaction in accordance with the 'Operation of each Underlying Swap Transaction' section of Part 4 (*Underlying Swap Transaction Terms*) of the iTraxx® Swaption Standard Terms Supplement or, as the case may be, section 5.2 (*Operation of Underlying Swap Transaction*) of the CDX Swaption Standard Terms Supplement, notwithstanding anything to the contrary in the iTraxx® Swaption Standard Terms Supplement or, as the case may be, the CDX Swaption Standard Terms Supplement, the Auction Settlement Date for any such Event Determination Date will be deemed to be the later of (i) the Auction Settlement Date that would be determined in accordance with Section 6.3 of the 2014 Credit Derivatives Definitions and (ii) the Transaction Business Day immediately following the Expiration Date.

8.4 Settlement Payment

Notwithstanding anything to the contrary in the iTraxx® Swaption Standard Terms Supplement or, as the case may be, the CDX Swaption Standard Terms Supplement, the Settlement Payment (or the absolute value thereof, as applicable) shall be payable on the Transaction Business Day immediately following the Expiration Date.

9. Calculation Agent

9.1 Appointment of Calculation Agent

The Calculation Agent in respect of any CCM Client Transaction shall be the CCM.

9.2 Calculations and Determinations of Calculation Agent

In the event that the Calculation Agent is entitled or required to make any calculation or determination in respect of a CCM Client Transaction in respect of a matter that has already been or will be determined in respect of and pursuant to the terms of the corresponding CCM Client Cleared Transaction, the Calculation Agent in respect of the CCM Client Transaction shall be obliged to make the same calculation or determination in respect of such CCM Client Transaction as the determination in respect of the corresponding CCM Client Cleared Transaction.

10219615309-v3 70-41024903

File No. SR-LCH SA-2022-007 Page 304 of 439

PUBLIC

Formatted: Header, Right

10. Amendments

The Mandatory Provisions may be amended from time to time pursuant to Section 11 (Amendments) of Part C of the CDS Clearing Supplement. The parties agree that any amendments made to the Mandatory Provisions in accordance with Section 11 (Amendments) of Part C of the CDS Clearing Supplement shall be deemed to apply automatically to the CCM Client Transaction(s) with effect from the date of such amendment to the Mandatory Provisions.

Page 305 of 439 Exhibit 5b



CDSClear

LCH SA CDS Clearing Procedures

Section 2 – Margin, NPV Payment and Price Alignment

[•]

Classification: Confidential

CONTENTS

SECTION 2 – MARGIN, NPV PAYMENT AND PRICE ALIGNMENT INTEREST

2.1	OVERVIEW	<u>307</u> 2
2.2	MARGIN	<u>307</u> 2
2.3	EXCESS COLLATERAL AND THE CLIENT COLLATERAL BUFFER	<u>310</u> 5
2.4	COLLATERAL, CASH PAYMENTS AND VARIATION MARGIN COL TRANSFERS	
2.5	PAYMENT OF THE MARGIN REQUIREMENT, VARIATION REQUIREMENT, NPV PAYMENT REQUIREMENT AND PROVISION O COLLATERAL AND CLIENT COLLATERAL BUFFER	F EXCESS
2.6	REPORTS	<u>31712</u>
2.7	INITIAL MARGIN	3 <u>18</u> 12
2.8	SELF-REFERENCING PROTECTION MARGIN	<u>319</u> 13
2.9	LIQUIDITY AND CONCENTRATION RISK MARGIN	<u>320</u> 14
2.10	ACCRUED FIXED AMOUNT LIQUIDATION RISK MARGIN	<u>320</u> 14
2.11	CREDIT EVENT MARGIN	<u>320</u> 14
2.12	LEGAL ENTITY IDENTIFIER MARGIN	<u>321</u> 15
2.13	ADDITIONAL MARGIN	<u>321</u> 15
2.14	STRESS TEST LOSS OVER ADDITIONAL MARGIN / NET CAPITA MARGIN	
2.15	VARIATION MARGIN	<u>322</u> 16
2.16	NPV AMOUNT	<u>323</u> 17
2.17	CONTINGENCY VARIATION MARGIN	<u>325</u> 19
2.18	PRICE ALIGNMENT INTEREST	<u>326</u> 20
2.19	PRICE ALIGNMENT AMOUNT	<u>327</u> 21
2.20	CREDIT QUALITY MARGIN	<u>329</u> 22
2.21	EXTRAORDINARY MARGIN	<u>329</u> 23

PUBLIC

Capitalised terms used in this Margin, NPV Payment and Price Alignment Procedure and not otherwise defined herein shall have the meaning given pursuant to the remainder of the CDS Clearing Documentation, as such term is defined in the document entitled "CDS Clearing Rule Book" published by LCH SA, as amended from time to time.

2.1 **OVERVIEW**

All Clearing Members are required to pay Margin to LCH SA.

Further information about the Margin components and the calculation methodology in respect of each such component set out in this Section 2 is available to Clearing Members on the secure section of the Website and through any reporting mechanism specified in a Clearing Notice and/or upon request from LCH SA's Risk Management Department on +33 1 70 37 10 43 or CDS.Risk.SA@lch.com. Such information will be updated and/or re-issued following consultation with the Risk Committee.

Section 3 of the Procedures provides further detail of how Collateral should be transferred, and Cash Payments and Variation Margin Collateral Transfers made, by Clearing Members to LCH SA.

2.2 MARGIN

(a) Margin Requirement

For each Margin Account of each Clearing Member, the Margin Requirement consists of the following components:

- (i) Spread Margin;
- (ii) Short Charge Margin;
- (iii) Recovery Risk Margin;
- (iv) Interest Rate Risk Margin;
- (v) Wrong Way Risk Margin;
- (vi) Vega Margin;
- (vii) Self-Referencing Protection Margin;
- (viii) Liquidity and Concentration Risk Margin;
- (ix) Accrued Fixed Amount Liquidation Risk Margin;
- (x) Credit Event Margin;
- (x) Legal Entity Identifier Margin;
- (xi) Additional Margin;
- (xii) Stress Test Loss Over Additional Margin / Net Capital Ratio Margin;

CDS Clearing Procedures

Margin, NPV Payment and Price Alignment

PUBLIC

- (xiii) Contingency Variation Margin;
- (xiv) in respect of the House Margin Requirement only: Credit Quality Margin; and
- (xv) Extraordinary Margin.

Details of each of these components are set out below.

(b) Timing for calculation of the Margin Requirement

LCH SA will calculate the Margin Requirement for each Margin Account of each Clearing Member on each Business Day:

- (i) by no later than 04:00 CET, which will be the Margin Requirement for the Morning Call;
- (ii) by no later than the Start of Real Time; and
- (iii) whenever a position corresponding to the relevant trade leg of an Eligible Intraday Transaction is pre-registered in the relevant Clearing Member's Account Structure, in accordance with Section 3.1.7 of the CDS Clearing Rule Book, and the Intraday Novation Margin Requirement is calculated. The most recently calculated Margin Requirement for each Margin Account will be the Margin Requirement used for each of the First Intraday Call and the Second Intraday Call.

LCH SA will notify each Clearing Member of the Margin Requirement for each of its Margin Accounts through the relevant Margin Requirements Results File(s), in each case in accordance with, subject to and at the times set out in Section 5 of the Procedures.

(c) Variation Margin Requirement and NPV Payment Requirement

Each Clearing Member is required, as applicable, to:

- (i) transfer Variation Margin to LCH SA (or is entitled to receive Variation Margin from LCH SA, as applicable) by way of Variation Margin Collateral Transfer to cover its Client Variation Margin Requirement(s) and/or its House Variation Margin Requirement, as detailed in Paragraph 2.15 below and in Section 3 of the Procedures; and/or
- (ii) pay to LCH SA (or is entitled to receive from LCH SA, as applicable) NPV Amounts by way of NPV Payments to cover its Client NPV Payment Requirement(s) and/or its House NPV Payment Requirement, as detailed in Paragraph 2.14 below and in Section 3 of the Procedures.
- (d) Aggregate Margin for Cleared Transactions which reference a single Reference Entity

Notwithstanding anything to the contrary in the CDS Clearing Documentation, the sum of the aggregate Variation Margin and/or NPV Amount, Initial Margin and the Credit Event Margin that can be called from a Clearing Member that is a CDS Seller in respect of a Cleared Transaction referencing a single Reference Entity may not exceed the Floating Rate Payer Calculation Amount in respect of such Cleared Transaction.

PUBLIC

(e) Additional Collateral in respect of Client Cleared Transactions of a "non-hedging nature" of an FCM/BD Clearing Member

Each FCM/BD Clearing Member shall ensure that with respect to a Client Cleared Transaction registered in its FCM/BD Client Trade Account(s) that is of a "non-hedging nature" (as such term is used in Part 39 of the CFTC Regulations), it shall collect additional Collateral from the relevant FCM/BD Client in respect of such Client Cleared Transaction at a level of 10% above the FCM/BD Client Margin Requirement that LCH SA would normally require for such Client Cleared Transaction.

In connection with article 6.2.6.1 of the CDS Clearing Rule Book and this Paragraph 2.2 (e), FCM/BD Clearing Members are not required to lodge such additional Collateral with LCH SA which is in excess of the relevant FCM/BD Client Margin Requirement.

(f) Determination of the Legally Segregated Value ascribed to each FCM/BD Client Financial Account

In accordance with Section 3.2 of the Procedures, LCH SA will calculate the FCM/BD Margin Balance with respect to each FCM/BD Client Margin Account of an FCM/BD Clearing Member by determining the Legally Segregated Value recorded in the relevant FCM/BD Client Financial Account.

LCH SA will determine the Legally Segregated Value of each FCM/BD Clearing Member at the following times:

- after the FCM/BD Clearing Member's satisfaction of the Morning Call to reflect any increase or decrease in the relevant FCM/BD Client Margin Requirement calculated for the purpose of the Morning Call in accordance with Article 6.2.5 of the CDS Clearing Rule Book;
- (ii) after the FCM/BD Clearing Member's satisfaction of any Collateral Call (other than the Morning Call) to reflect only any increase in the relevant FCM/BD Client Margin Requirement calculated for the purpose of such Collateral Call; and
- (iii) after each determination of the value of Collateral recorded in the FCM/BD Client Collateral Account provided that (x) there is a decrease in the value resulting from this determination; and (y) such decrease will reduce the Legally Segregated Value in accordance with Article 6.2.4.4 (ii) of the CDS Clearing Rule Book.

LCH SA will record the amended Legally Segregated Value resulting from such determination in the relevant FCM/BD Client Financial Account.

(g) Calculation of Margin and NPV Amount(s) following a Payment Failure or the issuance of a Default Notice in respect of a Clearing Member

Pursuant to Article 1.2.9.2 and Article 4.3.2.3 of the CDS Clearing Rule Book, LCH SA may withhold payments it would otherwise be obliged to make to a Clearing Member under the CDS Clearing Documentation.

Where LCH SA withholds the payment (or repayment or reimbursement) of Margin and/or NPV Amount(s) due in respect of a Client Margin Account of the Clearing Member, LCH SA shall nevertheless continue to calculate the Margin and/or NPV Amount(s) that is or are due, in accordance with the CDS Clearing Rules, and update the records which are attributable to the relevant Client Collateral Account. As appropriate, any Margin and/or NPV Amount(s) which is or are calculated to be due in respect of the relevant Client Margin Account shall be an "accrual", forming part of the Client Assets which will either be:

- (i) ported in accordance with Clause 4.3.1(ii) of the CDS Default Management Process;
- (ii) taken into account when calculating the Client Clearing Entitlement pursuant to Clause 4.4.3 of the CDS Default Management Process; or
- (iii) transferred to a Receiving Clearing Member pursuant to TITLE V, Chapter 3 for CCMs and TITLE VI, Chapter 3 for FCM/BD Clearing Members.

2.3 EXCESS COLLATERAL AND THE CLIENT COLLATERAL BUFFER

(a) House Excess Collateral

A Clearing Member is entitled (but not obliged) to maintain Collateral over and above that which is needed to satisfy its House Margin Requirement. Such House Excess Collateral will be maintained in its House Collateral Account, in which case it will be used for the novation of House Trade Legs and taken into account by LCH SA when it calculates the amount of Collateral which is needed for the House Margin Account, as part of the Notional and Collateral Check (as set out in Paragraph 2.5(b)(i) below).

(b) Client Excess Collateral

A CCM is entitled to maintain Collateral over and above that which is needed to satisfy the CCM Client Margin Requirement in respect of any of its CCM Client Margin Accounts. Such CCM Client Excess Collateral will be maintained in a CCM Client Collateral Account of a CCM Client Account Structure, in which case it will be used for the novation of Client Trade Legs and taken into account by LCH SA when it calculates the amount of Collateral which is needed for that CCM Client Account Structure, as part of the Notional and Collateral Check (as set out in Paragraph 2.5(b)(i) below).

An FCM/BD Clearing Member is not authorised to post additional Collateral over and above that which it needed to satisfy the FCM/BD Client Margin Requirement in respect of any of its FCM/BD Client Margin Accounts. Any FCM/BD Client Excess Collateral recorded in any of its FCM/BD Client Financial Accounts and resulting from any decrease of the Spread Margin in relation to the attached FCM/BD Client Margin Account during a Clearing Day will be used for the novation of Client Trade Legs and taken into account by LCH SA when it calculates the amount of Collateral which is needed for the relevant FCM/BD Client Margin Account, as part of the Notional and Collateral Check, until the next Morning Call.

(c) Client Collateral Buffer

A Clearing Member is entitled (but not obliged) to maintain:

- (i) in the case of a CCM, Collateral in CCM House Collateral Account; and
- (ii) in the case of an FCM/BD Clearing Member, an amount of Collateral recorded in its FCM/BD Buffer Financial Account,

specifically for the purpose of assisting Clients to satisfy the Notional and Collateral Check performed by LCH SA prior to novation of a Client Trade Leg of an Eligible Intraday Transaction.

A Clearing Member which is a CCM may not hold Collateral as CCM Client Collateral Buffer in an amount in excess of the CCM Client Collateral Buffer Threshold. Any Collateral delivered and held as CCM Client Collateral Buffer that is in excess of the CCM Client Collateral Buffer Threshold shall be treated as CCM House Excess Collateral.

Where a Clearing Member holds:

- (i) in the case of a CCM, Collateral recorded as CCM Client Collateral Buffer in its CCM House Collateral Account; and
- (ii) in the case of an FCM/BD Clearing Member, an amount of Collateral recorded in its FCM/BD Buffer Financial Account,

the Available Client Collateral Buffer (or portion thereof) will be allocated to:

- (A) in the case of a CCM, its Client Account Structure(s); and
- (B) in the case of an FCM/BD Clearing Member, its FCM/BD Client Margin Account(s),

on an automatic 'first in time' basis, meaning that whenever a Client Trade Leg of an Eligible Intraday Transaction is received by LCH SA and the Eligibility Controls and Client Transaction Checks have been successfully completed pursuant to Article 3.1.4.3, LCH SA will allocate the Available Client Collateral Buffer (or portion thereof) to the relevant CCM Client Account Structure in the case of a CCM, and to the relevant FCM/BD Client Margin Account in the case of an FCM/BD Clearing Member, where the relevant Client Excess Collateral is otherwise insufficient to satisfy the Notional and Collateral Check. For the avoidance of doubt, a Clearing Member has no discretion or ability to instruct LCH SA as to which CCM Client Account Structure(s) in the case of a CCM, and FCM/BD Client Margin Account(s) in the case of an FCM/BD Clearing Member, the Available Client Collateral Buffer should be allocated to.

LCH SA shall reflect how the Client Collateral Buffer has been allocated between the CCM Client Account Structure(s) of a CCM or as the case may be, FCM/BD Client Margin Account(s) of an FCM/BD Clearing Member in its books and records but the Collateral comprising the Client Collateral Buffer shall, at all times (save where the relevant Clearing Member is a Defaulting Clearing Member or, in respect of a CCM, following an LCH Default), remain, in the case of a CCM, in the CCM House Collateral Account, and in the case of an FCM/BD Clearing Member, the amount of such Collateral recorded in the FCM/BD Buffer Financial Account.

Where:

PUBLIC

- (i) Client Collateral Buffer has been allocated to a CCM Client Account Structure of a CCM or as the case may be to an FCM/BD Client Margin Account of an FCM/BD Clearing Member; and
- (ii) there is a decrease in the Client Margin Requirement(s) calculated in respect of such CCM Client Account Structure or as the case may be, of such FCM/BD Client Margin Account following the novation of the Client Trade Leg of an Eligible Intraday Transaction,

the amount of such allocated Client Collateral Buffer will be reduced by an amount equal to the decrease in such Client Margin Requirement(s) and such amount will then become Available Client Collateral Buffer.

Following the occurrence of an Event of Default or, in respect of a CCM, an LCH Default, an amount of Collateral equal to the Allocated Client Collateral Buffer for the relevant CCM Client Account Structure of a CCM or as the case may be, for the relevant FCM/BD Client Margin Account of an FCM/BD Clearing Member will be transferred:

- (i) in the case of a CCM, from the CCM House Collateral Account of the Defaulting Clearing Member to the relevant CCM Client Collateral Account of the relevant CCM Client Account Structure; or
- (ii) in the case of an FCM/BD Clearing Member, from the FCM/BD Buffer Financial Account of the Defaulting Clearing Member to the relevant FCM/BD Client Financial Account,

in accordance with Article 1.3.1.3(iv) of the CDS Clearing Rule Book or clause 4.2 of the CDS Default Management Process, as applicable.

(d) The House Excess Collateral Threshold and Client Collateral Buffer Threshold

Where a Clearing Member wishes to transfer additional Collateral to LCH SA with a view to maintaining House Excess Collateral and/or Client Buffer Collateral as described in Paragraphs 2.3(a) and (c) above, it must notify LCH SA of its:

- (i) House Excess Collateral Threshold; and/or
- (ii) Client Collateral Buffer Threshold.

To set its House Excess Collateral Threshold and/or Client Collateral Buffer Threshold for a Business Day (D), a Clearing Member must notify LCH SA by submitting the form (which is available on the Website) by email at the email address specified in Paragraph 2.1. The form must be received by LCH SA by no later than 12.00 CET on D-1. It is the Clearing Member's responsibility to ensure the due receipt by LCH SA of the relevant form. Accordingly, the Clearing Member should confirm its request no later than 12:00 CET by a phone call to LCH SA's Risk Management Department on +33 1 70 37 10 43, although a failure to do so shall not invalidate any request actually received by LCH SA.

Once a Clearing Member has notified LCH SA of its House Excess Collateral Threshold and/or Client Collateral Buffer Threshold, LCH SA will apply this in the context of each successive

PUBLIC

Collateral Call, until such time as the relevant Clearing Member notifies LCH SA of an amended House Excess Collateral Threshold and/or Client Collateral Buffer Threshold.

2.4 COLLATERAL, CASH PAYMENTS AND VARIATION MARGIN COLLATERAL TRANSFERS

(a) Types of Collateral and currencies for Cash Payments and Variation Margin Collateral Transfers

Section 3 of the Procedures sets out the Collateral types which a Clearing Member can transfer, and currencies in which Cash Payments and Variation Margin Collateral Transfers can be made, to LCH SA to satisfy its obligations in respect of NPV Payment(s) and Variation Margin and each of the Margin components listed in Paragraph 2.2 above and for the purposes of maintaining Excess Collateral and/or Client Collateral Buffer.

(b) Transferring Collateral and making Cash Payments and Variation Margin Collateral Transfers

Further details on the process for:

- (i) transferring Collateral to satisfy the Required Collateral Amount;
- (ii) transferring additional Collateral to LCH SA with a view to maintaining Excess Collateral in one or more Collateral Account(s), if applicable or substituting for another type of Collateral;
- (iii) making NPV Payments, to satisfy the NPV Payment Requirements of each Clearing Member; and
- (iv) Making Variation Margin Collateral Transfers, to satisfy the Variation Margin Requirements of each Clearing Member;

are set out in Section 3 of the Procedures.

(c) Repayment of Collateral

References, in this Section 2 of the Procedures, to the "repayment" or "reimbursement" of Margin shall mean that:

- (i) the amount called from a Clearing Member in respect of the relevant Margin component shall, from such point, be reduced to zero in respect of the relevant Open Positions; and
- (ii) the value of any Collateral that has been transferred to LCH SA in respect of such Margin component shall be taken into account by LCH SA in calculating the relevant Clearing Member's Margin Balance in accordance with Section 3 of the Procedures.

- 2.5 PAYMENT OF THE MARGIN REQUIREMENT, VARIATION MARGIN REQUIREMENT, NPV PAYMENT REQUIREMENT AND PROVISION OF EXCESS COLLATERAL AND CLIENT COLLATERAL BUFFER
- (a) Morning Call
 - (i) Scheduled Margin and NPV Payment Requirement Calculation Time

The Margin Requirement, Variation Margin Requirement and NPV Payment Requirement for each Margin Account of a Clearing Member are calculated on each Business Day by 07.45 CET.

The relevant Margin Requirement Results File, provided as part of the Backloading Transaction Reports, will notify each Clearing Member of its:

- (A) Margin Requirement for the Morning Call (and each component thereof);
- (B) Margin Balance;
- (C) Excess Collateral or Margin Shortfall (as the case may be);
- (D) Variation Margin Requirement and/or NPV Payment Requirement, as applicable; and
- (E) House Excess Collateral Threshold and Client Collateral Buffer Threshold,

for the relevant Margin Accounts in accordance with and subject to Section 5 of the Procedures.

Each Clearing Member is required to:

- (A) transfer Collateral to satisfy the Required Collateral Amount;
- (B) make Variation Margin Collateral Transfers to satisfy its House Variation Margin Requirement and its Client Variation Margin Requirement(s), if applicable; and
- (C) make NPV Payments in respect of its House NPV Payment Requirement and its Client NPV Payment Requirement(s), if applicable,

by such times as set out in Section 3 of the Procedures.

Following such transfer of Collateral and/or Cash Payments by an FCM/BD Clearing Member, any FCM/BD Client Excess Collateral recorded in any of its FCM/BD Client Financial Accounts will be moved into the FCM/BD Swaps Unallocated Client Collateral Financial Account or FCM/BD SBS Client Excess Collateral Financial Account, as applicable and recorded as FCM/BD Swaps Unallocated Client Excess Collateral or FCM/BD SBS Client Excess Collateral, as applicable and LCH SA will issue Intraday Call Reports, in accordance with and subject to Section 5 of the Procedures, to all Clearing Members, setting out the:

(A) Margin Requirement (and each component thereof);

PUBLIC

- (B) Margin Balance;
- (C) Excess Collateral or Margin Shortfall (as the case may be);
- (D) NPV Payment Requirement; and
- (E) House Excess Collateral Threshold and Client Collateral Buffer Threshold,

for the relevant Margin Accounts of each Clearing Member.

On the basis of these calculations, LCH SA will be able to credit Euro denominated Cash Collateral and/or make the Cash Payment(s) and/or make the Variation Margin Collateral Transfer(s) to the relevant Clearing Members in accordance with Section 3 of the Procedures.

- (b) Margin calculations during the Real Time Session
 - (i) Intraday Novation Margin Requirement

As part of the Notional and Collateral Checks performed by LCH SA in order to clear Intraday Transactions on a 'trade by trade' basis, LCH SA will calculate the Intraday Novation Margin Requirement.

LCH SA will calculate the Intraday Novation Margin Requirement in respect of the relevant Clearing Member's Margin Account when LCH SA pre-registers a position corresponding to the relevant trade leg of an Eligible Intraday Transaction in accordance with Section 3.1.7 of the CDS Clearing Rule Book. The calculation identifies the additional, or reduced, risk exposure (as applicable) which would be attributable to the relevant Margin Account following the novation of such pre-registered positions, and accordingly the Intraday Novation Margin Requirement may either be a positive or negative figure.

Following the calculation of such Intraday Novation Margin Requirement, LCH SA will recalculate the Available Client Collateral Buffer and, in respect of the relevant Margin Account, the Margin Requirement and the Excess Collateral for such Margin Account or in the case of a CCM Gross Omnibus Segregated Account Structure or a CCM Indirect Client Gross Segregated Account Structure, the Excess Collateral for all the CCM Client Margin Accounts attached to that CCM Gross Omnibus Segregated Account Structure or CCM Indirect Client Gross Segregated Account Structure. These calculations will be undertaken on the assumption that the relevant Eligible Intraday Transactions, accounted for in the calculation of the Intraday Novation Margin Requirement, will be novated as contemplated. If the relevant Eligible Intraday Transactions are not novated for any reason, LCH SA will refresh its calculations to determine the Available Client Collateral Buffer plus the Margin Requirement and Excess Collateral for the relevant Margin Account or in the case of a CCM Gross Omnibus Segregated Account Structure or a CCM Indirect Client Gross Segregated Account Structure, Excess Collateral for all the CCM Client Margin Accounts attached to that CCM Gross Omnibus Segregated Account Structure or CCM Indirect Client Gross Segregated Account Structure.

LCH SA shall only calculate the Intraday Novation Margin Requirement for a Margin Account during the Real Time Session. Where the relevant Business Day is a Clearing Day and the Real Time Session does not take place, no calculation of the Intraday Novation Margin Requirement will be performed by LCH SA on such Business Day.

In the event LCH SA determines that there is a positive Intraday Novation Margin Requirement for a Margin Account and there is insufficient:

- (A) House Excess Collateral; or
- (B) Client Excess Collateral and/or Available Client Collateral Buffer which can be allocated to the relevant Client Margin Account or in the case of a CCM Gross Omnibus Segregated Account Structure or a CCM Indirect Client Gross Segregated Account Structure, to that CCM Gross Omnibus Segregated Account Structure or CCM Indirect Client Gross Segregated Account Structure,

as applicable, to satisfy such Intraday Novation Margin Requirement, the relevant Eligible Intraday Transaction will become a Rejected Transaction.

For the avoidance of doubt, in the event LCH SA determines that there is a neutral or negative Intraday Novation Margin Requirement (due to the pre-registered position corresponding to the relevant trade leg of an Eligible Intraday Transaction being set off against Open Positions registered in the relevant Margin Account), none of the House Excess Collateral, Client Excess Collateral and/or Available Client Collateral, as applicable, will be used or applied for the purpose of satisfying the Notional and Collateral Check.

LCH SA will perform a reporting update in respect of each Clearing Member's Margin Account ten times per Business Day (by 10.00 CET, 10.55 CET, 12.30 CET, 13.30 CET, 14.15 CET, 14:55 CET, 16.15 CET, 17.00 CET, 18.00 CET and 19.00 CET) in order to inform such Clearing Member of the updated Margin Requirement for each Margin Account, level of Excess Collateral and/or Client Collateral Buffer recorded in, or allocated to, the relevant Collateral Accounts.

(ii) Intraday Call

LCH SA will perform an Intraday Call twice per Business Day (by 11.25 CET (the "First Intraday Call") and 15.25 CET (the "Second Intraday Call") in order, if necessary, to transfer Collateral to satisfy the Required Collateral Amount.

Where the relevant Business Day is a Clearing Day, the First Intraday Call and the Second Intraday Call will not be performed to the extent there is no Real Time Session, on that Clearing Day.

The Margin Requirement in respect of each Margin Account of a Clearing Member for an Intraday Call will be the latest Margin Requirement calculated on that Clearing Day.

First Intraday Call:

During the First Intraday Call, LCH SA will issue to each Clearing Member the relevant risk management and collateral management reports (as set out in Section 5 of the Procedures), which will notify each such Clearing Member of its House Excess Collateral Threshold and Client Collateral Buffer Threshold and the:

- (A) Margin Requirement for the First Intraday Call (and each component thereof);
- (B) Excess Collateral or Margin Shortfall (as the case may be); and
- (C) Margin Balance,

for the relevant Margin Accounts, in accordance with and subject to Section 5 of the Procedures.

Second Intraday Call:

During the Second Intraday Call, LCH SA will issue to each Clearing Member the relevant risk management and collateral management reports (as set out in Section 5 of the Procedures), which will notify each such Clearing Member of its House Excess Collateral Threshold and Client Collateral Buffer Threshold and the:

- (A) Margin Requirement for the Second Intraday Call (and each component thereof);
- (B) Excess Collateral or Margin Shortfall (as the case may be); and
- (C) Margin Balance;

for the relevant Margin Accounts, in accordance with and subject to Section 5 of the Procedures.

Each Clearing Member is required to transfer Collateral to satisfy its Required Collateral Amount, make Variation Margin Collateral Transfers in respect of its Variation Margin Requirement and to make NPV Payments in respect of its NPV Payment Requirement, as applicable, for the relevant Margin Accounts, as set out in Section 3 of the Procedures.

2.6 **REPORTS**

All files and reports, mentioned in this Section 2 of the Procedures, will be available to Clearing Members through the reporting mechanism. If the reporting mechanisms are, for any reason unavailable, LCH SA will otherwise make such reports available to Clearing Members at the requisite time.

Please see Section 5 of the Procedures for further details about the relevant files and reports.

2.7 INITIAL MARGIN

Initial Margin covers the potential costs caused by a Defaulting Clearing Member and/or a double Event of Default, i.e. a combined Credit Event of a Reference Entity and a Clearing Member Event of Default (in which the Clearing Member is a CDS Seller in respect of an Index Cleared Transaction or a Single Name Cleared Transaction, or a protection seller in respect of the Underlying Index Transaction of an Index Swaption Cleared Transaction).

(a) Spread Margin

Spread Margin is calculated using the Value-at-Risk (VaR) model which is based on the following principles: at the Margin Account level, a distribution of potential losses is built from simulated scenarios based on the joint credit spread and at-the-money volatility variations observed in the past. LCH SA then determines the Spread Margin based on a quantile of the worst losses that the Margin Account could bear in the case of unfavourable credit spread and at-the-money volatility fluctuations.

The Spread Margin calculated in respect of the House Margin Account covers the potential costs of liquidating House Cleared Transactions of the Defaulting Clearing Member whilst the Spread Margin calculated in respect of each Client Margin Account covers the potential costs of liquidating any Client Cleared Transactions attributable to such Client Margin Account. It covers the potential future credit spread and at-the-money volatility fluctuations in case of unfavourable market movements under normal circumstances.

(b) Spread Margin Floor

LCH SA may, by Clearing Notice, specify a Spread Margin Floor approved by the board of directors of LCH SA following consultation with the Risk Committee.

Where the calculation of Spread Margin would result in the Spread Margin for any Margin Account of a Clearing Member being less than the Spread Margin Floor, the Spread Margin requirement for such Margin Account shall be equal to the Spread Margin Floor.

(c) Short Charge Margin

Where a Clearing Member is acting as a CDS Seller in respect of an Index Cleared Transaction or a Single Name Cleared Transaction, or as a protection seller in respect of the Underlying Index Transaction of an Index Swaption Cleared Transaction, Short Charge Margin will be required to cover the risk that the Clearing Member is subject to an Event of Default at the same time as a Credit Event occurs with respect to a-one or more Reference Entityies.

The Short Charge Margin is calculated using an algorithm, approved by the board of directors of LCH SA following consultation with the Risk Committee, based on the Open Positions registered in the relevant Margin Account of the Clearing Member.

(d) Recovery Risk Margin

Recovery Risk Margin covers the risk of future price fluctuations in case of unfavourable recovery rate movements under normal circumstances and when liquidating a Defaulting

Clearing Member's portfolio of House Cleared Transactions or Non-Ported Cleared Transactions.

The Recovery Risk Margin is calculated using an algorithm, approved by the board of directors of LCH SA following consultation with the Risk Committee, based on the Open Positions registered in the relevant Margin Account of the Clearing Member.

(e) Interest Rate Risk Margin

Interest Rate Risk Margin covers the risk of future price fluctuations in case of unfavourable interest rate movements under normal circumstances and when liquidating a Defaulting Clearing Member's portfolio of House Cleared Transactions or Non-Ported Cleared Transactions.

The Interest Rate Risk Margin is calculated using an algorithm, approved by the board of directors of LCH SA following consultation with the Risk Committee, based on the Open Positions registered in the relevant Margin Account of the Clearing Member.

(f) Wrong Way Risk Margin

Wrong Way Risk Margin is required to cover the anticipated financial contagion effect that would arise in case of a Clearing Member being declared in default in accordance with Title IV Chapter 3 of the CDS Clearing Rule Book, such contagion effect triggering additional spread or correlation risk not currently captured by the Spread Margin.

The Wrong Way Risk Margin is calculated using an algorithm, approved by the board of directors of LCH SA following consultation with the Risk Committee, based on the Open Positions registered in the relevant Margin Account of the Clearing Member.

(g) Vega Margin

Vega Margin covers the risk of future price fluctuations of an Index Swaption Cleared Transaction in case of unfavourable deformations of the volatility surface, when liquidating a Default Clearing Member's portfolio of House Cleared Transactions or Non-Ported Cleared Transactions.

The Vega Margin is calculated using an algorithm, approved by the board of directors of LCH SA following consultation with the Risk Committee, based on the Open Positions registered in the relevant Margin Account of the Clearing Member.

2.8 SELF-REFERENCING PROTECTION MARGIN

Where a Clearing Member is acting as a CDS Seller in respect of an Index Cleared Transaction or a Single Name Cleared Transaction, or as a protection seller in respect of the Underlying Index Transaction of an Index Swaption Cleared Transaction, for which such Clearing Member is, or becomes, the Reference Entity, Self-Referencing Protection Margin will be required to cover the protection that would have to be paid by LCH SA with respect to this Clearing Member should the Clearing Member be subject to an Event of Default.

PUBLIC

The Self-Referencing Protection Margin is calculated using an algorithm, approved by the board of directors of LCH SA following consultation with the Risk Committee, based on the Open Positions registered in the relevant Margin Account of the Clearing Member.

2.9 LIQUIDITY AND CONCENTRATION RISK MARGIN

Liquidity and Concentration Risk Margin is required to cover the bid-ask spread incurred when liquidating the House Cleared Transactions or the Non-Ported Cleared Transactions of a Defaulting Clearing Member. The size of this bid-ask spread will increase if the positions exceed predetermined thresholds in respect of the relevant credit default swap index or Reference Entity. Further details of the thresholds are available on a Clearing Notice published on the Website and/or upon request from LCH SA's Risk Management Department on +33 1 70 37 10 43 or CDS.Risk.SA@lch.com.

Liquidity and Concentration Risk Margin is calculated:

- (a) in respect of the House Margin Account of a Clearing Member to cover the potential costs of hedging or liquidating the House Cleared Transactions in case an Event of Default occurs in respect of such Clearing Member; and
- (b) in respect of a Client Margin Account of a Clearing Member to cover the potential costs of hedging or liquidating the Non-Ported Cleared Transactions attributable to such Client Margin Account in case an Event of Default occurs in respect of such Clearing Member.

The Liquidity and Concentration Risk Margin is calculated using an algorithm (including thresholds) approved by the board of directors of LCH SA following consultation with the Risk Committee.

2.10 ACCRUED FIXED AMOUNT LIQUIDATION RISK MARGIN

Each Clearing Member acting as a CDS Buyer in respect of an Index Cleared Transaction or a Single Name Cleared Transaction, or as a protection buyer in respect of the Underlying Index Transaction of an Index Swaption Cleared Transaction where the exercise of that Index Swaption Cleared Transaction falls in the margin calculation time horizon, is required to pay Accrued Fixed Amount Liquidation Risk Margin in respect of the relevant Cleared Transactions to cover the risk that it is subject to an Event of Default and accrued Fixed Amounts are due during the period that the relevant House Cleared Transactions or Non-Ported Cleared Transactions, as applicable, are liquidated pursuant to the CDS Default Management Process.

The Accrued Fixed Amount Liquidation Risk Margin is calculated daily for each Margin Account of each Clearing Member and corresponds to the aggregate amount of daily Fixed Amounts for such Clearing Member pursuant to its Cleared Transactions during a rolling forward-looking period of 5 Business Days.

2.11 CREDIT EVENT MARGIN

Where a Credit Event occurs with respect to the Reference Entity which is the subject of the Cleared Transaction, each Clearing Member is required to pay Credit Event Margin in respect of the relevant Cleared Transactions to cover the risk of non-payment of Variation Margin by the CDS Seller or CDS Buyer in respect of an Index Cleared Transaction or a Single Name

Cleared Transaction, or the Index Swaption Seller or Index Swaption Buyer in respect of an Index Swaption Cleared Transaction, arising from a potential adverse change in the estimated recovery rate.

Credit Event Margin will be calculated by LCH SA for each Margin Account of each Clearing Member, on each Business Day from the date of the relevant DC Credit Event Announcement until the settlement process in respect of such Cleared Transaction has been completed (including Physical Settlement as set out in the CDS Clearing Supplement, or Auction Settlement, as applicable) or any disputes in relation thereto have been finally resolved. The calculation of the Credit Event Margin is based on an estimated recovery rate of the Affected Cleared Transaction or the Restructuring Cleared Transaction, as the case may be, and the exposure of LCH SA by reference to the notional amount of the Clearing Member's Cleared Transaction(s) affected by the Credit Event.

Credit Event Margin will be reimbursed on the Business Day following completion or resolution of the settlement process (including Physical Settlement, or Auction Settlement and/or index re-versioning, as applicable) or the day on which settlement can no longer occur in respect of such Credit Event (including without limitation because no relevant Credit Event Notice or Notice of Physical Settlement is delivered within the required timeframes).

In the event that a DC Credit Event Announcement made in relation to a Credit Event is reversed then LCH SA shall reimburse each Clearing Member with the amount of any Credit Event Margin on the next following Business Day in accordance with Section 3 of the Procedures.

2.12 LEGAL ENTITY IDENTIFIER MARGIN

Legal Entity Identifier Margin covers the risk of concentration arising when liquidating simultaneously some or all Cleared Transactions registered in the Account Structures of one or more Clearing Members identified by the same Legal Entity Identifier (LEI), which would not have been covered by any other Margins.

The Legal Entity Identifier Margin is calculated using an algorithm, approved by the board of directors of LCH SA following consultation with the Risk Committee, based on the Open Positions registered in the Margin Accounts of one or more Clearing Members identified by the same LEI.

2.13 ADDITIONAL MARGIN

A Clearing Member will be required to pay Additional Margin for some or all of its Margin Accounts where that Clearing Member's Group Member Uncovered Risk is greater than x% of the current value of the CDS Default Fund.

Additional Margin will be equal to the difference between x% of the current value of the CDS Default Fund and the relevant Group Member Uncovered Risk on such Business Day.

The number x will depend on the internal credit score that LCH SA attributes to each Clearing Member and will be the same for each Margin Account of the Clearing Member. Any change to the number x will be communicated to the Clearing Member.

PUBLIC

When Additional Margin is required to be paid to LCH SA, or the amount of Additional Margin payable is increased due to a change in the relevant Group Member Uncovered Risk, LCH SA will notify the Clearing Member in the Margin Requirements Results File pursuant to Section 5 of the Procedures.

2.14 STRESS TEST LOSS OVER ADDITIONAL MARGIN / NET CAPITAL RATIO MARGIN

Stress Test Loss Over Additional Margin / Net Capital Ratio Margin covers the risk that a Clearing Member might not have sufficient capital to support the materialisation of stress test losses.

The Stress Test Loss Over Additional Margin / Net Capital Ratio Margin is calculated using an algorithm, approved by the board of directors of LCH SA following consultation with the Risk Committee, based on the Open Positions registered in the relevant Margin Accounts of the Clearing Member.

2.15 VARIATION MARGIN

Variation Margin is an amount exchanged on each Cash Payment Day between the Clearing Member and LCH SA to account for the potential profit or loss on CTM Cleared Transactions and, as the case may be, Irrevocable Backloading CTM Transactions which are not Rejected Transactions due to the variation of the market value of a CDS or, as the case may be, an Index Swaption.

It covers price fluctuations which have occurred since the registration of each CTM Cleared Transaction and Irrevocable Backloading CTM Transactions which are not Rejected Transactions. LCH SA will calculate the Variation Margin transferable in respect of each CTM Cleared Transaction and, as the case may be, Irrevocable Backloading CTM Transactions which are not Rejected Transactions of each Clearing Member as the difference between:

- (a) the net present value on the current Cash Payment Day of the relevant CTM Cleared Transaction allocated to (or Irrevocable Backloading CTM Transaction pre-allocated to) a Margin Account at the time of calculation; and
- (b) the net present value on the immediately preceding Cash Payment Day of the CTM Cleared Transaction allocated to (or Irrevocable Backloading CTM Transaction preallocated to) a Margin Account.

LCH SA will calculate the net present value of each CTM Cleared Transaction or Irrevocable Backloading CTM Transaction using the End of Day Contributed Prices provided to LCH SA in accordance with Article 4.2.7.1 of the CDS Clearing Rule Book and Section 5 of the Procedures (or, where such End of Day Contributed Prices are not available to LCH SA, the prices/spreads as set out in Article 4.2.7.1 of the CDS Clearing Rule Book and Section 5 of the Procedures). Any time this calculation is performed for the first time in respect of any particular CTM Cleared Transaction or Irrevocable Backloading CTM Transaction that CTM Cleared Transaction or Irrevocable Backloading CTM Transaction shall be deemed to have had a net present value of zero at the time of the preceding valuation.

PUBLIC

Where the difference between the net present values of each CTM Cleared Transaction or Irrevocable Backloading CTM Transaction with respect to a Clearing Member's Margin Account is:

- (i) a negative amount: such Clearing Member owes Variation Margin to LCH SA (and will be considered a Variation Margin debtor in relation to such Margin Account); or
- (ii) a positive amount: LCH SA owes Variation Margin to such Clearing Member.

In respect of all CTM Cleared Transaction LCH SA shall, on each Cash Payment Day, and to the extent that the following amounts are payable in the same currency and are related to the same Margin Account:

- (A) aggregate the Variation Margin (if any) payable by the Clearing Member to LCH SA on such Cash Payment Day in accordance with this paragraph 2.15; and
- (B) aggregate the Variation Margin (if any) payable by LCH SA to the Clearing Member on such Cash Payment Day in accordance with this paragraph 2.15.

and the Variation Margin payable on a Cash Payment Day to one party (the Transferee) by the other party (the Transferor) under paragraphs 2.15 (A) or (B) above (as applicable) shall be reduced by setting-off such Variation Margin against the Variation Margin payable by the Transferee to the Transferor under paragraphs 2.15 (A) or (B) above (as applicable).

On the basis of these calculations LCH SA will determine:

- (x) the Client Variation Margin Requirement for each Client Margin Account of the Clearing Member; and/or
- (y) the House Variation Margin Requirement for the House Margin Account,

which will trigger Variation Margin Collateral Transfer(s) to be made by the Clearing Member and/or LCH SA in accordance with Article 4.2.3.2 of the CDS Clearing Rule Book and Section 3 of the Procedures.

The amount of Variation Margin transferred or received by LCH SA to or from a Clearing Member may be adjusted in accordance with Clause 7 of the CDS Default Management Process.

2.16 NPV AMOUNT

NPV Amount is an amount paid on each Cash Payment Day between the Clearing Member and LCH SA to account for the variation of the market value of the CDS or, as the case may be, Index Swaption that is an STM Cleared Transaction or Irrevocable Backloading STM Transaction .

LCH SA will calculate the NPV Amount payable on each Cash Payment Day in respect of each STM Cleared Transaction or Irrevocable Backloading STM Transaction of each Clearing

Margin, NPV Payment and Price Alignment

PUBLIC

Member on the basis of the change in the net present value of each STM Cleared Transaction and Irrevocable Backloading STM Transaction.

LCH SA will calculate the net present value of each STM Cleared Transaction or Irrevocable Backloading STM Transaction using the End of Day Contributed Prices provided to LCH SA in accordance with Article 4.2.7.1 of the CDS Clearing Rule Book and Section 5 of the Procedures (or, where such End of Day Contributed Prices are not available to LCH SA, the prices/spreads as set out in Article 4.2.7.1 of the CDS Clearing Rule Book and Section 5 of the Procedures). Unless otherwise agreed between the Clearing Member and LCH SA, the net present value of an STM Cleared Transaction or Irrevocable Backloading STM Transaction on the Trade Date (as such term is defined in the CDS Clearing Supplement) shall be equal to zero.

Immediately upon LCH SA making each such determination of the net present value of an STM Cleared Transaction or Irrevocable Backloading STM Transaction, a net present value reset (a "NPV Reset") of such STM Cleared Transaction or Irrevocable Backloading STM Transaction shall occur with respect to that STM Cleared Transaction or Irrevocable Backloading STM Transaction. For STM Cleared Transaction or Irrevocable Backloading STM Transaction registered in the Account Structure of an FCM/BD Clearing Member, such NPV Reset shall not be deemed to have occured if the resulting NPV Payment Requirement is not satisfied. However, if such NPV Payment Requirement is satisfied, the related NPV Reset will be deemed to have occurred immediately upon LCH SA making such determination of the net present value of the relevant STM Cleared Transaction or Irrevocable Backloading STM Transaction.

Upon the occurrence of an NPV Reset in relation to a STM Cleared Transaction or Irrevocable Backloading STM Transaction:

- (i) if LCH SA has determined that the net present value of the STM Cleared Transaction or Irrevocable Backloading STM Transaction has increased since the immediately preceding NPV Reset, an amount of cash (the NPV Amount) denominated in the currency of the STM Cleared Transaction or Irrevocable Backloading STM Transaction and equal to the amount of such increase, shall immediately become due and payable by LCH SA to the Clearing Member;
- (ii) if LCH SA has determined that the net present value of the STM Cleared Transaction or Irrevocable Backloading STM Transaction has decreased since the immediately preceding NPV Reset, an amount of cash (the NPV Amount) denominated in the currency of the STM Cleared Transaction or Irrevocable Backloading STM Transaction and equal to the amount of such decrease, shall immediately become due and payable by the Clearing Member to LCH SA;
- (iii) if LCH SA has determined that the net present value of the STM Cleared Transaction or Irrevocable Backloading STM Transaction has not changed since the immediately preceding NPV Reset, neither LCH SA nor the Clearing Member shall be obliged to make any payment in relation to such STM Cleared Transaction or Irrevocable Backloading STM Transaction; and
- (iv) the net present value of the STM Cleared Transaction or Irrevocable Backloading STM Transaction shall for all purposes be equal to zero.

PUBLIC

The Clearing Member and LCH SA hereby agree that:

- (a) for the avoidance of doubt, an "increase" in the net present value of an STM Cleared Transaction or Irrevocable Backloading STM Transaction shall mean that the net present value of that STM Cleared Transaction or Irrevocable Backloading STM Transaction has moved in favour of the Clearing Member since the immediately preceding NPV Reset; and
- (b) for the avoidance of doubt, a "decrease" in the net present value of an STM Cleared Transaction or Irrevocable Backloading STM Transaction shall mean that the net present value of that STM Cleared Transaction or Irrevocable Backloading STM Transaction has moved in favour of LCH SA since the immediately preceding NPV Reset.

Except as otherwise prescribed in the CDS Clearing Rules, the net present value calculated by LCH SA shall in no circumstances be called in question.

In respect of all STM Cleared Transactions LCH SA shall, on each Cash Payment Day, and to the extent that the following amounts are payable in the same currency and are related to the same Margin Account:

- (A) aggregate the NPV Amounts (if any) payable by the Clearing Member to LCH SA on such Cash Payment Day in accordance with this paragraph 2.16; and
- (B) aggregate the NPV Amounts (if any) payable by LCH SA to the Clearing Member on such Cash Payment Day in accordance with this paragraph 2.16;

and the NPV Amount payable on a Cash Payment Day to one party (the Payee) by the other party (the Payer) under paragraphs 2.16 (A)(A) or (B)(B) above (as applicable) shall be reduced by setting-off such NPV Amount against the NPV Amount payable by the Payee to the Payer under paragraphs 2.16 (A)(A) or (B)(B) above (as applicable).

On the basis of these calculations, LCH SA will determine:

- the Client NPV Payment Requirement for each Client Margin Account of the Clearing Member; and/or
- (y) the House NPV Payment Requirement for the House Margin Account,

which will trigger an NPV Payment to be made by the Clearing Member and/or LCH SA in accordance with Article 4.2.3.2 of the CDS Clearing Rule Book and Section 3 of the Procedures.

The NPV Amount paid or received by LCH SA to or from a Clearing Member may be adjusted in accordance with Clause 7 of the CDS Default Management Process.

2.17 **CONTINGENCY VARIATION MARGIN**

Contingency Variation Margin is calculated on any Business Day, in respect of:

PUBLIC

- (i) new Cleared Transactions arising from the novation of Eligible Intraday Transactions. It is intended to cover the risk that the Clearing Member fails to make NPV Payment(s) to meet the NPV Payment Requirement or Variation Margin Collateral Transfer(s) to meet the Variation Margin Requirement in respect of each of its Margin Accounts at the next Morning Call; and
- (ii) Cleared Transactions with a CDS Contractual Currency in USD provided that such Business Day is a day on which commercial banks in New York City are not open for business. It is intended to cover the risk that the Clearing Member fails to satisfy its NPV Payment(s) or Variation Margin Collateral Transfer(s) obligations in USD to meet the NPV Payment Requirement or Variation Margin Requirement, as applicable, in respect of each of its Margin Accounts at the Morning Call on the following Cash Payment Day.

Contingency Variation Margin is called from a Clearing Member in place of the Variation Margin or NPV Amount, as applicable, which LCH SA determines would have been owed by such Clearing Member had such Clearing Member been required to make a NPV Payment or Variation Margin Collateral Transfer to satisfy the NPV Payment Requirement or Variation Margin Requirement, as applicable, in relation to the relevant Margin Account at that point in time (being the time of the Morning Call, the First Intraday Call or the Second Intraday Call, as described below).

Contingency Variation Margin is called (as applicable) in relation to each Margin Account of a Clearing Member:

- (a) in respect of Eligible Intraday Transactions: during the First Intraday Call and/or the Second Intraday Call.
 - Contingency Variation Margin paid by a Clearing Member during the First Intraday Call and/or the Second Intraday Call is repaid to such Clearing Member at the Morning Call on the following Business Day if the required Variation Margin has been transferred or the required NPV Amount has been paid, as applicable, by such Clearing Member at the relevant Morning Call in accordance with Paragraph 2.12 above; and
- (b) in respect of Cleared Transactions with a CDS Contractual Currency in USD: during the Morning Call.

Contingency Variation Margin paid by a Clearing Member during the Morning Call is repaid to such Clearing Member on the following Cash Payment Day if the required Variation Margin has been transferred or the required NPV Amount has been paid, as applicable, by such Clearing Member at the relevant Collateral Call in accordance with Paragraph 2.12 above and Section 3 of the Procedures.

2.18 PRICE ALIGNMENT INTEREST

Each Clearing Member that has, on a cumulative net basis, received Variation Margin Collateral Transfers from LCH SA is required to pay Price Alignment Interest. LCH SA shall pay Price Alignment Interest to each Clearing Member that has, on a cumulative net basis, transferred Variation Margin in accordance with Paragraph 2.13 above. In a negative interest rate

Margin, NPV Payment and Price Alignment

PUBLIC

environment where the applicable Price Alignment Interest rate is negative, LCH SA will either (i) pay Price Alignment Interest if a Clearing Member has, on a cumulative net basis, received Variation Margin from LCH SA, or (ii) charge Price Alignment Interest if a Clearing Member has, on a cumulative net basis, transferred Variation Margin.

A Report published in accordance with and subject to Section 5 of the Procedures and received by each Clearing Member on each Cash Payment Day shall indicate the amount of Price Alignment Interest to be paid or received by a Clearing Member.

These payments are made in order to minimise distortion of pricing for Original Transactions cleared through LCH SA as a result of daily Variation Margin Collateral Transfers and changes in the net present value of CTM Cleared Transactions.

Price Alignment Interest is calculated for each Clearing Member by LCH SA using the interest rate prevailing on that Cash Payment Day, as published by LCH SA on the Website, on the basis of the net present value of the CTM Cleared Transactions allocated to the Margin Accounts of the Clearing Member on the immediately preceding Cash Payment Day.

2.19 PRICE ALIGNMENT AMOUNT

Upon LCH SA's determination of the Price Alignment Amount in relation to an STM Cleared Transaction:

- (i) if LCH SA has determined that the Cumulative Net Present Value is greater than zero, then, subject to (3) below, an amount of cash denominated in the currency of the STM Cleared Transaction equal to the Price Alignment Amount shall immediately become due and payable by the Clearing Member to LCH SA;
- (ii) if LCH SA has determined that the Cumulative Net Present Value is less than zero, then, subject to (3) below, an amount of cash denominated in the currency of the STM Cleared Transaction equal to the Price Alignment Amount shall immediately become due and payable by LCH SA to the Clearing Member; and
- (iii) if the Price Alignment Amount payable by a party on a Cash Payment Day is a negative amount, then the Price Alignment Amount payable by that party will be deemed to be zero, and the other party will pay to that party the absolute value of the negative Price Alignment Amount on such Cash Payment Day.

A Report published in accordance with and subject to Section 5 of the Procedures and received by each Clearing Member on each Cash Payment Day shall indicate the amount of Price Alignment Amount to be paid or received by a Clearing Member.

These payments are made in order to minimise distortion of pricing for Original Transactions cleared through LCH SA as a result of daily NPV Payments and changes in the net present value of STM Cleared Transactions.

For the purpose of determining the Cumulative Net Present Value of an STM Cleared Transaction that has been either (i) transferred to a Backup Clearing Member pursuant to the CDS Default Management Process, or (ii) transferred to a Receiving Clearing Member pursuant to Title V, Chapter 3 and Title VI, Chapter 3 of the CDS Clearing Rule Book the Trade Date of

PUBLIC

the STM Cleared Transaction that comes into existence immediately following such transfer shall be the Trade Date of the STM Cleared Transaction that was so transferred.

For the purpose of determining the Cumulative Net Present Value of an STM Cleared Transaction that has been converted from a CTM Cleared Transaction, pursuant to Article 3.1.10.8 of the CDS Clearing Rule Book and Section 5 of the Procedures, the Trade Date of the STM Cleared Transaction that comes into existence immediately following such conversion shall be the Trade Date of the CTM Cleared Transaction that was so converted.

For the purposes of this paragraph 2.17:

"Cumulative Net Present Value" means, in respect of an STM Cleared Transaction and a Cash Payment Day falling after the Trade Date, a hypothetical value, determined by LCH SA acting in its sole and absolute discretion, equal to:

- (a) the aggregate of NPV Amounts (if any) payable by the Clearing Member to LCH SA (expressed as a negative number) under paragraph 2.14 above from, and including, the Trade Date to, and including, that Cash Payment Day; plus
- (b) if such STM Cleared Transaction has been converted from a CTM Cleared Transaction, the aggregate of the amounts (if any) of Variation Margin determined to be payable by the Clearing Member to LCH SA (expressed as a negative number) from, and including, the Trade Date, to, and including, the STM Conversion Date (as defined in Section 5 of the Procedures) relating to such STM Cleared Transaction; plus
- the aggregate of NPV Amounts (if any) payable by LCH SA to the Clearing Member (expressed as a positive number) under paragraph 2.14 above, and including, the Trade Date, to , and including, that Cash Payment Day; plus
- (d) if such STM Cleared Transaction has been converted from a CTM Cleared Transaction, the aggregate of the amounts (if any) of Variation Margin determined to be payable by LCH SA to the Clearing Member (expressed as a positive number) from, and including, the Trade Date, to, and including, the STM Conversion Date relating to the STM Cleared Transaction.

"Price Alignment Amount" means, in respect of a Cash Payment Day falling after the Trade Date, the product of:

- (1) the absolute value of the Cumulative Net Present Value on such Cash Payment Day;
- (2) the applicable Price Alignment Amount Rate on such Cash Payment Day; and
- (3) the day count fraction determined by the LCH SA as being applicable to the currency of the STM Cleared Transaction.

"Price Alignment Amount Rate" means the applicable interest rate prevailing on that Cash Payment Day, as mentioned in the files and reports available to the Clearing Member.

PUBLIC

2.20 **CREDIT QUALITY MARGIN**

LCH SA may require a Clearing member to provide Credit Quality Margin when LCH SA determines that the credit quality of such Clearing Member has deteriorated, depending on the internal credit score that LCH SA attributes to the relevant Clearing Member: (i) following monitoring carried out in accordance with Article 2.3.2.1 of the CDS Clearing Rule Book; and/or (ii) in the circumstances set out in Article 4.2.1.2 of the CDS Clearing Rule Book.

On each Business Day, Credit Quality Margin will be equal to the higher of the amounts calculated as follows:

- (i) (Y-1)*IM where Y stands for the credit multiplier applied to the Clearing Member's Initial Margins and IM stands for the Clearing Member's Initial Margins; or
- (ii) X*(Stress Risk IM) where X stands for the stress risk percentage and (Stress Risk IM) stands for the Clearing Member's Group Member Uncovered Risk.

Credit multipliers and stress risk percentage are determined in accordance with the methodology established by LCH SA. Credit multipliers which can be applied range from 1 to 1.4, meaning that the additional liability for any Clearing Member is capped at 40% of the relevant Clearing Member's Initial Margin. The stress risk percentage ranges between 0% and 100% of the Group Member Uncovered Risk, meaning that the additional liability for any Clearing Member is capped at 100% of the relevant Clearing Member's Group Member Uncovered Risk.

LCH SA assesses the amount of the Credit Quality Margin across each of the Margin Accounts of a Clearing Member. Credit Quality Margin will only be called in relation to a Clearing Member's House Margin Account.

LCH SA may update a Clearing Member's credit multiplier or the stress risk percentage that should apply: (i) following monitoring carried out in accordance with Article 2.3.2.1 of the CDS Clearing Rule Book; and/or (ii) in the circumstances set out in Article 4.2.1.2 of the CDS Clearing Rule Book. LCH SA will notify a Clearing Member of the Credit Quality Margin that has been called in the Margin Requirements Results File in accordance with Section 5 of the Procedures.

2.21 EXTRAORDINARY MARGIN

LCH SA may require a Clearing Member to provide Extraordinary Margin to cover the risk of price/spread fluctuations occurring on an intraday basis or during a day that is a holiday in the TARGET2 calendar.

Extraordinary Margin is called (as applicable) in relation to each Margin Account of a Clearing Member at the time of the most appropriate time slot for the purpose of making a Collateral Call. LCH SA will notify the relevant Clearing Member of the amount of Extraordinary Margin in accordance with, subject to and at the times set out in the CDS Clearing Rules.